SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM S-8 REGISTRATION STATEMENT

UNDER
THE SECURITIES ACT OF 1933

SAIC, INC.

(Exact name of registrant as specified in its charter)

DELAWARE

(State or other jurisdiction of incorporation or organization)

20-3562868 (I.R.S. Employer Identification No.)

10260 CAMPUS POINT DRIVE SAN DIEGO, CALIFORNIA 92121 (858) 826-6000

(Address, including zip code, and telephone number, including area code of registrant's principal executive offices)

2006 Equity Incentive Plan
2006 Employee Stock Purchase Plan
Science Applications International Corporation Retirement Plan
Science Applications International Corporation Stock Compensation Plan
Science Applications International Corporation Management Stock Compensation Plan
Science Applications International Corporation Key Executive Stock Deferral Plan
Science Applications International Corporation 1999 Stock Incentive Plan

AMSEC Employees 401(k) Profit Sharing Plan
(Full title of the plans)

DOUGLAS E. SCOTT, ESQ.
Senior Vice President, General Counsel and Secretary
SAIC, Inc.
10260 Campus Point Drive
San Diego, California 92121
(858) 826-6000

(Name, address, including zip code, and telephone number, including area code, of agent for service)

CALCULATION OF REGISTRATION FEE

Title of Securities to be Registered	Amount to be Registered(1)(2)	Off	sed Maximum Fering Price or Share(3)	Proposed Maximum Aggregate Offering Price(3)	Amount of Registration Fee(3)(4)
Class A Preferred Stock, par value \$0.0001 per share(1)(2)	219,000,000 shares	\$	17.645	\$ 3,864,255,000	\$ 413,476
Common Stock, par value \$0.0001 per share(1) (2)					

- (1) This Registration Statement registers an aggregate of 219,000,000 shares of class A preferred and common stock of Registrant, of which 75,000,000 shares are issuable under the 2006 Equity Incentive Plan; 9,000,000 shares under the 2006 Employee Stock Purchase Plan; 45,000,000 shares under the Science Applications International Corporation Retirement Plan; 2,000,000 shares under the Science Applications International Corporation Management Stock Compensation Plan; 4,000,000 shares under the Science Applications International Corporation Key Executive Stock Deferral Plan; 82,000,000 shares under the Science Applications International Corporation 1999 Stock Incentive Plan; and 1,000,000 shares under the AMSEC Employees 401(k) Profit Sharing Plan (collectively the "Plans"). Shares of class A preferred stock or of common stock may be issued under the 2006 Equity Incentive Plan, the 2006 Employee Stock Purchase Plan, the Science Applications International Corporation Retirement Plan, the Science Applications International Corporation Management Stock Compensation Plan, the Science Applications International Corporation Management Stock Compensation Plan, the Science Applications International Corporation Key Executive Stock Deferral Plan and the AMSEC Employees 401(k) Profit Sharing Plan. Only shares of class A preferred stock may be issued under the Science Applications International Corporation 1999 Stock Incentive Plan. Shares of class A preferred stock are divided into series A-1, series A-2, series A-3 and series A-4 preferred stock, and are convertible into common stock subject to certain limitations set forth in Registrant's restated certificate of incorporation. In addition, this Registration Statement covers an indeterminate number of shares of common stock issuable on conversion of class A preferred stock. No separate consideration will be received for the common stock that is issuable upon conversion of the class A preferred stock.
- (2) In addition, pursuant to Rule 416(c) under the Securities Act of 1933, this Registration Statement covers an indeterminate amount of plan interests to be

- offered or sold pursuant to the Science Applications International Corporation Retirement Plan and the AMSEC Employees 401(k) Profit Sharing Plan. This Registration Statement also covers, pursuant to Rule 416(a) under the Securities Act of 1933, any additional shares of class A preferred stock and common stock which become issuable under the Plans by reason of any stock dividend, stock split, recapitalization or other similar transaction in accordance with the adjustment provisions of the Plans.
- (3) The class A preferred stock is not publicly traded, but each share is convertible into one share of common stock. Therefore, the registration fee for the class A preferred stock and the common stock has been determined, in accordance with Rule 457(c) and 457(h) under the Securities Act of 1933, on the basis of the average of the high and low prices of the common stock on the New York Stock Exchange on October 13, 2006.
- (4) Pursuant to Rule 457(p) under the Securities Act of 1933, Registrant hereby offsets from the registration fee required in connection with this Registration Statement an aggregate of \$147,920 in fees previously paid consisting of: (i) \$22,872 previously paid by Registrant in connection with the registration of 25,000,000 shares of class A preferred stock and common stock being offered pursuant to the Registration Statement on Form S-4 (Registration No. 333-128022) initially filed with the Securities and Exchange Commission on September 1, 2005, which shares are in excess of those to be used in connection with the merger described in the Explanatory Note, (ii) \$50,758 previously paid by Registrant in connection with the registration of 28,750,000 shares of common stock being offered pursuant to the Registration Statement on Form S-1 (Registration No. 333-128021) initially filed with the Securities and Exchange Commission on September 1, 2005, which shares are in excess of those being sold in Registrant's public offering of common stock, and (iii) \$74,290 previously paid by Science Applications International Corporation, a wholly-owned subsidiary of Registrant, in connection with the registration of its class A common stock pursuant to Registration Statements on Form S-8 (Registration No. 333-119145 filed on September 21, 2004 (which incorporated Registration Statement No. 333-128095 filed August 30, 2002), and Registration No. 333-123095 filed on March 2, 2005 (which incorporated Registration Statement No. 333-113202 filed on March 1, 2004)), which shares remain unsold after the completion of the offerings contemplated by such Registration Statements.

The Registration Statement shall become effective upon filing in accordance with Rule 462(a) under the Securities Act of 1933.

EXPLANATORY NOTE

We entered into an Agreement and Plan of Merger, or the merger agreement, as amended and restated as of July 24, 2006, with Science Applications International Corporation ("Old SAIC") and SAIC Merger Sub, Inc. Pursuant to the merger agreement, our newly created wholly-owned subsidiary, SAIC Merger Sub, Inc., merged with and into Old SAIC, as a result of which Old SAIC became our wholly-owned subsidiary. In the merger, each share of class A common stock of Old SAIC was converted into the right to receive two shares of our class A preferred stock and, subject to the exercise of appraisal rights, each share of class B common stock of Old SAIC was converted into the right to receive forty shares of our class A preferred stock. Our class A preferred stock is convertible on a one-for-one basis into shares of our common stock.

In connection with the merger, our board of directors and stockholder and the board of directors and stockholders of Old SAIC approved and adopted the 2006 Equity Incentive Plan and the 2006 Employee Stock Purchase Plan. The other Plans were either assumed by us or otherwise permit us to issue our class A preferred stock or common stock in lieu of the shares of class A common stock of Old SAIC previously issuable thereunder.

This Registration Statement registers shares of our class A preferred stock and common stock that may be purchased or issued upon the exercise of options or awards that have been or may be granted under the 2006 Equity Incentive Plan and the Science Applications International Corporation 1999 Stock Incentive Plan; or may be issued and delivered to a trustee or agent for the benefit of employees under the Science Applications International Corporation Retirement Plan; the Science Applications International Corporation Stock Compensation Plan; the Science Applications International Corporation Key Executive Stock Deferral Plan; the 2006 Employee Stock Purchase Plan; and the AMSEC Employees 401(k) Profit Sharing Plan.

PART I

INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

The Registration Statement is being filed by SAIC, Inc. with respect to the Plans referred to on the cover page. The document(s) containing the plan information required in Part I of this Registration Statement will be provided to each participant in the Plans, as specified by Rule 428(b)(1) promulgated by the Securities and Exchange Commission (the "Commission") under the Securities Act of 1933, as amended (the "Securities Act"). Such documents are not being filed with the Commission but constitute (together with the documents incorporated by reference into this Registration Statement pursuant to Item 3 of Part II hereof) a prospectus that meets the requirements of Section 10(a) of the Securities Act.

PART II

INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

Item 3. Incorporation of Documents by Reference

The following documents filed with the Commission by SAIC, Inc. and Science Applications International Corporation are hereby incorporated by reference in this Registration Statement:

Science Applications International Corporation

- (a) Registrant's Prospectus dated October 12, 2006, filed with the Commission on October 13, 2006 pursuant to Rule 424(b) under the Securities Act, relating to Registrant's Registration Statement on Form S-1, as amended (Registration No. 333-128021), which contains audited financial statements of Science Applications International Corporation for the fiscal year ended January 31, 2006;
- (b) Quarterly Report on Form 10-Q for the quarter ended April 30, 2006, filed with the Commission on June 9, 2006;
- (c) Quarterly Report on Form 10-Q for the quarter ended July 31, 2006, filed with the Commission on September 5, 2006;
- (d) Current Reports on Form 8-K filed with the Commission on February 1, 2006, March 1, 2006, March 20, 2006, March 27, 2006, April 3, 2006, April 6, 2006, May 8, 2006, May 23, 2006, June 26, 2006, July 14, 2006, July 25, 2006, August 23, 2006, August 28, 2006, September 27, 2006, October 4, 2006, and October 13, 2006;
- (e) Annual Report on Form 11-K of the Science Applications International Corporation 401(k) Profit Sharing Plan (now part of the Science Applications International Corporation Retirement Plan) for the plan year ended December 31, 2005, filed with the Commission on June 27, 2006; and
- (f) Annual Report on Form 11-K of the AMSEC Employees 401(k) Profit Sharing Plan for the plan year ended December 31, 2005, filed with the Commission on June 27, 2006.

SAIC, Inc.

- (a) Prospectus dated October 12, 2006, filed with the Commission on October 13, 2006 pursuant to Rule 424(b) under the Securities Act, relating to Registrant's Registration Statement on Form S-1, as amended (Registration No. 333-128021), which contains audited balance sheet of Registrant for the fiscal year ended January 31, 2006;
- (b) Quarterly Report on Form 10-Q for the quarter ended April 30, 2006, filed with the Commission on June 9, 2006;
- (c) Quarterly Report on Form 10-Q for the quarter ended July 31, 2006, filed with the Commission on September 5, 2006;
- (d) Current Report on Form 8-K filed with the Commission on September 27, 2006, October 13, 2006 and October 17, 2006;
- (e) Description of the class A preferred stock, par value \$.0001 per share, and common stock, par value \$.0001 per share contained in Amendment No. 5 to Registration Statement on Form S-1 (Registration No. 333-128021), which was filed with the Commission on October 2, 2006.

All documents filed by us with the Commission pursuant to Section 13(a), 13(c), 14 or 15(d) of the Securities Exchange Act of 1934, as amended, subsequent to the date of this Registration Statement and prior to the filing of a post-effective amendment which indicates that all securities offered have been sold or which deregisters all securities then remaining unsold, shall be deemed to be incorporated by reference in this Registration Statement and to be a part hereof from the date of filing of such documents. Any statement contained in a document incorporated or deemed to be incorporated by reference herein shall be deemed to be modified or

superseded for purposes of this Registration Statement to the extent that a statement contained herein or in any other subsequently filed document which also is, or is deemed to be, incorporated by reference herein modifies or supersedes any such statement. Any such statement so modified or superseded will not be deemed, except as so modified or superseded, to constitute a part of this Registration Statement.

Item 4. Description of Securities

Not applicable.

Item 5. Interests of Named Experts and Counsel

The validity of the issuance of the shares of class A preferred stock and common stock offered hereby has been passed upon for us by Douglas E. Scott, Esq., Senior Vice President, General Counsel and Secretary of the Registrant. As of October 17, 2006, Mr. Scott owned of record 128,485 shares of class A preferred stock, had the right to acquire an additional 659,989 shares pursuant to previously granted stock options and beneficially owned a total of 48,492 shares through the retirement plan and key executive stock deferral plan of the Registrant and its subsidiary.

Item 6. Indemnification of Directors and Officers

Section 145 of the General Corporation Law of the State of Delaware permits a Delaware corporation to indemnify its directors, officers, employees and agents, subject to certain limitations.

As permitted by the General Corporation Law of the State of Delaware, our restated certificate of incorporation includes a provision that eliminates the personal liability of our directors for monetary damages for breach of fiduciary duty as a director, except for liability (1) for any breach of the director's duty of loyalty to us or our stockholders, (2) for acts or omissions not in good faith or that involve intentional misconduct or a knowing violation of law, (3) under Section 174 of the General Corporation Law of the State of Delaware or (4) for any transaction from which the director derived an improper personal benefit.

As permitted by the General Corporation Law of the State of Delaware, our restated certificate of incorporation provides that (1) we shall indemnify our directors and officers to the fullest extent permitted by the General Corporation Law of the State of Delaware, subject to certain limited exceptions, (2) we may indemnify our other employees and agents as set forth in the General Corporation Law of the State of Delaware, (3) we are required to advance expenses, as incurred, to our directors and executive officers in connection with a legal proceeding to the fullest extent not prohibited by applicable law, subject to the receipt by us of an undertaking to repay such amounts to the extent required by law and (4) the rights conferred in the restated certificate of incorporation are not exclusive.

We have entered into indemnification agreements with each of our directors and executive officers to give such directors and officers additional contractual assurances regarding the scope of the indemnification set forth in our restated certificate of incorporation and to provide additional procedural protections. We also intend to enter into indemnification agreements with any new directors, executive officers and board-appointed officers in the future.

The exculpation and indemnification provisions in our restated certificate of incorporation and the indemnification provisions of indemnification agreements that have been entered into between us and each of our directors and executive officers may be sufficiently broad to permit indemnification of our directors and executive officers for liabilities arising under the Securities Act of 1933.

We also intend to maintain director and officer liability insurance, if available on reasonable terms, to insure our directors and officers against the cost of defense, the cost of settlement or payment of a judgment under certain circumstances.

Item 7. Exemption From Registration Claimed

Not applicable.

Item 8. Exhibits

Exhibit No.Description of ExhibitIncorporated by Reference From4.1Restated Certificate of Incorporation of Registrant.Incorporated by reference to Exhibit 3.1 to Registrant's CurrentReport on Form 8-K as filed on October 17, 2006 with the SEC.

4.2	Restated Bylaws of Registrant.	Incorporated by reference to Exhibit 3.2 to Registrant's Current Report on Form 8-K as filed on October 17, 2006 with the SEC.
4.3	Registrant's 2006 Equity Incentive Plan.	Incorporated by reference to Exhibit 10.24 to Registrant's Post-Effective Amendment No. 3 to Form S-4 Registration Statement No. 333-128022, filed on July 25, 2006.
4.4	Registrant's 2006 Employee Stock Purchase Plan.	Incorporated by reference to Exhibit 10.25 to Registrant's Post-Effective Amendment No. 3 to Form S-4 Registration Statement No. 333-128022, filed on July 25, 2006.
4.5	Science Applications International Corporation 1999 Stock Incentive Plan, as amended through August 15, 1999.	Incorporated by reference to Exhibit 10(e) to Science Application International Corporation's Annual Report on Form 10-K for the fiscal year ended January 31, 2000.
4.6	Science Applications International Corporation Stock Compensation Plan, as amended and restated effective January 1, 2005.	Incorporated by reference to Exhibit 10.3 to Science Application International Corporation's Current Report on Form 8-K as filed on April 3, 2006 with the SEC.
4.7	Science Applications International Corporation Management Stock Compensation Plan, as amended and restated effective January 1, 2005.	Incorporated by reference to Exhibit 10.2 to Science Application International Corporation's Current Report on Form 8-K as filed on April 3, 2006 with the SEC.
4.8	Science Applications International Corporation Key Executive Stock Deferral Plan, as amended and restated effective January 1, 2005.	Incorporated by reference to Exhibit 10.1 to Science Application International Corporation's Current Report on Form 8-K as filed on March 1, 2006 with the SEC.
4.9*	Science Applications International Corporation Retirement Plan (as amended and restated effective January 1, 2006).	
4.10*	AMSEC Employees 401(k) Profit Sharing Plan.	
5.1*	Opinion of Douglas E. Scott, Esq.	
23.1	Consent of Douglas E. Scott, Esq. (contained in Exhibit 5.1 hereto).	
23.2*	Consent of Independent Registered Public Accounting Firm, Deloitte $\&$ Touche LLP.	
24	Power of Attorney (included on the signature page of this Registration Statement).	

The Science Applications International Corporation Retirement Plan and the AMSEC Employees 401(k) Profit Sharing Plan have been or will be submitted to the Internal Revenue Service ("IRS") in a timely manner and all changes required by the IRS in order to qualify these plans under Section 401 of the Internal Revenue Code have been or will be made.

Item 9. Undertakings

- (a) The undersigned Registrant hereby undertakes:
 - (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this Registration Statement:
 - (i) To include any prospectus required by section 10(a)(3) of the Securities Act of 1933;
 - (ii) To reflect in the prospectus any facts or events arising after the effective date of this Registration Statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the Registration Statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20% change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement;

(iii) To include any material information with respect to the plan of distribution not previously disclosed in the Registration Statement or any material change to such information in the Registration Statement;

Provided, however,

Paragraphs (a)(1)(i) and (a)(1)(ii) of this section do not apply if the Registration Statement is on Form S-8, and the information required to be included in a post-effective amendment by those paragraphs is contained in reports filed with or furnished to the Commission by the Registrant pursuant to section 13 or section 15(d) of the Securities Exchange Act of 1934 that are incorporated by reference in the Registration Statement.

- (2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (4) That, for the purpose of determining liability of the Registrant under the Securities Act of 1933 to any purchaser in the initial distribution of the securities:

The undersigned Registrant undertakes that in a primary offering of securities of the undersigned Registrant pursuant to this Registration Statement, regardless of the underwriting method used to sell the securities to the purchaser, if the securities are offered or sold to such purchaser by means of any of the following communications, the undersigned Registrant will be a seller to the purchaser and will be considered to offer or sell such securities to such purchaser:

- (i) Any preliminary prospectus or prospectus of the undersigned Registrant relating to the offering required to be filed pursuant to Rule 424;
- (ii) Any free writing prospectus relating to the offering prepared by or on behalf of the undersigned Registrant or used or referred to by the undersigned Registrant;
- (iii) The portion of any other free writing prospectus relating to the offering containing material information about the undersigned Registrant or its securities provided by or on behalf of the undersigned Registrant; and
 - (iv) Any other communication that is an offer in the offering made by the undersigned Registrant to the purchaser.
- (b) The undersigned Registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the Registrant's annual report pursuant to section 13(a) or section 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference in the Registration Statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (c) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the Registrant certifies that it has reasonable grounds to believe that it meets all the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of San Diego, State of California, on October 18, 2006.

SAIC, INC.

By /s/ K.C. Dahlberg
K.C. Dahlberg
Chairman of the Board and
Chief Executive Officer

POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENT, that each individual, including representatives of the various Plans, whose signature appears below constitutes and appoints D.E. Scott and M.W. Sopp power of substitution, for him or her and in his or her name, place and stead, in any and all capacities to sign any and all amendments (including post-effective amendments) to this registration statement, and new registration statements relating to this Form S-8, and to file the same with all exhibits thereto, and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorneys-infact and agents, and each of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection therewith, as fully to all intents and purposes as each of them might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents, or either of them, or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this Registration Statement on Form S-8 has been signed below by the following persons in the capacities and on the dates indicated.

Title	Date
Chairman of the Board and Chief Executive Officer	October 18, 2006
(Principal Executive Officer)	
	October 18, 2006
(Principal Financial Officer)	
Controller	October 18, 2006
	October 10, 2000
(Frincipal Accounting Officer)	
Director	October 18, 2006
-	
_ Director	October 18, 2006
Director	October 18, 2006
- Director	October 10, 2000
Director	October 18, 2006
-	
	Chairman of the Board and Chief Executive Officer (Principal Executive Officer) Chief Financial Officer (Principal Financial Officer) Controller (Principal Accounting Officer) Director Director

Signature	Title	Date
/s/ A.K. Jones	Director	October 18, 2006
A.K. Jones		
/s/ H.M.J. Kraemer, Jr.	Director	October 18, 2006
H.M.J. Kraemer, Jr.		
/s/ E.J. Sanderson, Jr.	Director	October 18, 2006
E.J. Sanderson, Jr.		
/s/ L.A. Simpson	Director	October 18, 2006
L.A. Simpson	-	
/s/ J.P. Walkush	Director	October 18, 2006
J.P. Walkush	-	
/s/ A.T. Young	Director	October 18, 2006
A.T. Young		2 223001 20, 2000

Pursuant to the requirements of the Securities Act of 1933, the Committee for the Science Applications International Corporation Retirement Plan has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of San Diego, State of California on October 18, 2006.

SCIENCE APPLICATIONS INTERNATIONAL CORPORATION RETIREMENT PLAN

By /s/ Lucy K. Moffitt

Lucy K. Moffitt, Member Retirement Plans Committee

Pursuant to the requirements of the Securities Act of 1933, the Committee for the AMSEC Employees 401(k) Profit Sharing Plan has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Virginia Beach, State of Virginia on October 18, 2006.

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

By /s/ Joanne T. Brooks

Joanne T. Brooks, Chair AMSEC Retirement Plan Committee

EXHIBIT INDEX

Exhibit No. 4.1	Description of Exhibit Restated Certificate of Incorporation of Registrant.
4.2	Restated Bylaws of Registrant.
4.3	Registrant's 2006 Equity Incentive Plan.
4.4	Registrant's 2006 Employee Stock Purchase Plan.
4.5	Science Applications International Corporation 1999 Stock Incentive Plan, as amended through August 15, 1999.
4.6	Science Applications International Corporation Stock Compensation Plan, as amended and restated effective January 1, 2005.
4.7	Science Applications International Corporation Management Stock Compensation Plan, as amended and restated effective January 1, 2005.
4.8	Science Applications International Corporation Key Executive Stock Deferral Plan, as amended and restated effective January 1, 2005.
4.9*	Science Applications International Corporation Retirement Plan (as amended and restated effective January 1, 2006).
4.10*	AMSEC Employees 401(k) Profit Sharing Plan.
5.1*	Opinion of Douglas E. Scott, Esq.
23.1	Consent of Douglas E. Scott, Esq. (contained in Exhibit 5.1 hereto).
23.2*	Consent of Independent Registered Public Accounting Firm, Deloitte & Touche LLP.
24	Power of Attorney (included on the signature page of this Registration Statement).

^{*} Filed herewith.

Incorporated by Reference From

Incorporated by reference to Exhibit 4.1 to Registrant's Current Report on Form 8-K as filed on October 17, 2006 with the SEC.

Incorporated by reference to Exhibit 4.2 to Registrant's Current Report on Form 8-K as filed on October 17, 2006 with the SEC.

Incorporated by reference to Exhibit 10.24 to Registrant's Post-Effective Amendment No. 3 to Form S-4 Registration Statement No. 333-128022, filed on July 25, 2006.

Incorporated by reference to Exhibit 10.25 to Registrant's Post-Effective Amendment No. 3 to Form S-4 Registration Statement No. 333-128022, filed on July 25, 2006.

Incorporated by reference to Exhibit 10(e) to Science Application International Corporation's Annual Report on Form 10-K for the fiscal year ended January 31, 2000.

Incorporated by reference to Exhibit 10.3 to Science Application International Corporation's Current Report on Form 8-K as filed on April 3, 2006 with the SEC.

Incorporated by reference to Exhibit 10.2 to Science Application International Corporation's Current Report on Form 8-K as filed on April 3, 2006 with the SEC.

Incorporated by reference to Exhibit 10.1 to Science Application International Corporation's Current Report on Form 8-K as filed on March 1, 2006 with the SEC.

SCIENCE APPLICATIONS INTERNATIONAL CORPORATION

RETIREMENT PLAN

Amended and Restated January 1, 2006

TABLE OF CONTENTS

ARTIC	TEI	NAME EFFECTIVE DATE AND PURPOSES	Page 1
intic		Name and Effective Date	1
			1
ARTIC		Plan Purposes ELIGIBILITY AND PARTICIPATION	1
AKIIC			2
		Eligibility to Participate	2
		Commencement of Participation	2
ARTIC	CLE III	CONTRIBUTIONS	3
	3.1	Elective Deferrals	3
	3.2	Voluntary Contributions	6
	3.3	Company Matching Contributions	6
	3.4	Profit Sharing and ESOP Contributions	9
	3.5	Other Provisions Regarding Company Contributions	10
	3.6	Rollover Contributions	10
	3.7	Plan-to-Plan Transfers	10
	3.8	Investment of Accounts	11
	3.9	Valuation of Accounts	12
ARTIC	CLE IV	LIMITATION ON ANNUAL ADDITIONS	15
	4.1	Definitions	15
	4.2	Section 415 Limitation	15
	4.3	Annual Addition Limitations	15
	4.4	No Contractual Right to Excess Contributions	16
ARTIC	CLE V	VESTING	17
	5.1	Fully Vested Accounts	17
	5.2	Employer Matching Contributions, Profit Sharing Account and ESOP Account	17
	5.3	Alternate Payee Accounts	20
ARTIC	CLE VI	DISTRIBUTIONS	21
	6.1	Distribution Upon Termination of Employment	21
	6.2	Designation of Beneficiary; Distribution Upon Death or Disability	22

24

6.3

In-Service Withdrawals

TABLE OF CONTENTS (continued)

6.4	Distribution of MAPP Subaccount	<u>Page</u> 26
6.5	Special Rules Relating to Distributions	28
6.6	Direct Rollovers	31
ARTICLE V	TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS	33
7.1	Trust Fund	33
7.2	Permissible Types of Plan Investments	33
7.3	Form of Company Contributions	34
7.4	Valuation of Company Contributions in the Form of Company Stock	34
7.5	Company, Committee and Trustee Not Responsible for Adequacy of Trust Fund	34
7.6	Company Stock Transactions with Disqualified Persons	34
ARTICLE V	III ACCOUNTING AND ALLOCATION PROCEDURES	35
8.1	Allocation of Forfeitures	35
8.2	Reinstatement of Forfeitures	36
8.3	Allocation of Income or Loss on Unallocated Company Contributions; Company Contributions to Defray Plan Expenses	37
8.4	Accounting Procedures	37
8.5	Suspended Participants	37
8.6	Accounting for Interest of an Alternate Payee	37
ARTICLE I	OPERATION AND ADMINISTRATION OF THE PLAN; VOTING AND OTHER RIGHTS OF COMPANY STOCK	38
9.1	Members	38
9.2	Committee Action	38
9.3	Rights and Duties	38
9.4	Compensation Bonding Indemnity and Liability	42
9.5	Manner of Administration	43
9.6	Duty of Care	43
9.7	Reliance Upon Documents and Opinions	43
9.8	Requirement of Proof	43

TABLE OF CONTENTS (continued)

	9.9	Voting and Other Rights of Company Stock	<u>Page</u> 44
		Plan Expenses	45
		Multiple Fiduciary Capacity	46
		Allocation of Fiduciary Responsibility	46
		Section 404(c) Provisions	46
ARTIC		PLAN AMENDMENT AND TERMINATION	48
		Amendments	48
		Retroactive Amendments	48
	10.3	Discontinuance of Plan	48
	10.4	Failure to Contribute	49
	10.5	Merger or Consolidation	49
		Procedures on Termination	49
	10.7	Partial Termination	50
ARTIC	CLE XI	MISCELLANEOUS	51
	11.1	Contributions Not Recoverable	51
	11.2	Limitation on Participant's Rights	51
	11.3	Receipt or Release	51
	11.4	Alienation	51
	11.5	Nonconforming Distributions Under Court Order	52
	11.6	Authorized Participant Loans	52
	11.7	Lapsed Benefits	54
	11.8	Addresses	55
	11.9	Notices and Communications	55
	11.10	Governing Law	55
	11.11	Interpretation	56
	11.12	Withholding For Taxes	56
	11.13	Limitation on Company, Committee and Trustee Liability	56
	11.14	Successors and Assigns	56
	11.15	Counterparts	56
	11.16	Top-Heavy Plan Requirements	56

TABLE OF CONTENTS

(continued)

	Page
11.17 Rule 16b-3 Provisions	56
11.18 Military Service	56
11.19 Errors, Misstatements, and Payment of Interest	56
ARTICLE XII ESOP FEATURE	57
12.1 ESOP Contributions	57
12.2 Diversification Election	57
12.3 Dividends on Allocated Shares	57
12.4 Valuation of Stock	58
12.5 Tender for Stock	58
12.6 Voting of Stock	58
12.7 Put Right	58
ARTICLE XIII DEFINITIONS	60
13.1 Definitions	60

SCIENCE APPLICATIONS INTERNATIONAL CORPORATION Retirement Plan

ARTICLE I

NAME, EFFECTIVE DATE AND PURPOSES

1.1 Name and Effective Date. Science Applications International Corporation (the "Company") adopted the Science Applications International Corporation Profit Sharing Retirement Plan on February 1, 1970 (the "PSRP"). Effective January 1, 1995, the PSRP was merged with the Science Applications International Corporation Profit Sharing Plan II, which was originally adopted as the SAIC COMSYSTEMS Division Money Accumulation Pension Plan on February 1, 1977 and was subsequently merged with the TSC Profit Sharing Retirement Plan effective January 1, 1992. The PSRP was amended and restated in its entirety most recently effective January 1, 2001 and has been amended subsequently in several respects.

The Company adopted the Science Applications International Corporation Cash or Deferred Arrangement (the "CODA") effective January 1, 1983. The CODA was amended and restated in its entirety most recently effective as of January 1, 2001 and has been subsequently amended in several respects.

Science Applications, Inc. established the Science Applications, Inc. Stock Bonus Retirement Plan effective February 1, 1973 (the "Predecessor Plan"). In connection with the reorganization between Science Applications, Inc. and the Company effected October 1, 1981, the Company adopted the Predecessor Plan and the name of Predecessor Plan was changed to the Science Applications International Corporation Stock Bonus Retirement Plan. The Predecessor Plan was amended and restated in its entirety effective as of January 1, 1985, and has been amended in several respects and has undergone restatements and amendments subsequent to that date, most recently effective as of January 1, 2001. The resulting plan is now known as the Science Applications International Corporation Employee Stock Retirement Plan (the "ESRP").

Effective as of November 28, 2003, the PSRP was merged with the CODA, and the merged plan was renamed the Science Applications International Corporation 401(k) Profit Sharing Plan (this "Plan"). In addition, effective as of January 1, 2006, (i) the ESRP is merged with and into this Plan, and (ii) an additional component ("ESOP Feature") is added to this Plan. In connection with such merger, this Plan is hereby renamed the Science Applications International Corporation Retirement Plan. This Plan is hereby amended and restated in its entirety, effective as of January 1, 2006, except as may otherwise be provided herein.

1.2 <u>Plan Purposes</u>. This Plan is designed to foster and expand employee ownership of Company Stock through a tax-qualified profit sharing plan for purposes of Sections 401(a) and 501(a) of the Code and provides a vehicle whereby Eligible Employees may elect to defer the receipt of some portion of their income and to have those amounts contributed to the Plan under a "cash or deferred arrangement" within the meaning of Section 401(k) of the Code. In addition, the ESOP Feature of the Plan consists of both a stock bonus plan and an employee stock ownership plan intended to qualify under Sections 401(a) and 4975(e)(7) of the Code, and is designed as such to invest primarily in Company Stock.

ARTICLE II

ELIGIBILITY AND PARTICIPATION

2.1 Eligibility to Participate.

- (a) An Eligible Employee shall become eligible to participate in the Plan with respect to Elective Deferrals, Rollover Contributions and Company Matching Contributions upon his Employment Commencement Date. Eligible Employees who had not previously become eligible to participate with respect to the Plan features listed in this subsection (a) shall become eligible as of January 1, 2005.
- (b) An Eligible Employee shall become eligible to participate in the Plan with respect to Profit Sharing Contribution and ESOP Contribution on the date on which the Eligible Employee completes a Year of Service, provided that the Eligible Employee is an Eligible Employee on such date.

2.2 Commencement of Participation.

- (a) (1) An Eligible Employee shall commence participation with respect to Elective Deferrals and Company Matching Contributions for the payroll period commencing as soon as administratively feasible following his Employment Commencement Date (for Eligible Employees not previously eligible for such features, on January 1, 2005, if later), and with respect to Rollover Contributions as soon as administratively feasible following his Employment Commencement Date.
- (2) An Eligible Employee shall commence participation with respect to Profit Sharing Contribution and ESOP Contribution on the Entry Date coinciding with or next following the completion of the eligibility requirement set forth in Section 2.1(b).
- (b) A Participant who incurs a Break in Service and is later reemployed as an Eligible Employee, and a Suspended Participant who again becomes an Eligible Employee, shall resume participation immediately upon becoming an Eligible Employee, with Elective Deferrals commencing for the payroll period commencing as soon as administratively feasible following the resumption of service as an Eligible Employee.
- (c) In the event that service with a predecessor employer has been approved in accordance with Section (a)(v) of the definition of Hours of Service, the commencement of participation for Eligible Employees for whom such service has been approved shall be the payroll period commencing as soon as administratively feasible following such approval or, if later, and only with respect to Profit Sharing Contribution and ESOP Contribution, the Entry Date following the completion of a Year of Service in accordance with Section 2.1(b) above.

ARTICLE III

CONTRIBUTIONS

3.1 Elective Deferrals.

- (a) <u>Election to Defer</u>. Subject to the limitations of this Article III and Article IV, each Participant may elect to defer up to twenty-five percent (25%) (effective January 1, 2006, 75%) of his Compensation on a pre-tax basis for a Plan Year. The deferred amount shall be contributed directly by the Company to the Trust Fund as Elective Deferrals and, if applicable, Catch-Up Contributions. For purposes of this Section 3.1, the Participant's Compensation shall be determined prior to the reduction caused by the deferral made pursuant to this Section 3.1.
- (b) <u>Termination of, Change in Rate of, or Resumption of Deferrals</u>. A Participant's Elective Deferrals election shall remain in effect, notwithstanding any change in Compensation until the Participant elects to change the election. A Participant may at any time submit a request to the Committee to terminate, alter the rate of, or resume his Elective Deferrals.
- (c) <u>Contribution and Allocation of Elective Deferrals</u>. The Company shall reduce the Participant's Compensation in the amount of the Participant's Elective Deferrals and shall make a contribution to the Trustee equal to such reduction as soon as practicable after such amounts are withheld from the Participant's Compensation, but in any event within time prescribed under applicable regulations. Such contributions shall be allocated to Participants' Plan Accounts.

(d) 402(g) Limit on Elective Deferrals.

- (i) Elective Deferrals made on behalf of any Participant under this Plan and all other plans maintained by the Company or an Affiliated Company shall not exceed the limitation under Section 402(g)(1) of the Code for the taxable year of the Participant, as adjusted annually under Section 402(g)(4) of the Code.
- (ii) In the event that the dollar limitation provided for in subsection (d)(i) is exceeded, such Excess Deferrals shall be treated, to the extent permissible, as Catch-Up Contributions pursuant to Section 414(v) of the Code. If not so treated, the Participant is deemed to have requested a distribution of such Excess Deferrals by the first March 1 following the close of the Participant's taxable year. The Committee shall distribute such Excess Deferrals, and any income allocable to such amount, to the Participant by the first April 15 thereafter. In determining the Excess Deferrals distributable with respect to a Participant's taxable year, Excess Deferrals previously distributed or treated as Catch-Up Contributions for the Plan Year beginning with or within such taxable year shall reduce the amount otherwise distributable under this paragraph (ii).
- (iii) In the event that a Participant is also a participant in (i) another qualified cash or deferral arrangement as defined in Section 401(k) of the Code, (ii) a simplified

employee pension, as defined in Section 408(k) of the Code, or (iii) a salary reduction arrangement, within the meaning of Section 3121(a)(5)(D) of the Code, and the elective deferrals, as defined in Section 402(g)(3) of the Code, made under such other arrangement(s) and this Plan cumulatively exceed the dollar limit under this subsection (d) for such Participant's taxable year, the Participant may, not later than March 1 following the close of his taxable year, notify the Committee in writing of such Excess Deferrals and request that the Elective Deferrals made on his behalf under this Plan be reduced by an amount specified by the Participant. The Committee shall then distribute such Excess Deferrals in the same manner as provided in subsection (d)(ii).

(e) Section 401(k) Limitations on Elective Deferrals.

- (i) The Committee may estimate at such times as the Committee in its discretion determines, the extent, if any, to which the deferral of taxation on Elective Deferrals under Section 401(k) of the Code may not be available to any Participant or class of Participants. In accordance with any such estimate, the Committee may modify the limits in Section 3.1 (a) or set initial or interim limits for Elective Deferrals relating to any Participant or class of Participants. These rules may include provisions authorizing the suspension or reduction of Elective Deferrals above a specified dollar amount or percentage of Compensation, or any other means adopted by the Committee.
- (ii) For each Plan Year, an actual deferral percentage will be determined for each Employee meeting the requirements of Section 2.1. The actual deferred percentage shall equal the ratio of the total amount of Elective Deferrals, Qualified Matching Contributions and Additional Company Contributions made on behalf of or allocated to the Employee for the Plan Year divided by the Employee's Compensation for the Plan Year. In the case of any Participant who is a Highly Compensated Employee and who is eligible to participate in one or more other defined contribution plans of the Company or an Affiliated Company containing a cash or deferred arrangement described in Section 401(k) of the Code, the actual deferral percentage shall be determined by treating this Plan and all such other plans (other than those which may not be aggregated under Treasury Regulations) as a single plan. With respect to Employees who have met the requirements of Section 2.1, and who have made no Elective Deferrals under the Plan, such actual deferral percentage will be zero.
- (iii) The average of the actual deferral percentages for Highly Compensated Employees in the Plan Year (the "High Average") when compared with the average of the actual deferral percentages for Non-Highly Compensated Employees in the applicable year (the "Low Average") must meet one of the following requirements:
 - (A) The High Average is no greater than 1.25 times the Low Average; or
 - (B) The High Average is no greater than two times the Low Average, and the High Average is no greater than the Low Average plus two percentage points.

- (C) The "applicable year" under the prior year testing method is the Plan Year immediately before the Plan Year being tested. The "applicable year" under the current year testing method is the Plan Year being tested. The Plan shall use the current year testing method for the 1997 testing year and subsequent years unless and until changed by the Committee to the prior year method consistent with procedures prescribed by the Internal Revenue Service pursuant to Code Sections 401(k)(3)(A) and 401(m)(2)(A).
- (iv) If, at the end of the Plan Year, a Participant or class of Participants has excess Elective Deferrals, then the Committee may elect, at its discretion, to pursue either of the following courses of action or any combination thereof:
 - (A) Excess Elective Deferrals, and any earnings attributable thereunder, may be distributed to Participant(s) no later than 12 months following the close of the Plan Year to which the excess Elective Deferrals relate. Any such excess Elective Deferrals distributed from the Plan with respect to a Participant for a Plan Year shall be reduced by any amount previously distributed to such Participant under Section 3.1(d) for the Participant's taxable year ending with or within such Plan Year. The amount of any Participant's Excess Deferrals is determined under subsection (e)(v).
 - (B) The Company, in its discretion, may make an Additional Company Contribution to the Plan, which will be allocated as of a date within the applicable year among the Accounts of Non-Highly Compensated Employees who have met the requirements of Section 2.1, or may make a Qualified Matching Contribution to Non-Highly Compensated Participants who made Elective Deferrals. Such contributions shall be fully (100%) vested at all times and shall be subject to the withdrawal restrictions which are applicable to Elective Deferrals. Such contributions shall be considered "qualified non-elective contributions" or "qualified matching contributions" under applicable Treasury Regulations.
- (v) Excess Elective Deferrals shall be determined by the Committee in accordance with this Section 3.1(e)(v). The Committee shall calculate a tentative reduction amount to the Elective Deferrals of the Highly Compensated Employee(s) with the highest actual deferral percentage equal to the amount which, if they were actually reduced, would enable the Plan to meet the limits in Section 3.1(e)(iii) or to cause the actual deferral percentage of such Highly Compensated Employee(s) to equal the actual deferral percentage of the Highly Compensated Employee(s) with the next-highest actual deferral percentage. The process shall be repeated until the limits in Section 3.1(e)(iii) above are satisfied. The aggregate amount of the tentative reduction amounts in the preceding sentence shall constitute "Refundable Contributions." The entire aggregate amount of Refundable Contributions shall be refunded to Highly Compensated Employees. The amount to be refunded to each Highly Compensated Employee (which shall constitute his Excess Elective Deferrals) shall be determined as follows: (A) the Elective Deferrals of the Highly Compensated Employee(s) with the highest dollar amount of Elective Deferrals of such Highly Compensated Employee(s) to equal the dollar amount of Elective Deferrals of the Highly Compensated Employee(s) with the next-highest Elective Deferrals, and (B) the process in the foregoing clause shall be repeated until the total amount of Elective Deferrals refunded equals the

total amount of Refundable Contributions. The earnings attributable to excess Elective Deferrals will be determined in accordance with Treasury Regulations. The Committee will not be liable to any Participant (or his Beneficiary, if applicable) for any losses caused by inaccurately estimating or calculating the amount of any Participant's excess Elective Deferrals and earnings attributable to Elective Deferrals.

- (vi) In the event that the Committee determines that an amount to be deferred pursuant to the election provided in Section 3.1 would cause the Company Contributions under this and any other tax-qualified retirement plan maintained by the Company or any Affiliated Company to exceed the applicable deduction limitations contained in Code Section 404, or to exceed the maximum Annual Addition determined in accordance with Article IV, the Committee may treat such amount in accordance with the rules set forth above in Section 3.1(e)(i) above.
- (vii) In the discretion of the Committee, the tests described in this Section may be applied by aggregating the Plan with any other defined contribution plans permitted under the Code. For purposes of determining whether the Plan satisfies the requirements of this Section 3.1(e), all Elective Deferrals and elective contributions made under any other plan maintained by the Company which is aggregated with the Plan for purposes of Section 401 (a) or 410(b) of the Code (other than Section 410(b)(2)(A)(ii)) are to be treated as made under a single plan. Furthermore, if two or more plans are permissively aggregated for purposes of the test described in this subsection (e), the aggregated plans must also satisfy Sections 401(a)(4) and 410(b) of the Code as though they were a single plan.
- **3.2** <u>Voluntary Contributions</u>. No Voluntary Contributions are permitted after December 31, 1986. Voluntary Contributions made prior to such date shall be maintained in a Voluntary Account as set forth herein.

3.3 Company Matching Contributions.

- (a) <u>Amount of Company Matching Contributions</u>. Subject to the limitations of this Section 3.3 and Article IV, for each Plan Year the Company may, in its sole discretion, make a Company Matching Contribution. The amount, if any, of the Company Matching Contribution for each Fringe Rate Group shall be determined by the Company in its sole discretion. The amounts or percentages of Company Matching Contributions among Fringe Rate Groups may differ or may be zero. However, in no event shall the amount of a Company Matching Contribution for any Plan Year result in the allocation to any Participant's Account in excess of fifty percent (50%) of that portion of such Participant's Elective Deferrals which do not exceed six percent (6%) of such Participant's Match Eligible Compensation for the applicable Plan Year. Thus, with respect to a Participant, the Company Matching Contribution shall not exceed three percent (3%) of such Participant's Match Eligible Compensation. For avoidance of doubt, the Company shall not make Company Matching Contributions with respect to amounts treated as Catch-Up Contributions.
- (b) <u>Contribution and Allocation of Company Matching Contributions</u>. The Company shall pay to the Trustee the Company Matching Contributions for any Plan Year no

later than the due date for filing the federal income tax return (including extensions) of the Company for the taxable year with respect to which the contribution is made. The Company Matching Contributions with respect to each Fringe Rate Group for any Plan Year shall be allocated to the Company Matching Account maintained for the Participant on behalf of whom the contribution under Section 3.3(a) was made. For avoidance of doubt, the Company Matching Contribution for each Fringe Rate Group shall be separately allocated only to the Participants employed in the Fringe Rate Group for which such contribution was made.

(c) Section 40K(m) Limitations on Company Matching Contributions.

- (i) The Committee may estimate, at such times as the Committee in its discretion determines, the extent, if any, to which Company Matching Contributions may not be available to any Participant or class of Participants as a result of the limitations of Code Section 401(m). In accordance with any such estimate, the Committee may modify the percentages in Section 3.3(a) or set initial or interim limits or percentages for Company Matching Contributions relating to any Participant or class of Participants. After determining the amount of excess Elective Deferrals, if any, under Section 3.1(e), the Committee shall determine the aggregate contribution percentage under (ii) below.
- (ii) For each Plan Year, a contribution percentage will be determined for each Employee meeting the requirements of Section 2.1. The contribution percentage shall equal the ratio of the total amount of Company Matching Contributions made on behalf of or allocated to the Employee for the Plan Year divided by the Employee's Compensation for the Plan Year. In the case of any Participant who is a Highly Compensated Employee and who is eligible to participate in one or more other defined contributions plans of the Company or an Affiliated Company to which Company Matching Contributions are made, the contribution percentage shall be determined by treating this Plan and such other plans (other than those which may not be aggregated under Treasury Regulations) as a single plan. With respect to Employees who have met the requirements of Section 2.1 and for whom there were no Company Matching Contributions under the Plan, such contribution percentage will be zero.
- (iii) The average of the contribution percentages for Highly Compensated Employees in the Plan Year (the "High Average") when compared with the average of the contribution percentages for Non-Highly Compensated Employees in the applicable Year (the "Low Average") must meet one of the following requirements:
 - (A) The High Average is no greater than 1.25 times the Low Average; or
 - (B) The High Average is no greater than two times the Low Average, and the High Average is no greater than the Low Average plus two percentage points.
 - (C) The "applicable year" under the prior year testing method is the Plan Year immediately before the Plan Year being tested. The "applicable year" under the current year testing method is the Plan Year being tested. The Plan shall use the current year testing method for the 1997 testing year and subsequent years unless and until changed by the

Committee to the prior year method consistent with procedures prescribed by the Internal Revenue Service pursuant to Code Sections 401(k) (3)(A) and 401(m)(2)(A).

- (iv) If, at the end of the Plan Year, a Participant or class of Participants has excess contributions, then the Committee may elect, at its discretion, to pursue any of the following courses of action or any combination thereof:
 - (A) Excess Company Matching Contributions (and any earnings attributable thereto through the end of the Plan Year) attributable to Excess Elective Deferrals may be forfeited.
 - (B) Excess Company Matching Contributions (and any earnings attributable thereto through the end of the Plan Year) that are not vested may be forfeited.
 - (C) Excess Company Matching Contributions (and any earnings attributable thereto through the end of the Plan Year) may be distributed to the Participant no later than 12 months after the close of the Plan Year. The Committee may distribute (or forfeit pursuant to paragraph (B) above) Company Matching Contributions prior to distributing any Company Matching Contributions.
 - (D) Notwithstanding the foregoing, the conditions in this paragraph (D) must be met if there are Company Matching Contributions allocated to a Participant which are attributable to Excess Elective Deferrals. In such case, such Company Matching Contributions shall not be allocated to the Account of any Participant who had excess Company Matching Contributions in such Plan Year. In addition, Company Matching Contributions remaining in the Plan allocated to the Participant after satisfying this Section 3.3(c) cannot exceed the amount which may be allocated under Section 3.1 when taking into account only those Elective Deferrals remaining in the Plan after satisfying Section 3.1(d), 3.1(e) and this Section 3.3(c). Any such excess Company Matching Contributions (and earnings attributable thereto) must be forfeited or distributed pursuant to paragraphs (A), (B) or (C) immediately above.
- (v) Excess Company Matching Contributions shall be determined by the Committee in accordance with this Section 3.3(c)(v). The Committee shall calculate a tentative reduction amount to the Company Matching Contributions made with respect to the Highly Compensated Employee(s) with the highest contribution percentage equal to the amount which, if they were actually reduced, would enable the Plan to meet the limits in Section 3.3(c)(iii) above or to cause the contribution percentage of such Highly Compensated Employee(s) to equal the actual contribution percentage of the Highly Compensated Employee(s) with the next-highest actual contribution percentage. The process shall be repeated until the limits in Section 3.3(c)(iii) above are satisfied. The aggregate amount of the tentative reduction amounts in the preceding sentence shall constitute "Refundable Matching Contributions." The entire aggregate amount of Refundable Matching Contributions shall be refunded to Highly Compensated Employees. The amount to be refunded to each Highly Compensated Employee (which shall constitute his excess Company Matching Contributions) shall be determined as

follows: (A) the Company Matching Contributions made with respect to the Highly Compensated Employee(s) with the highest dollar amount of aggregate Company Matching Contributions shall be refunded to the extent that there are Refundable Matching Contributions or to the extent necessary to cause the dollar amount of Company Matching Contributions of such Highly Compensated Employee(s) to equal the dollar amount of Company Matching Contributions made with respect to the Highly Compensated Employee(s) with the next-highest aggregate Company Matching Contributions, and (B) the process in the foregoing clause shall be repeated until the total amount of Company Matching Contributions refunded equals the total amount of Refundable Matching Contributions. The earning attributable to excess Company Matching Contributions will be determined in accordance with Treasury Regulations. The Committee will not be liable to any Participant (or his Beneficiary, if applicable) for any losses caused by inaccurately estimating or calculating the amount of any Participant's excess Company Matching Contributions and earnings attributable to such Company Matching Contributions.

(d) <u>Forfeiture of Company Matching Contributions Attributable to Excess Deferrals</u>. Company Matching Contributions attributable to Elective Deferrals that exceed the limits of Section 3.1(d) or Section 3.1(e) shall be forfeited and allocated as set forth in Section 8.1.

3.4 Profit Sharing and ESOP Contributions.

- (a) For each Plan Year the Company may, in its sole discretion, contribute to the Trust Fund an amount or amounts to be determined by the Board of Directors in its discretion as a Profit Sharing Contribution and/or an ESOP Contribution. The amount, if any, of the Profit Sharing Contribution and ESOP Contribution for each Fringe Rate Group, shall be determined by the Company in its sole discretion. Such discretion may result in different amounts or percentages (including zero) of Profit Sharing Contributions and/or ESOP Contributions among Fringe Rate Groups. The Company's Profit Sharing Contribution and ESOP Contribution to the Trust Fund shall be made no later than the due date for filing the federal income tax return (including extensions) of the Company for the taxable year with respect to which the contribution is made. Any Profit Sharing Contributions and ESOP Contributions shall be subject to the requirements and restrictions of this Section 3.4 and subject also to the right of the Company to amend or terminate this Plan or to suspend or discontinue Profit Sharing Contributions and ESOP Contributions to this Plan.
- (b) All Profit Sharing Contributions and ESOP Contributions to the Trust Fund shall be held on an unallocated basis until allocated, as applicable, to Participants' Profit Sharing Accounts and ESOP Accounts as provided under this Plan or otherwise used or applied in accordance with the provisions of this Plan. Pending such allocation, Profit Sharing Contributions and ESOP Contributions shall be invested as directed, or under rules prescribed by the Committee. All gains or losses on such investments shall be allocated as provided in Section 8.3 and may be used for the payment of Plan expenses.
- (c) Profit Sharing Contributions and ESOP Contributions with respect to each Fringe Rate Group for any particular Plan Year (unadjusted for income, gain and loss, which shall be allocated separately pursuant to Section 8.3) shall be separately allocated only to the

Participants (A) employed in the Fringe Rate Group for which such contribution was made, (B) who completed 850 or more Hours of Service during the Plan Year and (C) who either are employed by the Company or an Affiliated Company on the last working day of the Plan Year or whose employment terminated during the Plan Year as a result of death, retirement on or after the Normal Retirement Date, Disability or involuntary lay-off (other than for cause, as determined by the Committee in its sole discretion) ("Eligible Participants"). The Profit Sharing Contribution and ESOP Contributions for each Fringe Rate Group shall be allocated, as applicable to the Profit Sharing Account or ESOP Account of each Eligible Participant in that Fringe Rate Group in the ratio that each such Eligible Participant's total Compensation for the Plan Year while a member of such Fringe Rate Group bears to all such Eligible Participants' total Compensation for such Plan Year while members of such Fringe Rate Group.

3.5 Other Provisions Regarding Company Contributions.

- (a) Expenses. The Company may, in its discretion, make contributions at any time to defray Plans expenses as provided in Section 8.3.
- (b) <u>Corrective Contributions</u>. If an error has been made in calculating the amount of any required Company Contributions or because of any other error there was a mistake which resulted in the Company contributing the incorrect amount of contributions for a particular Plan Year or Plan Years, the Company may adjust the amount of its contributions to the extent necessary to correct its mistake. The Company may make these corrections prospectively or retrospectively in its discretion.

3.6 Rollover Contributions.

An Eligible Employee may transfer to the Trust Fund a Rollover Contribution pursuant to procedures approved by the Committee. The Committee shall develop such procedures and may require such information as it deems necessary or desirable to determine that the proposed transfer will meet the requirements of a Rollover Contribution. Upon approval by the Committee, the amount transferred shall be deposited in the Trust Fund and shall be credited to a Rollover Account. Such Account shall be one hundred percent (100%) vested in the Employee, shall be valued in accordance with Section 3.9, but shall not share in allocation of Company Matching Contributions, Qualified Matching Contributions, forfeitures or Profit Sharing Contribution or ESOP Contribution. Upon termination of employment, the total amount of the Rollover Account shall be treated as part of the Employee's "Distributable Benefit" under Section 6.5 and distributed in accordance with Sections 6.1 through 6.4, as applicable. For avoidance of doubt, a Rollover Contribution shall not include the portion of any distribution that otherwise would not be includable in gross income.

3.7 Plan-to-Plan Transfers.

(a) In its sole discretion, the Committee may authorize the Plan to accept a transfer, from a plan qualified under Code Section 401(a), of a Participant's Account balance under such transferee plan. Any such transfer shall be allocated to a specified Account of the Participant as determined by the Committee.

(b) To the extent required by Code Section 411(d)(6), an Account balance transferred pursuant to this Section 3.7 shall retain optional forms of benefit payments that were applicable to such Account balance prior to the transfer to the Plan.

3.8 Investment of Accounts.

(a) Investment Direction by Participants.

- (i) All contributions to the Plan (whether Company Contributions, Voluntary Contributions, Rollover Contributions or plan-to-plan transfers) shall be invested as provided in this Section 3.8. The Committee shall establish rules setting forth the extent to which Participants may determine the manner in which their Accounts may be invested. Consistent with such rules, the Committee may establish a choice of investment alternatives from which each Participant may select in determining the manner in which the applicable portions of his Plan Account, Profit Sharing Account (if any), Voluntary Contributions (if any), Matching Contribution Account (if any) and Rollover Account (if any) will be invested. The Committee shall prescribe procedures for investment of amounts allocated to an Alternate Payee Account. To the extent permitted under rules established by the Committee, Company Contributions may be directed all or in part to be invested in a Company Stock Fund within the Trust; provided, however, that fifty percent (50%) of a Participant's Company Matching Contributions shall be invested in a Company Stock Fund.
- (ii) The Plan is designed to provide benefits largely through the long-term investment of Participant Accounts. It is not designed to serve as a vehicle for speculative investment, such as short-swing securities trades. Plan fiduciaries shall execute Participant-directed investment transactions, fund transfers, distributions and related transactions within a reasonable period of time after receiving proper instructions to do so, but neither the Plan nor Plan fiduciaries guarantee how promptly execution will commence or be completed.
- (iii) If such investment alternatives are established, each Participant may elect to invest the assets of his Accounts in such alternatives at such time, in such manner, and subject to such restrictions as the Committee shall specify.
- (iv) Separate funds within the Trust Fund shall be established to reflect the available alternatives, and separate subaccounts shall be established for each investment alternative selected by a Participant, and each such subaccount shall be valued separately.
- (v) The Committee, in its discretion, may permit Participants to transfer amounts from one investment alternative to one or more other investment alternatives. An election to transfer such amounts shall be made only at such time, in such manner, and subject to such restrictions as the Committee may specify. The Committee may provide that future contributions may be invested in a different investment alternative than amounts already accumulated in the Participant's Account(s).

- (vi) The Committee shall prescribe rules relating to the investment of the assets in the Accounts of a Participant who fails to make an effective election, for any reason whatsoever, as to how all or a portion of his Accounts shall be invested.
- (vii) The Committee shall arrange for reports to be delivered or made available to Participants regarding the investment of their funds pursuant to their investment elections. Failure of a Participant to notify the Committee regarding implementation of his investment election (or change therein) within thirty (30) days following any such notice shall be deemed to be an election to have the Accounts invested in the manner shown on such report, even if the manner of investment is different from that specified in the Participant's election form or investment instructions. If a Participant has not received a notice confirming his investment election (or change therein) and does not notify the Committee or its designated delegate within thirty (30) days of the date such election (or change) was to be effective, the Participant shall be deemed to have elected to have the Accounts invested in the manner in which they are in fact invested, even if that method differs from the Participant's election form or investment instructions.
- (b) <u>Timing of Transfers From or Into Company Stock Fund</u>. Any exchange or transfer into or out of any Company Stock Fund which results from any Participant-directed or Beneficiary-directed transaction or election (excluding any distribution due to (i) Disability in accordance with Section 6.2 or (ii) hardship in accordance with Section 6.3(b)) shall be made effective only on a Trade Date. This provision is intended to result in the classification of the periods in which Participants and Beneficiaries may not exchange or transfer into or out of a Company Stock Fund as falling within the blackout period exceptions of Sections 306(a)(4) and 306(b)(7) of the Sarbanes-Oxley Act of 2002, and the Plan and this Section 3.8(b) shall be interpreted and administered consistent with this result. The Committee shall prescribe such rules as may be appropriate to effectuate the provisions of this Section 3.8(b).
 - (c) Investment of Accounts of Participants Subject to Governmental Conflict of Interest Rules.
 - (i) If a Participant, by virtue of his spouse's employment with the government or otherwise, is subject to governmental conflict of interest rules that would, in the opinion of the General Counsel of the Company, require the Participant to divest himself of ownership of Company Stock, then, notwithstanding other Plan provisions, no portion of a Participant's Accounts may be invested in a Company Stock Fund and Company Contributions which would otherwise be directed to a Company Stock Fund within the Trust shall instead be invested from among the other Plan investment alternatives selected by the Participant.
 - (ii) The Committee may prescribe such procedures and rules as may be required or desirable to implement this Section 3.8(c).

3.9 Valuation of Accounts.

(a) The Trustee shall value the assets of the Trust on the basis of fair market values as of each Valuation Date.

- (b) The Committee shall value the Accounts of each Participant, Suspended Participant and Alternate Payee as of the applicable Valuation Date so as to reflect the current fair market value of each Account as of such Valuation Date. The valuation provisions of this Section 3.9 shall be applied and implemented in accordance with the rules established by the Committee. The Company, the Committee and Trustee do not in any manner or to any extent whatsoever warrant, guarantee or represent that the value of a Participant's Account shall at any time equal or exceed the amount previously contributed thereto, or that any valuation or accounting method or practice will continue to be applied.
- (c) Company Stock allocated or held on an unallocated basis in the Trust Fund shall be valued as of the applicable Valuation Date, according to the following rules:
 - (i) Company Stock acquired by the Trust Fund with cash shall initially be valued at the purchase price paid for such stock. On any subsequent Valuation Date, such Company Stock, as well as all other Company Stock held in, or contributed to, the Trust Fund, shall be valued in accordance with Section 3.9(c)(ii) or 3.9(c)(iii) below, as applicable.
 - (ii) If the Company Stock consists of a security other than a security (1) which is traded on a national securities exchange which is registered under Section 6 of the Securities Exchange Act of 1934 or (2) for which there is a generally recognized market as determined by bid and asked prices quoted by persons independent of the Company and of any "party in interest" to the Plan, as defined in Section 3(14) of ERISA, then the fair market value of the Company Stock shall be determined using the current stock price as determined pursuant to the valuation process described in the Company's most recent Annual Report on form 10-K (or other SEC filing describing the valuation process for Company Stock employed by the Company's Board of Directors), and confirmed by the Committee as consistent with the "adequate consideration" definition under Section 3(18) of ERISA. Notwithstanding the foregoing, Preferred Stock that is convertible into Company Stock described in Section 3.9(c)(iii) shall have the value established by an independent appraiser. A Participant, Beneficiary or Alternate Payee shall have no right to have a particular valuation or valuation process applied (or continue to be applied) to his Account(s).
 - (iii) If any Company Stock consists of securities listed on a national securities exchange, fair market value of such Company Stock shall be considered to be equal to the closing price of such Company Stock (as reported in the consolidated transaction reporting system, or if not so reported, as reported on the principal exchange market for such Company Stock by such exchange or on any system sponsored by such exchange) on the trading day immediately preceding the applicable Valuation Date. If any Company Stock consists of securities traded on a regular basis, as determined by the Company, in the over-the-counter market, the fair market value of such Company Stock shall be considered to be equal to the average between the high bid price and the low asked price quoted by the automatic quotation system of a securities association registered under the federal securities laws for the trading day immediately preceding the applicable Valuation Date.
- (d) Any Company Stock received by the Trustee as a stock split, dividend, or as a result of a reorganization or other recapitalization of the Company shall be allocated as of

the day on which the stock is received by the Trustee in the same manner as the Company Stock to which it is attributable is then allocated.

- (e) Allocation of Cash Dividends and Other Distributions Received in the Trust Fund.
- (i) All cash dividends paid to the Trustee with respect to Company Stock that has been allocated to an Account (if any) as of the date the dividend is received by the Trustee shall be allocated to such Account. If the Company Stock in the Trust Fund is held in a Company Stock Fund as an investment alternative pursuant to Section 3.9, such that Participants have an interest in such Company Stock only indirectly through an interest in such fund held in a subaccount, the cash dividends shall be allocated to such fund and shall thereafter be invested in accordance with the investment practices of such fund, and shall not be allocated directly to a Participant's Account or subaccount. Notwithstanding the foregoing, as determined by the Committee, cash dividends may be held in a short-term investment pending a determination regarding the extent to which the dividends will be distributed pursuant to Section 12.3.
 - (ii) All cash dividends paid to the Trustee with respect to unallocated Company Stock shall be allocated as provided in Section 8.3.
- (iii) Other distributions received by the Trustee with respect to investments of the Trust shall be allocated to the applicable fund(s) established pursuant to Section 3.8.

ARTICLE IV

LIMITATION ON ANNUAL ADDITIONS

4.1 Definitions.

For purposes of this Article IV, the following terms shall have the following meanings:

"**Defined Benefit Plan**" means a plan described in Code Sections 414(j) and 414(k)(2).

"Defined Contribution Plan" means a plan described in Code Sections 414(i) and 414(k)(2).

"Section 415 Compensation" shall mean a Participant's wages within the meaning of Code Section 3401(a) and all other payments of compensation to the Participant by the Company (in the course of the Company's business) for which the Company is required to provide the Participant a written statement under Code Sections 6041(d), 6051(a)(3) and 6052. Section 415 Compensation shall be determined without regard to any rules under Code Section 3401(a) that limit the remuneration included in wages based on the nature or location of the employment or the services performed. Compensation for any limitation year is the compensation actually paid or includible in gross income during such year. Effective January 1, 1998, "Section 415 Compensation" shall include elective deferrals as defined in Code Section 402(g)(3) of the Code and any amount which is contributed or deferred by the Company or an Affiliated Company at the election of an Employee and which is not includible in the gross income of the Employee by reason of Code Sections 125, 132(f)(4) or 457.

4.2 Section 415 Limitation.

Notwithstanding anything else contained herein, the Annual Additions to all Accounts of a Participant shall not exceed the lesser of \$40,000 (adjusted as permitted under Section 415(d)(1) of the Code and regulations issued thereunder) or 100% of the Participant's Section 415 Compensation (as defined below) from the Company and all Affiliated Companies during the Plan Year.

4.3 Annual Addition Limitations.

- (a) The compensation limitation of Section 4.2 of the Plan shall not apply to any contribution for medical benefits (within the meaning of Code Section 419A(f)(2)) after separation from service which is treated as an Annual Addition.
- (b) If any Company or any Affiliated Company contributes amounts, on behalf of Participants covered by the Plan, to other Defined Contribution Plans, the limitation on Annual Additions provided in this Article IV shall be applied to Annual Additions in the aggregate to the Plan and such other plans. Reduction of Annual Additions, where required, shall be accomplished by first refunding any voluntary contributions to Participants, then by reducing contributions under other Company retirement plans in the following order:
 - (i) Earnings on unallocated Company contributions under such other Company retirement plans;

- (ii) Earnings on unallocated Profit Sharing Contribution and ESOP Contribution under this Plan;
- (iii) Forfeitures under such other Company retirement plans;
- (iv) Return of Elective Deferrals and reduction of corresponding Company Matching Contributions;
- (v) Company contributions under such other Company retirement plans; and
- (vi) Profit Sharing Contribution and ESOP Contribution under this Plan.

To the extent allocations to a Participant are reduced under subsections (i)-(vi) above, such reduced amounts shall be allocated and reallocated to other Participants in the applicable Plan, except that Elective Deferrals shall be returned to the Participant.

(c) In the event the limitations of Section 4.2 or this Section 4.3 of the Plan are exceeded and the conditions specified in Treasury Regulations §1.415-6(b)(6) are met, the Committee may elect to apply the procedures set forth in Treasury Regulations §1.415-6(b)(6).

4.4 No Contractual Right to Excess Contributions.

If, in order to comply with the limitations of this Article IV, it becomes necessary to reduce a Participant's Account(s), to reduce or reallocate amounts previously allocated to such Accounts, or otherwise, such action(s) may be taken by the Committee and Trustee free of any contractual obligation to the Participant (or Beneficiary) affected based on prior Account balances or allocations.

ARTICLE V

VESTING

5.1 Fully Vested Accounts.

A Participant's Plan Account, Voluntary Account, Rollover Account, TRASOP Account, CODA Account and MAPP Subaccount, if any, shall be 100% vested and nonforfeitable.

5.2 Employer Matching Contributions, Profit Sharing Account and ESOP Account.

(a) Subject to the provisions of Section 5.2(b), the Vested Interest of each Participant or Suspended Participant shall be equal to the amount determined by multiplying the balance in his Matching Contribution Account, Profit Sharing Account and ESOP Account on the applicable date by the Vested Interest determined in accordance with the rules of Section 5.2(b) and the following schedule:

	Vested
Years of Service	Interest
Less than one year	0%
One year but less than two years	20%
Two years but less than three years	40%
Three years but less than four years	60%
Four years but less than five years	80%
Five years or more	100%

Vesting in a Participant's PSRP Profit Sharing Account, if any, shall be governed by the provisions of the PSRP in effect on February 1, 1990.

- (b) The determination of a Participant's or Suspended Participant's Vested Interest in his Matching Contribution Account, Profit Sharing Account and ESOP Account shall be subject to the following special rules:
 - (i) During an Employee's period of employment with the Company or an Affiliated Company (including periods while on an approved leave of absence or a Maternity or Paternity Absence), in the event of his death, Disability, attainment of Normal Retirement Date, or a judicial declaration of his mental incompetence, the Employee's Vested Interest shall become one hundred percent (100%), regardless of his number of Years of Service.
 - (ii) Forfeitures shall be determined as follows:
 - (A) A Participant who terminates employment prior to becoming one hundred percent (100%) vested in his Matching Contribution Account, Profit Sharing Account and ESOP Account shall immediately forfeit the non-vested portion of such Accounts upon receiving a distribution in accordance with Section 6.1.

- (B) In the case of a Participant with no Vested Interest in his Matching Contribution Account, Profit Sharing Account and ESOP Account, the forfeiture shall occur within 120 days after his termination of employment.
- (C) In the case of a Participant with a Vested Interest in his Matching Contribution Account, Profit Sharing Account and ESOP Account, but who terminates employment prior to becoming one hundred percent (100%) vested in such Accounts and incurs five consecutive Breaks in Service prior to receiving a distribution under Section 6.1, such Participant shall forfeit the non-vested portion of such Accounts within 120 days after the completion of the fifth consecutive Break in Service.
- (iii) A former Employee who is reemployed by the Company or an Affiliated Company, prior to incurring five consecutive Breaks in Service shall be subject to Section 5.2(c). If a former Employee incurs five consecutive Breaks in Service prior to reemployment, amounts forfeited from his Matching Contribution Account, Profit Sharing Account and ESOP Account shall remain forfeited and shall not be restored, and his Years of Service prior to such period of five consecutive Breaks in Service shall (subject to Section 5.2(c)) count only towards his Vested Interest applicable to allocations to his Matching Contribution Account credited after such period of five consecutive Breaks in Service.
- (iv) If an Employee whose Vested Interest is zero (and who has no balance in a MAPP Subaccount, Plan Account, Voluntary Account, Rollover Account, TRASOP Account or CODA Account) upon his initial Break in Service incurs five or more consecutive Breaks in Service, his Years of Service accumulated before the commencement of any such period of consecutive Breaks in Service shall not be taken into account for purposes of determining the Vested Interest in his Matching Contribution Account, his Profit Sharing Account and his ESOP Account at any time or for any purpose. An Employee's aggregate Years of Service shall not include any Years of Service not required to be taken into account under this Section 5.2(b) by reason of any Prior Break in Service.
- (v) No Employee shall be credited with any Years of Service performed prior to February 1, 1976, if the period of service would have been disregarded under the provisions of the Predecessor Plan in existence on the relevant date relating to continuity and interruptions of service and these rules requiring full-time service as a condition for participation in the Predecessor Plan.
- (vi) Prior to January 1, 2006, no Employee shall be given credit for any Years of Service performed before the computation period (as determined in accordance with the definition of "Compensation" in Section 13.1) during which the Employee attained the age of 18. Individuals who are Employees as of January 1, 2006 shall receive credit for any Years of Service performed before the computation period during which the Employee attained the age of 18; individuals who were not Employees as of such date shall not receive such credit. Furthermore, no amounts forfeited as a result of a termination of employment prior to January 1, 2006 shall be restored as a result of the crediting of Years of Service under this clause.

- (vii) In the event of a divestiture of an operating group, business unit or division, the Compensation Committee or the Compensation Committee's designee may, in their sole discretion, determine, with respect to Eligible Employees whose employment with the Company terminates as a result of such divestiture and in lieu of the otherwise applicable determination of Vested Interest specified in this Article V, (1) treat the Eligible Employees' Vested Interest as 100%, notwithstanding their Years of Service prior to termination; or (2) treat such Eligible Employees as Suspended Participants but credit Years of Service with the new employer to whom such group is divested for purposes of determining such Eligible Employees' Vested Interest. Any such determination for a particular group, business unit or division shall not bind the Company in any way with respect to any subsequent determination relating to a different group, business unit or division. In the event of a subsequent divestiture from the new employer, the Compensation Committee or the Compensation Committee designee may make a similar determination regarding vesting acceleration.
- (viii) In the event the Plan is amended to change any vesting schedule under the Plan, each Participant having no less than three Years of Service shall be permitted to elect, within a reasonable period after the adoption of such amendment, to have his Vested Interest determined under the Plan without regard to such amendment.
- (ix) The Vested Interest of a Participant who was an Employee as of December 31, 1994 in his Matching Contribution Account shall be 100%. Further, any Participant who was first employed before January 1, 1995 and is actively employed on November 28, 2003 shall become 100% vested his Profit Sharing Account and ESRP Plan Account on November 28, 2003. A Participant who was first employed before January 1, 1995 and terminated employment prior to November 28, 2003 shall become 100% vested in his Profit Sharing Account and ESRP Plan Account if he subsequently returns to active employment on or after November 28, 2003.
- (x) In order to comply with a government contract, or for other business reasons, the Chief Executive Officer, President, Chief Operating Officer, Corporate Executive Vice President and Chief Administrative Officer, or Senior Vice President and Director of Human Resources of the Company may determine (without any requirement of action by the Board of Directors), with respect to a designated category of Eligible Employees within a group or classification within the Company, including, without limitation, a group of newly Eligible Employees acquired through an acquisition, and in lieu of the otherwise applicable determination of such Employees' Vested Interest specified in this Article V, (1) to treat the Eligible Employees' Vested Interest as 100%, notwithstanding their Years of Service; or (2) to apply a more liberal vesting schedule than the schedule described in Section 5.2(a) for purposes of determining such Eligible Employees' Vested Interest. The designation of any such group or classification and the effective date of the method of determining the Vested Interest shall be made in writing and attached hereto as an exhibit and incorporated herein. Any such determination for a particular group shall not bind the Company in any way with respect to any subsequent determination relating to a different group.
- (xi) Notwithstanding anything to the contrary in this Section 5.2, a Participant who was first employed before January 1, 1995, terminated employment prior to

November 28, 2003 and whose Vested Interest in his Profit Sharing Account or ESRP Plan Account was less than 100% on the date of his termination shall forfeit the nonvested portion of his Profit Sharing Account and ESRP Plan Account effective as of November 28, 2003 even if such former Employee has not received a distribution in accordance with Article VI of the Plan. Any such forfeited amount shall be restored to the Participant's Profit Sharing Account and ESRP Plan Account (or ESOP Account, if the reemployment occurs after January 1, 2006) by the Company from existing forfeitures if the Participant is reemployed by the Company or an Affiliated Company and such restored amount shall become 100% vested in his Profit Sharing Account and ESRP Plan Account (or ESOP Account, if the reemployment occurs after January 1, 2006) as of his reemployment date.

(c) If a Participant who was partially vested under Section 5.2(a) is reemployed by the Company or an Affiliated Company on (or before) the Anniversary Date of the Plan Year in which his fifth consecutive Break in Service occurs, the Participant shall be entitled to have his Matching Contribution Account, Profit Sharing Account and ESOP Account reinstated by making the repayment described in Section 8.2.

5.3 Alternate Payee Accounts.

In the event that an Alternate Payee is awarded an interest in either the Matching Contribution Account, the Profit Sharing Account or the ESOP Account of a Participant whose Vested Interest in such Account is less than 100%, to the extent expressly required under the terms of the applicable qualified domestic relations order, the Vested Interest of the Alternate Payee at any subsequent time in that portion of the Alternate Payee Account attributable to such awarded interest shall be the same percentage as the Participant's Vested Interest in his applicable Matching Contribution Account, Profit Sharing Account or ESOP Account at that time, determined in accordance with Sections 5.1 and 5.2.

ARTICLE VI

DISTRIBUTIONS

- **6.1** <u>Distribution Upon Termination of Employment</u>. The following rules shall apply in the case of a Participant whose employment with the Company terminates other than due to his death or Disability:
- (a) A Participant's Distributable Benefit shall become distributable upon the Participant's termination of employment with the Company other than due to his death or Disability.
- (b) The Participant's Accounts shall continue to be valued pursuant to Section 3.9, but no further allocations of Company Contributions under Article III shall be made to such Accounts, except for an allocation of Company Contributions representing Elective Deferrals made prior to termination and Company Matching Contributions related thereto. Notwithstanding the foregoing, an Eligible Participant (as defined in Section 3.4(c)) shall receive a Profit Sharing Contribution and an ESOP Contribution.
- (c) If a Participant's Distributable Benefit exceeds \$5,000, distribution shall be made upon the Participant's termination of employment only if the Participant so requests or consents to such distribution in writing. An explanation of the Participant's right to defer distribution of his Distributable Benefit shall be provided to the Participant no less than 30 and no more than 90 days before such distribution is to be made (consistent with such regulations as the Secretary of the Treasury may prescribe). If the Distributable Benefit of a terminating Participant is a distribution to which Code Sections 401(a)(11) and 417 do not apply, such distribution may commence less than 30 days after the notice described in the preceding sentence is given, provided that: (1) the Committee clearly informs the Participant that the Participant has the right to a period of at least 30 days after receiving the notice to consider the decision of whether or not to elect a distribution, and (2) the Participant, after receiving the notice, affirmatively elects an immediate distribution. If a Participant does not request an immediate distribution of his Distributable Benefit upon a termination of employment, (unless Treasury Regulations otherwise provide and the Committee adopts different rules), a Participant's Distributable Benefit shall be distributed to him as soon as administratively practicable following the latest of:
 - (i) the Participant's termination of employment,
 - (ii) the earlier of the date the Participant consents in writing to such distribution or attains age 62, or
 - (iii) the date elected by the Participant, which shall not be later than the Required Payment Commencement Date.

Effective December 1, 2005, distribution to a Participant upon the Participant's Required Payment Commencement Date may, for any calendar year, if elected by the Participant, be the

minimum required distribution under Code Section 401(a)(9) and the regulations thereunder, as set forth in Section 6.5(j). A Participant making such an election may thereafter elect to receive any remaining amounts in his Accounts in a single sum.

- (d) Distributions shall be in the form of a cash lump sum, provided that a Participant may, upon or after reaching age 59 1 /2 as of the date of distribution, elect to receive the portion of his Distributable Benefit invested in a Company Stock Fund in the form of shares of Company Stock (plus cash for any fractional shares), or in the form of cash. The value of the amount distributed in the form of cash from the portion of a Participant's Distributable Benefit invested in a Company Stock Fund shall be the net proceeds at the sale of the Company Stock liquidated pending distribution, plus cash for any fractional share.
- (e) In lieu of receiving his entire Distributable Benefit in a single lump-sum distribution as provided in Section 6.1(d), a Participant upon or after attaining age 59 ½ may elect on or after May 1, 1998 to receive a distribution of that portion of his Account(s) that is not invested in a Company Stock Fund within the Trust, while leaving in the Plan the remaining portion which is invested in a Company Stock Fund until a later distribution date specified by the Participant, which in no event shall be a date after the Required Payment Commencement Date. Any such partial withdrawal shall be subject to such limitations and restrictions as maybe imposed by Committee rule. Notwithstanding the foregoing, with respect to any portion of the Participant's Account attributable to Company Stock acquired by the ESRP after December 31, 1986 (as determined under terms of the ESRP and applicable Treasury Regulations), the Participant may elect to have payment made in substantially equal annual payments over a period of five years, with the first such annual payment made not later than one year after the close of the Plan Year which is the fifth Plan Year following the Plan Year in which the Participant separated from service with the Company. Notwithstanding the foregoing, a Participant who elects such a Company Stock Distributable Benefit may, at any time prior to attaining age 59-1/2, request that the Committee pay him the Vested Interest in his Accounts in a cash lump sum and, in such event, the distribution shall be made not later than one hundred twenty (120) days (or as soon thereafter as is reasonable for administrative purposes) after the close of the Plan Year in which such request is made.

6.2 <u>Designation of Beneficiary; Distribution Upon Death or Disability.</u>

(a) On a form prescribed by and delivered to the Committee, each Participant shall have the right to designate in writing a Beneficiary or Beneficiaries whom such Participant desires to receive the benefits of this Plan in the event of such Participant's death. If, as of January 1, 2006, a Participant had Beneficiary designations on file for the ESRP and for this Plan, then the Beneficiary designation for the ESRP shall be disregarded and the Beneficiary designation for this Plan shall govern. If a Participant did not have a Beneficiary designation on file for this Plan, but had a Beneficiary designation for the ESRP, then the ESRP Beneficiary designation shall govern. The foregoing rules shall apply until such time as a new Beneficiary designation is filed under this Plan.

- (b) A Participant may from time to time change his designated Beneficiary or Beneficiaries without the consent of such Beneficiary or Beneficiaries by filing a new designation in writing with the Committee.
- (c) If a deceased Participant shall have failed to designate a Beneficiary, if the Committee shall be unable to locate a designated Beneficiary after reasonable efforts have been made, if for any reason the designation shall be legally ineffective, or if the Beneficiary shall have predeceased the Participant (and no legally effective contingent Beneficiary shall have been named), any distribution required to be made under the provisions of this Plan shall be payable to the Participant's surviving spouse, and if the Participant had no surviving spouse, to the Participant's estate, and the surviving spouse or estate shall be considered the Beneficiary under this Plan.
- (d) In the event that a Participant shall predecease his Beneficiary and on the subsequent death of the Beneficiary a remaining distribution is payable under the applicable provisions of this Plan, the distribution shall be payable to the Beneficiary's estate.
- (e) The Company, the Committee and the Trustee shall have no duty to determine whether a Beneficiary designation or spousal consent made pursuant to this Section 6.2 was an informed designation or consent or was freely given, and shall be entitled to rely upon the Beneficiary form filed with the Committee or its delegate, as well as such other documents as may be required pursuant to Section 6.2(i), and shall be under no duty or obligation to attempt to protect the rights of a spouse or former spouse of a Participant.
- (f) Upon the death of a Participant during his employment, or in the event that the Committee shall determine that a Participant has suffered a Disability while an Employee of the Company, the Committee shall direct the Trustee to make a distribution of the Participant's Distributable Benefit to the Participant's Beneficiary determined under this Section 6.2 (in the event of death), or to the disabled Participant (in the event of Disability).
- (g) Except as may be required in accordance with Section 6.4, the form of the Distributable Benefit shall be a cash lump-sum distribution, payable within one hundred twenty (120) days (or as soon thereafter as is reasonable for administrative purposes) after the close of the Plan Year in which the death of the Participant occurs, or in which he is determined to be Disabled, as the case may be, subject to proof of death or Disability satisfactory to the Committee. In no event shall the payment to a Participant who has suffered a Disability be made later than such Participant's Required Payment Commencement Date.
- (h) Upon the death of a former Participant after his retirement, Disability or other termination of employment, but prior to the distribution of his Distributable Benefit to which he is entitled, the Committee shall direct the Trustee to make a distribution of the balance to which the deceased Participant was entitled, to the Participant's Beneficiary determined under this Section 6.2, such payment to be made within one hundred twenty (120) days (or as soon thereafter as is reasonable for administrative purposes) after the close of the Plan Year in which the death of the Participant occurs, notwithstanding any elections previously made by the Participant.

(i) The Committee or Trustee, or both, may require (and rely upon) the execution and delivery of such documents, papers and receipts as the Committee or Trustee may determine necessary or appropriate in order to establish the fact of death of the deceased Participant and of the right and identity of any Beneficiary or other person or persons claiming any benefits.

6.3 In-Service Withdrawals.

- (a) <u>Age 59 ½ Withdrawals</u>, A Participant shall be entitled to make withdrawals from the vested portion of his Accounts including any earnings thereon, after the date on which he attains the age of 59-1/2 even though his employment with the Company has not yet been terminated.
- (b) <u>Hardship Withdrawal</u>. A Participant may withdraw amounts attributable to Elective Deferrals and Catch-Up Contributions made to this Plan prior to attaining age 59-1/2, upon incurring a hardship as determined by the Committee, provided that no amount representing earnings in such account after December 31, 1988, may be withdrawn. A withdrawal will be deemed on account of hardship only if the distribution is made on account of an immediate and heavy financial need and is necessary to satisfy such financial need. Any withdrawal made pursuant to this Section 6.3(b) shall be made in accordance with rules prescribed by the Committee that are consistent with Regulations under Code Section 401(k), and the following rules:
 - (i) The determination of whether an Employee has an immediate and heavy financial need is to be made by the Committee on the basis of all relevant facts and circumstances. A distribution will be deemed to be made on account of an immediate and heavy financial need of the Employee if the distribution is on account of:
 - (A) Medical expenses that would be deductible under Code Section 213(d) (determined without regard to whether the expenses exceed 7.5% of adjusted gross income);
 - (B) Purchase of the principal residence for the Employee (excluding mortgage payments);
 - (C) Payment of tuition and related education fees for the next twelve months of post-secondary education for the Employee or the Employee's spouse, children or dependents (as defined in Code Section 152, without regard to sections 152(b)(1), (b)(2) and (d)(1)(B));
 - (D) The need to prevent the eviction from the Employee's principal residence or foreclosure on the mortgage of the Employee's principal residence;
 - (E) Payments for burial or funeral expenses for the Employee's deceased parent, spouse, children or dependents (as defined in Code Section 152, without regard to section 152(d)(1)(B));

- (F) Expenses for the repair of damages to the Employee's principal residence that would qualify for the casualty deduction under Code Section 165 (determined without regard to whether the loss exceeds 10% of adjusted gross income); or
- (G) Any other event described in Treasury Regulations or rulings as an immediate and heavy financial need and approved by the Committee as a reason for permitting distribution under this section.
- (ii) To receive a hardship withdrawal, the following requirements must first be met by the applicant:
 - (A) The applicant must apply in the manner required by the Committee and including any required signatures and certifications;
 - (B) The applicant must have withdrawn, or must withdraw at the same time that an application for hardship withdrawal is submitted, all eligible in-service withdrawals from balances in this Plan and all other Company retirement plans;
 - (C) The applicant must have applied for all eligible loans from those Company retirement plans which permit Participant loans;
 - (D) The hardship withdrawal cannot be less than \$500.00; and
- (E) The applicant must suspend all Elective Deferrals to all qualified plans of deferred compensation maintained by the Company or an Affiliated Company for six (6) months after receiving the hardship withdrawal pursuant to rules prescribed by the Committee.
- (iii) In the event that a Participant has elected to have his Plan Account invested in more than one investment alternative pursuant to the rules of Section 3.8, then the withdrawal shall be made first pro rata from the subaccounts other than those invested in a Company Stock Fund in which his Account is invested, and after exhaustion of such subaccounts, from his subaccounts invested in a Company Stock Fund.
- (iv) The timing of the payment of the withdrawal shall be made as soon as practicable following the request in accordance with the rules established by the Committee.

(c) Withdrawal of Voluntary Accounts.

(i) A Participant shall be entitled to a distribution from his Voluntary Account (up to the full balance thereof on the Applicable Valuation Date) prior to the termination of his employment with the Company or an Affiliated Company, subject to reasonable restrictions imposed by the Committee with respect to the frequency and amount of such withdrawals.

- (ii) All withdrawals from Voluntary Accounts shall be paid in cash as soon as administratively practicable following the receipt of a request for withdrawal from the Participant.
- (iii) In the event that a Participant has elected to have his Voluntary Account invested in more than one investment alternative pursuant to the rules of Section 3.8, then a Participant withdrawing less than the entire balance in his Voluntary Account will receive a pro rata distribution from among such investment alternatives.
- (iv) If requested by the Participant, the Participant's Voluntary Account shall be distributed as soon as practicable following his termination of employment with the Company, notwithstanding the fact that a later distribution of his Distributable Benefit is provided pursuant to Section 6.1.

6.4 <u>Distribution of MAPP Subaccount</u>. Notwithstanding the preceding sections of this Article VI:

- (a) (i) A Participant's Distributable Benefits attributable to his MAPP Subaccount shall be paid in the form of a distribution of an annuity contract (purchased from a legal reserve life insurance company with the aggregate balance in the Participant's MAPP Subaccount) providing a joint and fifty percent (50%) survivor annuity ("Joint and Survivor Annuity") if the Participant was married at the time of retirement or at the time the Distributable Benefit would otherwise be paid pursuant to Section 6.1. The Participant, with the consent of his spouse, may, however, elect not to receive a Joint and Survivor Annuity as provided in this Section 6.4, in which case the Participant's benefit will be paid in accordance with the otherwise applicable provisions of this Article VI.
 - (ii) The Joint and Survivor Annuity payable under this Section 6.4 shall provide for payments to the Participant in equal monthly payments, beginning in the month following the month in which the Participant's Normal Retirement Date occurs (or would occur in the event of termination of employment prior to such date), and ending the payment for the calendar month in which his death occurs, and, if the Participant dies after his benefit commences and is survived by a spouse entitled to survivor benefits under such Joint and Survivor Annuity, such spouse shall receive monthly payments of fifth percent (50%) of the Participant's benefit beginning with the payment for the calendar month following the month in which the Participant died and ending with the payment for the calendar month in which the spouse dies.
 - (iii) A Participant may elect with the consent of his spouse, in the manner prescribed by the Committee, not to receive a Joint and Survivor Annuity in which case the Participant shall receive his benefit as otherwise provided in this Article VI. This election may be made or revoked at any time during the ninety (90)-day period ending on the annuity starting date but shall become irrevocable once the Participant's benefit commences (*i.e.*, once the annuity contract is distributed. The election may be made only with the consent of the spouse, which consent must acknowledge the effect of such election and must be given in writing witnessed by a member of the Committee, a representative thereof, or a notary public. Spousal

consent may be waived by the Committee only if it is established to the satisfaction of the Committee that such consent cannot be obtained because the Participant has no spouse, because the spouse cannot be located, or because of such other circumstances as are prescribed in regulations issued by the Secretary of the Treasury. Consent obtained or waived under this Section 6.4(a) shall be effective only with respect to the spouse to whom such consent applied.

- (b) If a Participant dies before commencement of payment of his Distributable Benefit attributable to his MAPP Subaccount, the Participant's surviving spouse shall (unless such surviving spouse elects to receive a lump sum distribution as set forth in this subsection (b)) receive a distribution of an annuity contract (provided from a legal reserve life insurance company with the aggregate balance in the Participant's Accounts) providing an immediate annuity for the life of the surviving spouse ("Preretirement Survivor Annuity"). The surviving spouse may waive the Preretirement Survivor Annuity and instead receive a lump sum distribution at any time before the benefit commences, but distribution in the form of an annuity shall become irrevocable once the benefit commences (i.e., once the annuity contract is distributed).
- (c) Survivor benefits under Section 6.4(a) or (b) shall not be paid to any surviving spouse (or former spouse) who was neither (i) married to the Participant throughout the one-year period ending on the earlier of the Participant's annuity starting date or the date of the Participant's death nor (ii) deemed to be married at such time pursuant to Code Section 417(d)(2). Instead, the amount attributable to the MAPP Subaccount shall be paid in a lump sum to the Participant's Beneficiary.

(d) Required Minimum Distributions.

- (i) <u>General</u>. All distributions required under Section 6.4 of the Plan will be determined and made in accordance with the Treasury Regulations under Code Section 401(a)(9), and the requirements of this Section 6.4(d) will take precedence over any inconsistent provisions of the Plan. Notwithstanding the foregoing, distributions maybe made under a designation made before January 1, 1984, in accordance with Section 242(b)(2) of the Tax Equity and Fiscal Responsibility Act (TEFRA) and the provisions of the Plan that relate to Section 242(b)(2) of TEFRA.
- (ii) <u>Death of Participant before Distributions Begin</u>. If the Participant dies before distribution begins, and the Participant's surviving spouse is the Participant's sole designated Beneficiary, then distributions to the surviving spouse will begin by December 31 of the calendar year immediately following the calendar year in which the Participant dies. Unless a lump sum distribution is elected pursuant to Section 6.4(b) or a lump sum is payable pursuant to Section 6.5(f), the Participant's interest shall be distributed in the form a Preretirement Survivor Annuity purchased from an insurance company. If the Participant's interest is distributed in the form of a Preretirement Survivor Annuity purchased from an insurance company, distributions thereunder will be made in accordance with the requirements of Code Section 401(a)(9) and the Treasury Regulations. Distributions to a Beneficiary other than the surviving spouse of the Participant shall be made pursuant to Section 6.1.

6.5 Special Rules Relating to Distributions.

- (a) <u>Form</u>. Subject to the Participant's right under Sections 6.1(d), 6.1(e) and Section 6.5(b) to elect to receive a Company Stock distribution after age 59 ½ calculated under the provisions of Section 6.5(e), a Participant's Distributable Benefit shall be distributed in the form of cash or, if elected in accordance with Section 6.6(b), by trustee-to-trustee transfer.
- (b) <u>Amount</u>. For purposes of determining the amount of Distributable Benefit that will be distributed to a Participant or Beneficiary pursuant to the rules of this Article VI, the value of the Participant's Account shall be determined in accordance with rules prescribed by the Committee. However, the value of the Participant's Account shall be increased or decreased (as appropriate) by any contributions or distributions properly allocable under the terms of this Plan to his Account that occurred on or after the Applicable Valuation Date or for any other reason were not otherwise properly reflected in the valuation of his Account on such Valuation Date.
- (c) <u>Liability</u>. Neither the Committee, the Company, nor the Trustee shall have any responsibility for any increase or decrease in the value of a Participant's Account as a result of any valuation made under the terms of this Plan after the date of his termination of employment and before the date of the distribution of his Account to him or his Beneficiary. Also, neither the Committee, the Company, nor the Trustee shall have any responsibility for failing to make any interim valuation of a Participant's Account between the date of distribution to the Participant of his Account and the immediately preceding Valuation Date, even though the Plan Assets may have been revalued in that interim for a purpose other than to revalue the Accounts under this Plan.
- (d) Notice of Right to Receive Company Stock. The Committee shall notify the Participant in advance of making the distribution of his right to elect distribution in the form of Company Stock upon reaching age 59 ¹/2. Upon being so notified the Participant shall have a reasonable period of time (at least fourteen days) in which to elect to receive his Distributable Benefit in the form of a Company Stock distribution as calculated under Section 6.5(e). This election by the Participant shall be made in writing or as otherwise permitted by Section 11.9, shall be irrevocable when made unless the Committee shall approve a revocation thereof, and shall operate to require the Committee to cause the Participant's Distributable Benefit to be made in the form of a Company Stock distribution as calculated under Section 6.5(e).
 - (e) <u>Distribution of Company Stock</u>. A Company Stock Distributable Benefit shall be governed by the following rules:
 - (i) The amount of such a distribution shall be the number of whole shares of Company Stock that can be purchased, either by a stated percentage or by the dollar value of the Participant's Plan balance (determined as of the Applicable Valuation Date), with the remainder of the value of the Participant's Plan balance distributable in the form of cash.
 - (ii) If more than one class of Company Stock is available for distribution to a Participant, the Participant must receive substantially the same proportion of each such class of stock ("Pro Rata Distribution"). The rule in the preceding sentence shall not

apply to the extent that the Participant elects, pursuant to rules to be prescribed by the Committee, to receive the distribution in a form other than the Pro Rata Distribution. The rules of this Section 6.5(e)(ii) shall be applied by the Committee in a manner not inconsistent with the provisions of Code Section 409(h). For purposes of the foregoing, Preferred Stock is not considered available for distribution because Preferred Stock is converted prior to distribution.

- (iii) Notwithstanding paragraphs (i) and (ii) above, a Company Stock Distributable Benefit shall not include Preferred Stock. Instead, any such Preferred Stock shall be converted prior to distribution in accordance with the provisions of the applicable Certificate of Incorporation.
 - (iv) Any Company Stock Distributable Benefit under this Plan is limited to, and shall be deemed made from, the assets of the ESOP Fund.
- (f) <u>Distributable Benefit Under \$5,000</u>. If the present value of the Participant's Distributable Benefit determined as of the Participant's termination date, death or any subsequent Valuation Date does not exceed \$5,000, the Participant (or Beneficiary in the event of the Participant's death) shall receive a single sum distribution of the entire value of the Participant's Distributable Benefit within twelve months (or as soon thereafter as is reasonable for administrative purposes) following the Participant's termination of employment, or as soon as practicable following receipt of notice of Participant's death or such subsequent Valuation Date, whichever is applicable; provided, however, that with respect to a Participant's MAPP Subaccount, no distribution shall be made under this subsection after the Participant's annuity starting date unless the Participant and spouse consent in writing to such distribution. For this purpose, the annuity starting date shall mean the first day of the first period for which an amount is payable as an annuity or, in the case of a disability benefit, the first day of the first period for which a benefit is to be received by reason of a Disability. Notwithstanding anything to the contrary contained in this Section 6.5(f), in the event that the present value of a Participant's Distributable Benefit is greater than \$1,000 but equal to or less than \$5,000, if the Participant does not affirmatively elect to have such distribution either (1) paid directly to an eligible retirement plan specified by the Participant in a direct rollover, or (2) to receive the distribution directly, then the Committee shall pay the distribution in a direct rollover to an individual retirement plan designated by the Committee. If the Participant's Distributable Benefit is zero, the Participant or the Participant's spouse.
- (g) <u>Rollover Account</u>. A Participant's Distributable Benefit from his Rollover Account, if any, shall be distributed in the same fashion as provided in Sections 6.1 through 6.3, as applicable.
- (h) <u>Facility of Payment</u>. If any payee under the Plan is a minor or if the Committee reasonably believes that any payee is legally incapable of giving a valid receipt and discharge for any payment due him, the Committee may have the payment, or any part thereof, made to the person (or persons or institution) whom it reasonably believes is caring for or supporting the payee, unless it has received due notice of claim therefore from a duly appointed guardian or committee of the payee. Any payment shall be a payment from the Accounts of the

payee and shall, to the extent thereof, be a complete discharge of any liability under the Plan to the payee.

(i) <u>Distribution to Alternate Payees</u>. If an Alternate Payee is entitled to a distribution of benefits from this Plan pursuant to a qualified domestic relations order, as defined in Section 11.5, the benefits payable to such Alternate Payee shall be distributed pursuant to such qualified domestic relations order under rules or procedures described by the Committee. If permitted by applicable law and regulations, the Committee may require or permit immediate distribution of such benefits to an Alternate Payee at any time following the determination by the Committee that such an order is a qualified domestic relations order. In the event that an Alternate Payee dies prior to receipt of the amounts due him from an Alternate Payee Account, such amounts shall be distributed to the estate of the Alternate Payee as soon as practicable following the date such amounts would have been distributed to such Alternate Payee unless the order provides that the amounts shall revert back to the Participant.

(j) Compliance with Section 401(a).

- (i) Notwithstanding anything to the contrary contained herein, the distribution options under this Plan shall comply with Section 401(a)(9) of the Code and regulations promulgated thereunder, which are hereby incorporated by this reference as a part of this Plan. Accordingly, unless otherwise permitted by law, the interest of each Participant shall commence to be distributed no later than such Participant's Required Payment Commencement Date.
- (ii) For purposes of determining the distributions under Section 6.1(c) to Participants who elect to receive a minimum required distribution for any distribution calendar year, the following shall apply:
 - (A) During the Participant's lifetime, the minimum amount that will be distributed for each distribution calendar year is the lesser of:
 - (1) the quotient obtained by dividing the Participant's account balance by the distribution period in the Uniform Lifetime Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations, using the Participant's age as of the Participant's birthday in the distribution calendar year; or
 - (2) if the Participant's sole designated Beneficiary for the distribution calendar year is the Participant's spouse, the quotient obtained by dividing the Participant's account balance by the number in the Joint and Last Survivor Table set forth in Section 1.401(a) (9)-9 of the Treasury regulations, using the Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the distribution calendar year.
 - (B) Required minimum distributions will be determined under this Article VI. beginning with the first distribution calendar year and up to and including the distribution calendar year that includes the Participant's date of death. Required minimum distributions following the Participant's death shall be made in a lump sum as set forth in Section 6.2.

- (C) The required minimum distribution for any distribution calendar year shall be made first pro rata from the subaccounts other than those invested in a Company Stock Fund in which his Account is invested, and after exhaustion of such subaccounts, from his subaccounts invested in a Company Stock Fund.
 - (D) For purposes of this Section 6.5(j), the following definitions shall apply:
 - (1) A "distribution calendar year" is a calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first distribution calendar year is the calendar year immediately preceding the calendar year which contains the Participant's required beginning date. The required minimum distribution for the Participant's first distribution calendar year will be made on or before the Participant's Required Payment Commencement Date. The required minimum distribution for other distribution calendar years, including the required minimum distribution for the distribution calendar year in which the Participant's Required Payment Commencement Date occurs, will be made on or before December 31 of that distribution calendar year.
 - (2) The "Participant's account balance" is the account balance as of the last valuation date in the calendar year immediately preceding the distribution calendar year (valuation calendar year) increased by the amount of any contributions made and allocated or forfeitures allocated to the account balance as of dates in the valuation calendar year after the valuation date and decreased by distributions made in the valuation calendar year after the valuation date. The account balance for the valuation calendar year includes any amounts rolled over or transferred to the Plan either in the valuation calendar year or in the distribution calendar year if distributed or transferred in the valuation calendar year.

6.6 Direct Rollovers.

- (a) Notwithstanding any provision of the Plan to the contrary that would otherwise limit a distributee's election under this Section 6.6, a "distributee" (as defined below) may elect, at the time and in the manner prescribed by the Committee, to have any portion of an "eligible rollover distribution" (as defined below) made payable directly to an "eligible retirement plan" (as defined below) specified by the distributee in a "direct rollover" (as defined below).
 - (b) For purposes of this Section 6.6:
 - (i) An "eligible rollover distribution" is any distribution of all or any portion of the balance to the credit of the distributee, except that an eligible rollover distribution does not include: (a) any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancies) of the distributee or the distributee's designated Beneficiary, or for a specified period of 10 years or

more; (b) any distribution to the extent such distribution is required under Code Section 401(a)(9); (c) the portion of any distribution that is not includible in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities) unless such portion is transferred in a direct trustee-to-trustee transfer to a qualified trust which is part of a plan which is a defined contribution plan and which agrees to separately account for amounts so transferred, including separate accounting for the portion which is includible in gross income and the portion which is not so includible; (d) any hardship distribution described in Section 401(k)(2)(B)(i)(IV) of the Code; and (e) any other type of distribution which the Internal Revenue Service announces (pursuant to regulation, notice, or otherwise) is not an eligible rollover distribution. If approved by the Committee in its sole discretion, an eligible rollover distribution may include active loans from the Participant's Accounts.

- (ii) An "eligible retirement plan" is a retirement plan that accepts the distributee's eligible rollover distribution and is an individual retirement account described in Code Section 408(a), an individual retirement annuity described in Code Section 408(b), an annuity plan described in Code Section 403(a), an annuity contract as described in Code Section 403(b), a qualified trust described in Code Section 401(a) or an eligible plan under Code Section 457 which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state. However, in the case of a distribution to a surviving spouse, or to a spouse or former spouse who is the Alternate Payee under a qualified domestic relations order, as defined in Code Section 414(p), an eligible retirement plan is an individual retirement account or individual retirement annuity.
- (iii) A "distributee" includes an employee or former employee. In addition, the employee's or former employee's surviving spouse and the employee's or former employee's spouse or former spouse who is the alternative payee under a qualified domestic relations order, as defined in Section 414(p) of the Code, are distributees with regard to the interest of the spouse or former spouse.
- (iv) A "direct rollover" is a payment by the Plan to the eligible retirement plan specified by the distributee. A portion of a distribution shall not fail to be an eligible rollover distribution merely because the portion consists of after-tax employee contributions which are not includible in gross income. However, such portion may be transferred only to an individual retirement account or annuity described in Code Sections 408(a) or (b), or to a qualified defined contribution plan described in Code Sections 401(a) or 403(a) that agrees to separately account for amounts so transferred, including separately accounting for the portion of such distribution which is includible in gross income and the portion of such distribution which is not so includible.

ARTICLE VII

TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS

- **7.1 Trust Fund.** The Company has established the Trust pursuant to a Trust Agreement under which the Trustee has agreed to hold and administer in trust the assets of the Plan.
 - 7.2 <u>Permissible Types of Plan Investments</u>. The assets of the Plan may be invested in the following types of assets as determined by the Committee:
 - (a) Class A Common Stock, except to the extent the Participant has instructed the Trustee to make, and the Trustee has made, a valid election to receive Class B Common Stock pursuant to the terms of the Plan of Reorganization and Agreement of Merger (the "1984 Plan of Reorganization") dated as of June 1, 1984 between the Company and its wholly owned subsidiary, Science Applications, Inc., in which event the Committee or its delegate shall keep records to reflect the number of shares of Class B Common Stock allocated to each Participant's Plan Account. No further allocation of Class B Common Stock shall be made by the Company to any such Class B Common Stock Account after the number of shares required to be allocated thereto pursuant to such election have been allocated in accordance with the terms of the 1984 Plan of Reorganization. The Committee, in its discretion, may direct the Trustee to hold up to one hundred percent (100%) of the assets of the Trust in:
 - (i) Company Stock;
 - (ii) Other "Qualifying Employer Securities," as that term is defined in ERISA Section 407(d)(5);
 - (iii) "Qualifying Employer Real Property," as that term is defined in ERISA Section 407(d)(4);
 - (iv) Cash; or
 - (v) Any other property that is a permissible plan investment under applicable law.

The Board of Directors may determine that Company Stock, or a fund within the Trust which holds Company Stock (a "Company Stock Fund"), shall not be an available investment option for any former Employee who, pursuant to Section 6.1, has deferred payment of his Distributable Benefit. In the event the Board of Directors makes such a determination, the Committee shall establish procedures to implement the foregoing restriction on investment in Company Stock or a Company Stock Fund, including the timing and procedures for any fund-to-fund transfers, appropriate default options and, if considered appropriate, any transition rules.

- **7.3** Form of Company Contributions. The Company Contributions to the Trust Fund shall be paid in cash, Company Stock, or such other property as the Board of Directors may from time to time determine.
- **7.4** <u>Valuation of Company Contributions in the Form of Company Stock</u>. Company Stock contributed by the Company to the Trust Fund shall be valued using the rules set forth in Section 3.9(c).

7.5 Company, Committee and Trustee Not Responsible for Adequacy of Trust Fund.

- (a) Neither the Company, the Committee nor the Trustee shall be liable or responsible for the adequacy of the Trust Fund to meet and discharge any or all payments and liabilities hereunder. All Plan benefits will be paid from the Trust assets, and neither the Company, the Committee nor the Trustee shall have any duty or liability to furnish the Trust with any funds, securities or other assets except as expressly provided in the Plan.
- (b) Except as required under the Plan or Trust or under Part 4 of Title I of ERISA, the Company shall not be responsible for any decision, act or omission of the Trustee, the Committee, or the Investment Manager (if applicable), and shall not be responsible for the application of any moneys, securities, investments or other property paid or delivered to the Trustee.
- (c) The Company expressly disavows any contractual obligation, implied or explicit, to make any contribution to the Plan or to contribute any specified amount.
- **7.6** Company Stock Transactions with Disqualified Persons. Acquisition or sale by the Plan of Company Stock or other qualifying employer securities (as defined in Section 407(a)(5) of ERISA) from or to a "disqualified person", as defined in Code Section 4975(e)(2), shall be at a price which represents "adequate consideration," as defined in Section 3(18) of ERISA or, in the event such Company Stock or other qualifying employer security is a marketable obligation, as defined in Section 407(e) of ERISA, at a price not less favorable to the Plan than the price determined under Section 407(e)(1) of ERISA. No commission shall be charged to the Plan in connection with any such sale or acquisition. The determination as to whether or not such a sale or acquisition satisfies the requirements of this Section 7.6 shall be made by the Committee.

ARTICLE VIII

ACCOUNTING AND ALLOCATION PROCEDURES

8.1 <u>Allocation of Forfeitures</u>. The treatment of all amounts that are forfeited by reason of the termination of the employment of a Participant shall be governed by the following rules:

(a) Matching Contributions,

Forfeitures of Matching Contribution Accounts occurring during a Plan Year shall be used to offset Company Matching Contributions for such Plan Year and allocated to the Matching Contribution Accounts of those Participants who are entitled to receive an allocation of Company Matching Contributions for the Plan Year according to the rules of Section 3.3.

(b) Profit Sharing Contributions and ESOP Contributions.

- (i) Forfeitures of Profit Sharing Accounts and PSRP Profit Sharing Accounts occurring during a Plan Year shall be added to Profit Sharing Contributions for such Plan Year and allocated to the Profit Sharing Accounts of all Eligible Participants (as defined in Section 3.4(c)) for the Plan Year in the proportion that the Compensation of each such Eligible Participant (regardless of Fringe Rate Group) bears to the total compensation of all such Eligible Participants.
- (ii) Forfeitures of ESOP Accounts occurring during a Plan Year shall be added to ESOP Contributions for such Plan Year and allocated to the ESOP Accounts of all Eligible Participants (as defined in Section 3.4(c)) for the Plan Year in the proportion that the Compensation of each such Eligible Participant (regardless of Fringe Rate Group) bears to the total compensation of all such Eligible Participants.
- (c) No forfeitures attributable to either Company Matching Contributions or under subparagraph (b) above shall be allocated to any Plan Account, Rollover Account, Voluntary Account, Alternate Payee Account, TRASOP Account, CODA Account or ESOP Alternate Payee Account.
- (d) Pending allocation, all forfeitures shall be accounted for in the same manner as Company Contributions and shall be adjusted for income, gain or loss on such forfeitures.
- (e) In the event that amounts are forfeited by reason of the termination of employment of a Participant, shares of Class B Common Stock, if any, which may be held in such Participant's Matching Contributions Account, Profit Sharing Contributions Account, ESOP Account (or, indirectly, through a Participant's interest in a Company Stock Fund in which his Account is invested) shall be sold by the Trustee to the Company for cash equal to its fair market value, determined as of the date of such sale, and the cash proceeds thereof shall be allocated with the other assets held in such Participant's Plan Account pursuant to the provisions

of this Section 8.1. Alternatively, as determined by the Committee, such shares shall be retained in the Trust and allocated pursuant to this Section 8.1. In determining whether Common Stock to be forfeited is Class A or Class B Common Stock, the first-in, first-out method shall be applied.

(f) Notwithstanding anything to the contrary herein, amounts forfeited under the Plan may be used by the Company to satisfy reinstatements due to military leave or amounts owed to the Plan as a result of an error in the operation of the Plan.

8.2 Reinstatement of Forfeitures.

- (a) If a Participant who was partially vested under Section 5.2(a) is reemployed by the Company or an Affiliated Company on (or before) the Anniversary Date of the Plan Year in which his fifth consecutive Break in Service occurs, the Participant shall be entitled to have the entire portion of his Matching Contribution Account, his Profit Sharing Account, his ESOP Account and his PSRP Profit Sharing Account (including the nonvested portion(s) of such Account(s)) reinstated by repaying the total amount distributed to him. Such reinstatement shall be made from current forfeitures or, if necessary, from Company Contributions and shall not be treated as an Annual Addition. However, this repayment must be made prior to the earlier of (1) five years from the date of reemployment or (2) five consecutive Breaks in Service after the distribution of the Vested Interest in his Plan Account following such termination of employment, provided he is an Eligible Employee during that period. If such repayment is not made, then the previously forfeited amounts shall not be restored.
 - (b) In the case of a repayment made pursuant to this Section 8.2:
 - (i) The Participant shall not be required to pay any interest charge upon the amounts repaid by him; and
 - (ii) The nonvested portion(s) of his applicable Account(s) as set forth in subparagraph (a) above (which was not distributed to him) shall not be adjusted for gains or losses during the period between the forfeiture and the repayment of the distributed amount.
 - (iii) In the case of a Participant with no Vested Interest in his applicable Account(s) as set forth in subparagraph (a) above who is reemployed prior to incurring five consecutive Breaks in Service, his entire nonvested Account(s) (unadjusted for gains or losses during the period between the date of his forfeiture and the date of his reemployment) shall be reinstated upon his reemployment, without regard to the repayment requirement of subsection (a) above.
 - (iv) In no event shall a Participant who has received a distribution which includes the balance in his Plan Account, Voluntary Account, TRASOP Account, CODA Account and/or Rollover Account be entitled either to repay the Plan or to have the balance in such Plan Account, Voluntary Account, TRASOP Account, CODA Account and/or Rollover Account reinstated upon reemployment by the Company or an Affiliated Company. However, if the previous distribution otherwise qualifies for a Rollover Contribution, the Participant may make a Rollover Contribution upon reemployment.

- **8.3** Allocation of Income or Loss on Unallocated Company Contributions; Company Contributions to Defray Plan Expenses. At the time Company Contributions are allocated to Plan Accounts, the income, gain or loss on unallocated Company Contributions, adjusted for any Plan expenses paid or accrued as of the end of the Plan Year preceding the actual allocation date (except for Plan expenses paid through Company Contributions pursuant to the following sentence), shall be allocated to those Participants eligible to receive an allocation of Profit Sharing Contribution or ESOP Contribution for such Plan Year, pro rata, according to each such Participant's entitlement to such allocation. The Company may, in its discretion, make a Plan Contribution at any time for the purpose of defraying Plan expenses. Such contribution shall be used to defray Plan expenses and shall not be allocated to Accounts of Participants.
- **8.4** Accounting Procedures. The Committee shall establish accounting procedures for the purpose of making the allocations, valuations and adjustments to Accounts provided for in this Article VIII, as well as the implementation of investment direction by Participants pursuant to Section 3.8 and transfers between or distributions from subaccounts established pursuant to Section 3.8. From time to time the Committee may modify such accounting procedures for the purpose of achieving equitable, nondiscriminatory, and administratively feasible allocations among the Accounts in accordance with the general concepts of the Plan and the provisions of this Article VIII.

A Participant, Beneficiary or Alternate Payee shall have no contractual or other right to have a particular accounting procedure or convention apply, or continue to apply, and the Committee shall be free to alter any such procedure or convention without obligation to any Participant, Beneficiary or Alternate Payee, consistent with the requirements of Code Section 411(d)(6).

- **8.5** <u>Suspended Participants</u>. The Accounts of each Suspended Participant shall be held intact and shall be valued on each Valuation Date as provided in Section 3.9, but shall not receive any allocation of Company Contributions or forfeitures; provided, however, that if the Participant completes, during the Plan Year in which he becomes a Suspended Participant, 850 or more Hours of Service during such Plan Year, his Profit Sharing Account shall participate in the allocation of Company Contributions and forfeitures of Profit Sharing Contribution or ESOP Contribution for such Plan Year, if he otherwise would have been eligible for such an allocation or if such suspended Participant's employment terminated during such Plan Year as a result of death, retirement on or after the Normal Retirement Date, Disability or involuntary lay-off (other than for cause) as determined by the Committee in its sole discretion.
- **8.6** <u>Accounting for Interest of an Alternate Payee</u>. In the event an Alternate Payee is awarded an interest in the Plan benefits of a Participant pursuant to a qualified domestic relations order, as defined in Section 11.5, such interest shall be separated into one or more separate Accounts and accounted for under rules prescribed by the Committee, pending distribution to the Alternate Payee.

ARTICLE IX

OPERATION AND ADMINISTRATION OF THE PLAN; VOTING AND OTHER RIGHTS OF COMPANY STOCK

- **9.1** Members. The Committee is the Science Applications International Corporation Retirement Plans Committee. The members of the Committee shall be appointed by the Board of Directors and shall hold office until resignation, death, removal by the Board of Directors or the date the member ceases to be an employee of the Company. A member of the Committee may resign by delivering a written notice of resignation to the Board of Directors. The Board of Directors may remove any member by delivering a certified copy of its resolution of removal to such member. Vacancies in the membership of the Committee shall be filled by the Board of Directors.
- **9.2** Committee Action. A majority of the members of the Committee as constituted at any time shall constitute a quorum, and any action by a majority of the members present at any meeting, or authorized by a majority of the members in writing without a meeting, shall constitute the action of the Committee and be binding upon the same as if all members had joined therein. A member of the Committee shall not vote or act upon any matter which relates solely to himself as a Participant. Any member or members of the Committee designated by the Committee may execute any certificate or other written direction on behalf of the Committee. The Trustee or any third person dealing with the Committee may conclusively rely upon any certificate or other written direction so signed until the Committee shall file with the Trustee a written revocation of the authorization of the designated members.

9.3 Rights and Duties.

- (a) The Company shall be the Plan Administrator (as defined in Section 3(16)(A) of ERISA). The Company delegates its duties under the Plan to the Committee. The Committee shall act as the Fiduciary with respect to control and management of the Plan for purposes of ERISA on behalf of the Participants and their Beneficiaries, shall enforce the Plan in accordance with its terms, shall be charged with the general administration of the Plan, and shall have all powers necessary to accomplish its purposes, including, but not limited to, the following:
 - (i) To allocate fiduciary responsibilities (other than trustee responsibilities) among the named Fiduciaries and to designate one or more other persons to carry out fiduciary responsibilities (other than trustee responsibilities). The term "trustee responsibilities" as used herein shall have the meaning set forth in Section 405(c) of ERISA. The preceding provisions of this Section 9.3(a) shall not limit the authority of the Committee to appoint one or more Investment Managers in accordance with Section 9.3(b).
 - (ii) To designate agents to carry out responsibilities relating to the Plan, other than fiduciary responsibilities.

- (iii) To employ such legal, actuarial, medical, accounting, clerical, administrative and ministerial and other assistance as it may deem appropriate in carrying out the provisions of this Plan, including one or more persons to render advice with regard to any responsibility any named Fiduciary or any other fiduciary may have under the Plan.
- (iv) To establish rules and regulations from time to time for the conduct of the Committee's business and the administration and effectuation of this Plan.
- (v) To administer, interpret, construe and apply this Plan and the Plan's claims procedure and to decide all questions which may arise or which may be raised under this Plan by any employee, Participant, former Participant, Beneficiary, Alternate Payee or other person whatsoever, including but not limited to all questions relating to eligibility to participate in the Plan, the amount of service of any Participant, and the amount of benefits to which any Participant or his Beneficiary may be entitled on or after the Effective Date hereof.
 - (vi) To determine the manner in which the assets of this Plan, or any part thereof, shall be distributed.
- (vii) To select alternative investment options from which Participants may select from in determining investment of their Accounts, and to establish rules and procedures regarding such investment options.
 - (viii) To satisfy accounting, auditing, record keeping, insurance, bonding and reporting and disclosure requirements.
- (ix) To establish claims procedures consistent with regulations of the Secretary of Labor for presentation of claims by Participants and Beneficiaries for Plan benefits, consideration of such claims, review of claim denials and issuance of a decision on review. Such claims procedures shall at a minimum consist of the following:
 - (A) The Committee shall notify Participants and, where appropriate, Beneficiaries of their right to claim benefits under the claims procedures, and may, if appropriate, make forms available for filing of such claims, and shall provide the name of the person or persons with whom such claims should be filed.
 - (B) The Committee shall establish procedures for action upon claims initially made and the communication of a decision to the claimant promptly and, in any event, not later than 90 days after the claim is received by the Committee, unless special circumstances require an extension of time for processing the claim. If an extension is required, written notice of the extension shall be furnished to the claimant prior to the end of the initial 90-day period, which notice shall indicate the reasons for the extension and the expected decision date. The extension shall not exceed 90 days. The claim may be deemed by the claimant to have been denied for purposes of further review described below in the event a decision is not furnished to the claimant within the period described in the three preceding sentences.
 - (C) Every claim for benefits which is denied shall be denied by written notice setting forth in a manner calculated to be understood by the claimant (i) the

specific reason or reasons for denial, (ii) specific references to any provisions of the Plan on which the denial is based, (iii) a description of any additional material or information necessary for the claimant to perfect his claim with an explanation of why such material or information is necessary, and (iv) an explanation of the procedure for further reviewing the denial of the claim under the Plan, including a statement of the right of the claimant to bring an action under Section 502(a) of ERISA following an adverse benefit determination on review.

- (D) The Committee shall establish a procedure for review of claim denials, such review to be undertaken by the Committee. The review given after denial of any claim shall be a full and fair review with the claimant or his duly authorized representative having 65 days after receipt of denial of his claim to request such review. Claimant shall have the right to submit documents, records, issues, comments and other information in writing which relates to the claim for benefits, all of which shall be taken into account regardless of whether it was submitted in the initial benefit determination. The claimant shall be provided upon request and at no charge reasonable access to, and copies of, all documents, records and other information relevant to the claimant's claim for benefits.
- (E) The Committee shall establish a procedure for issuance of a decision by the Committee not later than 60 days after receipt of a request for review from claimant unless special circumstances (i.e., such as the need to hold a hearing) require an extension of time for processing the claim. If the Committee determines that an extension of time for processing is required, written notice of the extension shall be furnished to the claimant prior to the termination of the initial 60-day period. The extension notice shall indicate the special circumstances requiring an extension of time and the date by which the Plan expects to render the determination on review. In no event shall such extension exceed a period of 60 days from the end of the initial period. The decision on review shall be in writing and shall include specific reasons for the decision written in a manner calculated to be understood by the claimant with specific reference to any provisions of the Plan on which the decision is based.
- (F) Notwithstanding any other provision of this Plan, no person claiming any benefit under this Plan may commence any legal action in connection therewith until all of the claim and appeal procedures specified in this Section 9.3(a)(ix) have been commenced and completed. In addition, no action may be commenced with respect to or arising from any claim for benefits against the Plan (or the Committee, the Company or any of its or their agents) more than one hundred eighty (180) days after the Participant, Beneficiary or other individual, as the case maybe ("Claimant") is first given a written notice of the denial of his or her appeal by the Committee. Unless the Committee specifically determines otherwise, this period shall not be extended even if the Committee again considers the matter after the initial denial. This limitations period shall apply to all actions arising out of or relating to a claim for benefits including, but not limited to, any action under Section 502(a)(1)(B) of ERISA and any action under Section 502(a)(3) of ERISA to the extent said claim relates to the provision of benefits or rights under this Plan.
- (x) To execute any amendment to the Plan which is necessary to maintain the qualification and tax exempt status of the Plan under the Code and ERISA, and any other amendments to the Plan which do not materially affect the financial obligation of the

Company nor the level of benefits provided to Participants and Beneficiaries as stated in Section 10.1.

(xi) To perform or cause to be performed such further acts as it may deem to be necessary, appropriate or convenient in the efficient administration of the Plan.

Any action taken in good faith by the Committee in the exercise of its authority conferred upon it by the Plan shall be conclusive and binding upon Participants and their Beneficiaries and any Alternate Payees.

- (b) The Committee shall have the power to direct the Trustee in writing with respect to the investment of the Trust Fund or any part thereof. Where investment authority, management and control of the Trust Fund have been delegated to the Trustee by the Committee, the Trustee shall be the Fiduciary with respect to the investment, management and control of the Trust Fund with full discretion in the exercise of such investment, management and control. Among the securities which the Committee may direct the Trustee to purchase are "employer securities" as defined in Section 409(1) of the Code or any successor statute thereto. Except as otherwise provided by law, the Committee may appoint one or more Investment Managers, as defined in Section 3(38) of ERISA, to manage all or a portion of the assets of the Plan. An Investment Manager, when appointed, shall have full power to manage the assets of the Plan for which it has responsibility, and neither the Company nor the Committee shall thereafter have any responsibility for the management of those assets, except to the extent such power or responsibility shall have been reserved to the Company or Committee in the documents governing the relationship between or among the Plan, the Company and the Investment Manager. Where investment authority, management and control of Trust assets is not specifically delegated to the Trustee, the Trustee shall not be a Fiduciary with respect to the investment, management and control of Trust assets and shall be subject to the direction of the Committee or the Investment Managers appointed by the Committee, if any, regarding the investment, management and control of such assets, and in such case the Committee, or the Investment Managers, as the case may be, shall be the Fiduciary with respect to the investment, management and control of such assets.
- (c) Each Fiduciary under the Plan and Trust shall be solely responsible for its own acts or omissions. Except to the extent required by ERISA or the Code, no Fiduciary shall have the duty to question whether any other Fiduciary is fulfilling any or all of the responsibilities imposed upon such other Fiduciary by ERISA or by any regulations or rulings issued thereunder. No Fiduciary shall have any liability for a breach of fiduciary responsibility of another Fiduciary with respect to the Plan or Trust unless he knowingly participates in such breach, knowingly undertakes to conceal such breach, has actual knowledge of such breach and fails to take reasonably remedial action to remedy said breach or, through his negligence in performing his own specific fiduciary responsibilities, has enabled such other Fiduciary to commit a breach of the latter's fiduciary responsibilities.

9.4 Compensation, Bonding, Indemnity and Liability.

- (a) Members of the Committee shall serve without compensation unless the Board of Directors shall otherwise determine. However, in no event shall any member of the Committee who is an Employee receive compensation from the Plan for his services as a member of the Committee. All members shall be reimbursed for any necessary or appropriate expenditures incurred in the discharge of duties as members of the Committee. The compensation or fees, as the case may be, of all officers, agents, counsel, the Trustee, or other persons retained or employed by the Committee shall be fixed by the Committee. Members of the Committee and any delegates shall be bonded to the extent required by Section 412(a) of ERISA and the regulations thereunder. Bond premiums and all expenses of the Committee or of any delegate who is an employee of the Company shall be paid by the Company and the Company shall furnish the Committee and any such delegate with such clerical and other assistance as is necessary in the performance of their duties. The Committee is authorized at the expense of the Company to employ such legal counsel as it may deem advisable to assist in the performance of its duties hereunder. No expense or fee of the Committee for the administration of the Plan and services in relation thereto shall be paid from the Trust assets, other than those expenses related to investments as are charged against an investment alternative established pursuant to Section 3.9.
- (b) To the extent permitted by law, the Company shall indemnify each member of the Board of Directors and the Committee, and any other Employee of the Company with duties under the Plan, against expenses (including any amount paid in settlement) reasonably incurred by him in connection with any claims against him by reason of his conduct in the performance of his duties under the Plan, except in relation to matters as to which he acted fraudulently or in bad faith in the performance of such duties. The preceding right of indemnification shall pass to the estate of such a person and shall be in addition to any other right to which the Board of Directors or Committee member or other person may be entitled as a matter of law or otherwise. The following additional provisions apply to the right of indemnification:
 - (i) The Company shall promptly pay all legal costs and expenses covered by this indemnification;
 - (ii) The Company's indemnification obligation survives the termination of the Plan, the termination of employment of any indemnified party and the termination of an indemnified party's fiduciary status under the Plan;
 - (iii) The Company's indemnification obligation is not reduced as applied to acts or omissions occurring prior to any reduction in future indemnification rights nor prior to a date at least 90 days after notice of such indemnification reduction is given to the affected indemnified party, unless such party consents in writing;
 - (iv) The Company's indemnification obligation exists until a final decision (by a court or arbitrator) is rendered to the effect that the individual acted fraudulently or

in bad faith; in which case the indemnified party shall be obligated to repay all benefits provided under this Section 9.4;

- (v) All related costs, claims and expenses, including those, if any, incurred in enforcing the indemnification provisions, are covered by this indemnification obligation.
- **9.5** Manner of Administration. The Committee shall have full discretion to construe and interpret the terms and provisions of the Plan and to make factual determinations with respect to the administration of the Plan, which interpretation, construction or determination shall be final and binding on all parties, including but not limited to the Company and any Participant or Beneficiary or Alternate Payee.
- **9.6** <u>Duty of Care</u>. In the exercise of the powers and duties of the Committee as Plan Administrator and Fiduciary with respect to control and management of the Plan, each member of the Committee shall use the care, prudence, and diligence under the circumstances then prevailing that a prudent person acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims.

9.7 Reliance Upon Documents and Opinions.

- (a) The members of the Committee, the Board of Directors, the Company and any Employee of the Company delegated under the provisions hereof to carry out any fiduciary responsible under the Plan ("Delegated Fiduciary"), shall be entitled to rely upon any tables, valuations, computations, estimates, certificates and reports furnished by any consultant, or firm or corporation which employs one or more consultants, upon any opinions furnished by legal counsel, and upon any reports furnished by the Trustee. The members of the Committee, the Board of Directors, the Company and any Delegated Fiduciary shall not be liable in any manner whatsoever for anything done or action taken or suffered in reliance from any such consultant or firm or corporation which employs one or more consultants, trustee, or counsel.
- (b) Any and all such things done or actions taken or suffered by the Committee, the Board of Directors, the Company and any Delegated Fiduciary shall be conclusive and binding on all Employees, Participants, Beneficiaries, Alternate Payees and any other persons whomsoever, except as otherwise provided by law.
- (c) The Committee and any Delegated Fiduciary may, but are not required to, rely upon all records of the Company with respect to any matter or thing whatsoever, and may likewise treat those records as conclusive with respect to all Employees, Participants, Beneficiaries, and any other persons whomsoever, except as otherwise provided by law.
- **9.8** Requirement of Proof. The Committee or the Company may, in its (or their) sole discretion, require satisfactory proof of any matter under this Plan from or with respect to any Employee, Participant, Beneficiary or Alternate Payee, and no benefits under this Plan need be paid until the required proof shall be furnished.

9.9 Voting and Other Rights of Company Stock.

- (a) All voting rights of Company Stock held in the Trust Fund shall be exercised in accordance with the following provisions:
- (i) Each Participant (which term shall include, for purposes of this Section 9.9, Beneficiaries and Alternate Payees having an interest in an Account or fund holding Company Stock) shall be given the opportunity to instruct the Trustee confidentially on a form prescribed and provided by the Company as to how to vote those shares (including fractional shares) of Company Stock allocated to his Account(s) under the Plan (directly or indirectly through an interest in a Company Stock Fund) on the date immediately preceding the record date for the meeting of shareholders of the Company. The Trustee shall not divulge to the Company the instructions of any Participant. The Company may require verification of the Trustee's compliance with such confidential voting instructions by an independent auditor elected by the Company.
- (ii) All Participants entitled to direct such voting shall be notified by the Committee (or the Company, pursuant to its normal communications with shareholders) of each occasion for the exercise of these voting rights within a reasonable time (but not less than the time period that may be required by any applicable state or federal law) before these rights are to be exercised. The notification shall include all information distributed by the Company to other shareholders regarding the exercise of such rights.
- (iii) The Participants shall be so entitled to direct the voting of fractional shares (or fractional rights to shares). However, the Committee may, to the extent possible, direct the Trustee to vote the combined fractional shares (or fractional rights to shares) so as to reflect the aggregate direction of all Participants giving directions with respect to fractional shares (or fractional rights to shares).
- (iv) In the event that a Participant shall fail to direct the Trustee, in whole or in part, as to the exercise of voting rights arising under any Company Stock allocated to his Account, then these voting rights, together with voting rights as to shares of Company Stock which have not been allocated, shall be exercised by the Trustee in the same proportion as the number of Shares of Company Stock for which the Trustee has received direction in such matter (e.g., to vote for, against or abstain from voting on a proposal, or to grant or withhold authority to vote for a director or directors), and the Trustee shall have no discretion in such matter, except as may be required by applicable law.
- (v) Except as provided in paragraph (b) below, all rights (other than voting rights) of Company Stock held in the Trust Fund shall be exercised in the same manner and to the same extent as provided above with respect to the voting rights of the Company Stock, subject to the rules prescribed by the Committee, which rules, among other matters, may prescribe that no action shall be taken with respect to shares as to which no direction is received from Participants. The Trustee shall have no discretion with respect to the exercise of any such rights, except as may be required by applicable law.

- (vi) Neither the Committee nor the Trustee shall make any recommendation to any Participant regarding the exercise of the Participant's voting rights or any other rights under the provisions of this Section 9.9, nor shall the Committee or Trustee make any recommendation as to whether any such rights should or should not be exercised by the Participant.
- (b) All responses to tender and exchange for Company Stock offers shall be made in accordance with the following provisions:
- (i) Each Participant shall be given the opportunity, to the extent that shares of Company Stock are allocated to his Account, to direct the Trustee in writing as to the manner in which to respond to a tender or exchange offer with respect to Company Stock, and the Trustee shall respond in accordance with the instructions so received. The Trustee shall not divulge to the Company the instructions of any Participant. The Committee shall utilize its best efforts to timely distribute or cause to be distributed to each Participant such information as will be distributed to shareholders of the Company in connection with any such tender or exchange offer, together with a form addressed to the Trustee requesting confidential instructions on whether or not such shares will be tendered or exchanged. If the Trustee shall not receive timely direction from a Participant as to the manner in which to respond to such a tender or exchange offer, the Trustee shall not tender or exchange any shares of Company Stock with respect to which such Participant has the right of direction, and the Trustee shall have no discretion in such matter, except as may be required by applicable law.
- (ii) Unallocated shares of Company Stock and shares of Company Stock held by the Trustee pending allocation to Participants' Accounts shall be tendered or exchanged (or not tendered or exchanged) by the Trustee in the same proportion as shares with respect to which Participants have been given the opportunity to direct the Trustee pursuant to subsection (i) above are tendered or exchanged, and the Trustee shall have no discretion in such matter, except as may be required by applicable law.

9.10 Plan Expenses.

- (a) Except as provided in subsection (b) below, all expenses incurred in the establishment, administration and operation of the Plan, including but not limited to the expenses incurred by the members of the Committee in exercising their duties, to the extent these expenses are not paid by the Company, shall be charged to the Trust Fund and accounted for pursuant to the provisions of Article VIII, unless paid by the Company.
- (b) Costs or expenses which are particular to a specific asset or group of assets in the Trust Fund, such as interest and brokerage charges which are included in the cost of securities purchased by the Trustee (or charged to proceeds in the case of sales), as determined by the Committee, shall be charged or allocated in a fair and equitable manner to the Accounts, subaccounts or funds to which those assets are allocated pursuant to rules prescribed by the Committee.

9.11 Multiple Fiduciary Capacity. Any person or group of persons may serve in more than one fiduciary capacity with respect to the Plan.

9.12 Allocation of Fiduciary Responsibility.

- (a) It is the intent of the Company to comply with Section 405(c) of ERISA and to have the limitation on liability set forth in Section 405(c)(2) of ERISA apply to the maximum extent allowed by law.
- (b) Pursuant to Section 405(c) of ERISA, the authority to control and manage the operation and administration of the Plan is allocated to the Committee. Except to the extent expressly provided to the contrary in this Plan document, and the Trust Agreement, the responsibilities allocated to the Committee include:
 - (i) Responsibilities identified as Committee authority and powers in Section 9.3(a); and
 - (ii) Responsibilities identified elsewhere in this Plan document as applicable to the Committee.
 - (c) The Board of Directors is allocated the following responsibilities, acting with the advice and assistance of the Committee:
 - (i) Appointing the Trustee;
 - (ii) Appointing members of the Committee;
 - (iii) Performing those duties specifically allocated to it elsewhere in this Plan document.
- (d) The Trustee shall have only those responsibilities which have been specifically allocated to it under this Plan document and related Trust Agreement, plus any "trustee responsibilities", under Section 405(c) of ERISA, which may not legally be allocated to another person or Fiduciary. Any Investment Manager appointed pursuant to Section 9.3(b) may be granted exclusive authority and discretion to manage and control all or any portion of the assets of the Plan, subject to such limitations as may be provided in the documents governing the relationship between or among the Plan, the Company (if applicable) and the Trustee or Investment Manager.

9.13 Section 404(c) Provisions.

(a) Except for the portion of this Plan invested in Company Stock, this Plan is intended to constitute a plan described in Section 404(c) of ERISA and the regulations thereunder. As a result, with respect to elections described in the Plan and any other exercise of control by a Participant or Beneficiary over assets in the Participant's Accounts, such Participant or Beneficiary shall be solely responsible for such actions, and neither the Trustee, the Company, nor any other person or entity which is otherwise a Fiduciary shall be

liable for any loss or liability which results from such Participant's or Beneficiary's exercise of control.

- (b) The Committee shall provide to each Participant or his Beneficiary the information described in Section 2550.404(c)-1(b)(2)(i)(B)(1) of the Department of Labor Regulations. Upon request by a Participant or his Beneficiary, the Committee shall provide the information described in Section 2550.404c-1(b)(2)(i)(B)(2) of the Department of Labor Regulations.
- (c) The Committee may take such other actions or implement such other procedures as it deems necessary or desirable in order that the Plan comply with Section 404(c) of ERISA.

ARTICLE X

PLAN AMENDMENT AND TERMINATION

- 10.1 Amendments. The Board of Directors in its sole discretion may at any time, and from time to time, amend the Plan. In addition, the Committee shall have the authority to execute any amendment to the Plan which is necessary to maintain the qualification and tax exempt status of the Plan under the Code and ERISA, and any other amendments to the Plan which do not materially affecting the financial obligation of the Company nor the level of benefits provided to Participants and Beneficiaries as noted in Section 9.3. Any amendment to the Plan shall be evidenced by an instrument in writing executed in the name of the Company by an officer or officers duly authorized to execute such instrument. Except as may be required to permit the Plan and Trust to meet the requirements for qualification and tax exemption under the Code, or the corresponding provisions of other or subsequent revenue laws or of ERISA, no amendment may be made which may:
 - (a) Cause any assets of the Trust Fund, at any time prior to the satisfaction of all liabilities with respect to Participants and their Beneficiaries, to be used for or diverted to purposes other than for the exclusive benefit of Participants or their Beneficiaries;
 - (b) Decrease the accrued benefit of any Participant or Beneficiary within the meaning of Section 411(d)(6) of the Code; and
 - (c) Increase the responsibilities or liabilities of a Trustee or an Investment Manager without its written consent.
- **10.2** Retroactive Amendments. The Plan may be amended prospectively or retroactively (as provided in Code Section 401(b)) to make the Plan conform to any provision of ERISA, any Code provisions dealing with tax-qualified employees' trusts, or any regulation under either.

10.3 Discontinuance of Plan.

- (a) It is the Company's expectation that the Plan and the payment of contributions hereunder will be continued indefinitely, but continuance of the Plan by the Company is not assumed as a contractual obligation, and the Company reserves the right to permanently discontinue contributions hereunder. In the event of the complete discontinuance of contributions by the Company, the entire interest of each Participant affected thereby shall immediately become 100% vested. The Company, the Board of Directors, the Committee, and the Employees, agents and shareholders of the Company shall not be liable for the payment of any benefits under the Plan and all benefits hereunder shall be payable solely from the assets of the Trust except as otherwise required by ERISA.
- (b) The Board of Directors may terminate the Plan at any time. Upon complete termination or partial termination of the Plan, the entire interest of each of the affected Participants shall become 100% vested. The Trustee shall thereafter, upon direction of the

Committee, distribute to the Participants the amounts in such Participant's Accounts in the same manner as set forth in Article VI, subject, where appropriate, to Section 403(d)(1) of ERISA and regulations of the Secretary of Labor thereunder as may affect allocation of assets upon termination of such Plan.

10.4 Failure to Contribute. The failure of the Company to contribute to the Trust in any year, shall not, in and of itself, constitute a complete discontinuance of contributions to the Plan.

10.5 Merger or Consolidation.

- (a) This Plan and its related Trust shall not in whole or in part merge or consolidate with, or transfer its assets or liabilities to any other plan or trust unless each affected Participant in this Plan would receive a benefit immediately after the merger, consolidation, or transfer (if the Plan then terminated) which is equal to or greater than the benefit he would have been entitled to receive immediately before the merger, consolidation, or transfer (if the Plan had then terminated).
- (b) In the event of a consolidation, merger, sale, liquidation, or other transfer of the operating assets of the Company to any other company, the ultimate successor or successors to the business of the Company shall automatically be deemed to have elected to continue this Plan in full force and effect, in the same manner as if the Plan had been adopted by resolution of its board of directors, unless the successor(s), by resolution of its board of directors, shall elect not to so continue this Plan in effect, in which case the Plan shall automatically be deemed terminated as of the applicable effective date set forth in the board resolution.
- (c) The Committee may, in its discretion, authorize a plan-to-plan transfer, provided such transfer will meet the requirements of Section 414(1) of the Code and that all other actions legally required are taken. In the event of a transfer of assets from the Plan pursuant to this subsection, any corresponding benefit liabilities shall also be transferred.
- (d) In the event that assets are transferred to this Plan from another qualified plan in a plan-to-plan transfer, such assets will be accounted for separately to the extent required to preserve optional forms of benefit or other attributes of the transfer or plan as may be required by law or as may be determined by the Committee to be desirable.

10.6 Procedures on Termination.

- (a) Upon termination of the Plan, the Committee shall determine whether to continue the Trust, to distribute the assets of the Trust to Participants, Beneficiaries and Alternate Payees, to transfer the assets in the Trust to another qualified plan maintained by the Company, or to take other action consistent with applicable law.
- (b) The Trustee and the Committee shall continue to function as such for such period of time as may be necessary for the winding up of this Plan and for the making of distributions in accordance with the provisions of this Plan.

10.7 Partial Termination.

- (a) In the event of a partial termination of the Plan within the meaning of Code Section 411(d)(3), the interest of affected Participants in the Trust Fund, as of the date of the partial termination, shall become nonforfeitable as of that date.
- (b) That portion of the assets of the Plan affected by the partial termination shall be used exclusively for the benefit of the affected Participants and their Beneficiaries, and no part thereof shall otherwise be applied.
- (c) With respect to Plan assets and Participants affected by a partial termination, the Committee and the Trustee shall follow the same procedures and take the same actions prescribed in this Article X in the case of a total termination of the Plan.

ARTICLE XI

MISCELLANEOUS

- 11.1 Contributions Not Recoverable. Except where contributions are required to be returned to the Company by the provisions of the Plan as permitted or required by ERISA or the Code, it shall be impossible for any part of the contributions made under the Plan to be used for, or diverted to, purposes other than the exclusive benefit of Participants or their Beneficiaries. Notwithstanding this or any other provision of the Plan, the Company shall be entitled to recover, and the Participants under the Plan shall have no interest in (i) any contributions made under the Plan by mistake of fact, so long as the contribution is returned within one year after payment and (ii) any contributions for which deduction is disallowed under Section 404 of the Code, so long as the contributions are returned to the Company within one year following such disallowance or as permitted or required by the Code or ERISA. In the event of such mistake of fact or disallowance of deductions, contributions shall be returned to the Company, subject to the limitations, if any, of Section 403(c) of ERISA.
- 11.2 <u>Limitation on Participant's Rights</u>. Participant in the Plan shall not give any Employee the right to be retained as an Employee of the Company or any right or interest under the Plan other than as herein provided. The Company reserves the right to dismiss any Employee without any liability for any claim either against the Trustee, the Trust except to the extent provided in the Trust, or against the Company. All benefits under the Plan shall be provided solely from the assets of the Trust.
- **11.3** <u>Receipt or Release</u>. Any payment to any Participant or Beneficiary in accordance with the provisions of the Plan shall, to the extent hereof, be in full satisfaction of all claims against the Trustee, the Committee and the Company. The Trustee may require such Participant or Beneficiary, as a condition precedent to such payment, to execute a receipt and release to such effect.
 - 11.4 Alienation. Except as otherwise provided by law and as otherwise provided by Sections 11.5 and 11.6:
 - (a) The interest of any Participant, Beneficiary or Alternate Payee in the income, benefits, payments, claims or rights hereunder, or in the Trust Fund shall not in any event be subject to sale, assignment, hypothecation, or transfer. Each Participant, Beneficiary or Alternate Payee is prohibited from anticipating, encumbering, assigning, or in any manner alienating his interest under the Trust Fund, and is without power to do so, except as may otherwise be provided for in the Trust Agreement. The interest of any Participant, Beneficiary or Alternate Payee shall not be liable or subject to his debts, liabilities, or obligations, now contracted, or which may be subsequently contracted. The interest of any Participant, Beneficiary or Alternate Payee shall be free from all claims, liabilities, bankruptcy proceedings, or other legal process now or hereafter incurred or arising; and the interest or any part thereof, shall not be subject to any judgment rendered against the Participant, Beneficiary or Alternate Payee.

- (b) In the event any person attempts to take any action contrary to this Article XI, that action shall be void and the Company, the Committee, the Trustee and all Participants, their Beneficiaries and Alternate Payees may disregard that action and are not in any manner bound thereby, and they, and each of them separately, shall suffer no liability for any disregard of that action, and shall be reimbursed on demand out of the Trust Fund for the amount of any loss, cost or expense incurred as a result of disregarding or of acting in disregard of that action.
- (c) The preceding provisions of this Section 11.4 shall be interpreted and applied by the Committee in accordance with the requirements of Code Section 401(a)(13) as construed and interpreted by authoritative judicial and administrative rulings and regulations.

11.5 Nonconforming Distributions Under Court Order.

- (a) The right to a benefit payable with respect to a Participant pursuant to a "qualified domestic relations order" (within the meaning of Section 414(p) of the Code) may be created, assigned or recognized. The Committee shall establish reasonable procedures to determine the qualified status of domestic relations orders and to administer distributions under such qualified orders. In the event a qualified domestic relations order exists with respect to a benefit payable under the Plan, the benefits otherwise payable to a Participant or Beneficiary shall be payable to the Alternate Payee specified in the qualified domestic relations order.
- (b) The Plan's obligations to the Participant and each Alternate Payee shall be discharged to the extent of any payment made pursuant to such qualified domestic relations order.
- (c) The reasonable expense of determining whether a domestic relations order satisfies the requirements of being a qualified domestic relations order shall be charged to the Account of the Participant; provided, however, that if a qualified domestic relations order so provides, such expense may, instead, be charged to the Alternate Payee's share of the Participant's Account or divided equally between the Participant and the Alternate Payee.

11.6 Authorized Participant Loans.

(a) Each Participant shall have the right, subject to prior approval by the Committee, to borrow from his Accounts. Application for a loan must be submitted by a Participant to the Committee according to the procedures established by the Committee. Approval shall be granted or denied in accordance with the rules established by the Committee to govern loans from the Plan, and the Committee shall have the power to modify such rules. Such rules are included in a separate document(s) and shall be considered a part of the Plan; provided, each rule and each loan shall be made only in accordance with the regulations and rulings of the Internal Revenue Service and Department of Labor and other applicable state or federal law. The Committee shall act in its sole discretion to ascertain whether the requirements of such regulations and rulings and the Plan loan rules have been met.

- (b) Notwithstanding anything contained in the rules established by the Committee pursuant to Section 11.6(a), each loan shall, by its terms, satisfy each of the following requirements:
 - (i) The amount of the loan shall not exceed the lesser of:
 - (A) \$50,000, reduced by the excess, if any, of a Participant's highest outstanding balance of all loans from the Plan during the preceding 12 months over the outstanding balance of such loans on the loan date, or
 - (B) 50% of the value of the Participant's Vested Interest in his Accounts, except that a Participant's MAPP Subaccount shall not be included in determining such value. For purposes of this Section 11.6(b), the Participant's Vested Interest and outstanding loan balances in all qualified employer plans (as such term is defined in Code Section 72(p)(4)) of the Company shall be aggregated to determine whether a loan shall be permissible hereunder and the maximum permissible amount thereof.
 - (ii) Each loan must be repaid within fifty-six (56) months (except that if the loan proceeds are being used to purchase the principal residence of a Participant, the Committee may, in its discretion, establish a term of up to 25 years for repayment).
 - (iii) Each loan must require substantially level amortization over the term of the loan, with payments not less frequently than quarterly (loan payments may be suspended during a Participant's active military service, as provided in Section 414(u)(4) of the Code.
 - (iv) Each loan must be adequately secured, with the security to consist of the balance of the Participant's Accounts other than the MAPP Subaccount. For this purpose, the amount of the security must be at least equal to the amount of the loan. A Participant's MAPP Subaccount shall not constitute security for any loan made after January 1, 2006.
 - (A) In the case of any Participant who is an active Employee, automatic payroll deductions shall be required as additional security. To the extent a Participant's loan is secured by the Participant's Accounts, the investment gain or loss attributable to the loan shall not be included in the calculation or allocation of the increase or decrease in fair market value of the general assets of the Plan pursuant to Section 3.9. Instead, the entire gain or loss (including any gain or loss attributable to interest payments or default) shall be allocated to the Accounts of the Participant.
 - (B) In the case of any other Participant, the outstanding loan balance may at no time exceed 50% of the outstanding vested balance of the Participant's Accounts; provided, however, that for any loan made after January 1, 2006, a Participant's MAPP Subaccount shall not be included in a Participant's Accounts for this purpose. If such limit is at any time exceeded, or if the Participant fails to make timely repayment, the loan will be treated as in default and become immediately payable in full, subject to rules established by the Committee.

- (c) If the loan, or a loan from another qualified retirement plan maintained by the Company, is to be secured by some or all of the Participant's MAPP Subaccount under the Plan, the Participant and his spouse, if any, must consent to the loan and the possible reduction in the MAPP Subaccount in the event of a setoff of the loan against the MAPP Subaccount balances as a result of nonpayment of the loan. Such consent must be given in writing within a ninety-day period before the Committee makes the loan. In the event the Participant defaults on the loan and Participant's MAPP Subaccount is security for the loan, the MAPP Subaccount balances will not be used to satisfy the loan obligation prior to the earlier of the Participant's termination of employment with the Company or an event resulting in a permissible distribution of his MAPP Subaccount under the Plan. In the event of default, the Company shall offset the amount owed by the Participant against any amounts owed by the Company to the Participant.
- (d) In connection with the making of any loan to a Participant pursuant to the provisions of this Section 11.6, the Participant receiving such a loan may be required to execute such documents as may be required by the Committee and/or Trustee.

11.7 Lapsed Benefits.

- (a) In the event that a benefit is payable under this Plan to a Participant or Beneficiary and after reasonable efforts such individual cannot be located for the purpose of paying the benefit after a reasonable time following the date payment would otherwise have been made, the benefit shall be forfeited and treated like other forfeitures pursuant to the provisions of Section 8.1. If the Participant or Beneficiary later makes a valid claim for the benefit, the claim shall be paid pursuant to directions by the Committee.
- (b) For purposes of this Section 11.7, the term "Beneficiary" shall include any person entitled under Section 6.2 to receive the interest of a deceased Participant or deceased designated Beneficiary and shall also include an Alternate Payee.
- (c) The Accounts of a Participant shall continue to be maintained until the amounts in the Accounts are paid to the Participant or his Beneficiary. In the event that the Plan is terminated and the Committee directs the Trustee to liquidate the Trust and distribute the assets of the Trust Fund, the following rules shall apply:
 - (i) All Participants and Beneficiaries (including Participants and Beneficiaries who have not previously claimed their benefits under the Plan) shall be notified of their right to receive a distribution of their interests in the Plan.
 - (ii) All Participants and Beneficiaries shall be given a reasonable length of time, which shall be specified in the notice, in which to claim their benefits.
 - (iii) All Participants (and their Beneficiaries) who do not claim their benefits within the designated time period shall be presumed to be dead. The Accounts of such Participants shall be forfeited at such time. These forfeitures shall be disposed of according to rules prescribed by the Committee, which rules shall be consistent with applicable law.

- (iv) The Committee shall prescribe such rules as it may deem necessary or appropriate with respect to the notice, deposit or forfeiture rules stated above.
- (d) Should it be determined that the preceding rules relating to forfeiture of benefits upon Plan termination are inconsistent with any of the provisions of the Code and/or ERISA. these provisions shall become inoperative without the need for a Plan amendment and the Committee shall prescribe rules that are consistent with the applicable provisions of the Code and/or ERISA.
- **11.8** Addresses. Each Participant shall be responsible for furnishing the Committee with his correct current address and the correct current name and address of his Beneficiary or Beneficiaries.

11.9 Notices and Communications.

- (a) All applications, notices, designations, elections, and other communications (collectively referred to in this Section 11.9 as "communications") from or to Participants, Beneficiaries and Alternate Payees shall be made in the manner prescribed by the Committee. The Committee may require or permit to be made by means of such written or electronic media as the Committee may prescribe.
- (b) To the extent written communication is provided for by the Committee, an item shall be deemed to have been delivered and received by the Participant, Beneficiary or Alternate Payee when it is deposited in the United States mail with postage prepaid, addressed to the Participant, Beneficiary or Alternate Payee at his last address of record with the Committee.
- (c) The Committee in its absolute discretion may accept or require communications given by facsimile, telex, telegram, telephone or any form of electronic communication that the Committee reasonably believes in good faith to be genuine. If the Committee chooses to accept one or more alternative methods of communication, the Company, any Affiliated Company, any Participant or Beneficiary or any other person providing communication to the Committee will be required to follow reasonable procedures adopted by the Committee for written confirmation. In addition, the Committee may record oral instructions. If any person providing communication to the Committee fails or refuses to comply with the Committee's confirmation procedures, the Committee will be entitled to refuse to comply with such communications without incurring any liability for such refusal.
- 11.10 Governing Law. All legal questions pertaining to the Plan shall be determined in accordance with the provisions of ERISA and, to the extent not preempted by ERISA, the laws of the State of California; provided, however, that if any provision is susceptible to more than one interpretation, such interpretation shall be given thereto as is consistent with the Plan remaining qualified within the meaning of Section 401(a) of the Code. If any provisions of this instrument shall be held by a court of competent jurisdiction to be invalid or unenforceable, the remaining provisions hereof shall continue to be fully effective. All contributions made hereunder shall be deemed to have been made in California.

- **11.11** <u>Interpretation</u>. Article and Section headings are for convenient reference only and shall not be deemed to be part of the substance of this instrument or in any way to enlarge or limit the content of any Article or Section. Unless the context clearly indicates otherwise, masculine gender shall include the feminine, and the singular shall include the plural and the plural the singular.
- 11.12 <u>Withholding For Taxes</u>. Any payments out of the Trust Fund may be subject to withholding for taxes as may be required by any applicable federal or state law.
- **11.13** <u>Limitation on Company, Committee and Trustee Liability</u>. Any benefits payable under this Plan shall be paid or provided for solely from the Trust Fund and neither the Company, the Committee nor the Trustee assume any responsibility for the sufficiency of the assets of the Trust to provide the benefits payable hereunder.
- **11.14** <u>Successors and Assigns</u>. This Plan and the Trust established hereunder shall inure to the benefit of, and be binding upon, the parties hereto and their successors and assigns.
- **11.15** <u>Counterparts</u>. This Plan document may be executed in any number of identical counterparts, each of which shall be deemed a complete original in itself and may be introduced in evidence or used for any other purpose without the production of any other counterparts.
- **11.16** <u>Top-Heavy Plan Requirements</u>. For any Plan Year for which the Plan is a Top-Heavy Plan, as defined in Section A.3 of Appendix A, attached hereto, and despite any other provisions of the Plan to the contrary, the Plan will be subject to the provisions of Appendix A.
- 11.17 <u>Rule 16b-3 Provisions</u>. The Committee may (but need not) adopt such rules and/or take such actions or implement such measures and/or limitations as it deems desirable in order to comply with 17 C.F.R. 140.16b-3, promulgated under Section 16 of the Securities Exchange Act of 1934, as amended (the "SEC Section 16"). Neither the Company, the Board of Directors, the Committee, the Trustee nor the Plan shall have any liability to any Participant in the event that any Participant has any liability under SEC Section 16 due to any rule so adopted, the failure to adopt any rule, any Plan provision (or lack thereof), and transaction in the Plan or otherwise.
- **11.18** <u>Military Service</u>. Notwithstanding any provision of the Plan to the contrary, contributions, benefits and service credit with respect to qualified military service shall be provided in accordance with Section 414(u) of the Code.
- 11.19 Errors, Misstatements, and Payment of Interest. In the event of any misstatement, error, or omission of fact resulting in payment of benefits in an incorrect amount, the Committee shall promptly cause the amount of future payments to be corrected upon discovery of the facts, and shall pay, or authorize others to pay, the Participant or his Beneficiary any underpayment, in cash in a lump sum, or to recoup, or authorize others to proceed against the Participant or his Beneficiary for recovery of any such overpayment. Except as otherwise specifically provided herein, no interest or investment earnings shall be paid on the payment of any corrected amounts.

ARTICLE XII

ESOP FEATURE

12.1 FSOP Contributions.

This Article XII adds an ESOP Feature to the Plan. This ESOP Feature is both a stock bonus plan and an employee stock ownership plan and is intended to qualify under Sections 401(a) and 4975(e)(7) of the Code. As such it is designed to invest primarily in employer securities described in Section 409(1) of the Code. The ESOP Feature shall become effective ("ESOP Effective Date") as of January 1, 2006. Up to 100% of the assets of the ESOP Fund may be invested in Company Stock. An ESOP Contribution is any contribution to the Plan which the Company designates as such. ESOP Contributions shall be wholly invested in Company Stock, unless otherwise directed by the Company or the Committee and except as provided otherwise due to an election under Section 12.2.

12.2 Diversification Election.

- (a) Notwithstanding any other provision of the Plan following the ESOP Effective Date, any Participant who has attained age 55 and completed at least ten (10) years of participation in the Plan (a "Qualified Participant") may elect to have up to 25% of the value in his ESOP Account invested in the funds referred to in Section 12.2(b). Any such Participant may make this election at least once during each Plan Year of his Qualified Election Period. During the sixth Plan Year in the Participant's Qualified Election Period, the Participant may so direct the investment of up to 50% of the value of amounts in his ESOP Account. For this purpose, the term "Qualified Election Period" shall mean the period beginning with the Plan Year in which the Participant becomes a Qualified Participant and ending upon the earlier of the Participant's Termination of Employment or the sixth Plan Year following diversification eligiblity. The percentage of amounts in a Participant's ESOP Account for which he may make investment directions in a Plan Year shall be the "Allocable Portion" of such Participant's Account at the time of the election under this Section 12.2.
- (b) A Participant may elect to have the dollar value of the Allocable Portion of his ESOP Account, on the Valuation Date following the date on which the election is filed, invested by the Trustee in any investments available under the Plan (other than Company Stock) which are maintained by the Trustee and which conform to the requirements of Code Section 401(a)(28)(B). Such election, which shall be made pursuant to procedures established by the Committee, shall be in accordance with any notice, rulings, or regulations or other guidance issued by the Internal Revenue Service with respect to Code Section 401(a)(28)(B). Once such election is made, the Participant cannot subsequently direct that such Allocable Portion (or any income thereon) be reinvested in any portion of the ESOP Fund.

12.3 Dividends on Allocated Shares.

As of and after the ESOP Effective Date, all cash dividends on Company Stock allocated to Participants', Beneficiaries' and Alternate Payees' Accounts may, as determined by the Company consistent with Section 404(k) of the Code, be retained in the Participants,'

Beneficiaries' and Alternate Payees' Accounts, or be paid out to the Participant, Beneficiary or Alternate Payee. The Committee may determine how such dividends may be applied for any Plan Year up to the time when such dividends are finally allocated to the Accounts of Participants as of the last day of the Plan Year.

12.4 Valuation of Stock.

Company Stock held in Participants,' Beneficiaries' and Alternate Payees' Accounts shall be valued as of each Valuation Date, or at the discretion of the Committee, more frequently. All valuations of Company Stock which is not readily tradable on an established securities market shall be made pursuant to Section 3.9(c).

12.5 Tender for Stock.

All tender or exchange decisions with respect to Company Stock held by the Trustee shall be made only by Participants, Beneficiaries and Alternate Payees acting in their capacity as Named Fiduciaries with respect to both allocated and unallocated Shares in accordance with the provisions of Section 9.9(b).

12.6 Voting of Stock.

All voting rights on Company Stock held by the Trustee shall be exercised by the Trustee in accordance with the provisions of Section 9.9(a).

12.7 Put Right.

If, at the time of distribution, Company Stock distributed from the ESOP Fund is not readily tradable on an established market within the meaning of Section 409(h) of the Code and the Treasury Regulations thereunder, such Company Stock shall be subject to a put right in the hands of a Qualified Holder by which such Qualified Holder may sell all or any part of the Stock distributed to him by the ESOP Fund to the Trustee. For this purpose, the determination of whether such Company Stock is readily tradable on an established market shall be made consistent with the rules set forth in Section 3.9(c). Should the Trustee decline to purchase all or any part of the Company Stock put to it by the Qualified Holder, the Company shall purchase the Company Stock that the Trustee declines to purchase. The put right shall be subject to the following conditions:

- (a) The term "Qualified Holder" shall mean the Participant, Beneficiary or Alternate Payee receiving the distribution of such Company Stock, any other party to whom the Company Stock is transferred by gift or by reason of death, and also any trustee of an individual retirement account (as defined under Code Section 408) to which all or any portion of the distributed Company Stock is transferred pursuant to a tax-free "rollover" transaction satisfying the requirements of Sections 402 and 408 of the Code.
- (b) During the 60-day period following any distribution of such Company Stock, a Qualified Holder shall have the right to require the Company to purchase all or a portion of the distributed Company Stock held by the Qualified Holder. The purchase price to be paid

for any such Stock shall be their fair market value determined (1) as of the Valuation Date coinciding with or next preceding the exercise of the put right under this Section 12.2 or, (2) in the case of a transaction between the Plan and a "disqualified person" within the meaning of Section 4975(e)(2) of the Code or a "party in interest" within the meaning of Section 3(14) of ERISA, as of the date of the transaction.

- (c) If a Qualified Holder shall fail to exercise his put right right under Section 12.2(b), the option right shall temporarily lapse upon the expiration of the 60-day period. As soon as practicable following the last day of the Plan Year in which the 60-day option period expires, the Company shall notify the non-electing Qualified Holder (if he is then a shareholder of record) of the valuation of the Company Stock as of that date, During the 60-day period following receipt of such valuation notice, the Qualified Holder shall again have the right to require the Company to purchase all or any portion of the distributed Company Stock. The purchase price to be paid therefor shall be fair market value determined (1) as of the Valuation Date coinciding with or next preceding the exercise of the put right under this Section 12.2(c) or, (2) in the case of a transaction between the Plan and a "disqualified person" within the meaning of Section 4975(e)(2) of the Code or a "party in interest" within the meaning of Section 3(14) of ERISA, as of the date of the transaction.
- (d) The foregoing put rights under Sections 12.2(b) and (c) hereof shall be effective solely against the Company and shall not obligate the Plan or Trust in any manner.
 - (e) In making the determination of fair market value, the Company shall apply the rules set forth in Section 3.9(c).
- (f) The period during which the put right is exercisable does not include any time when a Qualified Holder is unable to exercise it because the Company is prohibited from honoring it by applicable Federal or State laws.
- (g) Except as otherwise required or permitted by the Code, the put rights under this Section 12.2 shall satisfy the requirements of Sections 54.4975-7(b) and 54.4975-1 1(a)(7)(i) of the Treasury Regulations to the extent, if any, that such requirements apply to such put rights.
- (h) Unless otherwise expressly provided by the Committee, a Qualified Holder must exercise his put right in writing. If a Qualified Holder exercises his put right under this Section 12.2, payment for the Company Stock repurchased shall be made, in the case of a distribution of a Participant's Account within one taxable year, in substantially equal annual payments over a period beginning not later than 30 days after the exercise of the put right and not exceeding five years (provided that adequate security and reasonable interest are provided with respect to unpaid amounts) or, in the case of other distributions, not later than 30 days after such exercise.

ARTICLE XIII

DEFINITIONS

- 13.1 <u>Definitions</u>. Whenever the following terms are used in this Plan, with the first letter capitalized, they shall have the meanings specified below.
- "Account" or "Accounts" shall mean the accounts maintained by the Committee for each Participant that are credited with the amounts provided for herein. The following Accounts are maintained under this Plan: Plan Account, Matching Contribution Account, Rollover Account, Alternate Payee Account, Profit Sharing Account and Voluntary Account. In addition, the following Accounts have been transferred from the ESRP and, except for the ESRP Plan Account, are maintained under this Plan: ESRP Plan Account, TRASOP Account, CODA Account, PSRP Profit Sharing Account and ESOP Alternate Payee Account.
- "Additional Company Contribution" shall mean a contribution by the Company pursuant to Section 3.1 that is intended to qualify as a qualified nonelective contribution pursuant to Code Section 401(k)(3)(D)(ii)(II).

"Affiliated Company" shall mean:

- (a) Any corporation that is included in a controlled group of corporations, within the meaning of Code Section 414(b), that includes the Company;
- (b) Any trade or business that is under common control with the Company within the meaning of Code Section 414(c);
- (c) Any member of an affiliated service group, within the meaning of Code Section 414(m), that includes the Company; and
- (d) Any entity required to be included under Code Section 414(o).
- "Alternate Payee" shall mean an individual awarded a portion of a Participant's benefits under the Plan pursuant to a qualified domestic relations order, as defined in Code Section 414(p) and Section 11.5 of the Plan. Any limitation or condition imposed by the Plan upon a Participant or his rights hereunder shall, unless expressly indicated otherwise, also serve to limit or condition the rights of an Alternate Payee of the Participant's Account(s).
- "Alternate Payee Account" shall mean the account opened and maintained by the Committee for each Alternate Payee who is awarded an interest in a Participant's benefits under the Plan pursuant to the provisions of Sections 8.6 and 11.5.
 - "Anniversary Date" shall mean the last day of each Plan Year.
 - "Applicable Valuation Date" shall mean the most recent date on which the Trust Assets were valued in accordance with the rules of Article III.

"Annual Addition" shall mean the sum credited to a Participant's Accounts for any Plan Year of (i) Company contributions, (ii) voluntary contributions, (iii) forfeitures, (iv) amounts credited after March 31, 1984 to an individual medical account, as defined in Code Section 415(1)(2) which is part of a "Defined Benefit Plan" maintained by the Company, and (v) amounts derived from contributions paid or accrued after December 31, 1985, in taxable years ending after such date, which are attributable to post-retirement medical benefits allocated to the separate account required with respect to a key employee (as defined in Section A.3 of Appendix A to the Plan) under a welfare benefit plan (as defined in Code Section 419(e)) maintained by the Company.

"Beneficiary" or "Beneficiaries" means the person or persons designated in Section 6.2 to receive the interest of a deceased Participant.

"Board of Directors" shall mean the Board of Directors (or its delegate, to the extent the duties of the Board of Directors are delegated to such person) of Science Applications International Corporation (or its ultimate parent corporation, if any) as it may from time to time be constituted.

"Break in Service" shall mean, with respect to an Employee, a computation period (determined under the definition of "Year of Service" in this Section 13.1), in which the Employee completes no more than 425 Hours of Service.

"Catch-Up Contribution" shall mean a contribution paid to the Trustee by the Company at the election of a Participant who has attained at least age 50 before the last day of the Plan Year and which is not reclassified as an Elective Deferral pursuant to definition of "Elective Deferrals" in this Section 13.1 for the Plan Year. Effective January 1, 2002, an eligible Participant may elect to make a Catch-Up Contribution to be withheld from his Compensation by the Company in lieu of receiving such amount as current Compensation. Such amount shall be nonforfeitable when made. Catch-Up Contributions shall be made in accordance with, and subject to the limitations of, Code Section 414(v). Catch-Up Contributions shall not be taken into account for purposes of complying with the required limitations of Code Sections 402(g) and 415(c). The Plan shall not be treated as failing to satisfy the provisions of Code Sections 401(k)(3), 401(k)(11), 401(k)(12), 410(b), or 416, as applicable, by reason of the making of such Catch-Up Contributions.

"CODA Account" shall mean the CODA Account formerly maintained under the ESRP, which has been transferred to this Plan.

"Code" shall mean the Internal Revenue Code of 1986, as in effect on the date of execution of this Plan document and as thereafter amended from time to time.

"Committee" shall mean the Science Applications International Corporation Retirement Plans Committee or any successor thereof.

"Company" shall mean Science Applications International Corporation (or its ultimate parent corporation, if any), or any successor thereof, if its successor shall adopt this Plan. In

addition, unless the context indicates otherwise, as used in this Plan the term Company shall also mean and include any Affiliated Company (or other entity) that has been granted permission by the Board of Directors to participate in this Plan. This permission shall be granted upon such terms and conditions as the Board of Directors deems appropriate.

"Company Contributions" shall mean all amounts (whether in cash or other property, including Company Stock) paid by the Company into the Trust Fund established and maintained under the provisions of this Plan for the purpose of providing benefits for Participants and their Beneficiaries, and shall include Elective Deferrals, Qualified Matching Contributions, Additional Company Contributions, Company Matching Contributions, Catch-Up Contributions, Profit Sharing Contributions, ESOP Contributions, Voluntary Contributions and any Company Contributions to pay Plan expenses as provided in Section 8.3.

"Company Matching Contributions" shall mean an amount contributed to the Plan by the Company or by an Affiliated Company in accordance with Section 3.3 of the Plan

"Company Stock" shall mean Class A Common Stock par value \$.01 per share ("Class A Common Stock"), and to the extent accounted for pursuant to Section 7.2 of the Plan and solely for such purpose, shall also mean Class B Common Stock, par value \$.05 per share ("Class B Common Stock"), of the Company, or any other security of the Company or the Company's ultimate parent corporation, if any, designated as Company Stock by the Committee, including Preferred Stock.

"Company Stock Fund" shall have the meaning set forth in Section 7.2(a) of the Plan.

"Compensation" shall mean:

(a) Annual Additions and Top-Heavy Status.

- (i) For purposes of applying the limitation on Annual Additions pursuant to Article IV of the Plan and determining whether the Plan is top heavy under Appendix A, "Compensation" shall include all of the following:
 - (A) An Employee's wages, salaries, fees for professional services and other amounts received (without regard to whether or not an amount is paid in cash) for personal services actually rendered in the course of employment with the Company to the extent that the amounts are includible in gross income including, but not limited to, commissions paid salespersons, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, bonuses, fringe benefits, and reimbursements or other expense allowances under a nonaccountable plan, as described in Income Tax Regulations, section 1.62-2(c);
 - (B) Amounts described in Code Sections 104(a)(3), 105(a) and 105(h), but only to the extent that these amounts are includible in the gross income of the Employee;

- (C) Amounts paid or reimbursed by the Company for moving expenses incurred by an Employee, but only to the extent that at the time of payment it is reasonable to believe that these amounts are not deductible by the Employee under Code Section 217;
- (D) The value of a nonqualified stock option granted to an Employee by the Company, but only to the extent that the value of the option is includible in the gross income of the Employee for the taxable year in which granted;
 - (E) The amount includible in the gross income of the Employee upon making the election described in Code Section 83(b); and
- (F) Elective Deferrals under Code Section 402(g)(3), Catch-Up Contributions under Code Section 414(v), and amounts contributed under Code Sections 125 and 132(f) by the Company at the Employee's election that are not included in the Employee's gross income.

Paragraphs (A) and (B) above shall include foreign earned income as defined in Code Section 911(b), whether or not excludible from gross income under Code Section 911. Compensation described in paragraph (A) above is to be determined without regard to the exclusions from gross income in Code Sections 931 and 933. Similar principles will be applied with respect to income subject to Code Sections 931 and 933 in determining Compensation described in Paragraph (B) above.

- (ii) For purposes of this subsection (a), "Compensation" shall not include items such as:
- (A) Company contributions to a plan of deferred compensation that are not includible in the Employee's gross income for the taxable year in which contributed, or Company contributions under a simplified employee pension within the meaning of Code Section 408(k) to the extent such contributions are excludible from gross income by the Employee, or any distributions from a plan of deferred compensation; however, any amounts received by an Employee pursuant to an unfunded nonqualified plan of deferred compensation are permitted to be considered as Compensation for Code Section 415 purposes in the year the amounts are includible in the gross income of the Employee;
- (B) Amounts realized from the exercise of a nonqualified stock option, or when restricted stock (or property) held by the Employee either becomes freely transferable or is no longer subject to a substantial risk of forfeiture;
 - (C) Amounts realized from the sale, exchange or other disposition of stock acquired under a qualified or incentive stock option; and
- (D) Other amounts that received special tax benefits, such as premiums for group-term life insurance (but only to the extent that the premiums are not includible in the gross income of the Employee) or Company Contributions (whether or not under a salary reduction agreement) towards the purchase of an annuity described in Code Section 403(b) (whether or not the amounts are actually excludible from the gross income of the Employee).

- (b) <u>Highly Compensated Employees</u>. For purposes of determining who is a Highly Compensated Employee, "Compensation" shall be "Compensation" as defined in subsection (a) above, but determined without regard to Code Sections 125, 132(f), 402(g)(3) and 402(h)(1)(B).
 - (c) Testing. For purposes of Section 3.1(e) and Section 3.3(c), "Compensation" shall mean the amount determined under subsection (a).
- (d) <u>Elective Deferrals and Catch-Up Contributions</u>. For purposes of determining the amount that may be deferred as provided in Section 3.1 and the amount of the Company Matching Contributions under Section 3.3, "Compensation" shall mean the Participant's salary or wages (as otherwise would be reflected in his regular payroll stub before reductions by amounts the Participant elects to defer pursuant to Section 3.1) plus one hundred percent (100%) of any cash bonuses paid to the Participant (or payable, but for an election by a Participant pursuant to Section 3.1) and exclusive of (i) any stock bonus regardless of whether it was fully vested when awarded and (ii) any taxable assignment allowances paid after December 31, 1992 to any Employee who is on assignment away from normal work location(s) and entitled to receive one or more taxable allowances or adjustments to Compensation based on such assignment.
- (e) <u>Profit Sharing</u>. For purposes of determining the allocation of Profit Sharing Contribution and ESOP Contribution pursuant to Section 3.4 and forfeitures of Profit Sharing Contribution and ESOP Contribution pursuant to Section 8.1, "Compensation" shall mean the amount of compensation paid by the Company during a calendar year by reason of services performed by an Employee reflected as "wages, tips, other compensation" on the Employee's Form(s) W-2 for such year; plus
 - (i) Effective on and after January 1, 2004, Elective Deferrals under Code Section 402(g)(3), Catch-Up Contributions under Code Section 414(v), and amounts contributed under Code Sections 125 and 132(f) by the Company at the Employee's election that are not included in the Employee's gross income; plus
 - (ii) Any Compensation which, but for Code Section 3401(a)(8)(A) (dealing with the section 911 exclusion and income subject to foreign withholding) would be required to be reflected as "wages, tips, other compensation" on the Employee's Form(s) W-2; less
 - (iii) Any Compensation paid by reason of services performed during any period in which the Employee is not a Participant under this Plan or is not an Eligible Employee; overtime pay (which shall be deemed to include base pay and premium pay for time worked in excess of a normal day or week); bonuses; commissions; and amounts reflecting reimbursed expenses or fringe benefits (including any amount relating to the grant or exercise of stock options or disposition of shares through exercise of options) which have been included as "wages, tips, compensation" on the Employee's Form(s) W-2.
 - (iv) Notwithstanding the foregoing, effective January 1, 1993, for purposes of determining the allocation of Profit Sharing Contribution and ESOP Contribution pursuant to Section 3.4 and forfeitures pursuant of Profit Sharing Contribution and ESOP Contribution

pursuant to Section 8.1 with respect to any Employee who is on assignment away from the normal work location(s) and entitled to receive one or more taxable allowances or adjustments to Compensation based on such assignment, "Compensation" shall mean the amount of Compensation paid by the Company during a calendar year by reason of services performed by such Employee which constitutes base salary (including comprehensive leave, vacation and holiday pay) less any amounts deferred pursuant to an Employee's election under Section 3.1 and unadjusted for any taxable assignment allowances.

- (f) <u>Compensation Limit</u>. For Plan Years after 2001, the amount considered as Compensation for any purpose hereunder shall be limited to \$200,000, adjusted at the same time and in the same manner as under Section 401(a)(17) and 415(d) of the Code.
 - "Compensation Committee" shall mean the subcommittee of the Board of Directors with specified authority regarding Plan matters.
 - "Delegated Fiduciary" shall have the meaning set forth in Section 9.7(a).
- "Disability" or "Disabled" shall mean the status of disability determined conclusively by the Committee based on certification of disability by the Social Security Administration, effective upon receipt of such certification by the Committee.
- "Distributable Benefit" shall mean the Vested Interest of a Participant in this Plan that is determined and distributable to him in accordance with the provisions of Article VI and which shall include his Vested Interest in his Plan Account, Matching Contribution Account, Profit Sharing Account, Voluntary Account, Rollover Account, ESOP Account, TRASOP Account, CODA Account and PSRP Profit Sharing Account, if any. In the case of an Alternate Payee, the Distributable Benefit shall mean the Vested Interest in the Alternate Payee Account and ESOP Alternate Payee Account, if any.

"Effective Date" shall mean the original effective date of this Plan, January 1, 1983 (with the original effective dates of the constituent plans which merged to create this Plan as set forth in Section 1.1). The effective date of any prior or subsequent amendments or restatements is the date specified therein or in any accompanying resolutions adopting such amendment. The rights of an Employee who has terminated employment shall be governed by the terms of the Plan in effect at the time of such termination, unless otherwise specified in any subsequent amendment.

"Elective Deferrals" shall mean contributions paid over by the Company to the Trustee at the election of a Participant in lieu of cash Compensation pursuant to Section 3.1. Except as otherwise provided in the following sentence, no amount that was designated or treated at the time of deferral or contribution as a Catch-Up Contribution shall be treated as an Elective Deferral. If a Participant's Elective Deferrals for a Plan Year are less than both the dollar limitation of Code Section 402(g) and the percentage limitation of Plan section 3.1(a), then all or part of the Participant's Catch-Up Contribution shall be reclassified as an Elective Deferral to the extent of the unused dollar limitation or percentage limitation, whichever limit applies first.

"Eligible Employee" shall include any Employee except as follows:

- (a) Any Employee who is covered by a collective bargaining agreement to which the Company or an Affiliated Company is a party if there is evidence that retirement benefits were the subject of good faith bargaining between the Company (or an Affiliated Company) and the collective bargaining representative, unless the collective bargaining agreement provides for coverage under this Plan.
 - (b) Any Employee of an Affiliated Company who has not been granted permission by the Board of Directors to participate in this Plan.
- (c) Any Employee who is a nonresident alien and who receives no earned income (within the meaning of Code Section 911(d)(2)) from the Company which constitutes income from sources within the United States (within the meaning of Code Section 861(a)(3)), unless the Employee is within a group or classification of nonresident alien Employees designated as eligible to participate in the Plan by the Board of Directors or its delegate.
 - (d) Any Employee who is a Leased Employee.
- (e) Any individual who is not classified for the relevant eligibility period by the Company or Affiliated Company on its payroll records as an employee of the Company or Affiliated Company under Code Section 3121(d) (including, but not limited to an individual classified by the Company or Affiliated Company as an independent contractor, a non-employee consultant or as an employee of an entity other than the Company or Affiliated Company), even if such classification is determined to be erroneous, or is retroactively revised by a governmental agency, by court order or as a result of litigation, by the Company or otherwise. In the event the classification of an individual who was excluded from the definition of Eligible Employee under the preceding sentence is determined to be erroneous or is retroactively revised, the individual shall nonetheless continue to be excluded from treatment as an Eligible Employee for all periods prior to the date the Company determines that its classification of the individual was erroneous or should be revised or is otherwise required to revise such classification.

"Eligible Participant" shall have the meaning set forth in Section 3.4(c).

"Employee" shall mean each person currently employed as a common-law employee of the Company or Affiliated Company. For services performed after December 31, 1986 for purposes of determining the number or identity of Highly Compensated Employees and for purposes of the requirements of Code Sections 414(n)(3)(A) and (B), the term "Employee" shall include any person who is a Leased Employee unless (i) such Leased Employees constitute less than 20% of the Company's nonhighly compensated workforce within the meaning of Code Section 414(n)(5)(C)(ii) and (ii) such person is covered by a plan meeting the requirements of Code Section 414(n)(5)(B).

"Employment Commencement Date" shall mean each of the following:

(a) For New Employees. The date on which an Employee first performs an Hour of Service in any capacity for the Company or an Affiliated Company with respect to which the

Employee is compensated or is entitled to compensation by the Company or the Affiliated Company.

- (b) <u>For Rehired Employees</u>. In the case of an Employee whose employment is terminated and who is subsequently reemployed by the Company or an Affiliated Company, the term "Employment Commencement Date" shall mean one of the following:
 - (i) for an Employee reemployed before incurring a Break in Service, the day on which the Employee first performed an Hour of Service for the Company or an Affiliated Company with respect to which the Employee is compensated or entitled to compensation by the Company or the Affiliated Company; or
 - (ii) for an Employee reemployed following a Break in Service of one year or more, the first day following the termination of employment on which the Employee performs an Hour of Service for the Company or an Affiliated Company with respect to which he is compensated or entitled to compensation by the Company or Affiliated Company.
- (c) <u>Affiliated Company Employees</u>. Unless the Board of Directors or its delegate shall expressly determine otherwise, an Employee shall not, for the purposes of determining his Employment Commencement Date, be deemed to have commenced employment with an Affiliated Company prior to the effective date on which the entity became an Affiliated Company.
 - "Entry Date" shall mean January 1, April 1, July 1 and October 1 of each Plan Year.
 - "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time.
 - "ESOP Alternate Payee Account" shall mean the Alternate Payee Account formerly maintained under the ESRP, which has been transferred to this Plan.
 - "ESOP Account" shall mean that portion of the Accounts of a Participant, Beneficiary or Alternate Payee which is invested in Company Stock.
- **"ESOP Contribution"** shall mean all amounts, whether in cash or other property, including Company Stock, paid by the Company into the Trust Fund as ESOP Contributions in accordance with the provisions of Section 3.4.
- **"ESOP Feature"** shall mean that portion of the Plan consisting of an employee stock ownership plan intended to satisfy the requirements of Section 4975(e)(7) of the Code, as set forth in Article XII of the Plan.
- **"ESOP Fund"** shall mean that portion of the Plan which consists of Company Stock, and any income thereon. The ESOP Fund shall include all Company Stock contributed to or held under the Plan. The non-ESOP portion of the Plan shall consist of any assets of the Plan not maintained under the ESOP Fund.

"ESRP Plan Account" shall mean the Plan Account formerly maintained under the ESRP, which has been transferred to this Plan. Amounts credited to a Participant's ESRP Plan Account as of the time of the transfer to this Plan shall be credited to the Participant's ESOP Account in this Plan to the extent invested in Company Stock and shall otherwise be credited to the Participant's Profit Sharing Account in this Plan.

"Excess Compensation" shall mean, with respect to Plan Years beginning prior to January 1 2004, all of a Participant's Compensation as defined in subsection (e) of the definition of "Compensation" set forth in this Section 13.1) for a particular calendar year in excess of the contribution and benefit base (taxable wage base) under section 230 of the Social Security Act in effect at the beginning of the Plan Year. In determining whether any portion of an Employee's Compensation qualifies as Excess Compensation, (i) amounts paid to the Employee while he was not a Participant in this Plan shall be taken into account only to the extent those amounts were paid to the Employee while he was a Participant in another profit sharing plan maintained by the Company which is integrated with Social Security ("integrated profit sharing plan") and (ii) the Excess Compensation shall be attributed as between this Plan and such other integrated profit sharing plan based on the crediting of compensation on a first-in-time earned basis to both such plans, and attributing the Excess Compensation to the plan (or plans) in which the Participant participated when (and to the extent) he earned the Excess Compensation.

"Fiduciary" shall mean all persons defined in Section 3(21) of ERISA associated in any manner with the control, management, operation, and administration of the Plan or the assets of the Plan, and such term shall be construed as including the term "Named Fiduciary" with respect to those Fiduciaries named in the Plan or who are identified as Fiduciaries pursuant to procedures specified in the Plan.

"5% Owner" shall mean an individual who owns (or is considered as owning, within the meaning of Code Section 318) more than 5% of the outstanding stock of the Company or stock possessing more than 5% of the total combined voting power of all stock of the Company.

"Fringe Rate Group" shall mean the members of an operating group, a division, a subsidiary or another business unit of the Company and the Affiliated Companies, including but not limited to employees of a business unit working under the terms of a specific contract(s) with a client or customer of the Company, as designated by the Company in its sole discretion.

"Highly Compensated Employee" shall mean, with respect to Plan Years beginning after December 31, 1996 any Employee who:

- (a) was at any time a 5% Owner during the Plan Year or the preceding Plan Year or;
- (b) for the preceding Plan Year
 - (i) had Compensation in excess of \$80,000, as adjusted pursuant to Section 415(d) of the Code, and;
- (ii) if the Company elects the application of this clause for such preceding Plan Year, was in the "top-paid group" of Employees for such preceding Plan Year.

For purposes of this definition, the "top paid group" shall be the group consisting of the top 20 percent of the Employees when ranked on the basis of Compensation paid during such Plan Year.

Highly Compensated Employees shall also include any former Employee who was an active Highly Compensated Employee for either (1) the year in which he separated from service, or (2) at any time after such Employee's fifty-fifth birthday.

"Hour of Service"

- (a) "Hour of Service" of an Employee shall mean the following:
- (i) Each hour for which the Employee is paid by the Company or an Affiliated Company (while it is an Affiliated Company) or entitled to payment for the performance of services as an Employee. An Employee will not be considered as being entitled to payment, for purposes of determining the computation period to which hours are to be credited, until the date the Company or Affiliated Company, as applicable, would normally make payment to the Employee for such hour based on normal payroll practices.
- (ii) Each hour in or attributable to a period of time during which the Employee performs no duties (irrespective of whether he has terminated his Employment) due to a vacation, holiday, illness, incapacity (including pregnancy or disability), layoff, jury duty, military duty or a leave of absence, for which he is so paid or so entitled to payment, whether direct or indirect. However, no such hours shall be credited to an Employee if such Employee is directly or indirectly paid or entitled to payment for such hours and if such payment or entitlement is made or due under a plan maintained solely for the purpose of complying with applicable workmen's compensation, unemployment compensation or disability insurance laws or is a payment which solely reimburses the Employee for medical or medically related expenses incurred by him.
- (iii) Each hour for which an Employee is entitled to back pay, irrespective of mitigation of damages, whether awarded or agreed to by the Company or an Affiliated Company, provided that such Employee has not previously been credited with an Hour of Service with respect to such hour under paragraphs (i) or (ii) above.
- (iv) The term "Hour of Service" shall also include periods during which an Employee who was on an authorized noncompensated leave of absence as of December 31, 1987 continues on such authorized noncompensated leave of absence provided the Employee returns to the employ of the Company or an Affiliated Company immediately upon the termination of such leave of absence and provided that, for purposes of Section 5.1 (relating to vesting) no hours shall be credited pursuant to this paragraph (iv) until the Employee completes 850 Hours of Service (excluding any additional leaves of absence) in the twelve-month period beginning with the Employee's Employment Commencement Date immediately following such leave of absence.
- (v) The term "Hour of Service" shall also include (for those purposes designated by the applicable officer specified below) hours credited to an Employee for service

with a predecessor employer provided that such service has been approved by the President, Chief Operating Officer, Senior Vice President and Director of Human Resources or Chief Administrative Officer of the Company for recognition under this Plan, which approval shall apply on a nondiscriminatory basis to all Employees with service during the recognition period for such predecessor employer as calculated according to Committee rules.

- (b) Hours of Service under subsections (a)(ii), (a)(iii), and (a)(iv) shall be calculated in accordance with Department of Labor Regulation 29 C.F.R. Section 2530.200b-2(b). Hours of Service shall be credited to the appropriate computation period according to the Department of Labor Regulation 29 C.F.R. Section 2530.200b-2(c).
- (c) In the event that an Employee receives credit for Hours of Service for a period during which no duties are performed (including sick leave, jury duty, vacations or other paid non-work periods), the Employee shall be deemed to have completed Hours of Service for each day or portion thereof during that period on the basis of what he would have been credited if he had worked his normal work schedule during such non-work period.
- (d) To the extent not otherwise credited under this definition, Hours of Service determined with respect to a Maternity or Paternity Absence shall be credited as follows: the Employee shall be credited (solely for purposes of determining whether a Break in Service has occurred) with those Hours of Service that otherwise would normally have been credited to such Employee but for such absence, except that (i) the total number of Hours of Service so credited shall not exceed 426 and (ii) such Hours of Service shall be credited as Hours of Service in the Plan Year in which the absence from work commences only if the Employee would be prevented from incurring a Break in Service in such Plan Year solely by virtue of such crediting, and shall otherwise be credited in the Plan Year immediately following the Plan Year in which the absence from work commences.
- (e) All Hours of Service and Years of Service relating to periods prior to January 1, 1990, shall be determined under the provisions of the PSRP or the CODA, as applicable, as in effect prior to January 1, 1990.

"Investment Manager" shall mean the one or more Investment Managers, if any, appointed pursuant to Section 9.3(b).

- "Leased Employee" shall mean any person other than an Employee, who pursuant to an agreement between the Employer and a "leasing organization" has performed services for the Employer (or for the Employer and related persons determined in accordance with Code Section 414(n)(6)) on a substantially full time basis for a period of at least one year, and such services were performed under the primary direction or control of the Employer in accordance with Code Section 414(n)(2)(C). A Leased Employee shall not be considered an Employee of the Employer if:
 - (a) the Leased Employee is covered by a money purchase pension plan providing:
 - (i) a nonintegrated employer contribution rate of at least 10 percent of compensation, as defined in Code Section 415(c)(3), but including amounts contributed pursuant to a salary reduction agreement which are excludable from the employee's gross income under Code Sections 125, 402(e)(3), 402(h) (1)(B) or 403(b);

- (ii) immediate participation; and
- (iii) full and immediate vesting; and
- (b) Leased Employees do not constitute more than 20 percent of the Nonhighly Compensated Employees of the Employer.
- "Limitation Year" shall mean the calendar year.
- "MAPP Subaccount" shall mean the account maintained by the Committee containing the account of a Participant with a balance in his Plan Account under the Science Applications International Corporation Profit Sharing Plan II which became fully vested as of August 1, 1987 as a result of the conversion of such Plan into a profit sharing plan, and any subsequent allocations thereto pursuant to Section 6.4 of the Plan.
 - "Match Eligible Compensation" shall mean the compensation earned while in a Fringe Rate Group that is eligible for Company Matching Contributions.
- "Matching Contribution Account" shall mean the account opened and maintained by the Committee for a Participant who is or has been allocated Company Matching Contributions.
- "Maternity or Paternity Absence" shall mean an absence from work for any period by reason of (a) an Employee's pregnancy, (b) the birth of a child of such Employee, (c) the placement of a child with the Employee in connection with the adoption of such child by such Employee, or (d) the caring for a natural or adopted child for a period beginning immediately following such birth or placement.
 - "Nonhighly Compensated Employee" shall mean an Employee who is not a Highly Compensated Employee.
- "Normal Retirement" shall mean a Participant's termination of employment with the Company as a result of such Participant attaining his Normal Retirement Date.
- "Normal Retirement Date" shall be the day on which a Participant attains age 59-1/2; provided, however, that if a Participant continues to be employed by the Company beyond the age of 59-1/2, then the date on which the Participant's employment with the Company terminates shall be referred to as the Participant's Normal Retirement Date.
- **"Participant"** shall mean any Eligible Employee who has satisfied the eligibility requirements set forth in Section 2.1 and has been enrolled in this Plan in accordance with the provisions of Section 2.2.

- "Participation Commencement Date" shall mean the day on which an Employee's participation in this Plan commences in accordance with Section 2.2.
- **"Plan"** shall mean the Science Applications International Corporation Retirement Plan (formerly referred to as the Science Applications International Corporation 401(k) Profit Sharing Plan) herein set forth, and as it may be amended from time to time.
- **"Plan** Account" shall mean the account opened and maintained by the Committee for each Participant for purposes of holding and accounting for Company Contributions representing Elective Deferrals, Qualified Matching Contributions, Catch-Up Contributions or Additional Company Contributions held in the Plan Accounts and allocated to Participants.
- "Plan Administrator" shall mean the administrator of the Plan, within the meaning of Section 3(16)(A) of ERISA. The Plan Administrator shall be the Company.
 - "Plan Year" shall mean the calendar year.
- "Preferred Stock" shall mean preferred stock of the Company (or its ultimate parent corporation, if any) convertible into Company Stock in accordance with the rules set forth in Section 409(1) of the Code.
- **"Profit Sharing Account"** shall mean the account opened and maintained by the Committee for each Participant in the Plan who is or has been allocated separately Profit Sharing Contributions or ESOP Contributions. If applicable, the Profit Sharing Account shall be subdivided into subaccounts to reflect separately the MAPP Subaccount and all other profit sharing allocations. Furthermore, any such other subaccounts as are necessary or appropriate for the operation of the Plan may be established by the Committee.
- **"Profit Sharing Contribution"** shall mean all amounts, whether in cash or other property, including Company Stock, paid by the Company into the Trust Fund as Profit Sharing Contributions in accordance with the provisions of Section 3.4.
- "PSRP Profit Sharing Account" shall mean the Profit Sharing Account formerly transferred from the PSRP to the ESRP and subsequently maintained under the ESRP, which has now been transferred to this Plan.
- "Qualified Matching Contribution" shall mean a contribution made on account of a Participant's Elective Deferrals pursuant to Section 3.1 and which is intended to qualify as a matching contribution pursuant to Code Section 401(k)(3)(D)(ii)(I) and which is designated by the Committee to be included in the Deferral Percentage unless and to the extent designated by the Committee to be included, instead, in the Contribution Percentage.
- "Required Payment Commencement Date" in the case of a Participant other than a 5% Owner, shall be the sixtieth day after the close of the latest Plan Year in which occurs:
 - (a) The Participant's Normal Retirement Date;

- (b) The tenth anniversary of the date the Participant commenced participation in the Plan; or
- (c) The Participant's termination of employment with the Company or an Affiliated Company, unless a Participant elects, pursuant to procedures established by the Committee (which procedures may provide that no affirmative election is treated as an election to defer distribution), to defer distribution until no later than April 1 following the calendar year in which the Participant attains age 70-1/2.
- (d) In the case of a 5% Owner, the Required Payment Commencement Date shall be April 1 following the calendar year in which the Participant attains age 70-1/2, whether or not the Participant has retired. If a Participant becomes a 5% Owner after attaining age 70-1/2, the Required Payment Commencement Date shall not be later than the last day of the calendar year in which the Participant becomes a 5% Owner.

"Rollover Account" shall mean the account opened up and maintained by the Committee for a Participant who has made a Rollover Contribution or planto-plan transfer to the Plan.

"Rollover Contribution" shall mean a contribution to the Plan by an Eligible Employee of amounts which qualify for exclusion from gross income pursuant to Code Section 402(a), subject to the timing, maximum amount limitations, and other requirements of Code Section 402(a) from a qualified plan or an annuity contract described in Code Sections 403(a), or (b), from an eligible plan described in Code Section 457(b) that is maintained by a state, political subdivision of a state, or agency or instrumentality of a state or political subdivision of a state, and the portion of a distribution from an individual retirement account or annuity that is described in Code Sections 408(a) or (b) funded exclusively with a distribution from another retirement plan qualified under Code Sections 401 (a), 403(a) or (b), or 457(b) and earnings thereon (also known as a conduit IRA) that is eligible to be rolled over and would otherwise be includible in gross income. An Eligible Employee may also be the surviving spouse or a spouse or former spouse who is an Alternate Payee under a qualified domestic relations order, as defined in Code Section 414(p).

"Suspended Participant" shall mean any Participant who remains an Employee but who ceases to be eligible to participate in this Plan by virtue of ceasing to be an Eligible Employee. Status as a Suspended Participant shall commence as of the date such Participant ceases to be an Eligible Employee. A Suspended Participant shall not be deemed a Participant except for those purposes specified in this Plan or as required by law.

"Trade Date" shall mean, if there is a "limited market" for the Company Stock, one of the four predetermined times in each calendar year (one each calendar quarter) during which Company Stock may be bought or sold in accordance with the terms and conditions of the "limited market" as set forth in the Company's prospectus, or, if there is no "limited market" for the Company Stock, the four predetermined date(s) in each calendar year (one each quarter) during which the Committee, in its sole discretion, elects to conduct a transaction with respect to the assets of the Plan. If Company Stock is readily tradable on an established securities market (or is Preferred Stock convertible into Company Stock that is readily tradable on an established

securities market), then rules established by the Committee may designate any or all business days as Trade Dates.

"TRASOP Account" shall mean the TRASOP Account formerly maintained under the ESRP, which has been transferred to this Plan.

"Trust" or **"Trust** Fund" shall mean the one or more trusts created for funding purposes under the Plan. The Trust Fund maybe commingled for investment purposes with the assets of other qualified retirement plans maintained by the Company by investing through a master trust fund operated pursuant to a master trust agreement between the Company and the Trustee.

"Trustee" shall mean, as of February 1, 1998, Vanguard Fiduciary Trust Company or any successor or other corporation acting as a trustee of the Trust Fund.

"Valuation Date" shall mean the date as of which the Trustee shall determine the value of the Assets in the Trust Fund and the value of each Account, which shall be the last day of each Plan Year and such other dates as may be determined in rules prescribed by the Committee, which rules may prescribe different dates for the various investment funds comprising the Trust Fund.

"Vested Interest" shall mean the interest of **a** Participant's Accounts which has become vested in accordance with the rules of Article V. The Vested Interest of an Alternate Payee in his or her Alternate Payee Account shall be determined as set forth in Section 5.3.

"Voluntary Account" shall mean the account opened and maintained by the Committee for each Participant in the Plan who has made Voluntary Contributions to the PSRP.

"Voluntary Contributions" shall mean the nondeductible employee contributions made by the Participant to the Plan in accordance with the rules of Article III

"Year of Service" shall mean a computation period during which the Employee completes 850 or more Hours of Service. In no instance will an Employee receive more than one Year of Service with respect to services performed in a single computation period.

- (a) For purposes of determining eligibility to receive Profit Sharing Contributions and ESOP Contributions under Article III the relevant computation period shall be determined in accordance with the following rules:
 - (i) An Employee's initial computation period shall be the twelve-consecutive-month period beginning on the Employee's Employment Commencement Date.
 - (ii) An Employee's second (and all subsequent) computation periods shall be the calendar years commencing with the calendar year which includes (or starts on the same day as) the first anniversary of the Employee's Employment Commencement Date.
 - (b) For purposes of vesting, the relevant computation period in all cases shall be the calendar year.

APPENDIX A

TOP-HEAVY PROVISIONS

- **A.1** <u>Application</u>. If the Plan is or becomes top-heavy in any Plan Year, the provisions of this Appendix A will supersede any conflicting provisions in the Plan.
 - A.2 Criteria. The Plan shall be top-heavy for any Plan Year if any of the following conditions exist:
 - (a) The Top-Heavy Ratio for the Plan exceeds 60% and the Plan is not part of any Required Aggregation Group or Permissive Aggregation Group of Plans.
 - (b) The Plan is part of a Required Aggregation Group of plans, but not part of a Permissive Aggregation Group, and the Top-Heavy Ratio for the group of plans exceeds 60%.
 - (c) The Plan is part of a Required Aggregation Group and part of a Permissive Aggregation Group of plans and the Top-Heavy Ratio for the Permissive Aggregation Group exceeds 60%.
 - A.3 <u>Definitions</u>. As used in this Appendix A, the following terms shall have the meanings specified below.
 - (a) <u>Determination Date</u>: With respect to any Plan Year, (i) the Determination Date shall be the last day of the preceding Plan Year, or (ii) in the case of the first Plan year of the Plan, the last day of such Plan Year.
 - (b) <u>Key Employee</u>: Any Employee or former Employee (and the Beneficiaries of such Employees) who, pursuant to the rules of Code Section 416(i) and the Regulations thereunder, is or was:
 - (i) An officer of the Company having an annual Compensation greater than 50% of the dollar limitation under Code Section 415(b)(1)(A) (for Determination Dates on and after January 1, 2002, Compensation of greater than \$130,000 (as adjusted under Code Section 416(i)(1)));
 - (ii) For Determination Dates on or before December 31, 2001, one of the ten Employees having annual Compensation from the Company of more than the dollar limitation under Code Section 415(c)(1)(A), and owning (or considered as owning) under Code Section 318 the largest interest in the Company;
 - (iii) A 5% Owner of the Company; or
 - (iv) A 1% Owner of the Company having Annual Compensation from the Company of more than \$150,000.

The determination period is the Plan Year containing the Determination Date and the four preceding Plan Years. Effective January 1, 2002, the five-year look-back period is reduced to a one-year period ending on the Determination Date, with the exception of any distribution made for a reason other than separation from service, death, or disability.

- (c) <u>Permissive Aggregation Group</u>: The Required Aggregation Group of plans plus any other plan or plans of the Company that, when considered as a group with the Required Aggregation Group, would continue to satisfy the requirements of Code Sections 401(a)(4) and 410, and which are designated by the Company to constitute a Permissive Aggregation Group.
- (d) <u>Required Aggregation Group</u>: (1) Each plan of the Company in which a Key Employee is a Participant or was a Participant at any time during the determination period (regardless of whether the Plan has terminated) and (2) any other qualified plan of the Company that enables a plan described in (1) to meet the requirements of Code Section 401(a)(4) or 410.

(e) Top-Heavy Ratio:

- (i) If the Company maintains one or more defined contribution plans (including any Simplified Employee Pension) and the Company has not maintained any defined benefit plan that during the five-year period ending on the Determination Date has or has had accrued benefits, the Top-Heavy Ratio for this Plan alone or for the Required Aggregation Group or Permissive Aggregation Group, as appropriate, is a fraction, the numerator of which is the sum of the Account balances of Key Employees as of the Determination Date (including any part of any account balances (including any part of any account balances (including any part of any account balances distributed in the five-year period ending on the Determination Date), both computed in accordance with Code Section 416 and regulations thereunder. Both the numerator and denominator of the Top-Heavy Ratio are adjusted to reflect any contributions not actually made as of the Determination Date, but which is to be taken into account on that date under Code Section 416 and regulations thereunder.
- (ii) If the Company maintains one or more defined contribution plans (including any Simplified Employee Pension) and the Company maintains or has maintained one or more defined benefit plans that during the five-year period ending on the Determination Date has or has had any accrued benefits, the Top-Heavy Ratio for any Required or Permissive Aggregation Group, as appropriate, is a fraction, the numerator of which is the sum of Account balances under the aggregated defined contribution plan or plans for all Key Employees determined in accordance with (1) above and the present value of accrued benefits under the aggregated defined benefit plan or plans for all Key Employees as of the Determination Date, and the denominator of which is the sum of the Account balances under the aggregated defined contribution plan or plans for all Participants determined in accordance with (1) above, and the present value of accrued benefits under the defined benefit plan or plans for all Participants as of the Determination Date, all determined in accordance with Code Section 416 and the regulations thereunder. The accrued benefits under a defined benefit plan in both the numerator and denominator of the Top-Heavy Ratio are adjusted for any distribution of an accrued benefit made

in the five-year period ending on the Determination Date. Solely for the purpose of determining if the Plan, or any other plan included in a Required Aggregation Group is top-heavy (within the meaning of Code Section 416(g)), the accrued benefit of an Employee other than a Key Employee shall be determined under (i) the method, if any, that uniformly applies for accrual purposes under all plans maintained by the Company and any Affiliated Companies or (ii) if there is no such method, as if such benefit accrued not more rapidly than the slowest accrual rate permitted under the fractional accrual rate of Code Section 411(b)(1)(C).

- (iii) For purposes of (i) and (ii) above, the value of Account balances and the present value of accrued benefits will be determined as of the most recent Valuation Date that falls within or ends with the twelve-month period ending on the Determination Date, except as provided in Code Section 416 and regulations thereunder for the first and second Plan Years of a defined benefit plan. The Account balances and accrued benefits of a Participant (A) who is not a Key Employee but who was a Key Employee in a prior year or (B) who has not been credited with at least one Hour of Service with any Company maintaining the Plan at any time during the five-year period ending on the Determination Date will be disregarded. The calculation of the Top-Heavy Ratio, and the extent to which distributions, rollovers and transfers are taken into account, will be made in accordance with Code Section 416 and regulations thereunder. Voluntary deductible contributions will not be taken into account in computing the Top-Heavy Ratio. When aggregating plans, the value of account balances and accrued benefits will be calculated with reference to the Determination Dates that fall within the same calendar year.
- (iv) For purposes of establishing the present value in order to compute the Top-Heavy Ratio any benefit shall be discounted only for mortality and interest based on the interest rate that would be used as of the date of distribution by the Pension Benefit Guaranty Corporation to determine the present value of a lump-sum distribution on plan termination.
- (f) Valuation Date: The date as of which Account balances or accrued benefits are valued for purposes of calculating the Top-Heavy Ratio.
- **A.4** <u>Vesting Requirements</u>. If the Plan is determined to be a top-heavy plan in any Plan Year, then a Participant's rights to his Accounts derived from Company Contributions, determined as of the end of such Plan Year, shall vest in accordance with the following schedule, unless a more rapid vesting schedule is then in effect under the terms of the Plan:

Years of Vesting Service	Vesting Percentage
2	20%
3	40%
4	60%
5	80%
6 or more	100%

If the Plan ceases to be a top-heavy plan in any Plan Year, then the vesting schedule set forth in Section 5.2 shall apply for such Plan Year with respect to any portion of a Participant's Accounts that is forfeitable as of the beginning of such Plan Year; provided, however, that a Participant with three or more Years of Service shall be given the option of remaining under the vesting schedule set forth above.

A.5 Minimum Contribution. If this Plan is top-heavy in any Plan Year, the Company Contributions for such year for each "participant" (as defined for the purpose of providing mandatory minimum contributions under regulations) who is not a Key Employee shall not be less than three percent (3%) of such participant's compensation. If, however, the Plan does not enable a defined benefit plan to meet the requirements of Section 401(a)(4) or 410, the Company Contributions shall not exceed that percentage of each participant's compensation which is equal to the highest percentage of compensation at which Company Contributions are made for the Plan Year for any Key Employee (a) under the Plan or (b) if the Plan is part of an aggregation group, under any defined contribution plan in such group. For Plan Years beginning before January 1, 1989, for purposes of this Section A.5, Company Contributions attributable to a salary reduction or similar arrangement and contributions made pursuant to Chapter 21 of Title II of the Social Security Act shall be disregarded. For Plan Years beginning after December 31, 1988, Company Contributions attributable to a salary reduction or similar arrangement made by Employees other than Key Employees shall not be taken into account under this Section A.5. Effective January 1, 2002, Company Matching Contributions shall be taken into account in determining whether the minimum contribution requirement has been satisfied under Code Section 416(c)(2). Company Matching Contributions that are used to satisfy the minimum contribution requirements shall be treated as matching contributions for purposes of the average contribution percentage test and other requirements of section 401(m) of the Code.

IN WITNESS WHEREOF, in order to record the adoption of this document, SCIENCE APPLICATIONS INTERNATIONAL CORPORATION has caused this instrument to be executed by its duly authorized officer this 19 day of December, 2005.

SCIENCE APPLICATIONS INTERNATIONAL CORPORATION a Delaware Corporation

By: /s/ John H. Warner, Jr.

John H. Warner, Jr.
Corporate Executive Vice President &
Chief Administrative Officer

SCIENCE APPLICATIONS INTERNATIONAL CORPORATION RETIREMENT PLAN

Amendment No. 1

Effective on and after January 1, 2006 the Science Applications International Corporation 401(k) Retirement Plan is hereby amended as follows:

- 1. The definition of "ESOP Account" is amended by adding the following to the end of the definition:
 - "The ESOP Account shall consist of the 'Exchangeable ESOP Subaccount,' and the 'Non-Exchangeable ESOP Subaccount.' If, at the time it authorizes a contribution to the Plan in the form of Company Stock, the Board of Directors specifies that such Company Stock is subject to investment direction by Participants as provided in Section 3.8, then such Company Stock shall be credited to the Exchangeable ESOP Subaccount. Furthermore, if pursuant to rules established by the Committee under Section 3.8, a Participant directs that a portion of his Elective Deferrals, Catch-Up Contributions or Rollover Contribution be invested in Company Stock, then the Company Stock credited as a result of such contributions (along with that portion of any loan repayments by such Participant credited as Company Stock by reason of the Participant's direction to invest Elective Deferrals in Company Stock) shall be credited to the Exchangeable ESOP Subaccount. All other all Company Stock, including Company Stock credited to the ESRP Plan Account at the time of transfer to this Plan, shall be credited to the Non-Exchangeable ESOP Subaccount."
- 2. Section 3.8(a)(i) is amended to read as follows:
 - "All contributions to the Plan (whether Company Contributions, Voluntary Contributions, Rollover Contributions or plan-to-plan transfers) shall be invested as provided in this Section 3.8. The Committee shall establish rules setting forth the extent to which Participants may determine the manner in which their Accounts (other than Company Stock credited to the Non-Exchangeable ESOP Account) may be invested. Consistent with such rules, the Committee may establish a choice of investment alternatives from which each Participant may select in determining the manner in which the applicable portions of his Plan Account, Profit Sharing Account (if any), Voluntary Contributions (if any), Matching Contribution Account (if any) and Rollover Account (if any) will be invested. The Committee shall prescribe procedures for investment of amounts (other than Company Stock credited to the Non-Exchangeable ESOP Account) allocated to an Alternate Payee Account. To the extent permitted under rules established by the Committee, a Participant may direct that Company Contributions (other than Company Stock credited to the Non-Exchangeable ESOP Account) be invested all or in part in a Company Stock Fund within the Trust. Company Stock credited to the Non-Exchangeable ESOP Subaccount shall not be subject to the direction of investments by Participants except as provided in Section 3.8(d) with respect to Participants who are no longer Employees and as required for diversification under Section 12.2. Neither the Committee nor the Trustee shall cause such Company Stock to be invested other than as provided in Section 3.8(d)."

3. The following new Section 3.8(d) is hereby added to the Plan:

"(d) Non-Exchangeable ESOP Subaccount.

Except for Company Stock credited to the Exchangeable ESOP Subaccount, the Company's contributions made in the form of Company Stock are made with the expectation and understanding that such contributions shall remain invested in Company Stock until the Participant whose Account is credited with such Company Stock ceases to be an Employee, except to the extent that diversification is required under Section 12.2 or is otherwise expressly permitted under this Plan. Accordingly, the Non-Exchangeable ESOP Subaccount shall be invested exclusively in Company Stock (subject only to the power to hold cash pending immediate distribution to Participants, Beneficiaries, and Alternate Payees), and until the Participant whose Account is credited with such Company Stock ceases to be an Employee shall not be transferred to any other investment alternative, except as required under Section 12.2 or as otherwise expressly provided in this Plan. Notwithstanding any other provisions of the Plan, no fiduciary (including but not limited to the Committee and the Trustees) shall have the authority or duty to cause any Company Stock credited to the Non-Exchangeable ESOP Subaccount to be invested other than in Company Stock (except for cash held pending immediate distribution to Participants, Beneficiaries, or Alternate Payees)."

SCIENCE APPLICATIONS INTERNATIONAL CORPORATION RETIREMENT PLAN

Amendment No. 2

Effective on and after May 15, 2006 the Science Applications International Corporation 401(k) Retirement Plan is hereby amended by adding the following to the end of Section 3.8(d):

"(i) Notwithstanding the provisions of this Plan generally prohibiting transfers from the Non-Exchangeable ESOP Subaccount, subject to rules established by the Committee, and the Company's right to not purchase all or any portion of the shares of Company Stock offered for sale to it by the Plan, each Participant shall be entitled to direct for the Trade Date occurring on or about June 30, 2006 or any postponement thereof (and for no other Trade Date), that amounts credited to the Non-Exchangeable ESOP Account on the Participant's behalf as of such Trade Date be invested all or in any part in an investment alternative offered under the Plan other than the Company Stock Fund."

APPENDIX A

SCIENCE APPLICATIONS INTERNATIONAL CORPORATION

RETIREMENT PLAN

Amendment No. 3

Effective on and after July 1, 2006 the Science Applications International Corporation Retirement Plan is hereby amended by changing the definition of "Trade Date" in Section 13.1 of the Plan to read in full as follows:

'Trade Date' shall mean, (i) prior to July 1, 2006, if there is a 'limited market' for the Company Stock, one of the four predetermined times in each calendar year (one each calendar quarter) during which the Company Stock may be bought or sold in accordance with the terms and conditions of the 'limited market' as set forth in the Company's prospectus, or, if there is no 'limited market' for the Company Stock, the four predetermined dates in each calendar year (one each quarter) during which the Committee, in its sole discretion, elects to conduct a transaction with respect to the assets of the Plan, and (ii) on and after July 1, 2006, if there is a 'limited market' for the Company Stock, one of the four predetermined times in each fiscal year of the Company (one each fiscal quarter) during which the Company Stock may be bought or sold in accordance with the terms and conditions of the 'limited market' as set forth in the Company's prospectus, or, if there is no 'limited market' for the Company Stock, the four predetermined dates in each fiscal year of the Company (one each fiscal quarter) during which the Committee, in its sole discretion, elects to conduct a transaction with respect to the assets of the Plan. In accordance with the foregoing, there need not be a Trade Date in the third calendar quarter of 2006, and there shall be a Trade Date in the Company's third fiscal quarter of 2006.

In addition, with respect to those shares of Company Stock that are readily tradable on an established securities market (and those series of Preferred Stock, the shares of which are immediately convertible into Company Stock that is readily tradable on an established securities market), then rules established by the Committee may designate any or all business days as Trade Dates (for purposes of this definition, the 'Additional Trade Dates'). For avoidance of doubt, any Additional Trade Dates occurring while shares of Company Stock or Preferred Stock remain subject to covenants or restrictions preventing such shares from being traded on an established securities market shall not apply to such shares. The rules established for Additional Trade Dates shall include such rules or procedures as the Committee deems appropriate for identifying those shares subject to the Additional Trade Dates in each Participants' Account.

Furthermore, if all Company Stock is readily tradable on an established securities market (or is Preferred Stock, all series of which are immediately convertible into Company Stock that is readily tradable on an established securities market), then rules established by the Committee may designate any or all business days as Trade Dates."

SCIENCE APPLICATIONS INTERNATIONAL CORPORATION RETIREMENT PLAN

Amendment No. 4

Effective on and after January 1, 2006 the Science Applications International Corporation Retirement Plan is hereby amended by changing paragraphs (d) and (e) of the definition of "Compensation" in Section 13.1 of the Plan to read in full as follows:

- "(d) Elective Deferrals and Catch-Up Contributions. For purposes of determining the amount that may be deferred as provided in Section 3.1 and the amount of the Company Matching Contributions under Section 3.3, "Compensation" shall mean the Participant's salary or wages (as otherwise would be reflected in his regular payroll stub before reductions by amounts the Participant elects to defer pursuant to Section 3.1) plus one hundred percent (100%) of any cash bonuses paid to the Participant (or payable, but for an election by a Participant pursuant to Section 3.1) and exclusive of (i) any stock bonus regardless of whether it was fully vested when awarded and (ii) any taxable assignment allowances paid after December 31, 1992 to any Employee who is on assignment away from normal work location(s) and entitled to receive one or more taxable allowances or adjustments to Compensation based on such assignment. For avoidance of doubt and without implying the inclusion in Compensation of any amount not included as Compensation under the preceding sentence, Compensation under this subsection (d) shall not include amounts reflecting reimbursed expenses or fringe benefits (including any amount relating to the grant or exercise of stock options, disposition of shares through exercise of options, payment of dividends on non-vested stock or payments representing dividends on stock units or stock rights, and any distributions from a plan of deferred compensation).
- (e) Profit Sharing. For purposes of determining the allocation of Profit Sharing Contribution and ESOP Contribution pursuant to Section 3.4 and forfeitures of Profit Sharing Contribution and ESOP Contribution pursuant to Section 8.1, "Compensation" shall mean the amount of compensation paid by the Company during a calendar year by reason of services performed by an Employee reflected as "wages, tips, other compensation" on the Employee's Form(s) W-2 for such year; plus
 - (i) Effective on and after January 1, 2004, Elective Deferrals under Code Section 402(g)(3), Catch-Up Contributions under Code Section 414(v), and amounts contributed under Code Sections 125 and 132(f) by the Company at the Employee's election that are not included in the Employee's gross income; plus
 - (ii) Any Compensation which, but for Code Section 3401(a)(8)(A) (dealing with the section 911 exclusion and income subject to foreign withholding) would be required to be reflected as "wages, tips, other compensation" on the Employee's Form(s) W-2; less (iii) Any Compensation paid by reason of services performed during any period in which the Employee is not a Participant under this Plan or is not an Eligible Employee; overtime pay (which shall be deemed to include base pay and premium pay for time worked in excess of a normal day or week); bonuses; commissions; and amounts reflecting reimbursed expenses or fringe benefits (including any amount relating to the grant or

exercise of stock options, disposition of shares through exercise of options, payment of dividends on non-vested stock, payments representing dividends or dividend equivalents on stock units or stock rights, and any distributions from a plan of deferred compensation) which have been included as "wages, tips, compensation" on the Employee's Form(s)W-2.

(iv) Notwithstanding the foregoing, effective January 1, 1993, for purposes of determining the allocation of Profit Sharing Contribution and ESOP Contribution pursuant to Section 3.4 and forfeitures pursuant of Profit Sharing Contribution and ESOP Contribution pursuant to Section 8.1 with respect to any Employee who is on assignment away from the normal work location(s) and entitled to receive one or more taxable allowances or adjustments to Compensation based on such assignment, "Compensation" shall mean the amount of Compensation paid by the Company during a calendar year by reason of services performed by such Employee which constitutes base salary (including comprehensive leave, vacation and holiday pay) less any amounts deferred pursuant to an Employee's election under Section 3.1 and unadjusted for any taxable assignment allowances."

AMSEC

EMPLOYEES 401(k) PROFIT SHARING PLAN

Amended and Restated 1/1/97

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

TABLE OF CONTENTS

Artic	Article		
I.	NAM	NAME, EFFECTIVE DATE AND PURPOSES	
	1.1	Name and Effective Date	1
	1.2	Plan Purposes	1
II.	DEFI	DEFINITIONS	
	2.1	Accounts	5
	2.2	Additional Company Contribution	3
	2.3	Adjustment Factor	3
	2.4	Affiliated Company	3
	2.5	Alternate Payee	3
	2.6	Alternate Payee Account	3
	2.7	Anniversary Date	3
	2.8	Applicable Valuation Date	2
	2.9	Annual Addition	2
	2.10	Average Deferral Percentage	4
	2.11	Beneficiary	4
	2.12	Board of Directors	4
	2.13	Break in Service	4
	2.14	Code	
	2.15	Company	
	2.16	Company Contributions	4
	2.17	Company Contribution Account	5
	2.18	Company Stock	5
	2.19	Compensation	5
	2.20	Contribution Percentage	(
	2.21	Deferral Account	5
	2.22	Deferral Percentage	5
	2.23	Disability	5
	2.24	Distributable Benefit	7
	2.25	Effective Date	7
	2.26	Elective Deferrals	8
	2.27	Eligible Employee	8
	2.28	Employee	G
	2.29	Employment Commencement Date	G
	2.30	ERISA	10
	2.31	Excess Aggregate Contributions	10
	2.32	Excess Contributions	10
	2.33	Excess Deferrals	10
	2.34	5% Owner	10
	2.35	Highly Compensated Employee	10

i

1				
2.38 IRP Contributions 13 2.39 IRP Contributions 13 2.40 Limitation Year 14 2.41 Materinity of Patemity Absence 14 2.43 Nothing Contribution 14 2.44 Marcinity of Patemity Absence 14 2.43 Nonal Retirement Age 14 2.44 Normal Retirement Age 14 2.45 Normal Retirement Date 14 2.47 NNS 14 2.48 NNS 401(k) Plan 15 2.50 NNS Transfer After-Tax Account 15 2.51 NNS Transfer After-Tax Account 15 2.52 NNS Transfer Employee Account 15 2.51 NNS Transfer Employee Account 15 2.52 NNS Transfer Employee Account 15 2.53 Participant 15 2.54 Participant 15 2.55 Plan Year 15 2.55 Plan Year 15 2.57 Plan Year 16 2.50 Posti Sharing Contribution 16 </th <th></th> <th>2.36</th> <th>Hour of Service</th> <th>11</th>		2.36	Hour of Service	11
2.38 IRSA CONTINUTION 13 2.39 Lessel Employee 13 2.40 Limitation Year 14 2.41 Matching Contribution 14 2.42 Marching Contribution 14 2.43 Nonhighly Compensated Employee 14 2.44 Normal Retirement Age 14 2.45 Normal Retirement Date 14 2.46 Normal Retirement Date 14 2.47 NNS 14 2.48 NNS 401(k) Plan 15 2.49 NNS Transfer After-Tax Account 15 2.50 NNS Transfer Poferral Account 15 2.51 NNS Transfer Poferral Account 15 2.51 NNS Transfer Rollover Account 15 2.52 NS Transfer Rollover Account 15 2.53 Participation Commencement Date 15 2.55 Plan Administrator 15 2.55 Plan Near 15 2.57 Plan Vear 15 2.58 Profit Sharing Contribution 16 2.61 Rollover Account				
2.34				
2.44				
2.41 Matching Contribution			1 0	
2.42 Montingly Compensated Employee 14 2.43 Noningla Nocimpensated Employee 14 2.44 Normal Retirement Age 14 2.45 Normal Retirement Date 14 2.47 NNS man Retirement Date 14 2.47 NNS man Retirement Date 14 2.48 NNS 401(k) Plan 15 2.49 NNS Transfer Endove 15 2.50 NNS Transfer Endover Account 15 2.51 NNS Transfer Employee Account 15 2.52 NNS Transfer Rollover Account 15 2.53 Participant 15 2.54 Participant 15 2.55 Plan 15 2.55 Plan 15 2.57 Plan Administrator 15 2.57 Plan Year 15 2.58 Profit Sharing Contribution 16 2.50 Profit Sharing Contribution 16 2.61 Roll Quited Payment Commencement Date 16 2.62 Rol				14
2.44 Normal Retirement 14 2.45 Normal Retirement Age				14
14				14
14				14
14		2.45	Normal Retirement Age	14
2.48 NNS 401(k) Plan 15 2.49 NNS Transfer After-Tax Account 15 2.50 NNS Transfer Epeferal Account 15 2.51 NNS Transfer Employee Account 15 2.52 NNS Transfer Rollover Account 15 2.53 Participant 15 2.53 Participant 15 2.54 Participant on Commencement Date 15 2.55 Plan Administrator 15 2.55 Plan Administrator 15 2.56 Plan Administrator 15 2.57 Plan Eperity Starting Contribution 16 2.59 Protected Benefits 16 2.60 Required Payment Commencement Date 16 2.61 Rollover Account 16 2.62 Rollover Account 17 2.63 Valuation Date 17 2.64 Trust and Trust Fund 17 2.65 Vested Interest 17 2.67 Vested Interest 17 2.69 Voluntary Account 18 2.70 Year of Service			0	14
15		2.47	NNS	14
15		2.48	NNS 401(k) Plan	15
1.5				15
15		2.50	NNS Transfer Deferral Account	15
15		2.51	NNS Transfer Employee Account	15
2.54				
2.54		2.53	Participant	15
2.55 Plan				15
2.56 Plan Administrator 15 2.57 Plan Year 15 2.58 Profit Sharing Contribution 16 2.59 Protected Benefits 16 2.60 Required Payment Commencement Date 16 2.61 Rollover Account 16 2.62 Rollover Contribution 16 2.63 Suspended Participant 17 2.64 Trust and Trust Fund 17 2.65 Trustee 17 2.60 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 18 2.70 Vesr of Service 18 3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. TSUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BYPARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSON 21 4.1 Trust Fund 29 4.2 Permissible Types of Plan Investments 21				15
2.57 Plan Year 15 2.58 Profit Sharing Contribution 16 2.59 Protected Benefits 16 2.60 Required Payment Commencement Date 16 2.61 Rollover Account 16 2.62 Rollover Contribution 16 2.63 Suspended Participant 17 2.64 Trust and Trust Fund 17 2.65 Trustee 17 2.66 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 18 2.70 Year of Service 18 III. ELI-BILITY AND PARTICIPATION 20 3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. Trust FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 20 4.2 Permissible Types of Plan Investments 21		2.56	Plan Administrator	
2.59 Protected Benefits 16 2.60 Required Payment Commencement Date 16 2.61 Rollover Account 16 2.62 Rollover Contribution 16 2.63 Suspended Participant 17 2.64 Trust and Trust Fund 17 2.65 Trustee 17 2.66 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 17 2.69 Voluntary Account 18 2.70 Vested Interest 18 2.70 Vear of Service 18 3.1 Eligibility And Participate 20 3.2 Automatic Commencement of Participation 20 3.1 Eligibility of Participate 20 3.2 Automatic Commencement of Participation 20 4.1 Trust Fund And Company Stock transactions with disqualified persons 21 4.2 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 22 4.3 Permissible Types of Plan Investments 21 4.4 Permissible Types of Plan Investments 21 4.5 Permissible Types of Plan Investments 21 4.6 Permissible Types of Plan Investments 21 4.7 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.9 Permissible Types of Plan Investments 21 4.1 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 21 4.3 Permissible Types of Plan Investments 21 4.4 Permissible Types of Plan Investments 21 4.5 Permissible Types of Plan Investments 21 4.5 Permissible Types of Plan Investments 21 4.5 Permissible Types of Plan Investments 21 4.7 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.				15
2.59 Protected Benefits 16 2.60 Required Payment Commencement Date 16 2.61 Rollover Account 16 2.62 Rollover Contribution 16 2.63 Suspended Participant 17 2.64 Trust and Trust Fund 17 2.65 Trustee 17 2.66 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 17 2.69 Voluntary Account 18 2.70 Vested Interest 18 2.70 Vear of Service 18 3.1 Eligibility And Participate 20 3.2 Automatic Commencement of Participation 20 3.1 Eligibility of Participate 20 3.2 Automatic Commencement of Participation 20 4.1 Trust Fund And Company Stock transactions with disqualified persons 21 4.2 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 22 4.3 Permissible Types of Plan Investments 21 4.4 Permissible Types of Plan Investments 21 4.5 Permissible Types of Plan Investments 21 4.6 Permissible Types of Plan Investments 21 4.7 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.9 Permissible Types of Plan Investments 21 4.1 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 21 4.2 Permissible Types of Plan Investments 21 4.3 Permissible Types of Plan Investments 21 4.4 Permissible Types of Plan Investments 21 4.5 Permissible Types of Plan Investments 21 4.5 Permissible Types of Plan Investments 21 4.5 Permissible Types of Plan Investments 21 4.7 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.8 Permissible Types of Plan Investments 21 4.		2.58	Profit Sharing Contribution	16
2.60 Required Payment Commencement Date 16 2.61 Rollover Account 16 2.62 Rollover Contribution 16 2.63 Suspended Participant 17 2.64 Trust and Trust Fund 17 2.65 Trustee 17 2.66 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 18 2.70 Year of Service 18 III. ELIGIBILITY AND PARTICIPATION 20 IX TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21				16
2.62 Rollover Contribution 16 2.63 Suspended Participant 17 2.64 Trust and Trust Fund 17 2.65 Trustee 17 2.66 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 17 2.69 Voluntary Account 18 2.70 Year of Service 18 III. ELIGIBILITY AND PARTICIPATION 20 3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21		2.60	Required Payment Commencement Date	16
2.63 Suspended Participant 17 2.64 Trust and Trust Fund 17 2.65 Trustee 17 2.66 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 18 2.70 Year of Service 18 III. ELIGIBILITY AND PARTICIPATION 20 3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY ARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21			·	16
2.64 Trust and Trust Fund 17 2.65 Trustee 17 2.66 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 18 2.70 Year of Service 18 III. ELIGIBILITY AND PARTICIPATION 20 3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21		2.62	Rollover Contribution	16
2.64 Trust and Trust Fund 17 2.65 Trustee 17 2.66 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 18 2.70 Year of Service 18 III. ELIGIBILITY AND PARTICIPATION 20 3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21		2.63	Suspended Participant	17
2.66 Valuation Date 17 2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 18 2.70 Year of Service 18 III. ELIGIBILITY AND PARTICIPATION 20 3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21				17
2.67 Vested Interest 17 2.68 Voluntary Contribution 17 2.69 Voluntary Account 18 2.70 Year of Service 18 III. ELIGIBILITY AND PARTICIPATION 20 3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments		2.65	Trustee	17
2.68 Voluntary Contribution 17 2.69 Voluntary Account 18 2.70 Year of Service 18 III. ELIGIBILITY AND PARTICIPATION 20 3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21		2.66	Valuation Date	17
2.69 Voluntary Account 2.70 Year of Service III. ELIGIBILITY AND PARTICIPATION 3.1 Eligibility to Participate 3.2 Automatic Commencement of Participation IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 2.1 4.1 Trust Fund 4.2 Permissible Types of Plan Investments		2.67	Vested Interest	17
2.69 Voluntary Account 2.70 Year of Service III. ELIGIBILITY AND PARTICIPATION 3.1 Eligibility to Participate 3.2 Automatic Commencement of Participation IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 2.1 4.1 Trust Fund 4.2 Permissible Types of Plan Investments		2.68	Voluntary Contribution	17
III. ELIGIBILITY AND PARTICIPATION 3.1 Eligibility to Participate 3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 4.2 Permissible Types of Plan Investments 20 21 22 23 24 25 26 27 27 28 29 20 20 20 20 20 20 20 20 20				18
3.1 Eligibility to Participate 20 3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21		2.70	Year of Service	18
3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21	III.	ELIC	GIBILITY AND PARTICIPATION	20
3.2 Automatic Commencement of Participation 20 IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 21 4.1 Trust Fund 21 4.2 Permissible Types of Plan Investments 21		2.1	ייי אויין אין אין אין אין אין אין אין אין אין	20
IV. TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 4.1 Trust Fund 4.2 Permissible Types of Plan Investments 21				
BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS 4.1 Trust Fund 4.2 Permissible Types of Plan Investments 21		3.2	Automatic Commencement of Participation	20
4.2 Permissible Types of Plan Investments 21	IV.			21
4.2 Permissible Types of Plan Investments 21		4.1	Trust Fund	21

22

4.4

Company Contributions Form of Company Contributions

	4.5 4.6 4.7 4.8 4.9	Valuation of Company Contributions in the Form of Company Stock Investment of Company Contributions and Voluntary Contributions Irrevocability Company, Plan Administrator and Trustee Not Responsible for Adequacy of Trust Fund Company Stock Transactions With Disqualified Persons	22 22 24 24 25
V.	ELE	CTIVE DEFERRALS, ROLLOVER CONTRIBUTIONS, PLAN-TO-PLAN TRANSFERS, AND VOLUNTARY CONTRIBUTIONS	26
	5.1 5.2 5.3 5.4 5.5 5.6 5.7	Election Amount Subject to Election; Nondiscrimination Requirement and Distribution of Excess Contributions Matching Contributions; Nondiscrimination Requirement and Distribution of Excess Aggregate Contributions Termination of, Change in Rate of, or Resumption of Deferrals Rollover Contributions From Other Plans Plan-to-Plan Transfers Voluntary Contributions	26 26 30 34 34 34 35
VI.	ACC	COUNTING AND ALLOCATION PROCEDURES	36
	6.12 6.13 6.14 6.15 6.16 6.17	Deferral Account Company Contribution Account Voluntary Account Rollover Account Alternate Payee Account NNS Transfer Deferral Account NNS Transfer After-Tax Account NNS Transfer Rollover Account NNS Transfer Rollover Account NNS Transfer Employee Account Allocation of Company Contributions Valuation of Accounts Allocation of Forfeitures Allocation of Forfeitures Allocation of Income or Loss on Unallocated Company Contributions; Company Contributions to Defray Plan Expenses Accounting Procedures Suspended Participants Qualified Military Service Accounting for Interest of an Alternate Payee	36 36 36 36 36 36 36 37 38 41 42 42 43 43
VII.	VES	TING IN ACCOUNTS	44

	7.1	No Vested Rights Except as Herein Specified	44
	7.2	Participant's Vested Interest—General Rule	44
	7.3	Vested Percentage—Special Rules	44
	7.4	Nonforfeitable Accounts	46
	7.5	Alternate Payee Accounts	46
	7.6	Participants Vested Interest	46
VIII.	PAY	MENT OF PLAN BENEFITS	47
	8.1	Retirement	47
	8.2	Other Termination of Employment	47
	8.3	Withdrawals of Voluntary Contributions and from NNS Transfer Accounts; Hardship Withdrawals	47
	8.4	Withdrawals After Age 59-1/2	49
	8.5	Distributable Benefit	50
	8.6	Optional Forms of Payment	50
	8.7	Direct Rollover	51
	8.8	Timing on Distributions	52
	8.9	Distribution Rules	53
	8.10	Designation of Beneficiary	57
	8.11	Facility of Payment	58
	8.12	Distribution to Alternate Payees	58
	8.13	Additional Documents	59
	8.14	Transfer of Distributed Company Stock—Right of First Refusal for SAIC and the Plan	59
IX.	OPE	RATION AND ADMINISTRATION OF THE PLAN; VOTING AND OTHER RIGHTS OF COMPANY STOCK	60
	9.1	Plan Administration	60
	9.2	Plan Administrator Powers	60
	9.3	Investment Manager	62
	9.4	Periodic Review	62
	9.5	Reliance Upon Documents and Opinions	62
	9.6	Requirement of Proof	63
	9.7	Multiple Fiduciary Capacity	63
	9.8	Limitation on Liability	63
	9.9	Indemnification	64
		Allocation of Fiduciary Responsibility	64
		Prohibition Against Certain Actions	65
		Bonding and Insurance	66
		Voting and Other Rights of Company Stock	66
	9.14	Plan Expenses	68
х.	MEF	RGER OF COMPANY: MERGER OF PLAN	69
		Effect of Reorganization or Transfer of Assets	69
	10.2	Merger Restriction	69

XI.	PLAN TERMINATION AND DISCONTINUANCE OF CONTRIBUTIONS	70
	 11.1 Plan Termination 11.2 Discontinuance of Contributions 11.3 Rights of Participants 11.4 Trustee's Duties on Termination 11.5 Partial Termination 11.6 Failure to Contribute 	70 70 71 71 72 72
XII.	APPLICATION FOR BENEFITS	73
	 12.1 Application for Benefits; Claims Procedure 12.2 Action on Application 12.3 Appeals 	73 73 73
XIII.	LIMITATIONS ON ANNUAL ADDITIONS	75
	 13.1 Maximum Annual Additions 13.2 Effect of Participation in Other Company Plans 13.3 Incorporation by Reference of Code Section 415 13.4 No Contractual Right to Excess Contributions 	75 75 75 75
XIV.	RESTRICTION ON ALIENATION; PARTICIPANT LOANS	76
	 14.1 General Restrictions Against Alienation 14.2 Nonconforming Distributions Under Court Order 14.3 Authorized Participant Loans 	76 76 78
XV.	PLAN AMENDMENTS	81
	15.1 Amendments15.2 Retroactive Amendments	81 81
XVI.	TOP-HEAVY PROVISIONS	82
	 16.1 Application 16.2 Criteria 16.3 Definitions 16.4 Adjustment to Fractions 16.5 Vesting Requirements 16.6 Minimum Contribution 16.7 Ceiling on Includable Compensation 	82 82 85 85 85 86
XVII.	MISCELLANEOUS	87
	 17.1 No Enlargement of Employee Rights 17.2 Mailing of Payments; Lapsed Benefits 17.3 Addresses 	87 88 88

88

89

17.4 Notices and Communications

17.5 Reporting and Disclosure

17.6	Governing Law	89
17.7	Interpretation	89
17.8	Withholding For Taxes	89
17.9	Limitation on Company, Plan Administrator and Trustee Liability	89
17.10	Successors and Assigns	89
17.11	Counterparts	90
17.12	No Implied Rights or Obligations	90

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

ARTICLE I

NAME, EFFECTIVE DATE AND PURPOSES

- **1.1** Name and Effective Date. Effective September 1, 1999, the AMSEC Corporation Employee 401(k) Profit Sharing Plan of American Systems Engineering Corporation shall be re-named the AMSEC Employee 401(k) Profit Sharing Plan (the "Plan") and shall be sponsored by AMSEC LLC (the "Company"). This document amends and restates the Plan in its entirety, effective as of January 1, 1997, except where a different date is specifically provided.
- **1.2** <u>Plan Purposes</u>. This Plan is designed to constitute a tax-qualified profit sharing plan for purposes of Code sections 401(a) and 501(a) which provides a vehicle whereby Eligible Employees of the Company may elect to defer the receipt of some portion of their income and instead to have those amounts contributed to the Plan on their behalf under a "cash or deferred arrangement" within the meaning of Code section 401(k).

ARTICLE II

DEFINITIONS

- 2.1 Accounts. The term "Accounts" shall include the following Accounts that are maintained pursuant to the terms of this Plan:
 - (a) The Deferral Account maintained for each Participant who has made Elective Deferrals pursuant to the provisions of Section 5.1.
- (b) Company Contribution Account maintained for each Participant on whose behalf Matching Contributions or Profit Sharing Contributions are made pursuant to Sections 5.3 or 4.3.
 - (c) The Voluntary Account maintained for each Participant who has made Voluntary Contributions pursuant to the provisions of Section 5.7.
 - (d) The Rollover Account maintained for each Participant who has made Rollover Contributions pursuant to Section 5.5.
- (e) The Alternate Payee Account maintained for each Alternate Payee who is awarded an interest in a Participant's benefits under the Plan pursuant to the provisions of Sections 6.17 and 14.2.
- (f) The IRP Account maintained for periods prior to January 1, 1991 for each Participant who made IRP Contributions under the provisions of the Plan in effect prior to January 1, 1987.
- (g) NNS Transfer Deferral Account maintained for each Participant who formerly participated in the NNS 401(k) Plan and had an elective deferral account in that plan which was transferred to the Plan pursuant Section 5.6.
- (h) NNS Transfer After-Tax Account maintained for each Participant who formerly participated in the NNS 401(k) Plan and had an after-tax account in that plan which was transferred to the Plan pursuant to Section 5.6.
- (i) NNS Transfer Rollover Account maintained for each Participant who formerly participated in the NNS 401(k) Plan and had a rollover account in that plan which was transferred to the Plan pursuant to Section 5.6.

- (j) NNS Transfer Employer Account maintained for each Participant who formerly participated in the NNS 401(k) Plan and had a matching or stock grant account in that plan which was transferred to the Plan pursuant to Section 5.6.
- **2.2** <u>Additional Company Contribution</u>. "Additional Company Contribution" shall mean a contribution made by the Company pursuant to Section 4.3 which is intended to qualify as a qualified nonelective contribution pursuant to Code section 401(k)(3)(D)(ii)(II).
- **2.3** <u>Adjustment Factor</u>. "Adjustment Factor" shall mean the cost of living adjustment factor or method prescribed by the Secretary of the Treasury under Code section 415(d) for years beginning after December 31, 1987, as applied to the items and in the manner prescribed by the Secretary of the Treasury.
 - **2.4** <u>Affiliated Company</u>. "Affiliated Company" shall mean:
 - (a) Any corporation that is included in a controlled group of corporations, within the meaning of Code section 414(b), that includes the Company;
 - (b) Any trade or business that is under common control with the Company within the meaning of Code section 414(c);
 - (c) Any member of an affiliated service group, within the meaning of Code section 414(m), that includes the Company; and
 - (d) Any entity required to be included under Code section 414(o).
- **2.5** <u>Alternate Payee</u>. "Alternate Payee" shall mean an individual awarded a portion of a Participant's benefits under the Plan pursuant to a qualified domestic relations order, as defined in Code section 414(p) and Section 14.2 of the Plan. Any limitation or condition imposed by the Plan upon a Participant or his rights hereunder shall, unless expressly indicated otherwise, also serve to limit or condition the rights of an Alternate Payee of the Participant's Account(s).
 - 2.6 Alternate Payee Account. "Alternate Payee Account" shall mean the account maintained to reflect the interest of an Alternate Payee under the Plan.
 - 2.7 Anniversary Date. "Anniversary Date" shall mean the last day of each Plan Year.

- **2.8** <u>Applicable Valuation Date</u>. "Applicable Valuation Date" shall mean the most recent date on which the Trust assets were valued in accordance with the rules of Article VI.
 - 2.9 Annual Addition. "Annual Addition" shall mean "annual addition" as defined in Code section 415(c)(2).
- **2.10** <u>Average Deferral Percentage</u>. "Average Deferral Percentage" shall mean the average of the Deferral Percentages of the Participants who are in the Highly Compensated Employee group or the Nonhighly Compensated Employee group, as applicable.
- **2.11** <u>Beneficiary</u>. "Beneficiary" or "Beneficiaries" means the person or persons designated pursuant to Section 8.10 to receive the interest of a deceased Participant.
- **2.12 Board of Directors.** "Board of Directors" shall mean the Board of Directors (or its delegate, to the extent the duties of the Board of Directors are delegated to such person) of the Company as it may from time to time be constituted.
- **2.13 Break in Service**. "Break in Service" shall mean for vesting and participation purposes, with respect to an Employee, a Plan Year in which the Employee completes no more than 500 Hours of Service.
- **2.14** Code. "Code" shall mean the Internal Revenue Code of 1986, as in effect on the date of execution of this Plan document and as thereafter amended from time to time.
- **2.15** Company. "Company" shall mean AMSEC LLC, or any successor thereof, if its successor shall adopt this Plan. In addition, unless the context indicates otherwise, as used in this Plan the term Company shall also mean and include any Affiliated Company (or similar entity) which has been granted permission by the Board of Directors to participate in this Plan. This permission shall be granted upon such terms and conditions as the Board of Directors deems appropriate.
- **2.16** Company Contributions. "Company Contributions" shall mean all amounts (whether in cash or other property, including Company Stock) paid by the Company into the Trust Fund established and maintained under the provisions of this Plan for the purpose of providing benefits for Participants and their Beneficiaries, and shall include Elective Deferrals, Profit Sharing Contributions, Matching Contributions, Additional Company Contributions and any contributions to pay Plan expenses as provided in Section 6.13.

- **2.17** Company Contribution Account. "Company Contribution Account" shall mean the Account opened up and maintained for the Participant pursuant to Section 6.2.
- **2.18** Company Stock. "Company Stock" shall mean Class A Common Stock par value \$.01 per share ("Class A Common Stock"), and to the extent accounted for pursuant to Section 4.2 of the Plan and solely for such purpose, shall also mean Class B Common Stock, par value \$.05 per share ("Class B Common Stock"), of Science Applications International Corporation.

2.19 Compensation.

- (a) For purposes of determining the allocation of Profit Sharing Contributions pursuant to Sections 4.3 and 6.10(e) and the amount of the Participant's permissible Elective Deferrals under Section 5.2, "Compensation" shall mean the amounts described in subsection (b) below paid by the Company during a Plan Year by reason of services performed by an Employee less any compensation paid by reason of services performed during any period in which the Employee is not a Participant under this Plan, but including any elective contributions that are made by the Company on behalf of an Employee pursuant to an arrangement described in Code sections 125, 132(f)(4), 402(a)(8), 402(h) or 403(b).
- (b) For purposes of applying the limitation on Annual Additions pursuant to Article XIII of the Plan, "Compensation" shall mean a Participant's earned income, wages, salaries and fees for professional services and other amounts received for personal services actually rendered in the course of employment with the Company (including, but not limited to, commissions paid salespersons, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips and bonuses), and excluding the following:
 - (i) Company contributions to a plan of deferred compensation that are not includable in the Employee's gross income for the taxable year in which contributed, or Company contributions under a simplified employee pension (within the meaning of Code section 408(k)) to the extent such contributions are excludable from gross income by the Employee, or any distributions from a plan of deferred compensation;
 - (ii) Amounts realized from the exercise of a nonqualified stock option, or when restricted stock (or property) held by the Employee either becomes freely transferable or is no longer subject to a substantial risk of forfeiture;

- (iii) Amounts realized from the sale, exchange or other disposition of stock acquired under a qualified or incentive stock option; and
- (iv) Other amounts that received special tax benefits, or Company contributions (whether or not under a salary reduction agreement) towards the purchase of an annuity described in Code section 403(b) (whether or not the amounts are actually excludable from the gross income of the Employee).
- Notwithstanding the foregoing, effective for Plan Years beginning after December 31, 1997, "Compensation" for purposes of this subsection 2.19(b) shall include Elective Deferrals under Code section 402(g)(3) and amounts contributed under Code sections 125 and 132(f)(4), by the Company at the Employee's election that are not included in the Employee's gross income.
- (c) For purposes of determining whether an Employee is a Highly Compensated Employee, "Compensation" shall be "Compensation" as defined in subsection (b) above, but determined without regard to Code sections 125, 132(f)(4), 402(e)(3) and 402(h)(1)(B) and, in the case of Company Contributions made pursuant to a salary reduction agreement, without regard to Code section 403(b). Thus, Compensation for purposes of this subsection 2.19(c) includes elective or salary reduction contributions to any Company section 125 flex plans,132(f)(4) transit reimbursement arrangements, this Plan, or any tax-sheltered annuity, even for periods prior to January 1, 1998 (when such items are also includible under subsection 2.19(b)).
- (d) For purposes of determining the Deferral Percentage and Contribution Percentage of a Participant, "Compensation" shall mean the amounts specified in subsection (a) above provided that for an Employee's first year of participation in the Plan, Compensation shall include only amounts earned while a Participant.
- (e) The amount considered as "Compensation" for any purpose hereunder shall be limited to \$200,000, multiplied by the Adjustment Factor, for Plan Years after 1988 and prior to 1994. Commencing with Plan Years beginning in 1994, "Compensation" shall be limited to \$150,000, adjusted as provided in Code section 401(a)(17)(B).
- **2.20** <u>Contribution Percentage</u>. "Contribution Percentage" shall mean the ratio (expressed as a percentage) of (i) a Participant's Matching Contributions plus any Additional Company Contributions allocated to the Participant's Deferral Account that are taken into account in the nondiscrimination test

described in Section 5.3 for the Plan Year and (ii) the Participant's Compensation for the Plan Year. Notwithstanding the foregoing, for any Plan Year in which Matching Contributions are treated as Qualified Matching Contributions, such amounts shall not be included in clause (i) of the foregoing sentence for purposes of calculating the ratio.

- **2.21** <u>Deferral Account</u>. "Deferral Account" shall mean the Account opened up and maintained for each Employee who has made Elective Deferrals pursuant to Section 5.1.
- **2.22** <u>Deferral Percentage.</u> "Deferral Percentage" shall mean the ratio (expressed as a percentage) of (i) a Participant's Elective Deferrals (including Excess Deferrals) plus any Qualified Matching Contributions the Committee elects to treat as an Elective Contribution, plus any Additional Company Contributions allocated to the Participant's Deferral Account that are taken into account in the nondiscrimination test described in Section 5.2(d) for the Plan Year to (ii) the Participant's Compensation for the Plan Year.
- **2.23** <u>Disability</u>. "Disability" or "Disabled" shall mean the status of disability determined conclusively by the Plan Administrator based on certification of disability by the Social Security Administration, effective upon receipt of such certification by the Plan Administrator, or based on other reasonable evidence of disability.
- **2.24** <u>Distributable Benefit</u>. "Distributable Benefit" shall mean the interest of a Participant in this Plan that is determined and distributable to him in accordance with the provisions of Article VIII and which shall include his Vested Interest in his Deferral Account, Company Contribution Account, Voluntary Account, Rollover Account, IRP Account, NNS Transfer Deferral Account, NNS Transfer After-Tax Account, NNS Transfer Rollover Account and NNS Transfer Employee Account, if any. In the case of an Alternate Payee, the Distributable Benefit shall mean the Vested Interest in the Alternate Payee Account.
- **2.25** Effective Date. The original effective date of this Plan was July 1, 1981. The effective date of this amendment and restatement of the Plan is January 1, 1997, except where a different Effective Date is specified. The effective date of any prior or subsequent amendments is the date specified therein or in any accompanying resolutions adopting such amendment. The rights of an Employee who has terminated employment shall be governed by the terms of the Plan in effect at the time of such termination, unless otherwise specified in any subsequent amendment.

2.26 Elective Deferrals. "Elective Deferrals" shall mean contributions paid over by the Company to the Trustee at the election of a Participant in lieu of cash Compensation pursuant to Section 5.1.

2.27 Eligible Employee. "Eligible Employee" shall include any Employee except the following:

- (a) Any Employee who is covered by a collective bargaining agreement to which the Company or an Affiliated Company is a party if there is evidence that retirement benefits were the subject of good faith bargaining between the Company (or an Affiliated Company) and the collective bargaining representative, unless the collective bargaining agreement provides for coverage under this Plan.
- (b) Any Employee within a group or classification within the Company designated by the Plan Administrator as ineligible for participation in this Plan.
 - (c) Any Employee of an Affiliated Company which has not been granted permission by the Board of Directors to participate in this Plan.
- (d) Any Employee who becomes an Employee on or after January 1, 1992 and who is hired or becomes an Employee as part of a division, operating unit, geographical location or other identified unit of the Company, unless such new division, unit or location has been determined by the Plan Administrator to be eligible for participation in this Plan.
- (e) Any Employee who is a nonresident alien and who receives no earned income (within the meaning of Code section 911(d)(2)) from the Company which constitutes income from sources within the United States (within the meaning of Code section 861(a)(3)), unless the Employee is within a group or classification of nonresident alien Employees designated as eligible to participate in the Plan by the Board of Directors or its delegate.
- (f) Any individual who is not classified for the relevant eligibility period by the Company or Affiliated Company on its payroll records as an employee of the Company or Affiliated Company under Code section 3121(d) (including, but not limited to an individual classified by the Company or Affiliated Company as an independent contractor, a non-employee consultant or as an employee of an entity other than the Company or Affiliated Company), even if such classification is determined to be erroneous, or is retroactively revised by a governmental agency, by court order or as a result of litigation, or

otherwise. In the event the classification of an individual who was excluded from the definition of Eligible Employee under the preceding sentence is determined to be erroneous or is retroactively revised, the individual shall nonetheless continue to be excluded from treatment as an Eligible Employee for all periods prior to the date the Company determines that its classification of the individual was erroneous or should be revised or is otherwise required to revise such classification.

(g) Any Leased Employee.

2.28 Employee. "Employee" shall mean each person currently employed as a common-law employee of the Company or an Affiliated Company. For services performed after December 31, 1986 for purposes of determining the number or identity of Highly Compensated Employees and for purposes of the requirements of Code sections 414(n)(3)(A) and (B), the term "Employee" shall include any person who is a leased employee within the meaning of Code section 414(n)(2) unless (i) such leased employees constitute less than 20% of the Company's nonhighly compensated workforce within the meaning of Code section 414(n)(5)(C)(ii) and (ii) such person is covered by a plan meeting the requirements of Code section 414(n)(5)(B). A leased employee includes any person (other than an employee of the recipient) who pursuant to an agreement between the recipient and any other person ("leasing organization") has performed services for the recipient (or for the recipient and related persons determined in accordance with Code section 414(n)(6)) on a substantially full time basis for a period of at least one year, and such services are performed under the primary direction or control by the recipient employer. Contributions or benefits provided a leased employee by the leasing organization which are attributable to services performed for the recipient employer shall be treated as provided by the recipient employer.

2.29 Employment Commencement Date. "Employment Commencement Date" shall mean each of the following:

- (a) The date on which an Employee first performs an Hour of Service in any capacity for the Company or an Affiliated Company with respect to which the Employee is compensated or is entitled to compensation by the Company or the Affiliated Company.
- (b) In the case of an Employee whose employment is terminated and who is subsequently reemployed by the Company or an Affiliated Company, the term "Employment Commencement Date" shall also mean the first day following the termination of employment on which the Employee performs an Hour of Service for the Company or an Affiliated Company with respect to which he is

compensated or entitled to compensation by the Company or Affiliated Company.

Unless the Board of Directors or its delegate shall expressly determine otherwise, an Employee shall not, for the purposes of determining his Employment Commencement Date, be deemed to have commenced employment with an Affiliated Company prior to the effective date on which the entity became an Affiliated Company.

- **2.30** ERISA. "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time.
- **2.31** Excess Aggregate Contributions. "Excess Aggregate Contributions" shall mean the aggregate amount of contributions with respect to a Plan Year which are taken into account in computing the Contribution Percentages of Highly Compensated Employees and which are actually made on behalf of such Employees in excess of the maximum amount of such contributions permitted under the nondiscrimination test set forth in Section 5.3 of the Plan.
- **2.32** Excess Contributions. "Excess Contributions" shall mean the aggregate amount of contributions with respect to a Plan Year which are taken into account in computing the Deferral Percentages of Highly Compensated Employees and which are actually made on behalf of such Employees in excess of the maximum amount of such contributions permitted under the nondiscrimination test set forth in Section 5.2(d) of the Plan.
- **2.33** Excess Deferrals. "Excess Deferrals" shall mean the contributions made on behalf of a Participant under any qualified cash or deferred arrangements described in Code sections 401(k), 408(k) or 403(b) for a calendar year which exceed the limitation set forth in Section 5.2(c) of the Plan and which are includable in the Participant's gross income under Code section 402(g).
- **2.34** 5% Owner. "5% Owner" shall mean an individual who owns (or is considered as owning, within the meaning of Code section 318) more than 5% of the outstanding stock of the Company or stock possessing more than 5% of the total combined voting power of all stock of the Company.
 - 2.35 <u>Highly Compensated Employee</u>. "Highly Compensated Employee" shall mean, for Plan Years beginning after December 31, 1996:
 - (a) The following Employees:
 - (i) Any Employee who was a 5% Owner at any time during the Plan Year or the preceding Plan Year, or

- (ii) For the preceding Plan Year
 - A) Any Employee who had Compensation from the Employer in excess of \$80,000 multiplied by the Adjustment Factor, and
 - B) If the Employer so elects by Plan amendment, any Employee in the "top-paid group" of Employees for such preceding Plan Year.
- (b) For purposes of this Section 2.35, the "top-paid group" shall be the group consisting of the top twenty percent (20%) of Employees when ranked on the basis of Compensation during the applicable year.
- (c) For purposes of this Section 2.35, "Compensation" shall mean Compensation within the meaning of Code section 415(c)(3), plus elective or salary reduction contributions to a cafeteria plan, cash or deferred arrangement or tax-sheltered annuity.

2.36 Hour of Service.

- (a) "Hour of Service" of an Employee shall mean the following:
- (i) Each hour for which the Employee is paid by the Company or an Affiliated Company (while it is an Affiliated Company) or entitled to payment for the performance of services as an Employee. An Employee will not be considered as being entitled to payment, for purposes of determining the computation period to which hours are to be credited, until the date the Company or Affiliated Company, as applicable, would normally make payment to the Employee for such hour based on normal payroll practices.
- (ii) Each hour in or attributable to a period of time during which the Employee performs no duties (irrespective of whether he has terminated his Employment) due to a vacation, holiday, illness, incapacity (including pregnancy or disability), layoff, jury duty, military duty or a leave of absence, for which he is so paid or so entitled to payment, whether direct or indirect. However, no such hours shall be credited to an Employee if such Employee is directly or indirectly paid or entitled to payment for such hours and if such payment or

entitlement is made or due under a plan maintained solely for the purpose of complying with applicable workmen's compensation, unemployment compensation or disability insurance laws or is a payment which solely reimburses the Employee for medical or medically related expenses incurred by him.

- (iii) Each hour for which an Employee is entitled to back pay, irrespective of mitigation of damages, whether awarded or agreed to by the Company or an Affiliated Company, provided that such Employee has not previously been credited with an Hour of Service with respect to such hour under paragraph (i) or (ii) above.
- (iv) Absence while on an authorized leave of absence shall count as employment with the Company provided such leave of absence is not of more than two years duration. Absence from employment on account of active duty with the Armed Forces of the United States shall be counted as employment with the Company. If the Employee does not return to active employment with the Company, his employment with the Company shall be deemed to have ceased on the date the Plan Administrator receives notice that such Employee will not return to active employment with the Company. Hours shall be credited for the computation period or periods in which the authorized leave takes place. However, no more than 501 hours shall be credited under this subsection for each such computation period.
- (v) The term "Hour of Service" shall also include (for those purposes designated by the applicable officer specified below) hours credited to an Employee for service with a predecessor employer provided that such service has been approved by the President, Chief Operating Officer, or Senior Vice President for Administration of the Company for recognition under this Plan, which approval shall apply on a nondiscriminatory basis to all Employees with service during the recognition period for such predecessor employer.
- (b) Hours of Service under subsections (a)(ii), (a)(iii), and (a)(iv) shall be calculated in accordance with Department of Labor regulation, 29 C.F.R. section 2530.200b-2(b). Hours of Service shall be credited to the appropriate computation period according to the Department of Labor regulation, 29 C.F.R. section 2530.200b-2(c).
- (c) To the extent not otherwise credited under this Section 2.36, Hours of Service determined with respect to a Maternity or Paternity Absence shall be credited as follows: the Employee shall be credited (solely for purposes of Section 2.13) with those Hours of Service that otherwise would normally have been credited to such Employee but for such absence, except that

- (i) the total number of Hours of Service so credited shall not exceed 501 and (ii) such Hours of Service shall be credited as Hours of Service in the Plan Year in which the absence from work commences only if the Employee would be prevented from incurring a Break in Service in such Plan Year solely by virtue of such crediting, and shall otherwise be credited in the Plan Year immediately following the Plan Year in which the absence from work commences.
- 2.37 Investment Manager. "Investment Manager" shall mean the one or more Investment Managers, if any, appointed pursuant to Section 9.3.
- **2.38 IRP Contributions**. "IRP Contributions" shall mean voluntary amounts contributed by a Participant prior to January 1, 1987 which the Participant designated in writing were eligible for a tax deduction under Code section 219. All such IRP Contributions have been distributed by January 1, 1991 in accordance with the provisions regarding IRP Contributions in the Plan in effect as of October 1, 1989.
- **2.39** <u>Leased Employee</u>. "Leased Employee" shall mean any person, other than an Employee, who has performed services for the Company, if the services were:
 - (a) Substantially full time for at least one year;
 - (b) According to an agreement between the Company and another person (the "leasing organization"); and
 - (c) Performed under the primary direction or control of recipient Employer.

Contributions or benefits attributable to services performed for the Company that the leasing organization makes on behalf of the Leased Employee will be treated as if provided by the Company.

A Leased Employee shall be deemed to be an Employee unless the following conditions are satisfied:

- (i) the Leased Employee is covered by a money purchase pension plan providing:
- (A) a nonintegrated employer contribution rate of at least 10% of Compensation (as defined in Code section 415(c)(3) but including amounts contributed pursuant to a salary reduction agreement that are excludable from the Leased Employee's income pursuant to Code sections 125, 402(e)(3), 402(h)(1)(b), or 403(b));

- (B) immediate participation; and
- (C) full and immediate vesting; and
- (ii) Leased Employees do not constitute more than 20% of the Nonhighly Compensated Employees of the Company.
- 2.40 Limitation Year. "Limitation Year" shall mean the calendar year.
- **2.41** <u>Matching Contribution</u>. "Matching Contribution" shall mean a contribution paid by the Company (other than Profit Sharing Contributions, Elective Deferrals or Additional Company Contributions) that is made on account of a Participant's Elective Deferrals pursuant to Section 5.3 and which is intended to qualify as a Matching Contribution pursuant to Code section 401(m)(4)(A). Such Matching Contributions, and the Plan provisions regarding them, became effective as of January 1, 1990.
- **2.42** <u>Maternity or Paternity Absence</u>. "Maternity or Paternity Absence" shall mean an absence from work for any period by reason of (a) an Employee's pregnancy, (b) the birth of a child of such Employee, (c) the placement of a child with the Employee in connection with the adoption of such child by such Employee, or (d) the caring for a natural or adopted child for a period beginning immediately following such birth or placement.
 - 2.43 Nonhighly Compensated Employee. "Nonhighly Compensated Employee" shall mean an Employee who is not a Highly Compensated Employee.
- **2.44** <u>Normal Retirement</u>. "Normal Retirement" shall mean a Participant's termination of employment with the Company as a result of such Participant attaining his Normal Retirement Date.
 - 2.45 Normal Retirement Age. "Normal Retirement Age" shall be the day on which a Participant attains age 65.
- **2.46** Normal Retirement Date. "Normal Retirement Date" means the first day of the month coinciding with or next following the date a Participant attains his Normal Retirement Age.
 - 2.47 NNS" shall mean Newport News Shipbuilding, Inc.

- 2.48 NNS 401(k) Plan. "NNS 401(k) Plan" shall mean the Newport News Shipbuilding, Inc. 401(k) Investment Plan for Salaried Employees which was effective December 16, 1996.
- **2.49** NNS Transfer After-Tax Account. "NNS Transfer After-Tax Account" shall mean the Account opened up and maintained for a Participant who formerly participated in the NNS 401(k) Plan and had an after-tax account in that plan which was transferred to the Plan.
- **2.50** NNS Transfer Deferral Account. "NNS Transfer Deferral Account" shall mean the Account opened up and maintained for a Participant who formerly participated in the NNS 401(k) Plan and had an elective deferral account in that plan which was transferred to the Plan.
- **2.51** NNS Transfer Employer Account. "NNS Transfer Employer Account" shall mean the Account opened up and maintained for a Participant who formerly participated in the NNS 401(k) Plan and had a matching or stock account in that plan which was transferred to the Plan.
- **2.52** NNS Transfer Rollover Account. "NNS Transfer Rollover Account" shall mean the Account opened up and maintained for a Participant who formerly participated in the NNS 401(k) Plan and had a rollover account in that plan which was transferred to the Plan.
- **2.53** <u>Participant.</u> "Participant" shall mean any Eligible Employee who has satisfied the eligibility requirements set forth in Section 3.1 and has been enrolled in this Plan in accordance with procedures established by the Plan Administrator.
- **2.54** <u>Participation Commencement Date</u>. "Participation Commencement Date" shall mean the day on which an Employee's participation in this Plan commences in accordance with the provisions of Article III.
 - **2.55 Plan.** "Plan" shall mean the AMSEC Employees 401(k) Profit Sharing Plan herein set forth, and as it may be amended from time to time.
- **2.56** <u>Plan Administrator</u>. "Plan Administrator" shall mean the administrator of the Plan, within the meaning of Section 3(16)(A) of ERISA. For Plan Years beginning on or after January 1, 1989, the Plan Administrator shall be the Company.
 - 2.57 Plan Year. "Plan Year" shall mean the calendar year.

- **2.58** <u>Profit Sharing Contribution</u>. "Profit Sharing Contribution" shall mean the contributions paid by the Company (other than Additional Company Contributions, Elective Deferrals or Matching Contributions) to the Trustee as provided in Article IV.
- **2.59** <u>Protected Benefits</u>. "Protected Benefits" shall mean the benefits, rights and features that are required to be preserved by Code section 411(d)(6), including but not limited to certain optional forms of benefit payments, such as annuities.
- **2.59A Qualified Matching Contribution.** "Qualified Matching Contribution" shall mean a contribution made on account of a Participant's Elective Deferrals pursuant to Section 5.3 and which are intended to qualify as matching contributions pursuant to Code section 401(k)(3)(D)(ii)(I).
- **2.60** Required Payment Commencement Date. "Required Payment Commencement Date" in the case of a Participant other than a 5% Owner, shall be the sixtieth day after the close of the latest Plan Year in which occurs:
 - (a) The Participant's Normal Retirement Date;
 - (b) The tenth anniversary of the date the Participant commenced participation in the Plan; or
 - (c) The Participant's termination of employment with the Company or an Affiliated Company, unless a Participant who retires on or after his Normal Retirement Date elects to defer distribution for a period of up to five years following his Normal Retirement Date; but in no event beyond April 1 following the calendar year in which the Participant attains age 70-1/2.

In the case of a 5% Owner, the Required Payment Commencement Date shall be the earlier of the date specified in the preceding sentence and April 1 following the calendar year in which the Participant attains age 70-1/2, whether or not the Participant has retired. If a Participant becomes a 5% Owner after attaining age 70-1/2, the Required Payment Commencement Date shall not be later than the last day of the calendar year in which the Participant becomes a 5% Owner.

- **2.61** Rollover Account. "Rollover Account" shall mean the Account opened up and maintained for an Employee making a Rollover Contribution as provided in Section 5.5 or on whose behalf a plan-to-plan transfer is made pursuant to Section 5.6.
 - **2.62** Rollover Contribution. "Rollover Contribution" shall mean a contribution to the Plan by an Eligible Employee of

amounts which qualify for exclusion from gross income pursuant to Code section 402(c), subject to the timing, maximum amount limitations, and other requirements of Code section 402(c).

- **2.63** <u>Suspended Participant.</u> "Suspended Participant" shall mean any Participant who remains an Employee but who ceases to be eligible to participate in this Plan by virtue of ceasing to be an Eligible Employee. Status as a Suspended Participant shall commence as of the date such Participant ceases to be an Eligible Employee. A Suspended Participant shall not be deemed a Participant except for those purposes specified in this Plan or as required by law.
- **2.64** <u>Trust and Trust Fund.</u> "Trust" or "Trust Fund" shall mean the one or more trusts or SubTrusts created for funding purposes under the Plan. The Trust Fund may be commingled for investment purposes with the assets of other qualified retirement plans maintained by the Company by investing through a master trust fund operated pursuant to a master trust agreement between the Company and the Trustee.
- **2.65 Trustee.** "Trustee" shall mean the individual or individuals or corporation named by the Board of Directors to act as Trustee or any successor to such individual or individuals or corporation.
- **2.66** <u>Valuation Date</u>. "Valuation Date" shall mean the date as of which the Trustee shall determine the value of the assets in the Trust Fund and the value of each Account, which shall be the last day of each Plan Year and such other dates as may be determined in rules prescribed by the Plan Administrator, which rules may prescribe different dates for the various investment funds comprising the Trust Fund.
- **2.67** <u>Vested Interest</u>. "Vested Interest" shall mean the nonforfeitable interest of a Participant in his Profit Sharing, Matching Contribution and NNS Transfer Employee Accounts which has become vested in accordance with Article VII and the entire balance of his Deferral Account, Voluntary Account, Rollover Account, NNS Transfer Deferral Account, NNS Transfer After-Tax Account and NNS Transfer Rollover Account, if any. The Vested Interest of an Alternate Payee in his Alternate Payee Account shall be determined as set forth in Article VII.
- **2.68** <u>Voluntary Contribution</u>. "Voluntary Contribution" shall mean a nondeductible employee contribution made by a Participant to the Plan in accordance with the rules of Article V.

- **2.69** <u>Voluntary Account</u>. "Voluntary Account" shall mean the Account in which a Participant's Voluntary Contributions to the Plan are held pursuant to Section 6.3.
- **2.70** Year of Service. "Year of Service" means a period of twelve consecutive months during which an employee completes at least 1,000 Hours of Service.
 - (a) For purposes of determining a Participant's Years of Service and Breaks in Service for vesting purposes, the twelve-consecutive-month period shall be the Plan Year. The Participant shall not be required to be an Employee on the last day of the Plan Year in order to receive credit for a Year of Service for such year for vesting purposes. If a Participant completes less than 1,000 Hours of Service during a Plan Year, but remains an employee of the Company, his vested percentage shall not be increased for such Plan Year. If an individual terminates employment with the Company, and is subsequently rehired as an Employee, upon his participation or subsequent participation, his vesting percentage shall take into account all Years of Service subject to subsection (c).
 - (b) In determining the vested percentage of an Employee, all Years of Service shall be taken into account except (i) Plan Years during which a Participant does not complete at least 1,000 Hours of Service, (ii) Years of Service completed prior to a Participant's Breaks in Service if the Participant had no Vested Interest in his Company Contribution Account at the time he terminated employment with the Company, and the number of Breaks in Service incurred by the Participant exceed the greater of (A) five and (B) the Participant's Years of Service completed before the Breaks in Service, (iii) for Plan Years after 1991, all Years of Service completed after a Participant has incurred five consecutive Breaks in Service for purposes of determining the Vested Interest of the Participant's Company Contribution Account attributable to contributions made prior to such Breaks, and (iv) Years of Service completed prior to July 1, 1981.
 - (c) Service with a predecessor organization of the Company shall be treated as service with the Company in any case in which the Company maintains the plan of such predecessor organization. Employees who transferred to the Company from Science Applications International Corporation, who had a hire date with the Company on or after September 8, 1987, shall receive credit for Hours of Service and Years of Service with Science Applications International Corporation under the Plan. Employees who transfer to the Company from corporations or business divisions as designated by the Plan Administrator shall be credited with five Years of Service for purposes of this Plan.

(d) For purposes of this Section 2.70, if any Plan Year is less than twelve consecutive months, the number of Hours of Service required to accrue a Year of Service (or to incur a Break in Service) in such short Plan Year shall bear the same ratio to 1,000 (or, in the case of a Break in Service, (500) as the number of days in the short Plan Year bears to 365.

ARTICLE III

ELIGIBILITY AND PARTICIPATION

3.1 Eligibility to Participate.

- (a) Each Employee who was a Participant as of the Effective Date shall continue participation in the Plan.
- (b) Each other Eligible Employee shall become a Participant as provided in Section 3.2.

3.2 <u>Automatic Commencement of Participation</u>.

- (a) Each Eligible Employee described in Section 3.1(b) shall be entitled to commence participation in this Plan as of his applicable Participation Commencement Date, which shall be the later of his Employment Commencement Date or the date he becomes an Eligible Employee.
- (b) A former Employee who is reemployed by the Company as an Eligible Employee, or a Suspended Participant who again becomes an Eligible Employee, shall commence participation as of the date he is reemployed as an Eligible Employee or again becomes an Eligible Employee, as applicable.
- (c) In the case of an acquisition by the Company of a new division or geographical unit of Employees, or any other acquisition of an identifiable group of Employees, the Plan Administrator may determine that Employees of such division, geographical unit or other new unit shall be immediately eligible to participate in the Plan effective on the date they become Employees of the Company (or a later date specified by the Plan Administrator, on or before the date participation would otherwise commence under this Section 3.2).

ARTICLE IV

TRUST FUND AND COMPANY CONTRIBUTIONS; VALUATION OF COMPANY STOCK CONTRIBUTED; INVESTMENT DIRECTION BY PARTICIPANTS; COMPANY STOCK TRANSACTIONS WITH DISQUALIFIED PERSONS

- **4.1** <u>Trust Fund</u>. The Company has established the Trust pursuant to a Trust Agreement under which the Trustee has agreed to hold and administer in trust all amounts previously accumulated under the Plan together with the additional funds deposited with the Trustee pursuant to the terms of this Plan. The Company shall have the authority to select the Trustee to act under the Trust Agreement and to enter into new or amended Trust Agreements as it deems advisable.
- **4.2** <u>Permissible Types of Plan Investments</u>. The assets of the Plan may be invested in the following types of assets as determined by the Plan Administrator and described in a notification to the Trustee:
 - (a) Company Stock. The Plan Administrator, in its discretion, may direct the Trustee to hold up to one hundred percent (100%) of the assets of the Trust in Company Stock;
 - (b) Other "Qualifying Employer Securities," as that term is defined in ERISA section 407(d)(5);
 - (c) "Qualifying Employer Real Property," as that term is defined in ERISA section 407(d)(4);
 - (d) Cash; or
 - (e) Any other property that is a permissible plan investment under applicable law.

The board of directors of Science Applications International Corporation ("SAIC") may determine that, in the event the board of directors makes such a determination, in accordance with SAIC's philosophy of remaining, to the maximum extent feasible, an employee-owned company and of restricting stock ownership (both direct and indirect through SAIC's various benefit plans) to Employees and designated consultants to SAIC, Company Stock, or a fund within the Trust which holds Company Stock, shall not be an available investment option for any former Employee who, pursuant to Sections 8.1, 8.2 or 8.5, has deferred payment of his Distributable Benefit. In the event the board of directors makes such a determination, the Plan Administrator shall establish procedures to implement the foregoing restriction on investment

in Company Stock or Company Stock fund, including the timing and procedures for any fund-to-fund transfers, appropriate default options and, if considered appropriate, any transition rules.

- **4.3** Company Contributions. The Company shall pay over to the Trustee in cash an amount equal to the aggregate of the amounts that each Participant elects (pursuant to Article V) to have contributed to the Plan on his behalf as Elective Deferrals. Such contributions shall be made no later than 15 business days after the month that such amounts are withheld from the Participant's Compensation. If necessary or appropriate, the Company, in its sole discretion, may make contributions to the Trust in excess of the amounts required under the preceding sentence as Profit Sharing Contributions or Additional Company Contributions. In addition, the Company shall pay over to the Trustee Matching Contributions in the amount determined under Section 5.3. With respect to Participants who formerly participated in the NNS 401(k) Plan and had their accounts in that plan transferred to the Plan pursuant to Section 5.6, the Company, in its sole discretion, may make Profit Sharing Contributions, Additional Company Contributions, or Matching Contributions to those Participants alone and in an amount that is different than the amount of similar Contributions available to other Participants. Company Contributions shall be made no later than the due date for filing the federal income tax return (including extensions) of the Company for the taxable year with respect to which the contribution is made. The Company may, in its discretion, make contributions at any time to defray Plan expenses as provided in Section 6.13.
- **4.4** Form of Company Contributions. The Company Contributions to the Trust Fund shall be paid in cash, Company Stock, or such other property as the Board of Directors may from time to time determine.
- **4.5** <u>Valuation of Company Contributions in the Form of Company Stock.</u> Company Stock contributed by the Company to the Trust fund shall be valued as of the date of contribution using the rules set forth in Section 6.11(b)(ii), treating the date of contribution as the Valuation Date.
- **4.6** <u>Investment of Company Contributions and Voluntary Contributions</u>. All contributions to the Plan (whether Company Contributions, Voluntary Contributions, Rollover Contributions or plan-to-plan transfers) shall be invested as provided in this Section 4.6.
 - (a) The Plan Administrator may establish a choice of investment alternatives that each Participant may select from in determining the manner in which his Account(s) will be invested.

The Plan Administrator shall prescribe procedures for investment of amounts allocated to an Alternate Payee Account. Company Contributions or a specific portion thereof may be directed to be invested in a Company Stock fund within the Trust, subject to the direction of the Plan Administrator.

- (b) If such investment alternatives are established, each Participant may elect to invest the assets of his Accounts in such alternatives at such time, in such manner, and subject to such restrictions as the Plan Administrator shall specify.
- (c) Separate funds within the Trust Fund shall be established to reflect the available alternatives, and separate subaccounts shall be established for each investment alternative selected by a Participant, and each such subaccount shall be valued separately.
- (d) The Plan Administrator, in its discretion, may permit Participants to transfer amounts from one investment alternative to one or more other investment alternatives. An election to transfer such amounts shall be made only at such time, in such manner, and subject to such restrictions as the Plan Administrator may specify which in any event shall not be less frequent than once each month. The Plan Administrator may provide that future contributions may be invested in a different investment alternative than amounts already accumulated in the Participant's Account(s).
- (e) The Plan Administrator shall prescribe rules relating to the investment of the assets in the Accounts of a Participant who fails to make an effective election, for any reason whatsoever, as to how all or a portion of his Accounts shall be invested.
- (f) The Plan Administrator shall provide notice by arranging for reports to Participants regarding the investment of their funds pursuant to their investment elections. Failure of a Participant to notify the Plan Administrator regarding implementation of his investment election within thirty (30) days following such notice shall be deemed to be an election to have the Accounts invested in the manner shown on such report, even if the manner of investment is different from that specified in the Participant's election form or investment instructions. If a Participant has not received a notice confirming his investment election (or change therein) and does not notify the Plan Administrator or its designated delegate within thirty (30) days of the date such election (or change) was to be effective, the Participant shall be deemed to have elected to have the Accounts invested in the manner in which they are in fact invested, even

if that method differs from the Participant's election form or investment instructions.

- **4.7** <u>Irrevocability.</u> The Company shall have no right or title to, nor interest in, the contributions made to the Trust Fund, and no part of the Trust Fund shall revert to the Company except that funds may be returned to the Company as follows:
 - (a) In the case of a Company Contribution which is made by a mistake of fact, that contribution shall be returned to the Company within one (1) year after such contribution is made.
 - (b) If the Commissioner of the Internal Revenue Service determines that the Plan is not initially qualified under the Code, any contribution made incident to that initial qualification by the Company shall be returned to the Company within one year after the date the initial qualification is denied, but only if the application for the qualification is made by the time prescribed by law for filing the Company's return for the taxable year in which the Plan is adopted, or such later date as the Secretary of the Treasury may prescribe.
 - (c) All Company Contributions to the Plan are conditioned upon the deductibility of those contributions under Code section 404. To the extent a deduction is disallowed, the contribution shall, at the election of the Company, be returned to the Company within one year after the disallowance.

4.8 Company, Plan Administrator and Trustee Not Responsible for Adequacy of Trust Fund.

- (a) Neither the Company nor the Trustee shall be liable or responsible for the adequacy of the Trust Fund to meet and discharge any or all payments and liabilities hereunder. All Plan benefits will be paid from the Trust assets, and neither the Company nor the Trustee shall have any duty or liability to furnish the Trust with any funds, securities or other assets except as expressly provided in the Plan.
- (b) Except as required under the Plan or Trust or under Part 4 of Title I of ERISA, the Company shall not be responsible for any decision, act or omission of the Trustee or the Investment Manager (if applicable), and shall not be responsible for the application of any moneys, securities, investments or other property paid or delivered to the Trustee.
- (c) The Company expressly disavows any contractual obligation, implied or explicit, to make any contribution to the Plan or to contribute any specified amount.

4.9 Company Stock Transactions with Disqualified Persons. Acquisition or sale by the Plan of Company Stock or other qualifying employer securities (as defined in ERISA section 407(a)(5)), from or to a "disqualified person" (as defined in Code section 4975(e)(2)), shall be at a price which represents "adequate consideration" (as defined in ERISA section 3(18)) or, in the event such Company Stock or other qualifying employer security is a marketable obligation (as defined in ERISA section 407(e)), at a price not less favorable to the Plan than the price determined under ERISA section 407(e)(1). No commission shall be charged to the Plan in connection with any such sale or acquisition. The determination as to whether or not such a sale or acquisition satisfies the requirements of this Section 4.9 shall be made by the Plan Administrator.

ARTICLE V

ELECTIVE DEFERRALS, ROLLOVER CONTRIBUTIONS, PLAN-TO-PLAN TRANSFERS AND VOLUNTARY CONTRIBUTIONS

5.1 <u>Election</u>. At dates established by the Plan Administrator, each Participant may elect, on forms prescribed by the Plan Administrator, to defer the receipt of a portion of his Compensation for that Plan Year and to have the deferred amount contributed directly by the Company to the Trust Fund as Elective Deferrals.

5.2 Amount Subject to Election; Nondiscrimination Requirement and Distribution of Excess Contributions.

- (a) The amount of an individual's Compensation that may be deferred subject to the election provided in Section 5.1 shall be a percentage of the individual's Compensation, not to exceed the maximum percentage established by the Plan Administrator, except with respect to Participants who formerly participated in the NNS 401(k) Plan and whose accounts in that plan are transferred to the Plan pursuant to Section 5.6. With regard to such Participants who formerly participated in the NNS 401(k) Plan, the amount of such employees' Compensation that may be deferred, subject to the election provided in Section 5.1, shall be an amount equal to not less than one percent (1%) or such percentage of Compensation greater than 12 percent (12%) as may be specified from time to time by the Plan Administrator, provided that the percentage rate for determining such contributions shall be a whole number.
- (b) For purposes of Section 5.2(a), (i) the amount of the individual's Compensation shall be determined prior to the reduction caused by the deferral (if any) of a portion of his Compensation pursuant to this Article V and (ii) Compensation shall not include bonuses, income from stock options, vesting stock or non-cash remuneration.
- (c) The maximum amount of Elective Deferrals contributed on behalf of a Participant under the Plan shall not exceed \$7,000 (multiplied by the Adjustment Factor) during any calendar year. If a Participant's Elective Deferrals to this Plan exceed the limitation described in the preceding sentence, the Excess Deferral may be distributed to such Participant as Permitted by Code section 402(g)(A)(ii). Effective for Plan Years beginning on and after January 1, 1992, if, during a calendar year, (i) the Employee participates in this Plan and one or more other plans with a qualified cash or deferred arrangement described in Code sections 401(k), 408(k)(6), 125 or 403(b),

- (ii) the Employee's Elective Deferrals under all such plans (including this Plan) exceed \$7,000 (multiplied by the Adjustment Factor), and (iii) all or any portion of such Employee's Elective Deferrals under this Plan constitute Excess Deferrals by virtue of an allocation and notification by the Employee pursuant to Code section 402(g)(2)(A)(i), such Excess Deferrals shall not be distributed to such Employee as permitted by Code section 402(g)(2)(A)(i), notwithstanding that the Employee may request such distribution, but rather shall be held in the Trust Fund in the same manner as such Employee's Elective Deferrals which do not constitute Excess Deferrals.
- (d) The Company shall distribute Excess Contributions to Highly Compensated Employees as provided in subsection (e) below to the extent necessary to satisfy one of the deferral percentage tests described below:
 - (i) The Average Deferral Percentage of Highly Compensated Employees who are Participants is not more than the product of 1.25 and the Average Deferral Percentage of all Non-Highly Compensated Employees for the current Plan Year.
 - (ii) The Average Deferral Percentage of Highly Compensated Employees who are Participants is not more than the lesser of (A) 200% of the Average Deferral Percentage of all Non-Highly Compensated Employees for the current Plan Year and (B) the Average Deferral Percentage of all Non-Highly Compensated Employees for the current Plan Year, plus two percentage points.

In determining the Average Deferral Percentages, the Deferral Percentages of all "Qualified Employees" shall be taken into account. For this purpose, a Qualified Employee is any Employee who is directly or indirectly eligible under the Plan to make Elective Deferrals for all or a portion of a Plan Year and includes:

- (A) an Employee who would be a Participant but for the failure to make Elective Deferrals;
- (B) an Employee whose eligibility to make Elective Deferrals has been suspended because of an election not to participate or receipt of a distribution or a loan; and
- (C) an Employee who is unable to make an Elective Deferral because his Compensation is less than a stated dollar amount. In the case of a Qualified Employee who makes no Elective Deferrals, the Deferral Percentage that is to be included in determining the Average Deferral Percentage shall be zero.

- (e) Excess Contributions for the preceding Plan Year plus any income allocable to such contributions shall be distributed no later than the last day of each Plan Year to the Highly Compensated Employees to whose accounts such Excess Contributions were allocated. Excess Contributions shall be treated as Annual Additions. The income allocable to such contributions shall include the income accrued during the Plan Year for which the Excess Contributions were made and income for the period between the end of the Plan Year and the date of the distribution. Such income shall be determined pursuant to procedures established by the Plan Administrator consistent with Code section 401(k)(8)(A) and Treasury regulation section 1.401(k)-1(f)(4).
- (f) Distribution of Excess Contributions (plus any income or minus any loss allocable thereto) shall be made to Highly Compensated Employees on the basis of the amount of the Excess Contributions by, or on behalf of, each Highly Compensated Employee. Excess Contributions shall be distributed from the Participant's Deferral Account as follows:
 - (i) Calculate the dollar amount of Excess Contributions for each Highly Compensated Employee consistent with the provisions of Code section 401(k)(8)(B).
 - (ii) Add the dollar amounts calculated in subsection (i) above to arrive at the total dollar amount of the Excess Contributions for the Plan Year, and distribute such total amount in accordance with subsections (iii)-(v) below.
 - (iii) The Deferral Account of the Highly Compensated Employee with the highest dollar amount of Elective Deferrals for the Plan Year shall be reduced by the dollar amount required to cause such Highly Compensated Employee's Elective Deferrals for the Plan Year to equal the dollar amount of the Elective Deferrals of the Highly Compensated Employee with the next highest dollar amount of Elective Deferrals. Such reduction amount shall be distributed from the Deferral Account of the Highly Compensated Employee. However, if a lesser reduction, when added to the total dollar amount already distributed pursuant to this subsection (iii) would equal the total Excess Contributions as determined pursuant to subsection (iii) above, then the lesser reduction amount shall be distributed from such Highly Compensated Employee's Deferral Account.
 - (iv) If the total amount distributed in subsection (iii) is less than the total Excess Contributions determined pursuant to subsection (ii) above, then the reduction and distribution process in subsection (iii) above shall be repeated.

(v) No distributions of Excess Contributions (plus income) shall be made to any Highly Compensated Employee as long as any other Highly Compensated Employee has a higher amount of Elective Deferrals.

Any reduction in the amount of Elective Deferrals to be made on behalf of a Participant or any distribution of Excess Contributions under this subsection (e) will also be effective for purposes of determining the amount of Matching Contributions, if any, allocated to such Participant under Section 5.3.

- (g) The Deferral Percentage for any Participant who is a Highly Compensated Employee for the Plan Year and who is eligible to have salary deferrals described in Code section 402(g)(3)(A) or qualified nonelective contributions described in Code section 401(m)(4)(C) allocated on his behalf under two or more plans or arrangements described in Code section 401(k) that are maintained by the Company or any Affiliated Company shall be determined as if all such deferrals and contributions were made under a single arrangement. Notwithstanding the foregoing, certain plans shall be treated as separate if mandatorily disaggregated under regulations under Code section 401(k).
- (h) The Plan Administrator may prescribe such rules as it deems necessary or appropriate regarding the maximum amount that a Participant may elect to defer and the timing of such an election. These rules may prescribe a maximum percentage of Compensation that may be deferred, or may provide that the maximum percentage of Compensation that a Participant may defer will be a lower percentage of his Compensation above a certain dollar amount of Compensation than the maximum deferral percentage below that dollar amount of Compensation. These rules shall apply to all individuals eligible to make the election described in Section 5.1, except to the extent that the Plan Administrator prescribes special or more stringent rules applicable only to Highly Compensated Employees.
- (i) The Plan Administrator may, in its discretion, estimate, as soon as is reasonably possible prior to the close of the Plan Year, the extent (if any) to which deferral treatment under Code section 401(k) may not be available to any Participant (or class of Participants). If, pursuant to these estimates by the Plan Administrator, a Participant (or a class of Participants) is not eligible for deferral treatment for any or all of the amounts deferred pursuant to the election set forth above in Section 5.1, then the Plan Administrator may authorize a suspension of Elective Deferrals made pursuant to the provisions of Article V. This suspension shall be made pursuant to rules promulgated by the Plan Administrator. These rules may include

provisions authorizing the suspension of Elective Deferrals above a specified dollar amount of Compensation, or any other means that the Plan Administrator believes will help to ensure that deferral treatment will be available for all Participants.

(j) In the event that the Plan Administrator determines that an amount to be deferred pursuant to the election provided in Section 5.1 would cause the Company Contributions under this and any other tax-qualified retirement plan maintained by the Company or any Affiliated Company to exceed the applicable deduction limitations contained in Code section 404, or to exceed the maximum Annual Addition determined in accordance with Article XIII, the Plan Administrator may treat such amount in accordance with the rules set forth above in Section 5.2(i).

5.3 Matching Contributions; Nondiscrimination Requirement and Distribution of Excess Aggregate Contributions.

- (a) Effective for Plan Years beginning on or after January 1, 1990, the Company may, in its discretion, make a Matching Contribution in an amount equal to a percentage (determined each year in the Company's discretion) for each \$1 of a Participant's Elective Deferrals up to a maximum of \$5,000 of such Elective Deferrals, subject to the limitations specified in Article XIII, except with respect to Participants who formerly participated in the NNS 401(k) Plan and whose accounts in that plan are transferred to the Plan pursuant to Section 5.6. With regard to such Participants who formerly participated in the NNS 401(k) Plan, the Company, in its sole discretion, may make a Matching Contribution to those Participants alone and in an amount which is different than that available to other Participants, subject to the limitations specified in Article XIII.
- (b) The contribution described in Section 5.3(a) shall be paid to the Trustee on the last day of the calendar quarter, or as soon as possible thereafter, but in any event no later than the date which is prescribed by law for filing the Company's federal income tax return, including any extensions for the taxable year with respect to which such contributions are made. A Matching Contribution shall be allocated as provided in Section 6.10(b) and made to each Participant, including former NNS employees, who is performing duties as a wage earning Employee for the Company as of the last day of a calendar quarter and any Participant who dies, retires or becomes disabled during such quarter. Notwithstanding the above, for the period of January 1, 1995 through June 30, 1995, Matching Contributions will be paid to the Trustee and allocated to each Participant's account as of June 30, 1995 for those Participants who are employed on June 30, 1995.

- (c) Excess Aggregate Contributions, plus any income or minus any loss allocable thereto shall be forfeited, if forfeitable, or if not forfeitable, distributed to the Highly Compensated Employees to whose Accounts they were allocated, as set forth in subsections (d) and (e) below to the extent necessary to satisfy one of the following contribution percentage tests:
 - (i) The Average Contribution Percentage of Highly Compensated Employees who are Participants is not more than the product of 1.25 and the Average Contribution Percentage of all other Participants for the current Plan Year.
 - (ii) The Average Contribution Percentage of Highly Compensated Employees who are Participants is not more than the lesser of (A) 200 percent of the Average Contribution Percentage of all other Participants for the current Plan Year and (B) the Average Contribution Percentage of all other Participants for the current Plan Year plus two points.
- (d) In determining the Average Contribution Percentages, the Contribution Percentages of all "Qualified Employees" shall be taken into account. For this purpose, a Qualified Employee is any Employee who is directly or indirectly eligible under the Plan to receive an allocation of Matching Contributions for all or a portion of a Plan Year and includes:
 - (i) an Employee who would be a Plan Participant but for the failure to make Elective Deferrals;
 - (ii) an Employee whose eligibility to make Elective Deferrals has been suspended because of an election not to participate or receipt of a hardship distribution;

In the case of a Qualified Employee who makes no Elective Deferrals, and thus receives no Matching Contributions, the Contribution Percentage that is to be included in determining the Average Contribution Percentage shall be zero.

(e) Excess Aggregate Contributions (plus any income or minus any loss allocable thereto) for a Plan Year, to the extent not vested pursuant to Article VII, shall be forfeited and to the extent vested, they shall be distributed no later than the last day of the next succeeding Plan Year to those Highly Compensated Employees to whose accounts such Excess Aggregate Contributions were allocated. Excess Aggregate Contributions shall be treated as Annual Additions.

- (f) Distribution of Excess Aggregate Contributions (plus any income or minus any loss allocable thereto) shall be made to Highly Compensated Employees on the basis of the amount of the Matching Contributions on behalf of each Highly Compensated Employee. Excess Contributions shall be distributed from the Participant's Matching Contribution Account as follows:
 - (i) Calculate the dollar amount of Excess Aggregate Contributions for each Highly Compensated Employee consistent with the provisions of Code section 401(m)(6)(D).
 - (ii) Add the dollar amounts calculated in subsection (i) above to arrive at the total dollar amount of the Excess Aggregate Contributions for the Plan Year, and distribute such amount in accordance with subsections (iii)-(vi) below.
 - (iii) The Matching Contribution Account of the Highly Compensated Employee with the highest dollar amount of Matching Contributions for the Plan Year shall be reduced by the dollar amount required to cause such Highly Compensated Employee's Matching Contributions for the Plan Year to equal the dollar amount of the Matching Contributions of the Highly Compensated Employee with the next highest dollar amount of Matching Contributions. Such reduction amount shall be distributed from the Matching Contribution Account of the Highly Compensated Employee. However, if a lesser reduction, when added to the total dollar amount already distributed pursuant to this subsection (iii) would equal the total Excess Aggregate Contributions as determined pursuant to subsection (ii) above, then the lesser reduction amount shall be distributed from such Highly Compensated Employee's Matching Contribution Account.
 - (iv) If the total amount distributed in subsection (iii) is less than the total Excess Aggregate Contributions determined pursuant to subsection (ii) above, then the reduction and distribution process in subsection (iii) above shall be repeated.
 - (v) Income or loss attributable to such Excess Aggregate Contributions shall be determined based on the income or loss in the Company Contribution Account of each Highly Compensated Employee to whom such Excess Aggregate Contributions are distributed (or from whose accounts such Excess Aggregate Contributions are forfeited) for the Plan Year for which the Excess Aggregate Contributions were made. Such income or loss shall be determined pursuant to procedures established by the Plan Administrator consistent with Code section 401(m)(6) and Treasury regulation section 1.401(m)-1(e)(3).

- (vi) No distributions of Excess Aggregate Contributions (plus income) shall be made to any Highly Compensated Employee as long as any other Highly Compensated Employee has a higher amount of Matching Contributions.
- (g) The Contribution Percentage of a Highly Compensated Employee who is eligible to participate in two or more plans described in Code section 401(a) or arrangements described in Code section 401(k) maintained by the Company or an Affiliated Company to which Matching Contributions or employee after-tax contributions are made shall be aggregated for purposes of determining such Employee's Contribution Percentage.
- (h) Any Excess Aggregate Contributions that are forfeited shall be added to Company Contributions and allocated as provided in Section 6.12(a); provided, however, that forfeitures of Excess Aggregate Contributions shall not be allocated to the Account of any Highly Compensated Employee whose Matching Contributions Account is reduced pursuant to Subsection 5.3(f).
- (i) If one or more Highly Compensated Employees participates in a cash or deferred arrangement subject to the nondiscrimination test in Code section 401(k) and a plan subject to the nondiscrimination test in Code section 401(m) that are maintained by the Company or an Affiliated Company, and the sum of the Average Deferral Percentage and Average Contribution Percentage of the Highly Compensated Employees subject to either or both of those tests exceeds the aggregate limit defined below, the Matching Contribution Account of the Highly Compensated Employees who also participate in the cash or deferred arrangement shall be reduced (beginning with the Highly Compensated Employee whose Matching Contributions for the Plan Year are the highest) to the extent necessary to come within the Applicable Limit. The amount by which each Highly Compensated Employee's Matching Contribution Account is reduced shall be treated as an Excess Aggregate Contribution. The Average Deferral Percentage and Average Contribution Percentage of the Highly Compensated Employees shall be determined after any corrections required to meet the Average Deferral Percentage test and the Average Contribution Percentage test. Multiple use does not occur if either the Average Deferral Percentage or the Average Contribution Percentage of the Highly Compensated Employee does not exceed 1.25 multiplied by the Average Deferral Percentage and Average Contribution Percentage of the Nonhighly Compensated Employees for the current Plan Year. Effective for Plan Years ending on or before December 31, 2001, the provisions of Reg. Section 1.401(m)-2(b) shall apply for the purpose of applying the test for multiple use of the alternative limitation.

5.4 Termination of, Change in Rate of, or Resumption of Deferrals.

- (a) A Participant may submit on form(s) prescribed by the Plan Administrator a request to the Plan Administrator that he wishes to terminate, alter the rate of, or resume his Elective Deferrals made pursuant to this Article V. Such changes may be made once each month.
- (b) This request for termination, alteration, or resumption shall be effective as soon as practicable, subject to Section 5.2(i), and shall be effective on the later of the date specified in the notice of the change and the date which is fifteen days after the notice is received by the Plan Administrator.
- 5.5 Rollover Contributions From Other Plans. An Eligible Employee, regardless of whether he has satisfied the participation requirements of Section 3.1, may contribute to the Trust Fund a Rollover Contribution pursuant to procedures approved by the Plan Administrator. The Plan Administrator shall develop such procedures and may require such information from an Employee desiring to make such a contribution as it deems necessary or desirable to determine that the proposed contribution will meet the requirements of a Rollover Contribution. Upon approval by the Plan Administrator, the amount contributed shall be deposited in the Trust Fund and shall be credited to a Rollover Account. Such account shall be one hundred percent (100%) vested in the Employee, shall be valued in accordance with Section 6.11, but shall not share in allocations of any Profit Sharing Contributions. Upon termination of employment, the total amount of the Voluntary Account shall be distributed as soon as practicable following termination of employment or, if the Employee so elects, treated as part of the Employee's "Distributable Benefit" under Section 8.5 and distributed in accordance with Sections 8.1 through 8.7, as applicable.

Upon such contribution by an Employee who is otherwise eligible to participate in the Plan, but who has not yet completed the participation requirements of Section 3.1, his Rollover Account shall represent his sole interest in the Plan until he becomes a Participant.

5.6 <u>Plan-to-Plan Transfers</u>. In its sole discretion, the Plan Administrator may authorize (subject to procedures established by the Plan Administrator) the Trustee to accept a transfer, from a plan qualified under Code section 401(a), of a Participant's account(s) balance under such transferee plan but

only to the extent such account(s) balances are in cash and satisfy each of the following requirements:

- (a) If it is made to this Plan from a plan which contained Protected Benefits, such Protected Benefits shall continue to be maintained with respect to the amounts transferred whether in the Participant's NNS Transfer Accounts or in the Participant's other Accounts, depending upon the character of the contribution under the transferee plan.
- (b) To the extent required by Code section 411(d)(6), and pursuant to procedures established by the Company, an amount transferred to the Participant's Accounts, including a Participant's NNS Transfer Accounts, if any, pursuant to this Section 5.6 shall retain the Protected Benefits that were applicable to such Account balance prior to the transfer to the Plan. Notwithstanding the foregoing, the Company reserves the right to eliminate all or any of such Protected Benefits if and to the extent permitted by Code section 411(d)(6) and any regulations or guidance promulgated thereunder.
- **5.7 Voluntary Contributions.** No Voluntary Contributions are permitted subsequent to December 31, 1986. Voluntary Contributions made prior to such date shall be maintained in a Voluntary Account as set forth herein.

ARTICLE VI

ACCOUNTING AND ALLOCATION PROCEDURES

- **6.1** <u>Deferral Account</u>. The Plan Administrator shall open and maintain a separate Deferral Account for each Participant in the Plan who has made Elective Deferrals.
- **6.2** <u>Company Contribution Account</u>. The Plan Administrator shall open up and maintain a separate Company Contribution Account for each Participant in the Plan who has been allocated a Profit Sharing Contribution or Matching Contribution.
- **6.3** <u>Voluntary Account</u>. The Plan Administrator shall continue to maintain a separate Voluntary Account for each Participant who has made a Voluntary Contribution to the Plan.
- **6.4** Rollover Account. The Plan Administrator shall open and maintain a separate Rollover Account for each Participant in the Plan who has made a Rollover Contribution to the Plan.
- **6.5** <u>Alternate Payee Account</u>. The Plan Administrator shall open and maintain a separate Alternate Payee Account for each Alternate Payee who is awarded benefits under this Plan pursuant to a qualified domestic relations order within the meaning of Code section 414(p).
- **6.6** NNS Transfer Deferral Account. The Company shall open and maintain a separate NNS Transfer Deferral Account for each Participant who formerly participated in the NNS 401(k) Plan and had an elective deferral account in that plan which was transferred to the Plan.
- **6.7** NNS Transfer After-Tax Account. The Company shall open and maintain a separate NNS Transfer After-Tax Account for each Participant who formerly participated in the NNS 401(k) Plan and had a after-tax account in that plan which was transferred to the Plan.
- **6.8** NNS Transfer Rollover Account. The Company shall open and maintain a separate NNS Transfer Rollover Account for each Participant who formerly participated in the NNS 401(k) Plan and had a rollover account in that plan which was transferred to the Plan.
- **6.9** NNS Transfer Employee Account. The Company shall open and maintain a separate NNS Transfer Employee Account for each Participant who formerly participated in the NNS 401(k) Plan and

had a matching or stock grant account in that plan which was transferred to the Plan.

6.10 Allocation of Company Contributions.

- (a) All Company Contributions on account of Elective Deferrals shall be allocated to Participants' Deferral Accounts as soon as practicable following their contribution to the Trust, but in no event later than ninety (90) days after such amounts were deferred.
- (b) All Company Contributions which constitute Matching Contributions shall be allocated to Participants' Company Contribution Accounts within ninety (90) days following their contribution to the Trust. An allocation shall be made on behalf of a Participant who is otherwise eligible to receive a Matching Contribution only if such Participant is a wage-earning Employee of the Company on the last day of the calendar quarter, or if such Participant dies, retires or becomes disabled during such quarter.
- (c) Any Company Contributions which constitute Profit Sharing Contributions shall be allocated to the Company Contribution Account of each Participant who is performing duties as a wage-earning Employee of the Company on the last day of the Plan Year. An allocation also shall be made for any Participant who dies, retires or becomes disabled during the Plan Year. The Profit Sharing Contribution shall be allocated to each Participant eligible to receive a share of such contribution (except for those Participants who formerly participated in the NNS 401(k) Plan) in the proportion that the Compensation paid to each such Participant during the Plan Year bears to the Compensation paid to all such Participants. With respect to Participants who formerly participated in the NNS 401(k) Plan and whose accounts in that plan were transferred to the Plan pursuant to Section 5.6, any Profit Sharing Contribution shall be allocated among each such Participant who is otherwise eligible to receive a share of such contribution in the proportion that the Compensation paid to each such Participants during the Plan Years bears to the Compensation paid to all such Participants who formerly participated in the NNS 401(k) Plan.
- (d) Any Company Contributions for a Plan Year which constitute Additional Company Contributions shall be held on an unallocated basis until allocated to Participants' Deferral Accounts as provided under this Plan. Prior to allocation, these contributions shall not be adjusted for any income, gain or loss thereon, which shall be allocated separately pursuant to Section 6.13.

- (e) Company Contributions in the form of Additional Company Contributions shall be allocated to Participants' Deferral Accounts so as to allow the Plan to satisfy the nondiscrimination test set forth in Code section 401(k)(3)(A)(ii) and Section 5.2(d) of the Plan as applied to Elective Deferrals and the nondiscrimination test set forth in Code section 401(m)(2) and Section 5.3 of the Plan as applied to Matching Contributions. Subject to the limitations of Article XIII of this Plan and to the requirements of Code sections 401(k)(3)(A)(ii) and 401(m)(2), effective January 1, 1992, Additional Company Contributions shall be allocated as an equal dollar amount to Deferral Accounts of all Eligible Employees who are Nonhighly Compensated Employees who are taken into account in the nondiscrimination tests set forth in Sections 5.2 and 5.3 of the Plan.
- (f) Company Contributions made pursuant to Section 6.13 shall not be allocated to Accounts of Participants, but (together with any gains or income thereon) shall be applied to defray Plan expenses.
 - (g) In no event shall amounts be allocated which would cause the limitation on Annual Additions set forth in Article XIII to be exceeded.
- (h) Company Contributions in the form of Company Stock shall be allocated to the Accounts of the Participants based on the fair market dollar value of such contributed Company Stock as determined under the provisions of Section 6.11(b)(ii), unless a different valuation method shall be required under the applicable Treasury regulations.

6.11 Valuation of Accounts.

(a) As of each Valuation Date and within sixty (60) days after the removal or resignation of the Trustee, and at such other times as determined by the Plan Administrator, the Trustee shall value the assets of the Trust on the basis of fair market values.

- (b) As soon as is reasonably possible after receipt of these valuations from the Trustee, the Plan Administrator shall value the Accounts of each Participant, Suspended Participant and Alternate Payee as of the applicable Valuation Date so as to reflect the current fair market value of each Account as of such Valuation Date. The valuation provisions of this Section 6.11 shall be applied and implemented in accordance with the following rules:
 - (i) If separate subaccounts have been established for separate investment alternatives, each subaccount shall be valued separately and the total value of a Participant's Account(s) shall equal the total value of his interest in each of the respective subaccounts in which his Account(s) have been invested.
 - (ii) Company Stock allocated and credited to an Account or subaccount, or to a separate fund within the Trust Fund in which Participants' Accounts or subaccounts are invested as provided in Section 4.6, as well as Company Stock held on an unallocated basis in the Trust Fund, shall be valued as of the applicable Valuation Date, according to the following rules:
 - (A) Company Stock acquired by the Trust Fund with cash shall initially be valued at the purchase price paid for such stock. On any subsequent Valuation Date, such Company Stock, as well as all other Company Stock held in, or contributed to, the Trust Fund, shall be valued in accordance with Section 6.11(b)(ii)(B) or 6.11(b)(ii)(C) below, as applicable.
 - (B) If any Company Stock does not consist of securities listed on a national securities exchange, or traded on a regular basis, as determined by the Company, in the over-the-counter market, the fair market value of such stock shall be determined using the formula price for such stock, as described in the August 24, 1987 Prospectus for Science Applications International Corporation (or the most recent prospectus that supersedes that prospectus), on the applicable Valuation Date. The Plan Administrator may at any time, and from time to time, change the method of determining the fair market value of Company Stock, provided that the replacement method is consistent with applicable provisions of ERISA and the Code. A Participant, Beneficiary or Alternate Payee shall have no right to have a particular valuation method applied (or continue to be applied) to his Account(s).
 - (C) If any Company Stock consists of securities listed on a national securities exchange, fair market value of such Company Stock shall be considered to be equal to the closing price of such Company Stock (as reported in the consolidated transaction reporting system, or if not so reported, as reported on the principal exchange market for such Company Stock by such exchange or on any system sponsored by such exchange) on the trading day immediately preceding the day applicable Valuation Date. If any Company Stock consists of securities traded on a regular basis, as determined by the Company, in the over-the-counter market, the fair market value of such Company Stock shall be considered to be equal to the average between the high bid price and the low asked price quoted by the

- automatic quotation system of a securities association registered under the federal securities laws for the trading day immediately preceding the applicable Valuation Date.
- (iii) The fair market value of any guaranteed interest contract, trust or fund holding such a contract, or similar program entered into between an insurance company and the Plan shall be determined based on the principal amount of such contract or program, plus the amount of the guaranteed interest or other increase in value which is paid or credited to the Plan pursuant to such contract or program. The provisions of this subparagraph (iii) shall apply to any investment alternative established under Section 4.6 which is invested in such a contract or program.
- (iv) To the extent that a Participant's Account is invested in a regulated investment company offered as an investment alternative under the Trust, the value of that portion of the Account shall be valued, pursuant to rules prescribed by the Plan Administrator, based on the unit or share value of the regulated investment company on the applicable Valuation Date.
- (c) The Company, the Plan Administrator and Trustee do not in any manner or to any extent whatsoever warrant, guarantee or represent that the value of a Participant's Account shall at any time equal or exceed the amount previously contributed thereto, or that any valuation or accounting method or practice will continue to be applied.
- (d) <u>Allocation of Company Stock Received Pursuant to Stock Dividends, Splits, Recapitalizations, Etc</u>. Any Company Stock received by the Trustee as a stock split, dividend, or as a result of a reorganization or other recapitalization of the Company shall be allocated as of the day on which the stock is received by the Trustee in the same manner as the Company Stock to which it is attributable is then allocated.

(e) Allocation of Stock Rights, Warrants or Options.

- (i) In the event any rights, warrants or options are issued on Company Stock held in the Trust Fund, the Trustee shall exercise them for the acquisition of additional Company Stock as directed by the Plan Administrator and to the extent that cash is then available in the Trust Fund.
- (ii) Any Company Stock acquired in this fashion shall be treated as Company Stock purchased by the Trustee for the net price paid and shall be allocated in the same manner as the funds used to purchase the Company Stock were or would be

allocated under the provisions of this Plan, pursuant to directions of the Plan Administrator.

- (iii) Any rights, warrants, or options on Company Stock which cannot be exercised for lack of cash may, as directed by the Plan Administrator, be sold by the Trustee and the proceeds allocated in accordance with the source of the Company Stock with respect to which the rights, warrants or options were issued.
- (f) Allocation of Cash Dividends and Other Distributions Received in the Trust Fund.
- (i) All cash dividends paid to the Trustee with respect to Company Stock that has been allocated to an Account (if any) as of the date the dividend is received by the Trustee shall be allocated to such Account. If the Company Stock in the Trust Fund is held in a Company Stock fund as an investment alternative pursuant to Section 4.6, such that Participants have an interest in such Company Stock only indirectly through an interest in such fund held in a subaccount, the cash dividends shall be allocated to such fund and shall thereafter be invested in accordance with the investment practices of such fund, and shall not be allocated directly to a Participants' Account or subaccount.
 - (ii) All cash dividends paid to the Trustee with respect to unallocated Company Stock shall be allocated as provided in Section 6.13.
- (iii) Other distributions received by the Trustee with respect to investments of the Trust shall be allocated to the applicable fund(s) established pursuant to Section 4.6, as prescribed by the Plan Administrator.
- **6.12** <u>Allocation of Forfeitures</u>. The treatment of all amounts that are forfeited by reason of the termination of the employment of a Participant shall be governed by the following rules, effective January 1, 1992:
 - (a) Forfeitures shall be used to offset the Company's obligation to make Company Contributions and shall be allocated to the Matching Contribution Accounts of those Participants who are entitled to receive an allocation of Matching Contributions for the Plan Year in the manner prescribed in Section 6.10(b).
 - (b) No forfeitures shall be allocated to any Voluntary Account, Rollover Account, Deferral Account or Alternate Payee Account.

- (c) The forfeitures to be allocated shall be the amount of forfeitures occurring since the next preceding allocation under this Section 6.12 and prior to the date prescribed by the Plan Administrator as the cutoff date for such allocation, which shall be a date falling between the Anniversary Date and the actual date in which the allocations are made. Such date may be changed from time to time by the Plan Administrator.
- (d) Pending allocation, forfeitures shall be accounted for in the same manner as unallocated Company Contributions and shall not be adjusted for income, gain or loss on such forfeitures. Such income, gain or loss shall be considered and accounted for in the same fashion as income, gain or loss on unallocated Company Contributions.
 - (e) For periods prior to January 1, 1991, forfeitures occurring during a Plan Year were allocated in the same manner as Profit Sharing Contributions.
- **6.13** Allocation of Income or Loss on Unallocated Company Contributions; Company Contributions to Defray Plan Expenses. At the time Company Profit Sharing Contributions are allocated to Company Contribution Accounts, the income, gain or loss on unallocated Company Contributions (determined as of the end of the Plan Year preceding the actual allocation date), adjusted for any Plan expenses paid or accrued as of the end of the Plan Year preceding the actual allocation date (except for Plan expenses paid through Company Contributions pursuant to the following sentence), shall be allocated to those Participants eligible to receive an allocation of Profit Sharing Contributions for such Plan Year, pro rata, according to each such Participant's entitlement to such allocation. The Company may, in its discretion, make a Plan contribution at any time for the purpose of defraying Plan expenses. Such contribution shall be used to defray Plan expenses and shall not be allocated to Accounts of Participants.
- **6.14** Accounting Procedures. The Plan Administrator shall establish accounting procedures for the purpose of making the allocations, valuations and adjustments to Accounts provided for in this Article VI, as well as the implementation of investment direction by Participants pursuant to Section 4.6 and transfers between or distributions from subaccounts established pursuant to Section 4.6(b). From time to time the Plan Administrator may modify such accounting procedures for the purpose of achieving equitable, nondiscriminatory, and administratively feasible allocations among the Accounts in accordance with the general concepts of the Plan and the provisions of this Article VI.

A Participant, Beneficiary or Alternate Payee shall have no contractual or other right to have a particular accounting procedure or convention apply, or continue to apply, and the Plan Administrator shall be free to alter any such procedure or convention without obligation to any Participant, Beneficiary or Alternate Payee, consistent with the requirements of Code section 411(d)(6).

- **6.15** Suspended Participants. The Accounts of each Suspended Participant shall be held intact and shall be valued on each Valuation Date as provided in Section 6.11, but shall not receive any allocation of Company Contributions; provided, however, that if the Participant completes, during the Plan Year in which he becomes a Suspended Participant, 1,000 or more Hours of Service during such Plan Year, his Company Contribution Account shall participate in the allocation of Profit Sharing Contributions for such Plan Year, if he otherwise would have been eligible for such an allocation.
- **6.16 Qualified Military Service.** Effective December 12, 1994, notwithstanding any provision of this Plan to the contrary, Company Contributions, benefits and service credit with respect to qualified military service will be provided in accordance with Section 414(u) of the Code.
- **6.17** <u>Accounting for Interest of an Alternate Payee</u>. In the event an Alternate Payee is awarded an interest in the Plan benefits of a Participant pursuant to a qualified domestic relations order, as defined in Section 14.2, such interest shall be separated into one or more separate Accounts and accounted for under rules prescribed by the Plan Administrator, pending distribution to the Alternate Payee.

ARTICLE VII

VESTING IN ACCOUNTS

7.1 No Vested Rights Except as Herein Specified. No Participant, Beneficiary or Alternate Payee shall have any vested right or interest to, or any right of payment of, any assets of the Trust Fund, except as provided in this Plan. Neither the making of any allocations nor the crediting of any amounts to the Account of a Participant, Beneficiary or Alternate Payee shall vest in any Participant, Beneficiary or Alternate Payee any right, title, or interest in or to any assets of the Trust Fund

7.2 <u>Participant's Vested Interest—General Rule</u>. Subject to the provisions of Section 7.3 and 7.6, the Vested Interest of each Participant or Suspended Participant in his Company Contribution Account shall be determined by multiplying the balance in each such account as of the applicable date by the Vested Percentage determined in accordance with the rules of Section 7.3 and the following schedule:

Years of Service	Vested Percentage
Less than three years	0
Three years but less than five years	50%
Five years or more	100%

- **7.3** <u>Vested Percentage—Special Rules</u>. The determination of a Participant's or Suspended Participant's Vested Percentage in his Company Contribution Account shall be subject to the following special rules:
 - (a) During an Employee's period of employment with the Company or an Affiliated Company (including periods while on an approved leave of absence or a Maternity or Paternity absence), in the event of his death, Disability, attainment of Normal Retirement Age, or a judicial declaration of his mental incompetence, the Employee's Vested Percentage shall become one hundred percent (100%), regardless of his number of Years of Service.
 - (b) For Plan Years beginning on and after January 1, 1992, the following rules shall apply:
 - (i) If a Participant terminates employment with the Company before becoming fully vested in his Accounts, and does not elect to receive a distribution, the nonvested portion of his Accounts shall be forfeited at the time the Participant incurs five consecutive Breaks in Service. If the Participant resumes employment with the Company prior to incurring five

consecutive Breaks in Service, his Vested Percentage shall be determined as if he had not terminated employment.

- (ii) If a Participant terminates employment with the Company before becoming fully vested in his Company Contribution Account, and voluntarily elects to receive a distribution at termination, the nonvested portion of such Participant's Company Contribution Account shall be forfeited at the time the Participant incurs a Break in Service.
- (iii) If the nonvested portion of a Participant's Company Contribution Account is forfeited as described in clause (ii), and the Participant returns to employment prior to incurring five consecutive Breaks in Service, the amount forfeited shall be restored to his Company Contribution Account if the Participant repays to the Plan the full amount of the distribution attributable to contributions made by the Company before the earlier of (A) five years after the first date on which the Participant is subsequently reemployed by the Company, or (B) the date on which the Participant incurs five consecutive Breaks in Service. Reinstatement shall be made on the last day of the Plan Year coinciding with or following the Employee's Reemployment Commencement Date. Restoration shall be made from available funds in the following order: first from forfeitures occurring on the last day of the Plan Year coinciding with or following the Employee's Reemployment Commencement Date, and second, from extraordinary contributions made by the Company.
- (iv) If a Participant terminates employment with the Company when the value of the Participant's Distributable Benefit is \$5,000 or less, the Participant shall receive such Distributable Benefit as soon as practicable following termination of employment. The nonvested portion of such Participant's Accounts shall be forfeited at the date the Participant receives distribution of his Distributable Benefit. If the Participant resumes employment with the Company prior to incurring five consecutive Breaks in Services, and repays the amount distributed, as described in clause (iii), the nonvested portion shall be restored as provided in clause (iii). If a Participant has no vested interest in his Accounts, the Participant shall be deemed to have received a distribution of his Distributable Benefit.
- (v) An Employee's aggregate Years of Service shall not include any Years of Service not required to be taken into account under this Section 7.3(b) by reason of any prior Break in Service.
- (c) An Employee's Vested Percentage as of January 1, 1992 shall be equal to his Vested Percentage determined as of

December 31, 1991, under the terms of the Amended and Restated Plan as in effect on September 30, 1989.

- (d) In the event of a divestiture of an operating group or division, the Plan Administrator or the Plan Administrator's designee may, in their sole discretion, determine, with respect to Eligible Employees whose employment with the Company terminates as a result of such divestiture and in lieu of the otherwise applicable determination of Vested Percentage specified in this Article VII, (1) treat the Eligible Employees' Vested Percentage as 100%, notwithstanding their Years of Service prior to termination; or (2) treat such Eligible Employees as Suspended Participants but credit Years of Service with the new employer to whom such group is divested for purposes of determining such Eligible Employees' Vested Percentage. Any such determination for a particular group or division shall not bind the Company in any way with respect to any subsequent determination relating to a different group or division. In the event of a subsequent divestiture from the new employer, the Plan Administrator or the Plan Administrator's designee may make a similar determination regarding vesting acceleration.
- **7.4** Nonforfeitable Accounts. The Vested Interest of each Participant in his Deferral Account, Voluntary Account and Rollover Contributions Account shall at all times be 100 percent.
- **7.5** Alternate Payee Accounts. In the event that an Alternate Payee is awarded an interest in the Accounts of a Participant whose Vested Percentage in any such Account is less than 100%, the Vested Percentage at any time of the Alternate Payee in that portion of the Alternate Payee Account attributable to such awarded interest shall be the same percentage as the Participant's Vested Percentage in his Account at that time, determined in accordance with Sections 7.1 through 7.3.

7.6 Participants Vested Interest.

- (a) Notwithstanding the above, effective January 1, 1997, the Vested Interest in the Company Contribution of each Participant actively employed on January 1, 1997 shall be at all times 100%.
- (b) Notwithstanding the above, the Vested Interest in the Company Contribution of each Participant who formerly participated in the NNS 401(k) Plan and had accounts of that plan transferred to the Plan shall be at all times 100%.

ARTICLE VIII

PAYMENT OF PLAN BENEFITS

8.1 Retirement.

- (a) A Participant who attains Normal Retirement Age shall be fully vested in his Accounts and shall be entitled to receive a distribution of his Distributable Benefit as of his Normal Retirement Date. The value of his Distributable Benefit shall be based on the value of his Account as of the Applicable Valuation Date.
- (b) If the Participant continues in the service of the Company beyond his Normal Retirement Date with the consent of the Company consistent with applicable legal requirements, he shall continue to participate in the Plan in the same manner as Participants who have not reached their Normal Retirement Dates. At the subsequent termination of the Participant's employment with the Company, his Distributable Benefit shall be based upon the value of his Accounts as of the Applicable Valuation Date determined with reference to his date of termination of employment as though that were his Normal Retirement Date.
- **8.2** Other Termination of Employment. A Participant who terminates employment with the Company for any other reason other than retirement on or after Normal Retirement Age, shall be entitled to receive a distribution of his Distributable Benefit as soon as reasonably practicable but in no event later than 120 days following the end of the Plan Year in which the Participant terminates his employment. A Participant shall be deemed to have terminated employment on the date the Plan Administrator determines the Participant has incurred a Disability. The value of his Distributable Benefit shall be based on the value of his Account as of the Applicable Valuation Date.

8.3 Withdrawals of Voluntary Contributions and from NNS Transfer Accounts; Hardship Withdrawals.

(a) A Participant (except a Participant who formerly participated in the NNS 401(k) Plan) shall be entitled to make a withdrawal from his Voluntary Account at any time, subject to procedures established by the Plan Administrator. A Participant who formerly participated in the NNS 401(k) Plan and had accounts in that plan transferred to the Plan, shall be entitled to make a withdrawal from his NNS Transfer After-Tax Account, NNS Transfer Employer Account, or NNS Transfer Rollover Account at any time, subject to procedures established by the Plan Administrator.

- (b) Further, a Participant (except a Participant who formerly participated in the NNS 401(k) Plan), may withdraw amounts from his Deferral Account that are attributable to Elective Deferrals, excluding all earnings attributable thereto, upon incurring a financial hardship as determined by the Plan Administrator. A Participant who formerly participated in the NNS 401(k) Plan shall be entitled to make withdrawals from his Deferral Account and his NNS Transfer Deferral Account that are attributable to Elective Deferrals, excluding all earnings attributable thereto, upon incurring a financial hardship as determined by the Plan Administrator. A withdrawal will be deemed made on account of financial hardship only if the distribution is made on account of an immediate and heavy financial need and is necessary to satisfy such financial need. Any withdrawal made pursuant to this paragraph shall be made in accordance with rules prescribed by the Plan Administrator that are consistent with regulations under Code section 401(k), and the following rules:
 - (i) The determination of whether an Employee has an immediate and heavy financial need is to be made by the Plan Administrator on the basis of all relevant facts and circumstances. A distribution will be deemed to be made on account of an immediate and heavy financial need of the Employee if the distribution is on account of:
 - (A) Medical expenses of the Employee or his dependent(s) not covered by insurance;
 - (B) Purchase of the principal residence for the Employee;
 - (C) Payment of tuition for the next twelve months of post-secondary education for the Employee or his dependent(s);
 - (D) The need to prevent the eviction from the Employee's principal residence or foreclosure on the mortgage of the Employee's principal residence; or
 - (E) Other financial circumstances as the Plan Administrator may determine, consistent with applicable regulations and rulings of the Treasury Department and the Internal Revenue Service.

- (ii) To receive a hardship withdrawal, the following requirements must first be met by the applicant:
 - (A) The applicant must sign the forms provided by the Plan Administrator and certify all information requested on that form;
 - (B) The applicant must have withdrawn, or must withdraw at the same time that an application for hardship withdrawal is submitted, all voluntary account balances in other Company retirement plans, if any;
 - (C) The applicant must have applied for all eligible loans from this Plan or from any other Company retirement plans which permit Participant loans;
 - (D) The hardship withdrawal normally cannot be less than \$500.00;
 - (E) The applicant must suspend all Elective Deferrals for twelve (12) months after receiving the hardship withdrawal pursuant to rules prescribed by the Plan Administrator; and
 - (F) The maximum amount of Elective Deferrals for the calendar year following the year of the hardship withdrawal will be the maximum amount permitted by Code section 402(g) for such calendar year less the amount of any Elective Deferrals made by the Participant during the calendar year in which such Hardship withdrawal is made.
- (iii) In the event that a Participant has elected to have his Deferral Account invested in more than one investment alternative, then the withdrawal shall be made pro rata from the subaccounts in which his Accounts are invested.
- (iv) The timing of the payment of the withdrawal shall be made as soon as practicable following the request in accordance with the rules established by the Plan Administrator.
- (c) Upon fifteen (15) days' written notice to the Plan Administrator, and not more than once each Plan Year, a Participant may withdraw up to one hundred percent (100%) of the balance in his Voluntary Account as soon as administratively feasible following submission of the withdrawal request.
 - (d) Any withdrawal made under this Section 8.3 shall be subject to the restrictions on distributions set forth in Sections 8.6 through 8.8.
- **8.4** Withdrawals After Age 59 ½. On or after a Participant reaches the age of 59 ½, he may make unlimited withdrawals of any amount from any or all of his Accounts as set forth in Section 2.1.

8.5 Distributable Benefit.

- (a) A Participant's "Distributable Benefit" shall be his entire nonforfeitable interest in his Accounts. This amount shall be distributable in the forms described in Section 8.6, subject to the restrictions on distributions set forth in Sections 8.7 and 8.8.
- (b) For purposes of determining the amount of Distributable Benefit that will be distributed to a Participant or Beneficiary pursuant to the rules of this Article VIII, the value of the Participant's Accounts shall be determined in accordance with rules prescribed by the Plan Administrator. However, the value of the Participant's Accounts shall be increased or decreased (as appropriate) by any contributions or distributions properly allocable under the terms of this Plan to his Accounts that occurred on or after the Applicable Valuation Date or for any other reason were not otherwise properly reflected in the valuation of his Account on such Valuation Date.
- (c) Neither the Plan Administrator, the Company, nor the Trustee shall have any responsibility for any increase or decrease in the value of a Participant's Account as a result of any valuation made under the terms of this Plan after the date of his termination of employment and before the date of the distribution of his Distributable Benefit to him or his Beneficiary. Also, neither the Plan Administrator, the Company, nor the Trustee shall have any responsibility for failing to make any interim valuation of a Participant's Accounts between the date of distribution to the Participant of his Distributable Benefit and the immediately preceding Valuation Date, even though the Plan assets may have been revalued in that interim for a purpose other than to revalue the Accounts under this Plan.

8.6 Optional Forms of Payment.

- (a) Each Participant may elect, with his spouse's consent if required and subject to (b) below, to receive his Distributable Benefit in one of the following forms, or a combination of such forms, subject to the provisions of Sections 8.7 and 8.8:
 - (i) A single sum.
 - (ii) 10 or 15 year certain and life annuity.

- (iii) Contingent annuity 10 year certain and life annuity.
- (iv) Flexible installment refund annuity.
- (v) Installment refund annuity.
- (b) Any distribution not made in a single sum shall be made over a term that does not exceed any one (or a combination) of the following periods:
 - (i) A period certain not extending beyond the life expectancy of the Participant.
 - (ii) A period certain not extending beyond the joint life and last survivor expectancy of the Participant and a designated Beneficiary.
 - (iii) The Participant's life.
 - (iv) The lives of the Participant and the Participant's designated Beneficiary.
- (c) With respect to any Participant whose Plan Accounts contain no allocations made prior to January 1, 1992, the only available benefit option is a single sum distribution.

8.7 Direct Rollover.

- (a) This Section 8.7 applies to distributions made on or after January 1, 1993. Notwithstanding any provision of the Plan to the contrary that would otherwise limit a "distributee's" election under this Section 8.7, a distributee may elect, at the time and in the manner prescribed by the Plan Administrator, to have any portion of an "eligible rollover distribution" paid directly to a "eligible retirement plan" specified by the distributee in a direct rollover.
- (b) "Eligible rollover distribution" means any distribution of all or any portion of the balance of the credit of the distributee, except that an eligible rollover distribution does not include: any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the distributee or the joint lives (or joint life expectancies) of the distributee and the distributee's designated beneficiary, or for a specified period of ten years or more; any distribution to the extent such distribution is required under section 401(a)(9) of the Code; any hardship distribution made under Code section 401(k)(2)(B)(i)(IV) for distributions after December 31, 1998;

and the portion of any distribution that is not includable in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities).

- (c) "Eligible retirement plan" means an individual retirement account described in section 408(a) of the Code, an individual retirement annuity described in section 408(b) of the Code, an annuity plan described in section 403(a) of the Code, or a qualified trust described in section 401(a) of the Code, that accepts the distributee's eligible rollover distribution. However, in the case of an eligible rollover distribution to the surviving spouse, an eligible retirement plan is an individual retirement account or individual retirement annuity.
- (d) A "distributee" includes an employee or former employee. In addition, the employee's or former employee's surviving spouse and the employee's or former employee's spouse or former spouse who is an Alternate Payee are distributees with regard to the interest of the spouse or former spouse.
 - (e) A "direct rollover" is a payment by the Plan to the eligible retirement plan specified by the distributee.
- **8.8** <u>Distribution Restrictions on Plan Account and Matching Account</u>. Notwithstanding the prior provisions of this Article VIII to the contrary, a Participant's Plan Account and Matching Account Balances shall not be distributable earlier than:
 - (a) Upon termination of the Participant's employment with the Company on or after the date he or she attains Normal Retirement Date or by reason of death;
 - (b) Upon a determination by the Committee that the Participant has become Disabled;
 - (c) Upon termination of the Participant's employment with the Company and Affiliated Companies for any reason other than those specified in paragraph (a) above, but only if such termination constitutes a separation from service (within the meaning of Code section 401(k)(2)(B)(i)(I);
 - (d) Upon the Committee's approval of the Participant's application for a hardship or in-service withdrawal, but only to the extent provided in Section 8.7;
 - (e) In accordance with and to the limited extent provided in Sections 5.2(d) and 5.6(a);

- (f) If and to the extent permitted by Code section 401(k)(10) in connection with the Company's disposition of corporate assets or a subsidiary; and
- (g) Upon the creation or recognition of an Alternate Payee's right to all or a portion of a Participant's Plan and Matching Contribution Accounts under a domestic relations order which the Committee determines is a qualified domestic relations order as provided in Section 14.2, but only as to the portion which is payable to the Alternate Payee.
- **8.9** <u>Distribution Rules</u>. The provisions of Section 8.8(a) shall apply to the Distributable Benefit of a Participant who does not elect, and has not previously elected, a life annuity under the Plan. The provisions of Section 8.8(b) shall apply to the Distributable Benefit of a Participant who elects, or has previously elected, a life annuity under the Plan.
 - (a)(i) Upon termination of employment with the Company for any reason other than death, a Participant who does not elect to receive a life annuity under the Plan shall be entitled to a single sum payment of his entire Distributable Benefit. Payment shall be made in the form and at the time determined under Sections 8.6 and 8.7.
 - (ii) If the Participant has a Distributable Benefit remaining at his death, and has not elected a life annuity under the Plan, the Participant's benefit shall be paid to the Participant's surviving spouse. If there is no surviving spouse, or if the surviving spouse has consented in a manner conforming to the requirements of a Qualified Election as described below to a different Beneficiary designation, the Participant's Distributable Benefit shall be paid to the Participant's designated Beneficiary. The Participant's spouse may waive the spousal death benefit described above at any time, but the waiver shall not be effective unless it satisfies the requirements for a Qualified Election with respect to a Qualified Preretirement Survivor Annuity. The Plan Administrator shall direct the Trustee to pay out the Participant's benefits in the form elected by the Beneficiary subject to the provisions of Sections 8.6 and 8.7.
 - (b)(i) If a married Participant who has elected to receive a life annuity under the Plan dies prior to his Annuity Starting Date, is survived by a spouse and has not made a Qualified Election within the Election Period, the Participant's Distributable Benefit shall be used to provide a Qualified Preretirement Survivor Annuity providing for payments over the

life of the Participant's surviving spouse. The surviving spouse may elect to have distributions commence within a reasonable period after the Participant's death or to defer distribution as provided in Section 8.6.

- (ii) If a married Participant is alive on his Annuity Starting Date and does not make a Qualified Election within the Election Period, the Participant's Distributable Benefit shall be used to provide a Qualified Joint and Survivor Annuity providing payments over the joint life of the Participant and the Participant's spouse and the life of the survivor. The Participant shall designate on the form provided by the Plan Administrator the amount of the survivor annuity expressed as a percentage of the amount payable during the joint life of the Participant and his spouse. The Participant may elect to have such annuity distributed upon attainment of the earliest date on which the Participant may elect to receive retirement benefits.
- (iii) If the Participant is unmarried on his Annuity Starting Date, or has certified that his spouse cannot be located, and does not make a Qualified Election within the Election Period, the Participant's Distributable Benefit shall be used to provide a single life annuity providing payments over the life of the Participant.
- (iv) Within the Applicable Notice Period, the Plan Administrator shall provide each Participant with a written explanation of the terms and conditions of the Qualified Preretirement Survivor Annuity and the Qualified Joint and Survivor Annuity; the Participant's right to make, and the effect of, an election to waive such annuities; the requirement for spousal consent to the election to waive such annuities; the material features and optional forms of benefit; and the right to make, and the effect of, a revocation of an election to waive such annuities.
- (v) If the Participant makes a Qualified Election within the Election Period, the Participant, or the Participant's designated Beneficiary in the event of the Participant's death, may elect to receive benefits in one of the forms described in Section 8.6.
- (c) The following definitions shall apply with respect to this Section 8.9 in addition to the definitions in Article II of the Plan:
 - (i) "Annuity Starting Date" means the first day of the first period for which an amount is payable as an annuity or in any other form.

(ii) "Applicable Notice Period" means:

- (A) For a Qualified Preretirement Survivor Annuity, whichever of the following periods ends last:
- (1) The period beginning on the first day of the Plan Year in which the Employee attains age thirty-two (32) and ending with the close of the Plan Year preceding the Plan Year in which the Employee attains age thirty-five (35).
 - (2) A reasonable period ending after the individual becomes a Participant.
- (3) Notwithstanding clause (i), a reasonable period ending after the date the Participant terminates employment with the Company, if the Participant terminates employment before attaining age thirty-five (35).
- (B) For a Qualified Joint and Survivor Annuity, a reasonable period which shall not be less than thirty (30) days nor more than ninety (90) days before the Participant's Annuity Starting Date.
- (C) For purposes of clause (ii), a "reasonable period" shall be the end of the two-year period beginning one year before the described event occurs and ending one year after that date. For purposes of clause (iii), a "reasonable period" shall include the two-year period beginning one year before the Participant terminates employment with the Participating Employer and ending one year after such termination. If the Participant returns to employment with the Participating Employer, the Applicable Notice Period shall be redetermined.

(iii) "Election Period" means:

(A) For a Qualified Preretirement Survivor Annuity, the period which begins on the first day of the Plan Year in which the Participant attains age thirty-five (35) and ends on the date of the Participant's death. If a Participant terminates employment with the Company before the first day of such period, the Election Period shall begin on the date on which the Participant terminates employment with the Company with respect to benefits previously accrued. A Participant who will not yet attain age thirty-five (35) as of the end the current Plan Year may make a special Qualified Election to waive the Qualified Preretirement Survivor Annuity for the period beginning on the date of such election and ending on the first day of the Plan Year in which the Participant will

attain age thirty-five (35). Such election shall not be valid unless the Participant receives a written explanation of the Qualified Preretirement Survivor Annuity in terms similar to those described in subsection (c) above. Qualified Preretirement Survivor Annuity coverage shall be reinstated as of the first day of the Plan Year in which the Participant attains age thirty-five (35). Any waiver after such date must meet the requirements of a Qualified Election in order to be valid.

- (B) For a Qualified Joint and Survivor Annuity, the ninety-day period ending on the Participant's Annuity Starting Date.
- (iv) "Qualified Election" means an election by a Participant to waive a Qualified Joint and Survivor Annuity or Qualified Preretirement Survivor Annuity which meets the following requirements:
 - (A) The Participant's spouse consents in writing to the election.
 - (B) The election designates a specific Beneficiary, including any class of Beneficiaries or contingent Beneficiaries, which may not be changed without spousal consent, or the spouse expressly permits designations by the Participant without further spousal consent.
 - (C) The election designates a form of benefit payment that may not be changed without spousal consent (or the spouse expressly permits designation by the Participant without any further spousal consent). If it is established to the satisfaction of a Plan representative that there is no spouse, or that the spouse cannot be located, a waiver shall be deemed a Qualified Election.
 - (D) The spouse's consent acknowledges the effect of the election.
 - (E) The spouse's consent is witnessed by a Plan representative or a notary public.
 - (F) The Participant has received notice as described in clause (iv) within the Applicable Notice Period. Any consent made by a spouse to a Participant's election to waive an annuity shall be effective only with respect to such spouse. A consent which permits designation by the Participant without any requirement of further consent by such spouse shall acknowledge that the spouse has the right to limit consent to a specific Beneficiary, and a specific form of benefit where applicable, and that the spouse voluntarily elects to relinquish

- either or both such rights. A revocation of a prior waiver may be made by a Participant without the consent of the spouse at any time before the commencement of benefits. The number of revocations shall not be limited.
- (v) "Qualified Joint and Survivor Annuity" means an immediate annuity for the life of the Participant with a survivor annuity for the life of the spouse which is not less than 50 percent and not more than 100 percent of the amount of the annuity that is payable during the joint life of the Participant and the spouse. The percentage of the survivor annuity shall be 50 percent unless the Participant designates a different percentage.
 - (vi) "Qualified Preretirement Survivor Annuity" means an immediate annuity for the life of the surviving spouse of the Participant.

8.10 <u>Designation of Beneficiary</u>.

- (a) Subject to the provisions of Section 8.10(e), each Participant shall have the right to designate a Beneficiary or Beneficiaries to receive his Distributable Benefit. This designation is to be made on the form prescribed by and delivered to the Plan Administrator or its delegate.
- (b) Subject to the provisions of Section 8.10(e), a Participant shall have the right to change or revoke any such designation by filing a new designation or notice of revocation with the Plan Administrator. Subject to the provisions of Section 8.10(e), no notice to any Beneficiary nor consent by any Beneficiary shall be required to effect any such change or revocation.
- (c) If a deceased Participant shall have failed to designate a Beneficiary, if the Plan Administrator shall be unable to locate a designated Beneficiary after reasonable efforts have been made, if for any reason the designation shall be legally ineffective, or if the Beneficiary shall have predeceased the Participant (and no legally effective contingent Beneficiary shall have been named), any distribution required to be made under the provisions of this Plan shall be payable to the Participant's estate (except as provided in Section 8.10(e)), and the estate shall be considered the Beneficiary under this Plan.
- (d) In the event that the deceased Participant was not a resident of Virginia at the date of his death, the Plan Administrator, in its discretion, may require the establishment of ancillary administration in Virginia. In the event that a Participant shall predecease his Beneficiary and on the

subsequent death of the Beneficiary a remaining distribution is payable under the applicable provisions of this Plan, the distribution shall be payable to the Beneficiary's estate.

- (e) If a Participant shall be married at the time of his death, the designation by the Participant under Section 8.10 of a person other than the current spouse as his Beneficiary shall not take effect (and the entire Distributable Benefit shall be paid to such spouse) (i) unless the spouse of the Participant consents in writing to such designation, and the spouse's consent acknowledges the effect of such designation and is witnessed by the Plan Administrator (or its delegate), a notary, or (ii) unless it is established to the satisfaction of the Plan Administrator that such consent is not required because there is no spouse, because the spouse cannot be located, or because of such other circumstances as the Secretary of the Treasury may by regulations prescribe.
- (f) The Company, the Plan Administrator and the Trustee shall have no duty to determine whether a Beneficiary designation or spousal consent made pursuant to this Section 8.10 was an informed designation or consent or was freely given, and shall be entitled to rely upon the Beneficiary form filed with the Plan Administrator, as well as such other documents as may be required pursuant to Section 8.13, and shall be under no duty or obligation to protect the rights of a spouse or former spouse of a Participant, except as may be required by law.
- **8.11** Facility of Payment. If any payee under the Plan is a minor or if the Plan Administrator reasonably believes that any payee is legally incapable of giving a valid receipt and discharge for any payment due him, the Plan Administrator may have the payment, or any part thereof, made to the person (or persons or institution) whom it reasonably believes is caring for or supporting the payee, unless it has received due notice of claim therefor from a duly appointed guardian or committee of the payee. Any payment shall be a payment from the Accounts of the payee and shall, to the extent thereof, be a complete discharge of any liability under the Plan to the payee.
- **8.12** <u>Distribution to Alternate Payees</u>. If an Alternate Payee is entitled to a distribution of benefits from this Plan pursuant to a qualified domestic relations order, as defined in Section 14.2, the benefits payable to such Alternate Payee shall be distributed pursuant to such qualified domestic relations order under rules or procedures described by the Plan Administrator. If permitted by applicable law and regulations, the Plan Administrator may require or permit immediate distribution of such benefits to an Alternate Payee at any time following the determination by the Plan Administrator that such

an order is a qualified domestic relations order. In the event that an Alternate Payee dies prior to receipt of the amounts due him from an Alternate Payee Account, such amounts shall be distributed to the estate of the Alternate Payee as soon as practicable following the date such amounts would have been distributed to such Alternate Payee.

8.13 Additional Documents.

- (a) The Plan Administrator or Trustee, or both, may require (and rely upon) the execution and delivery of such documents, papers and receipts as the Plan Administrator or Trustee may determine necessary or appropriate in order to establish the fact of death of the deceased Participant and of the right and identity of any Beneficiary or other person or persons claiming any benefits under this Article VIII.
- (b) The Plan Administrator or the Trustee, or both, may, as a condition precedent to the payment of death benefits hereunder, require an inheritance tax release and/or such security as the Plan Administrator or Trustee, or both, may deem appropriate as protection against possible liability for state or federal death taxes attributable to any death benefits.

8.14 Transfer of Distributed Company Stock—Right of First Refusal for SAIC and the Plan.

- (a) Any Company Stock distributed from the Plan which is Class B Common stock shall be subject to a right of first refusal by Science Applications International Corporation ("SAIC"). The terms and conditions of the right of first refusal shall be those applied to Class A Common stock by the SAIC's certificate of incorporation, as in effect from time to time.
- (b) Any Company Stock distributed from the Plan, in addition to the right of first refusal imposed by the SAIC's certificate of incorporation, shall also be subject to a right of first refusal in favor of the Plan, subordinate to that of SAIC, the terms and conditions of which (except for the right being in favor of the Plan) shall be the same as the SAIC right of first refusal, modified as necessary to comply with applicable Treasury regulations.
- (c) In administering the rights of first refusal with respect to such Company Stock, the rights shall be modified, as necessary, to comply with applicable Treasury regulations.

ARTICLE IX

OPERATION AND ADMINISTRATION OF THE PLAN; VOTING AND OTHER RIGHTS OF COMPANY STOCK

9.1 Plan Administration.

- (a) The Plan Administrator has authority to control and manage the operation and administration of the Plan.
- (b) For purposes of ERISA section 402(a), the Plan Administrator, the Trustee and any Investment Manager appointed pursuant to Section 9.3 shall be Named Fiduciaries of this Plan.
- (c) The Plan Administrator shall cause to be maintained in its office for the purpose of inspection an accurate schedule listing the names of all persons from time to time serving as members of any committee to which the Plan Administrator delegates its duties and all Named Fiduciaries of the Plan.
- **9.2** <u>Plan Administrator Powers</u>. The Plan Administrator shall have all powers necessary to supervise the administration of the Plan and control its operations. In addition to any powers and authority conferred on the Plan Administrator elsewhere in the Plan or by law, the Plan Administrator shall have, by way of illustration but not by way of limitation, the following powers and authority:
 - (a) To allocate fiduciary responsibilities (other than trustee responsibilities) among the Named Fiduciaries and to designate one or more other persons to carry out fiduciary responsibilities (other than trustee responsibilities). However, no allocation or delegation under this Section 9.2(a) shall be effective until the person or persons to whom the responsibilities have been allocated or delegated agree to assume the responsibilities. The term "trustee responsibilities" as used herein shall have the meaning set forth in ERISA section 405(c). The preceding provisions of this Section 9.2(a) shall not limit the authority of the Plan Administrator to appoint one or more Investment Managers in accordance with Section 9.3:
 - (b) To designate agents to carry out responsibilities relating to the Plan, other than fiduciary responsibilities.
 - (c) To employ such legal, actuarial, medical, accounting, clerical, administrative and ministerial and other assistance as it may deem appropriate in carrying out the provisions of this Plan, including one or more persons to render

advice with regard to any responsibility any Named Fiduciary or any other fiduciary may have under the Plan.

- (d) To establish rules and regulations from time to time for the conduct of the Plan Administrator's business and the administration and effectuation of this Plan.
- (e) To administer, interpret, construe and apply this Plan and the Plan's claims procedure and to decide all questions which may arise or which may be raised under this Plan by any employee, Participant, former Participant, Beneficiary, Alternate Payee or other person whatsoever, including but not limited to all questions relating to eligibility to participate in the Plan, the amount of service of any Participant, and the amount of benefits to which any Participant or his Beneficiary may be entitled on or after the Effective Date hereof.
 - (f) To determine the manner in which the assets of this Plan, or any part thereof, shall be distributed.
- (g) To direct the Trustee, in writing, from time to time, to invest and reinvest the Trust Fund, or any part thereof, or to purchase, exchange, or lease any property, real or personal, which the Plan Administrator may designate. This shall include the right to direct the investment of all or any part of the Trust in any one security or any one type of securities permitted hereunder. Among the securities which the Plan Administrator may direct the Trustee to purchase are "employer securities" as defined in Code section 409(1) or any successor statute thereto.
- (h) To select alternative investment options from which Participants may select from in determining investment of their Accounts, and to establish rules and procedures regarding such investment options.
 - (i) To satisfy accounting, auditing, record keeping, insurance, bonding and reporting and disclosure requirements.
- (j) To perform or cause to be performed such further acts as it may deem to be necessary, appropriate or convenient in the efficient administration of the Plan.

Any action taken in good faith by the Plan Administrator in the exercise of authority conferred upon it by this Plan shall be conclusive and binding upon the Participants and their Beneficiaries and any Alternate Payees. All discretionary powers conferred upon the Plan Administrator shall be absolute, but shall be exercised in a uniform and nondiscriminatory manner.

9.3 Investment Manager.

- (a) The Plan Administrator, by action reflected in the minutes thereof, may appoint one or more Investment Managers, as defined in ERISA section 3(38), to manage all or a portion of the assets of the Plan.
- (b) An Investment Manager shall discharge its duties in accordance with applicable law and in particular in accordance with ERISA section 404(a) (1).
- (c) An Investment Manager, when appointed, shall have full power to manage the assets of the Plan for which it has responsibility, and neither the Company nor the Plan Administrator shall thereafter have any responsibility for the management of those assets, except to the extent such power or responsibility shall have been reserved to the Company or Plan Administrator in the documents governing the relationship between or among the Plan, the Company and the Investment Manager.

9.4 Periodic Review.

- (a) At periodic intervals, not less frequently than annually, the Plan Administrator shall review the long-run and short-run financial needs of the Plan and shall determine a funding policy for the Plan consistent with the objectives of the Plan and the minimum funding standards of ERISA, if applicable. In determining the funding policy, the Plan Administrator shall take into account, at a minimum, not only the long-term investment objectives of the Trust Fund consistent with the prudent management of the assets thereof, but also the short-run needs of the Plan to pay benefits.
- (b) All actions taken by the Plan Administrator with respect to the funding policy of the Plan, including the reasons therefor, shall be fully reflected in the minutes of the Plan Administrator's actions.

9.5 Reliance Upon Documents and Opinions.

(a) The Plan Administrator, the Board of Directors, the Company and any Employee of the Company delegated under the provisions hereof to carry out any fiduciary responsible under the Plan ("Delegated Fiduciary"), shall be entitled to rely upon any tables, valuations, computations, estimates, certificates and reports furnished by any consultant, or firm or corporation which employs one or more consultants, upon any opinions furnished by legal counsel, and upon any reports furnished by the Trustee. The Plan Administrator, the Board of Directors, the Company and

any Delegated Fiduciary shall not be liable in any manner whatsoever for anything done or action taken or suffered in reliance from any such consultant or firm or corporation which employs one or more consultants, trustee, or counsel.

- (b) Any and all such things done or actions taken or suffered by the Plan Administrator, the Board of Directors, the Company and any Delegated Fiduciary shall be conclusive and binding on all Employees, Participants, Beneficiaries, Alternate Payees and any other persons whomsoever, except as otherwise provided by law.
- (c) The Plan Administrator and any Delegated Fiduciary may, but are not required to, rely upon all records of the Company with respect to any matter or thing whatsoever, and may likewise treat those records as conclusive with respect to all Employees, Participants, Beneficiaries, and any other persons whomsoever, except as otherwise provided by law.
- **9.6** Requirement of Proof. The Plan Administrator or the Company may, in its (or their) sole discretion, require satisfactory proof of any matter under this Plan from or with respect to any Employee, Participant, Beneficiary or Alternate Payee, and no benefits under this Plan need be paid until the required proof shall be furnished.
 - 9.7 Multiple Fiduciary Capacity. Any person or group of persons may serve in more than one fiduciary capacity with respect to the Plan.

9.8 Limitation on Liability.

- (a) Except as provided in Part 4 of Title I of ERISA, neither the Company, the Board of Directors (or any member thereof), nor the Plan Administrator shall be subject to any liability with respect to his duties under the Plan unless he or it acts fraudulently or in bad faith.
- (b) Neither the Company, the Board of Directors (or any member thereof) nor the Plan Administrator shall be liable for any breach of fiduciary responsibility resulting from the act or omission of any other fiduciary or any person to whom fiduciary responsibilities have been allocated or delegated, except as provided in Part 4 of Title I of ERISA.
- (c) Neither the Company, the Board of Directors (or any member thereof), the Plan Administrator, nor the Trustee shall be liable to the extent relief from liability is provided pursuant to ERISA section 404(c).

(d) The Company in this Plan document does not intend to create additional fiduciary liability, or to characterize actions or responsibilities as fiduciary in nature, beyond that required by ERISA or other applicable law.

9.9 Indemnification.

- (a) To the extent permitted by law, the Company hereby indemnifies each member of the Board of Directors, and any Employee of the Company with duties under the Plan, against expenses (including any amount paid in settlement) reasonably incurred by him in connection with any claims against him by reason of his conduct in the performance of his duties under the Plan, except in relation to matters as to which he acted fraudulently or in bad faith in the performance of such duties. The preceding right of indemnification shall pass to the estate of such a person.
- (b) The preceding right of indemnification shall be in addition to any other right to which the Board of Directors or other person may be entitled as a matter of law or otherwise.

9.10 Allocation of Fiduciary Responsibility.

- (a) ERISA section 405(c) permits the division, allocation and delegation among Plan fiduciaries of the fiduciary responsibilities owed to the Plan Participants and Beneficiaries. Under this concept, each fiduciary, including a Named Fiduciary, is accountable only for its own functions, except to the extent of his co-fiduciary liability under ERISA section 405. It is the intent of the Company in establishing this Plan to comply with Section 405(c) and to have the limitation on liability set forth in ERISA section 405(c)(2) apply to the maximum extent allowed by law.
- (b) Pursuant to ERISA section 405(c), the authority to control and manage the operation and administration of the Plan is allocated to the Plan Administrator. Except to the extent expressly provided to the contrary in this Plan document, and the Trust Agreement, the responsibilities allocated to the Plan Administrator include:
 - (i) Responsibilities identified as Plan Administrator authority and powers in Sections 9.2(a) (j); and
 - (ii) Responsibilities identified elsewhere in this Plan document as applicable to the Plan Administrator.

- (c) The Board of Directors is allocated the following responsibilities, acting with the advice and assistance of the Plan Administrator:
 - (i) Appointing the Trustee;
 - (ii) Adopting Plan amendments;
 - (iii) Determining the amount of Company Contributions;
 - (iv) Determining whether to terminate the Plan or suspend contributions thereto;
- (v) Determining which Affiliated Companies shall participate in the Plan, and the conditions on which any such Affiliated Company shall participate;
 - (vi) Determining the form of Company Contributions; and
 - (vii) Performing those duties specifically allocated to it elsewhere in this Plan document.
- (d) The Trustee shall have only those responsibilities which have been specifically allocated to it under this Plan document and related Trust Agreement, plus any "trustee responsibilities," under ERISA section 405(c), which may not legally be allocated to another person or fiduciary. Any Investment Manager appointed pursuant to Section 9.3 may be granted exclusive authority and discretion to manage and control all or any portion of the assets of the Plan, subject to such limitations as may be provided in the documents governing the relationship between or among the Plan, the Company (if applicable) and the Trustee or Investment Manager.

9.11 Prohibition Against Certain Actions.

- (a) To the extent prohibited by law, in administering this Plan the Plan Administrator shall not discriminate in favor of any class of Employees and particularly it shall not discriminate in favor of Highly Compensated Employees.
- (b) The Plan Administrator shall not cause the Plan to engage in any transaction that constitutes a nonexempt prohibited transaction under Code section 4975(c) or ERISA section 406(a).
 - (c) All individuals who are fiduciaries with respect to the Plan (as defined in ERISA section 3(21)) shall discharge

their fiduciary duties in accordance with applicable law, and in particular, in accordance with the standards of conduct contained in ERISA section 404.

9.12 Bonding and Insurance.

- (a) Except as provided in ERISA section 412, as may be required under any other applicable law, or as may be required by the Plan Administrator in its sole discretion, no bond or other security shall be required by any member of any fiduciary under this Plan.
- (b) For purposes of satisfying its indemnity obligations under Section 9.9, the Company may (but need not) purchase and pay premiums for one or more policies of insurance. However, this insurance shall not release the Company of its liability under the indemnification provisions.

9.13 Voting and Other Rights of Company Stock.

- (a) All voting rights of Company Stock held in the Trust Fund shall be exercised in accordance with the following provisions:
- (i) Each Participant (which term shall include, for purposes of this Section 9.13, Beneficiaries and Alternate Payees having an interest in an Account or fund holding Company Stock) shall be given the opportunity to instruct the Trustee confidentially on a form prescribed and provided by the Company as to how to vote those shares (including fractional shares) of Company Stock allocated to his Account(s) under the Plan (directly or indirectly through an interest in a Company Stock fund) on the date immediately preceding the record date for the meeting of shareholders of the Company. The Trustee shall not divulge to the Company the instructions of any Qualified Holder. The Company may require verification of the Trustee's compliance with such confidential voting instructions by an independent auditor elected by the Company.
- (ii) All Participants entitled to direct such voting shall be notified by the Plan Administrator (or the Company, pursuant to its normal communications with shareholders) of each occasion for the exercise of these voting rights within a reasonable time (but not less than the time period that may be required by any applicable state of federal law) before these rights are to be exercised. The notification shall include all information distributed by the Company to other shareholders regarding the exercise of such rights.

- (iii) The Participants shall be so entitled to direct the voting of fractional shares (or fractional rights to shares). However, the Plan Administrator may, to the extent possible, direct the Trustee to vote the combined fractional shares (or fractional rights to shares) so as to reflect the aggregate direction of all Qualified Holders giving directions with respect to fractional shares (or fractional rights to shares).
- (iv) In the event that a Participant shall fail to direct the Trustee, in whole or in part, as to the exercise of voting rights arising under any Company Stock allocated to his Account, then these voting rights, together with voting rights as to shares of Company Stock which have not been allocated, shall be exercised by the Trustee in the same proportion as the number of shares of Company Stock for which the Trustee has received direction in such matter (e.g., to vote for, against or abstain from voting on a proposal, or to grant or withhold authority to vote for a director or directors), and the Trustee shall have no discretion in such matter.
- (v) Except as provided in paragraph (b) below, all rights (other than voting rights) of Company Stock held in the Trust Fund shall be exercised in the same manner and to the same extent as provided above with respect to the voting rights of the Company Stock, subject to the rules prescribed by the Plan Administrator, which rules, among other matters, may prescribe that no action shall be taken with respect to shares as to which no direction is received from Qualified Holders. The Trustee shall have no discretion with respect to the exercise of any such rights.
- (vi) Neither the Plan Administrator nor the Trustee shall make any recommendation to any Qualified Holder regarding the exercise of the Participant's voting rights or any other rights under the provisions of this Section 9.13, nor shall the Plan Administrator or Trustee make any recommendation as to whether any such rights should or should not be exercised by the Participant.
- (b) All responses to tender and exchange for Company Stock offers shall be made in accordance with the following provisions:
- (i) Each Participant shall be given the opportunity, to the extent that shares of Company Stock are allocated to his Account, to direct the Trustee in writing as to the manner in which to respond to a tender or exchange offer with respect to Company Stock, and the Trustee shall respond in accordance with the instructions so received. The Trustee shall

not divulge to the Company the instructions of any Participant. The Plan Administrator shall utilize its best efforts to timely distribute or cause to be distributed to each Participant such information as will be distributed to shareholders of the Company in connection with any such tender or exchange offer, together with a form addressed to the Trustee requesting confidential instructions on whether or not such shares will be tendered or exchanged. If the Trustee shall not receive timely direction from a Participant as to the manner in which to respond to such a tender or exchange offer, the Trustee shall not tender or exchange any shares of Company Stock with respect to which such Participant has the right of direction, and the Trustee shall have no discretion in such matter.

(ii) Unallocated shares of Company Stock and shares of Company Stock held by the Trustee pending allocation to Participants' Accounts shall be tendered or exchanged (or not tendered or exchanged) by the Trustee in the same proportion as shares with respect to which Participants have been given the opportunity to direct the Trustee pursuant to paragraph (i) above are tendered or exchanged, and the Trustee shall have no discretion in such matter.

9.14 Plan Expenses.

- (a) Except as provided in Section 9.14(b), all expenses incurred in the establishment, administration and operation of the Plan, to the extent these expenses are not paid by the Company, shall be charged to the Trust Fund and accounted for pursuant to the provisions of Article VI.
- (b) Costs or expenses which are particular to a specific asset or group of assets in the Trust Fund, such as interest and brokerage charges which are included in the cost of securities purchased by the Trustee (or charged to proceeds in the case of sales), as determined by the Plan Administrator, shall be charged or allocated in a fair and equitable manner to the Accounts, subaccounts or funds to which those assets are allocated pursuant to rules prescribed by the Plan Administrator.

ARTICLE X

MERGER OF COMPANY: MERGER OF PLAN

10.1 Effect of Reorganization or Transfer of Assets. In the event of a consolidation, merger, sale, liquidation, or other transfer of the operating assets of the Company to any other company, the ultimate successor or successors to the business of the Company shall automatically be deemed to have elected to continue this Plan in full force and effect, in the same manner as if the Plan had been adopted by resolution of its board of directors, unless the successor(s), by resolution of its board of directors, shall elect not to so continue this Plan in effect, in which case the Plan shall automatically be deemed terminated as of the applicable effective date set forth in the board resolution.

10.2 Merger Restriction. This Plan and its related Trust shall not in whole or in part merge or consolidate with, or transfer its assets or liabilities to any other plan or trust unless each affected Participant in this Plan would receive a benefit immediately after the merger, consolidation, or transfer (if the Plan then terminated) which is equal to or greater than the benefit he would have been entitled to receive immediately before the merger, consolidation, or transfer (if the Plan had then terminated).

ARTICLE XI

PLAN TERMINATION AND DISCONTINUANCE OF CONTRIBUTIONS

11.1 Plan Termination.

- (a) Subject to the following provisions of this Section 11.1, the Company may terminate the Plan and the Trust agreements at any time and for any reason by an instrument in writing executed in the name of the Company by an officer or officers duly authorized to execute such an instrument, and delivered to the Trustee. The Company expressly disavows any contractual obligation, implied or otherwise, to continue this Plan.
- (b) The Plan and Trust agreements may terminate if the Company merges into any other corporation, if as a result of the merger the entity of the Company ceases, and the Plan is terminated pursuant to the rules of Section 10.1.
- (c) Upon and after the effective date of the termination, the Company shall make no further contributions under the Plan and no contributions need be made by the Company applicable to the Plan Year in which the termination occurs, except as may otherwise be required by law.
- (d) The rights of all affected Participants to benefits accrued to the date of termination of the Plan, to the extent funded as of the date of termination, shall automatically become fully vested as of that date.

11.2 Discontinuance of Contributions.

(a) The Company by resolution of its Board of Directors may discontinue contributions to the Plan at any time and for any reason in the Board's sole discretion. Upon and after the effective date of this discontinuance, the Company shall make no further Company contributions under the Plan and no Company contributions need be made by the Company with respect to the Plan Year in which the discontinuance occurs, except as may otherwise be required by law.

- (b) The discontinuance of Company contributions on the part of the Company shall not terminate the Plan as to the funds and assets then held by the Trustee, or operate to accelerate any payments of distributions to or for the benefit of Participants, Beneficiaries or Alternate Payees, and the Trustee shall continue to administer the Trust Fund in accordance with the provisions of the Plan until all of the obligations under the Plan shall have been discharged and satisfied.
- (c) However, if this discontinuance of Company contributions shall cause the Plan to lose its status as a qualified plan under Code section 401(a), the Plan shall be terminated in accordance with the provisions of this Article XI.
- (d) On and after the effective date of a discontinuance of Company contributions, the rights of all affected participants to benefits accrued to that date, to the extent funded as of that date, shall automatically become fully vested as of that date.
- **11.3** <u>Rights of Participants</u>. In the event of the termination of the Plan, for any cause whatsoever, all assets of the Plan, after payment of expenses, shall be used for the exclusive benefit of Participants and their Beneficiaries and no part thereof shall be returned to the Company, except as provided in Section 4.7 of this Plan or as otherwise permitted by law.

11.4 Trustee's Duties on Termination.

- (a) Upon termination of the Plan, the Plan Administrator shall determine whether to continue the Trust, to distribute the assets of the Trust to Participants, Beneficiaries and Alternate Payees, to transfer the assets in the Trust to another qualified plan maintained by the Company, or to take other action consistent with applicable law.
- (b) If so directed by the Plan Administrator upon termination of this Plan, the Trustee shall proceed as soon as possible to reduce all of the assets of the Trust Fund to cash and/or common stock and other securities in such proportions as the Plan Administrator shall determine (after approval by the Internal Revenue Service, if necessary or desirable, with respect to any portion of the assets of the Trust Fund held in common stock or securities of the Company). After first deducting the estimated expenses for liquidation and distribution chargeable to the Trust Fund, and after setting aside a reasonable reserve for expenses and liabilities (absolute or contingent) of the Trust, the Plan Administrator shall make the allocations required under Article VI, where applicable, with the same effect as though the date of completion of liquidation were an Anniversary Date of the

Plan. Following these allocations, the Trustee shall promptly, after receipt of appropriate instructions from the Plan Administrator, distribute in accordance with such instructions to each former Participant, or Beneficiary or Alternate Payee, a benefit equal to the amount credited to his Accounts as of the date of completion of the liquidation.

(c) The Trustee and the Plan Administrator shall continue to function as such for such period of time as may be necessary for the winding up of this Plan and for the making of distributions in accordance with the provisions of this Plan.

11.5 Partial Termination.

- (a) In the event of a partial termination of the Plan within the meaning of Code section 411(d)(3), the interest of affected Participants in the Trust Fund, as of the date of the partial termination, shall become nonforfeitable as of that date.
- (b) That portion of the assets of the Plan affected by the partial termination shall be used exclusively for the benefit of the affected Participants and their Beneficiaries, and no part thereof shall otherwise be applied.
- (c) With respect to Plan assets and Participants affected by a partial termination, the Plan Administrator and the Trustee shall follow the same procedures and take the same actions prescribed in this Article XI in the case of a total termination of the Plan.
- **11.6** Failure to Contribute. The failure of the Company to contribute to the Trust in any year, shall not, in and of itself, constitute a complete discontinuance of contributions to the Plan.

ARTICLE XII

APPLICATION FOR BENEFITS

12.1 <u>Application for Benefits; Claims Procedure</u>. The Plan Administrator may require any person claiming benefits under the Plan to submit an application therefor, together with such documents and information as the Plan Administrator may require. In the case of any person suffering from a disability which prevents the claimant from making personal application for benefits, the Plan Administrator may, in its discretion, permit another person acting on his behalf to submit the application.

12.2 Action on Application.

- (a) Within ninety (90) days following receipt of an application and all necessary documents and information, the Plan Administrator's authorized delegate reviewing the claim shall furnish the claimant with written notice of the decision rendered with respect to the application.
 - (b) In the case of a denial of the claimant's application, the written notice shall set forth:
 - (i) The specific reasons for the denial, with reference to the Plan provisions upon which the denial is based;
 - (ii) A description of any additional information or material necessary for perfection of the application (together with an explanation why the material or information is necessary); and
 - (iii) An explanation of the Plan's claim review procedure.
- (c) A claimant who wishes to contest the denial of his application for benefits or to contest the amount of benefits payable to him shall follow the procedures for an appeal of benefits as set forth in Section 12.3 below, and shall exhaust such administrative procedures prior to seeking any other form of relief.

12.3 Appeals.

(a)(i) A claimant who does not agree with the decision rendered with respect to his application may appeal the decision to the Plan Administrator.

- (ii) The appeal shall be made, in writing, within sixty-five (65) days after the date of notice of the decision with respect to the application.
- (iii) If the application has neither been approved nor denied within the ninety-day period provided in Section 12.2 above, then the appeal shall be made within sixty-five days (65) after the expiration of the ninety-day period.
- (b) The claimant may request that his application be given full and fair review by the Plan Administrator. The claimant may review all pertinent documents and submit issues and comments in writing in connection with the appeal.
- (c) The decision of the Plan Administrator shall be made promptly, and not later than sixty (60) days after the Plan Administrator's receipt of a request for review, unless special circumstances require an extension of time for processing, in which case a decision shall be rendered as soon as possible, but not later than one hundred twenty (120) days after receipt of a request for review.
- (d) The decision on review shall be in writing and shall include specific reasons for the decision, written in a manner calculated to be understood by the claimant with specific reference to the pertinent Plan provisions upon which the decision is based.

ARTICLE XIII

LIMITATIONS ON ANNUAL ADDITIONS

13.1 <u>Maximum Annual Additions</u>. The Annual Additions of a Participant shall not exceed the maximum permissible amount specified in Code section 415(c)(1). To the extent necessary to reduce an excess amount in a Participant's Accounts, the Plan Administrator may distribute such Participant's Elective Deferrals to the Participant. Any such amounts shall be disregarded for purposes of the nondiscrimination test in Section 5.2 of the Plan and for purposes of determining whether the Participant has any Excess Deferrals for the Plan Year.

13.2 Effect of Participation in Other Company Plans.

If a Participant in this Plan is also a Participant in another defined contribution plan maintained by the Company, the aggregate Annual Additions of the Participant under this Plan and such other plan(s) shall not exceed the maximum permissible amount specified in Code section 415(c)(1). The Plan Administrator shall prescribe such rules as may be necessary or appropriate with respect to applying this limit to the respective plans involved so as to ensure that the aggregate limit or Annual Additions is not exceeded.

- **13.3** <u>Incorporation by Reference of Code Section 415</u>. In order to ensure compliance with Code section 415, the Plan hereby incorporates said Section by reference as though it were set out as part of this Plan. In applying Code section 415 to this Plan, the Plan shall include each grandfather or transition rule provided by such Section or any law amending such Section, in order to allow the largest benefit otherwise payable hereunder, or under other plans maintained by the Company, to be paid.
- **13.4** No Contractual Right to Excess Contributions. If, in order to comply with the limitations of this Article XIII, it becomes necessary to reduce a Participant's Account(s), to reduce or reallocate amounts previously allocated to such Accounts, or otherwise, such action(s) may be taken by the Plan Administrator and Trustee free of any contractual obligation to the Participant (or Beneficiary) affected based on prior Account balances or allocations.

ARTICLE XIV

RESTRICTION ON ALIENATION; PARTICIPANT LOANS

- 14.1 General Restrictions Against Alienation. Except as otherwise provided by law and as otherwise provided by Sections 14.2 and 14.3:
- (a) The interest of any Participant, Beneficiary or Alternate Payee in the income, benefits, payments, claims or rights hereunder, or in the Trust Fund shall not in any event be subject to sale, assignment, hypothecation, or transfer. Each Participant, Beneficiary or Alternate Payee is prohibited from anticipating, encumbering, assigning, or in any manner alienating his interest under the Trust Fund, and is without power to do so, except as may otherwise be provided for in the Trust Agreement. The interest of any Participant, Beneficiary or Alternate Payee shall not be liable or subject to his debts, liabilities, or obligations, now contracted, or which may be subsequently contracted. The interest of any Participant, Beneficiary or Alternate Payee shall be free from all claims, liabilities, bankruptcy proceedings, or other legal process now or hereafter incurred or arising; and the interest or any part thereof, shall not be subject to any judgment rendered against the Participant, Beneficiary or Alternate Payee.
- (b) In the event any person attempts to take any action contrary to this Article XIV, that action shall be void and the Company, the Plan Administrator, the Trustee and all Participants, their Beneficiaries and Alternate Payees may disregard that action and are not in any manner bound thereby, and they, and each of them separately, shall suffer no liability for any disregard of that action, and shall be reimbursed on demand out of the Trust Fund for the amount of any loss, cost or expense incurred as a result of disregarding or of acting in disregard of that action.
- (c) The preceding provisions of this Section 14.1 shall be interpreted and applied by the Plan Administrator in accordance with the requirements of Code section 401(a)(13) as construed and interpreted by authoritative judicial and administrative rulings and regulations.
- **14.2** <u>Nonconforming Distributions Under Court Order</u>. Benefits shall be paid to an Alternate Payee if such payment is required pursuant to a qualified domestic relations order, as defined in Code sections 401(a)(13) and 414(p).

In the event that the Plan receives a domestic relations order, the Plan Administrator or its delegate shall promptly

notify the Participant and any Alternate Payee (i.e., spouse, former spouse, child or other dependent of a participant who is recognized by a domestic relations order as having a right to receive all, or a portion of, the benefits payable under the Plan with respect to such Participant) of the receipt of such order and the Plan's procedures for determining the qualified status of such orders, and within a reasonable period of time after receipt of such order, the Plan Administrator shall determine whether such order is a qualified domestic relations order and notify the Participant and each Alternate Payee of such determination. In determining the qualified status of a domestic relations order and in administering distributions under such qualified orders, the Plan Administrator shall follow the following procedures:

- (a) When the Plan receives a domestic relations order affecting Plan benefits, the Plan Administrator or its delegate shall promptly notify each person specified in the order as entitled to benefits under the Plan (using the address(es) included in the domestic relations order) of the Plan procedure as set forth herein (and as supplemented, if necessary, by Plan Administrator procedures).
- (b) The Plan shall permit an Alternate Payee to designate a representative for receipt of copies of notices that are sent to the Alternate Payee with respect to a domestic relations order.
- (c) The Plan Administrator shall review any domestic relations order to determine if it satisfies the requirements of being a qualified domestic relations order. In making such determination, the Plan Administrator may seek the advice of legal counsel to the Plan and may rely upon the legal opinion of such counsel in determining the qualified status of domestic relations orders and appropriate measures to resist or implement such orders. The Plan Administrator may, but need not, enter an appearance on behalf of the Plan in the domestic relations lawsuit, if any, and may pursue such legal remedies as may be desirable for resisting unqualified orders or in modifying proposed orders.
- (d) During any period in which the issue of whether a domestic relations order is a qualified domestic relations order is being determined by the Plan Administrator, by a court of competent jurisdiction, or otherwise, the Plan Administrator shall segregate in a separate account in the Plan or in an escrow account the amounts, if any, which would have been payable to the Alternate Payee during such period if the order had been determined to be a qualified domestic relations order. If, within eighteen (18) months it is determined that the order is not a qualified domestic relations order or the issue as to

whether such order is a qualified domestic relations order is not resolved, than the Plan Administrator shall pay the segregated amounts (plus any interest thereon) to the person or persons who would have been entitled to such amounts if there had been no order. Any determination that an order is a qualified domestic relations order which is made after the close of the aforementioned eighteen (18) month period shall be applied prospectively only, should there be any undistributed benefits of the Participant to which the order related.

- (e) If the Plan Administrator or other fiduciary of the Plan acts in accordance with the foregoing procedures in treating a domestic relations order as being (or not being) a qualified domestic relations order or taking action to segregate an account and ultimately make payment thereof in accordance with subparagraph (d) above, then the Plan's obligations to the Participant and each Alternate Payee shall be discharged to the extent of any payment made pursuant to such act.
- 14.3 <u>Authorized Participant Loans</u>. The Plan Administrator may authorize one loan from the Trust Fund to Participants (including, for this purpose, Suspended Participants) pursuant to rules prescribed by the Plan Administrator, except with respect to Participants who formerly participated in the NNS 401(k) Plan. Such Participants who formerly participated in the NNS 401(k) Plan will be permitted to maintain up to two loans if such loans are outstanding as of September 1, 1999, but after such loans are repaid, such Participants will only be entitled to one loan from the Trust Fund pursuant to the rules prescribed by the Plan Administrator. These rules shall be designed to ensure that these Participant loans satisfy the requirements of Code sections 4975(d)(1), 72(p), and any other provision of law that is, or may become applicable. These rules shall provide that:
 - (a) The loans are available to all Participants on a reasonably equivalent basis.
 - (b) The loans are not made available to Highly Compensated Employees in amounts greater than the amounts made available for other Employees. For this purpose, the rules prescribed by the Plan Administrator may restrict the amount of the loan to a percentage of the Participant's Account balance or to use different percentages depending upon the amount of the loan, provided the percentages are applicable to all Participants. The Plan Administrator may also prescribe rules pursuant to which the portion of an individual's Account that is invested in Company Stock (or a fund within the Trust to which Company Stock is allocated) may (or may not) be taken into account in determining the maximum loan he may obtain.

- (c) The loans bear a reasonable rate of interest.
- (d) The loans are adequately secured. For this purpose, the amount of the security must be at least equal to the amount of the loan. The rules to be prescribed by the Plan Administrator may permit a Participant to use some or all of his interest under the Plan as security for the loan, provided that the Participant's interest in his Accounts attributable to Company Stock may not be used as collateral for a loan.
- (e) If the loan, or a loan from another qualified retirement plan maintained by the Company, is to be secured by some or all of the Participant's Accounts under the Plan, the Participant and his spouse, if any, must consent to the loan and the possible reduction in the Accounts in the event of a set-off of the loan against the Account balances as a result of nonpayment of the loan. Such consent must be given in writing within a ninety-day period before the Plan Administrator makes the loan. In the event the Participant defaults on the loan and Participant's Accounts are security for the loan, the Account balances will not be used to satisfy the loan obligation prior to the earlier of the Participant's termination of employment with the Company or an event resulting in a permissible distribution of his Accounts under the Plan.
- (f) The loan must state the date upon which the loan must be repaid, which may not exceed five years (except in the case of loans used to acquire a dwelling unit which, within a reasonable time after the loan is made, is to be used as the principal residence of the Participant). The loan must be repayable in substantially level payments, with payments not less frequently than quarterly. However, if a Participant is on an approved, unpaid leave of absence, the loan repayments may be suspended for a period not longer than one year. In the case of this exception to the level payments requirement, the loan must be repaid within the five year period and the remaining installment payments cannot be less than those required under the terms of the original loan. Notwithstanding the foregoing, loan repayments will be suspended under this Plan as permitted under Section 414(u)(4) of the Code.
- (g) In connection with the making of any loan to a Participant pursuant to the provisions of this Section 14.3, the Participant receiving such a loan may be required to execute such documents as may be required by the Plan Administrator and/or Trustee.

- (h) The amount of the loan may not exceed the lesser of:
- (i) \$50,000 (reduced by the excess of the highest outstanding balance of loans from the Plan during the one-year period ending on the date preceding the date on which such loan is made); or
- (ii) One-half of the present value of the Participant's vested interest in his Accounts. For purposes of this Section 14.3(h), the Participant's vested interest and outstanding loan balances in all qualified employer plans (as such term is defined in Code section 72(p)(3)) of the Company shall be aggregated to determine whether a loan shall be permissible hereunder and the maximum permissible amount thereof.

The decision as to whether or not any Participant Loans shall be made under this Section 14.3 shall be made in the sole discretion of the Plan Administrator, and the Participant shall not have a contractual right to obtain a loan hereunder.

- (i) In the event the Participant dies prior to distribution of his Distributable Benefit, the amount payable to his Beneficiary or spouse, as applicable, shall be reduced by the amount of the security interest in the Participant's vested interest held by the Plan by reason of a loan outstanding to such Participant.
- (j) If a Participant is subject to the requirements of Section 8.6(a), a loan shall be granted to a Participant only if the Participant obtains the consent of his spouse, if any, to use his Accounts as security for the loan. Spousal consent shall be obtained no earlier than the beginning of the 90-day period that ends on the date the loan is to be secured. The consent must be in writing, must acknowledge the effect of the loan and must be witnessed by a plan representative or notary public. Such consent shall thereafter be binding with respect to the consenting spouse or any subsequent spouse with respect to that loan. A new consent shall be required if the Participant's Accounts are used for renegotiation, extension, renewal or other revision of the loan.

ARTICLE XV

PLAN AMENDMENTS

- **15.1** <u>Amendments</u>. The Board of Directors in its sole discretion may at any time, and from time to time, amend the Plan by an instrument in writing executed in the name of the Company by an officer or officers duly authorized to execute such instrument, and delivered to the applicable Trustee. However, except as provided in Section 15.2 no amendment shall be made at any time, the effect of which would be:
 - (a) To cause any assets of the Trust Fund to be used for or diverted to purposes other than providing benefits to the Participants and their Beneficiaries, and defraying reasonable expenses of administering the Plan, except as provided in Section 4.7 or as otherwise permitted by law;
 - (b) To have any retroactive effect so as to deprive any Participant or Beneficiary of any benefit to which he would be entitled under this Plan if his employment were terminated immediately before the amendment; or
 - (c) To increase the responsibilities or liabilities of a Trustee or an Investment Manager without its written consent.
- **15.2 Retroactive Amendments.** The Plan may be amended prospectively or retroactively (as provided in Code section 401(b)) to make the Plan conform to any provision of ERISA, any Code provisions dealing with tax-qualified employees' trusts, or any regulation under either.

ARTICLE XVI

TOP-HEAVY PROVISIONS

- **16.1** <u>Application</u>. If the Plan is or becomes top heavy in any Plan Year, the provisions of this Article XVI will supersede any conflicting provisions in the Plan.
 - **16.2 Criteria.** The Plan shall be top heavy for any Plan Year if any of the following conditions exist:
 - (a) The Top-Heavy Ratio for the Plan exceeds 60% and this Plan is not part of any Required Aggregation Group or Permissive Aggregation Group of Plans.
 - (b) This Plan is part of a Required Aggregation Group of plans, but not part of a Permissive Aggregation Group, and the Top-Heavy Ratio for the group of plans exceeds 60%.
 - (c) This Plan is a part of a Required Aggregation Group and part of a Permissive Aggregation Group of plans and the Top-Heavy Ratio for the Permissive Aggregation Group exceeds 60%.
 - 16.3 <u>Definitions</u>. For purposes of this Article XVI, the following terms shall have the following meanings:
 - (a) Determination Date: With respect to any Plan Year, (i) the Determination Date shall be the last day of the preceding Plan Year, or (ii) in the case of the first Plan Year of the Plan, the last day of such Plan Year.
 - (b) Key Employee: Any Employee or former Employee (and the Beneficiaries of such Employees) who, pursuant to the rules of Code section 416(i) and the regulations thereunder, is or was:
 - (i) An officer of the Company having an annual Compensation greater than 50% of the dollar limitation under Code section 415(b)(1)(A);
 - (ii) One of the ten Employees having annual Compensation from the Company of more than the dollar limitation under Code section 415(c)(1) (A), and owning (or considered as owning) under Code section 318 the largest interest in the Company;
 - (iii) A 5% Owner of the Company; or
 - (iv) A 1% Owner of the Company having annual Compensation from the Company of more than \$150,000.

The determination period is the Plan Year containing the Determination Date and the four preceding Plan Years.

- (c) Permissive Aggregate Group: The Required Aggregation Group of plans plus any other plan or plans of the Company that, when considered as a group with the Required Aggregation Group, would continue to satisfy the requirements of Code sections 401(a)(4) and 410, and which are designated by the Company to constitute a Permissive Aggregate Group.
- (d) Required Aggregation Group: (1) Each plan of the Company in which a Key Employee is a Participant or was a Participant at any time during the determination period (regardless of whether the Plan has terminated) and (2) any other qualified plan of the Company that enables a plan described in (1) to meet the requirements of Code section 401(a)(4) or 410.

(e) Top-Heavy Ratio:

- (i) If the Company maintains one or more defined contribution plans (including any Simplified Employee Pension) and the Company has not maintained any defined benefit plan that during the five-year period ending on the Determination Date has or has had accrued benefits, the Top-Heavy Ratio for this Plan alone or for the Required Aggregation Group or Permissive Aggregation Group, as appropriate, is a fraction, the numerator of which is the sum of the Account balances of Key Employees as of the Determination Date (including any part of any Account balance distributed in the five-year period ending on the Determination Date), and the denominator of which is the sum of all Account balances (including any part of any Account balance distributed in the five-year period ending on the Determination Date), both computed in accordance with Code section 416 and regulations thereunder. Both the numerator and denominator of the Top-Heavy Ratio are adjusted to reflect any contributions not actually made as of the Determination Date, but which is to be taken into account on that date under Code section 416 and regulations thereunder.
- (ii) If the Company maintains one or more defined contribution plans (including any Simplified Employee Pension) and the Company maintains or has maintained one or more defined benefit plans that during the five-year period ending on the Determination Date has or has had any accrued benefits, the Top-Heavy Ratio for any Required or Permissive Aggregation Group, as appropriate, is a fraction, the numerator of which is the sum of Account balances under the aggregated defined contribution plan or plans for all Key Employees determined in accordance with (1) above and the present value of accrued benefits under the

aggregated defined benefit plan or plans for all Key Employees as of the Determination Date, and the denominator of which is the sum of the account balances under the aggregated defined contribution plan or plans for all Participants determined in accordance with (1) above, and the present value of accrued benefits under the defined benefit plan or plans for all Participants as of the Determination Date, all determined in accordance with Code section 416 and the regulations thereunder. The accrued benefits under a defined benefit plan in both the numerator and denominator of the Top-Heavy Ratio are adjusted for any distribution of an accrued benefit made in the five-year period ending on the Determination Date. Solely for the purpose of determining if the Plan, or any other plan included in a Required Aggregation Group is top-heavy (within the meaning of Code section 416(g)), the accrued benefit of an Employee other than a Key Employee shall be determined under (i) the method, if any, that uniformly applies for accrual purposes under all plans maintained by the Company and any Affiliated Companies or (ii) if there is no such method, as if such benefit accrued not more rapidly than the slowest accrual rate permitted under the fractional accrual rate of Code section 411(b)(1) (C).

- (iii) For purposes of (i) and (ii) above, the value of Account balances and the present value of accrued benefits will be determined as of the most recent Valuation Date that falls within or ends with the 12-month period ending on the Determination Date, except as provided in Code section 416 and regulations thereunder for the first and second Plan Years of a defined benefit plan. The Account balances and accrued benefits of a Participant (A) who is not a Key Employee but who was a Key Employee in a prior year or (B) who has not been credited with at least one Hour of Service with any Company maintaining the Plan at any time during the five-year period ending on the Determination Date will be disregarded. The calculation of the Top-Heavy Ratio, and the extent to which distributions, rollovers and transfers are taken into account, will be made in accordance with Code section 416 and regulations thereunder. Voluntary deductible contributions will not be taken into account in computing the Top-Heavy Ratio. When aggregating plans, the value of Account balances and accrued benefits will be calculated with reference to the Determination Dates that fall within the same calendar year.
- (iv) For purposes of establishing the present value in order to compute the Top-Heavy Ratio any benefit shall be discounted only for mortality and interest based on the interest rate that would be used as of the date of distribution by the Pension Benefit Guaranty Corporation to determine the present value of a lump-sum distribution on Plan termination.

- (f) Valuation Date: The date described in Section 2.1 as of which Account balances or accrued benefits are valued for purposes of calculating the Top-Heavy Ratio.
- **16.4** <u>Adjustment to Fractions</u>. In any Plan Year in which the Plan is Top-Heavy, in applying the limitations of Code section 415, the denominator of the Defined Benefit Fraction shall be computed using 100% of the dollar limitation instead of 125%, and the maximum aggregate amount used with respect to the denominator of the Defined Contribution Fraction shall be computed by using 100% of the dollar limitation instead of 125%.
- **16.5** <u>Vesting Requirements</u>. If the Plan is determined to be a Top-Heavy Plan in any Plan Year, then a Participant's right to his Accounts derived from Company Contributions, determined as of the end of such Plan Year, shall vest in accordance with the following schedule, unless a more rapid vesting schedule is then in effect under the terms of the Plan:

Years of Vesting Service	Vesting Percentage
2	20%
3	40%
4	60%
5	80%
6 or more	100%

If the Plan ceases to be a Top-Heavy Plan in any Plan Year, then the vesting schedule set forth in Article VII shall apply for such Plan Year with respect to any portion of a Participant's Accounts that is forfeitable as of the beginning of such Plan Year; provided, however, that a Participant with five or more years of vesting service shall be given the option of remaining under the vesting schedule set forth above.

16.6 Minimum Contribution. If this Plan is a Top-Heavy Plan in any Plan Year, the Company Contributions for such year for each "participant" (as defined for the purpose of providing mandatory minimum contributions under regulations) who is not a Key Employee shall not be less than three percent (3%) of such participant's compensation. If, however, the Plan does not enable a defined benefit plan to meet the requirements of Section 401(a)(4) or 410, the Company Contributions shall not exceed that percentage of each participant's compensation which is equal to the highest percentage of compensation at which Company Contributions are made for the Plan Year for any Key Employee (a) under the Plan or (b) if the Plan is part of an Aggregation Group, under any defined contribution plan in such Group. The percentage of compensation at which Company Contributions are made for a Key Employee shall be computed

without regard to compensation in excess of the ceiling on includable compensation set forth in Section 16.7 of this Article XVI. For Plan Years beginning before January 1, 1989, for purposes of this Section 16.6, Company Contributions attributable to a salary reduction or similar arrangement and contributions made pursuant to Chapter 21 of Title II of the Social Security Act shall be disregarded. For Plan Years beginning after December 31, 1988, Company Contributions attributable to a salary reduction or similar arrangement made by Employees other than Key Employees shall not be taken into account under this Section 16.6.

16.7 <u>Ceiling on Includable Compensation</u>. If this Plan is determined to be a Top-Heavy Plan in any Plan Year, then only the first \$150,000 of a Participant's Compensation shall be taken into account in determining the allocation to the Accounts of such Participant for the Plan Year. The \$150,000 limit shall automatically be adjusted for such Plan Years and to such extent as is permitted by the Secretary of the Treasury.

ARTICLE XVII

MISCELLANEOUS

17.1 No Enlargement of Employee Rights.

- (a) This Plan is strictly a voluntary undertaking on the part of the Company and shall not be deemed to constitute a contract between the Company and any Employee, or to be consideration for, or an inducement to, or a condition of, the employment of any Employee.
- (b) Nothing contained in this Plan or the Trust shall be deemed to give any Employee the right to be retained in the employ of the Company or to interfere with the right of the Company to discharge or retire any Employee at any time.
- (c) No Employee, or any other person, shall have any right to or interest in any portion of the Trust Fund other than as specifically provided in this Plan, and no Employee or any other person shall be entitled to rely upon any representations, whether oral or in writing, including representations made in the summary plan description, any prospectus or other document, which are inconsistent with this Plan document.

17.2 Mailing of Payments; Lapsed Benefits.

- (a) All payments under the Plan shall be delivered in person or mailed to the last address of the Participant or Beneficiary furnished pursuant to Section 17.3 below.
- (b) In the event that a benefit is payable under this Plan to a Participant or Beneficiary and after reasonable efforts such individual cannot be located for the purpose of paying the benefit during a period of three consecutive years following the date payment would otherwise have been made, the benefit shall be forfeited and allocated in the same manner as Additional Company Contributions. If the Participant or Beneficiary later makes a valid claim for the benefit, the claim shall be paid pursuant to directions by the Plan Administrator.
- (c) For purposes of this Section 17.2, the term "Beneficiary" shall include any person entitled under Section 8.10 to receive the interest of a deceased Participant or deceased designated Beneficiary and shall also include an Alternate Payee.
- (d) The Accounts of a Participant shall continue to be maintained until the amounts in the Accounts are paid to the Participant or his Beneficiary. In the event that the Plan is

terminated and the Plan Administrator directs the Trustee to the Trust and distribute the assets of the Trust Fund, the following rules shall apply:

- (i) All Participants and Beneficiaries (including Participants and Beneficiaries who have not previously claimed their benefits under the Plan) shall be notified of their right to receive a distribution of their interests in the Plan.
- (ii) All Participants and Beneficiaries shall be given a reasonable length of time, which shall be specified in the notice, in which to claim their benefits.
- (iii) All Participants (and their Beneficiaries) who do not claim their benefits within the designated time period shall be presumed to be dead. The Accounts of such Participants shall be forfeited at such time. These forfeitures shall be disposed of according to rules prescribed by the Plan Administrator, which rules shall be consistent with applicable law. Alternatively the Plan Administrator may, but shall not be required to, deposit such funds in an applicable state unclaimed property or similar fund, pursuant to applicable state law.
- (iv) The Plan Administrator shall prescribe such rules as it may deem necessary or appropriate with respect to the notice, deposit or forfeiture rules stated above.
- (e) Should it be determined that the preceding rules relating to forfeiture of benefits upon Plan termination are inconsistent with any of the provisions of the Code and/or ERISA, these provisions shall become inoperative without the need for a Plan amendment and the Plan Administrator shall prescribe rules that are consistent with the applicable provisions of the Code and/or ERISA.
- 17.3 <u>Addresses</u>. Each Participant shall be responsible for furnishing the Plan Administrator with his correct current address and the correct current name and address of his Beneficiary or Beneficiaries.

17.4 Notices and Communications.

(a) All applications, notices, designations, elections, and other communications from Participants, Beneficiaries and Alternate Payees shall be in writing, on forms prescribed by the Plan Administrator and shall be mailed or delivered to the office designated by the Plan Administrator, and shall be deemed to have been given when received by that office.

- (b) Each notice, report, remittance, statement and other communication directed to a Participant, Beneficiary or Alternate Payee shall be in writing and may be delivered in person or by mail. An item shall be deemed to have been delivered and received by the Participant, Beneficiary or Alternate Payee when it is deposited in the United States mail with postage prepaid, addressed to the Participant, Beneficiary or Alternate Payee at his last address of record with the Plan Administrator.
- **17.5** <u>Reporting and Disclosure</u>. The Plan Administrator shall be responsible for the reporting and disclosure of information required to be reported or disclosed by the Plan Administrator pursuant to ERISA or any other applicable law.
- **17.6** <u>Governing Law</u>. All legal questions pertaining to the Plan shall be determined in accordance with the provisions of ERISA and, to the extent not preempted by ERISA, the laws of the State of Virginia. All contributions made hereunder shall be deemed to have been made in Virginia.

17.7 Interpretation.

- (a) Article and Section headings are for convenient reference only and shall not be deemed to be part of the substance of this instrument or in any way to enlarge or limit the content of any Article or Section. Unless the context clearly indicates otherwise, masculine gender shall include the feminine, and the singular shall include the plural and the plural the singular.
- (b) The provisions of this Plan shall in all cases be interpreted in a manner that is consistent with this Plan satisfying the requirements of Code section 401(a).
- 17.8 Withholding For Taxes. Any payments out of the Trust Fund may be subject to withholding for taxes as may be required by any applicable federal or state law.
- **17.9** <u>Limitation on Company, Plan Administrator and Trustee Liability</u>. Any benefits payable under this Plan shall be paid or provided for solely from the Trust Fund and neither the Company, the Plan Administrator nor the Trustee assume any responsibility for the sufficiency of the assets of the Trust to provide the benefits payable hereunder.
- **17.10** <u>Successors and Assigns</u>. This Plan and the Trust established hereunder shall inure to the benefit of, and be binding upon, the parties hereto and their successors and assigns.

17.11 Counterparts. This Plan document may be executed in any number of identical counterparts.	terparts, each of which shall be deemed a complete original in
itself and may be introduced in evidence or used for any other purpose without the production of a	ny other counterparts.
17.12 No Implied Rights or Obligations. The Company, in establishing and maintaining the disavows the creation of any rights in Employees, Participants, Beneficiaries or Alternate Payees of Administrator, except as expressly provided herein.	3 3 1 3
IN WITNESS WHEREOF, in order to record the adoption of this document, AMSEC LLC officer this day of, 20	has caused this instrument to be executed by its duly authorized
I	AMSEC LLC
a	a Delaware Limited Liability Company
F	Ву:
90	

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

Amendment No. 1

Efective as of January 1, 1997, or such other dates as noted herein, the AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN (the "Plan") is hereby amended as follows:

1. The second sentence of Section 1.1 is amended to read as follows:

This document amends and restates the Plan in its entirety, effective as of January 1, 1997, except where a different date is specifically provided.

2. The definition of "Compensation" in Section 2.19 is amended to add to subsections (a), (b), and (c) a reference to Code section 132(f)(4) and reads as follows:

2.19 Compensation.

- (a) For purposes of determining the allocation of Profit Sharing Contributions pursuant to Sections 4.3 and 6.10(e) and the amount of the Participant's permissible Elective Deferrals under Section 5.2, "Compensation" shall mean the amounts described in subsection (b) below paid by the Company during a Plan Year by reason of services performed by an Employee less any compensation paid by reason of services performed during any period in which the Employee is not a Participant under this Plan, but including any elective contributions that are made by the Company on behalf of an Employee pursuant to an arrangement described in Code sections 125, 132(f)(4), 402(a)(8), 402(h) or 403(b).
- (b) For purposes of applying the limitation on Annual Additions pursuant to Article XIII of the Plan, "Compensation" shall mean a Participant's earned income, wages, salaries and fees for professional services and other amounts received for personal services actually rendered in the course of

employment with the Company (including, but not limited to, commissions paid salespersons, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips and bonuses), and excluding the following:

- (i) Company contributions to a plan of deferred compensation that are not includable in the Employee's gross income for the taxable year in which contributed, or Company contributions under a simplified employee pension (within the meaning of Code section 408(k)) to the extent such contributions are excludable from gross income by the Employee, or any distributions from a plan of deferred compensation;
- (ii) Amounts realized from the exercise of a nonqualified stock option, or when restricted stock (or property) held by the Employee either becomes freely transferable or is no longer subject to a substantial risk of forfeiture;
 - (iii) Amounts realized from the sale, exchange or other disposition of stock acquired under a qualified or incentive stock option; and
- (iv) Other amounts that received special tax benefits, or Company contributions (whether or not under a salary reduction agreement) towards the purchase of an annuity described in Code section 403(b) (whether or not the amounts are actually excludable from the gross income of the Employee).

Notwithstanding the foregoing, effective for Plan Years beginning after December 31, 1997, "Compensation" for purposes of this subsection 2.19(b) shall include Elective Deferrals under Code section 402(g)(3) and amounts contributed under Code sections 125 and 132(f)(4) by the Company at the Employee's election that are not included in the Employee's gross income.

- (c) For purposes of determining whether an Employee is a Highly Compensated Employee, "Compensation" shall be "Compensation" as defined in subsection (b) above, but determined without regard to Code sections 125, 132(f)(4), 402(e)(3) and 402(h)(1)(B), and, in the case of Company Contributions made pursuant to a salary reduction agreement, without regard to Code section 403(b). Thus, Compensation for purposes of this subsection 2.19(c) includes elective or salary reduction contributions to any Company section 125 flex plans, section 132(f)(4) transit reimbursement arrangements, this Plan, or any tax-sheltered annuity, even for periods prior to January 1, 1998 (when such items are also includible under subsection 2.19(b)).
- 3. The second sentence of Section 2.25 is amended to read as follows:

The effective date of this amendment and restatement of the Plan is January 1, 1997, except where a different Effective Date is specified.

4. Section 2.35(a)(ii)(B) is amended to read as follows:

If the Employer so elects by Plan amendment, any Employee in the "top-paid group" of Employees for such preceding Plan Year.

5. Section 5.2(c) is amended to add a new second sentence to read as follows:

If a Participant's Elective Deferrals to this Plan exceed the limitation described in the preceding sentence, the Excess Deferral may be distributed to such Participant as permitted by Code section 402(g)(A)(ii).

6. Section 5.3(i) is amended effective December 31, 2001 by adding a new final sentence to read as follows:

Effective for Plan Years ending on or before December 31, 2001, the provisions of Reg. section 1.401(m)-2(b) shall apply for the purpose of applying the test for multiple use of the alternative limitation.

7. Section 6.16 is amended to add the clause "Effective December 12, 1994" to the beginning of this Section.

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

Amendment No. 2

Effective as of January 1, 2002, or such other dates as noted herein, the AMSEC Employees 401(k) Profit Sharing Plan (the "Plan") is hereby amended as follows:

- 1. Section 2.1(a) is amended to include a reference to Catch-up Contributions and reads as follows in its entirety:
- (a) The Deferral Account maintained for each Participant who has made Elective Deferrals or Catch-up Contributions to the provisions of Section 5.1.
- 2. A new Section 2.13A is added to read as follows in its entirety: $\frac{1}{2}$
 - 2.13A Catch-up Contribution. "Catch-up Contribution" shall mean a contribution paid to the Trustee by the Company at the election of

a Participant who has attained age 50 before the last day of the Plan Year as provided in Section 5.1. Effective as of January 1, 2002, an eligible Participant may elect to make a Catch-up Contribution to be withheld from his or her Compensation by the Company in lieu of receiving such amount as current Compensation. Such amount shall be nonforfeitable when made. Catch-up Contributions shall be made in accordance with, and subject to the limitations of, section 414(v) of the Code. Catch-up Contributions shall not be taken into account for purposes of complying with the required limitations of Section 402(g) and 415(c) of the Code. The Plan shall not be treated as failing to satisfy the provisions of section 401(k)(3), 401(k)(11), 401(k)(12), 410(b), or 416 of the Code, as applicable, by reason of the making of such Catch-up Contributions.

- 3. A new Section 2.14A is added to read as follows in its entirety:
 - 2.14A Committee. "Committee" shall mean the AMSEC Retirement Plans Committee described in Article IX.
- 4. Section 2.16 is amended to include a reference to Catch-up Contributions and reads as follows in its entirety:
- **2.16** Company Contributions. "Company Contributions" shall mean all amounts (whether in cash or other property, including Company Stock) paid by the Company into the Trust Fund established and maintained under the provisions of this Plan for the purpose of providing benefits for Participants and their Beneficiaries, and shall include Elective Deferrals, Profit Sharing Contributions, Matching Contributions, Additional Company Contributions, Catch-up Contributions and any contributions to pay Plan expenses as provided in Section 6.13.
- 5. Effective as of January 1, 2002, the last paragraph of Section 2.19(b) is amended to refer to Catch-up Contributions to read as follows in its entirety:

"Notwithstanding the foregoing, effective for Plan Years beginning after December 31, 1997, "Compensation" for purposes of this subsection 2.19(b) shall include Elective Deferrals under Code section 402(g)(3), Catch-up Contributions under Code section 414(v), and amounts contributed under Code sections 125 and 132(f)(4), by the Company at the Employee's election that are not included in the Employee's gross income."

- 6. Section 2.19(e) is amended to read as follows in its entirety:
- (e) The amount considered as "Compensation" for any purpose hereunder shall be limited to \$200,000 or such other amount as provided by Code section 401(a)(17), as adjusted from time to time.

- 7. Section 2.21 is amended to include a reference to Catch-up Contributions and reads as follows in its entirety:
- **2.21 Deferral Account.** "Deferral Account" shall mean the Account opened up and maintained for each Employee who has made Elective Deferrals and, effective as of January 1, 2002, Catch-up Contributions pursuant to Section 5.1.
- 8. Section 2.26 is amended to include a reference to Catch-up Contributions and reads as follows in its entirety:
- **2.26** Elective Deferrals. "Elective Deferrals" shall mean contributions paid over by the Company to the Trustee at the election of a Participant in lieu of cash Compensation pursuant to Section 5.1. No amount that was designated or treated at the time of deferral or contribution as a Catch-up Contribution shall be treated as an Elective Deferral.
- 9. Section 2.58 is amended to include a reference to Catch-up Contributions and reads as follows in its entirety:
- **2.58 Profit Sharing Contribution**. "Profit Sharing Contribution" shall mean the contributions paid by the Company (other than Additional Company Contributions, Elective Deferrals, Catch-up Contributions or Matching Contributions) to the Trustee as provided in Article IV.
- 10. Section 2.62 is amended to read as follows in its entirety:
- **2.62** Rollover Contribution. "Rollover Contribution" shall mean a contribution to the Plan by an Eligible Employee of amounts which qualify for exclusion from gross income (a) pursuant to Code section 402(a), subject to the timing, maximum amount limitations, and other requirements of Code section 402(a), and (b) effective January 1, 2002, from an annuity contract described in Code section 403(b), and an eligible plan described in Code section 457(b) that is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state. Effective January 1, 2002 an Eligible Employee may also be the surviving spouse.
- 11. Section 3.2 is amended by adding the following paragraphs (d) and (e) to the end thereof to read as follows:
- (d) The Committee may adopt rules and procedures regarding the actual pay period in which Elective Deferrals and Catch-up Contributions commence to be deducted from Compensation to accommodate administrative and payroll practices and issues.

- (e) Notwithstanding the foregoing provisions of Section 3.2(c) to the contrary, the entry date for such Eligible Employees may be deferred by the Committee in the exercise of its discretion to allow a reasonable period of time for administrative procedures to become effective with respect to such Eligible Employees.
- 12. Section 4.3 is amended by adding the phrase "and, if applicable, Catch-up Contributions" to the end of the first sentence.
- 13. Section 5.1 is amended to include references to the Committee and Catch-up Contributions and reads as follows in its entirety:

"At dates established by the Committee, each Participant may elect, on forms prescribed by the Committee, to defer the receipt of a portion of his Compensation for that Plan Year, and to have the deferred amount contributed directly by the Company to the Trust Fund as Elective Deferrals and, if and to the extent applicable, Catch-up Contributions."

14. Section 5.2(c) is amended to include a reference to Catch-up Contributions, and to the dollar limitation imposed by reference to Code section 402(g), and reads as follows in its entirety:

"The maximum amount of Elective Deferrals contributed on behalf of a Participant under the Plan shall not exceed the maximum amount permitted pursuant to Code section 402(g) (multiplied by the Adjustment Factor) during any calendar year. If a Participant's Elective Deferrals to this Plan exceed the limitation described in the preceding sentence, the Excess Deferral may be recharacterized, to the extent permissible, as Catch-up Contributions pursuant to Code section 4.14(v) and, if not so recharacterized, then distributed to such Participant as Permitted by Code section 402(g)(A)(ii). If, during a calendar year, (i) the Employee participates in this Plan and one or more other plans with a qualified cash or deferred arrangement described in Code sections 401(k), 408(k)(6), or 403(b), (ii) the Employee's Elective Deferrals under all such plans (including this Plan) exceed the maximum amount permitted pursuant to Code section 402(g) (multiplied by the Adjustment Factor), and (iii) all or any portion of such Employee's Elective Deferrals under this Plan constitute Excess Deferrals by virtue of an allocation and notification by the Employee pursuant to Code section 402(g)(2)(A)(i), then to the extent such Excess Deferrals are not recharacterized as Catch-up Contributions pursuant to Code section 414(v), they shall be distributed to such Employee as permitted by Code section 402(g)(2)(A)(ii)."

- 15. Section 6.1 is amended by adding the clause "and, if eligible, Catch-up Contributions" to the end thereof.
- 16. Section 6.10(a) is amended to include a reference to Catch-up Contributions and reads as follows in its entirety:
- (a) All Company Contributions on account of Elective Deferrals and Catch-up Contributions shall be allocated to Participants' Deferral Accounts as soon as practicable following their contribution to the Trust, but in no event later than ninety (90) days after such amounts were deferred.
- 17. Section 8.3(b)(ii)(E) is modified by substituting the suspension period "six (6) months" for the period "twelve (12) months" effective for hardship withdrawals made on and after January 1, 2002.
 - 18. Section 8.6 is modified to read as follows in its entirety:

8.6 Optional Forms of Payment.

- (a) Subject to Section 8.6(d), each Participant may elect, with his spouse's consent if required and subject to (b) below, to receive his Distributable Benefit in one of the following forms, or a combination of such forms, subject to the provisions of Sections 8.7 and 8.8:
 - (i) A single sum.
 - (ii) 10 or 15 year certain and life annuity.
 - (iii) Contingent annuity 10 year certain and life annuity.
 - (iv) Flexible installment refund annuity.
 - (v) Installment refund annuity.
- (b) Any distribution not made in a single sum shall be made over a term that does not exceed any one (or a combination) of the following periods:
 - (i) A period certain not extending beyond the life expectancy of the Participant.
 - (ii) A period certain not extending beyond the joint life and last survivor expectancy of the Participant and a designated Beneficiary.

- (iii) The Participant's life.
- (iv) The lives of the Participant and the Participant's designated Beneficiary.
- (c) With respect to any Participant whose Plan Accounts contain no allocations made prior to January 1, 1992, the only available benefit option is a single sum distribution.
- (d) Any distribution form under the Plan other than a lump-sum option shall be eliminated at the "effective time" described below. Thereafter, the only distribution form permitted with respect to amounts in a Participant's Account shall be a lump-sum distribution; provided, however, that any payment in a lump-sum shall be made in a manner that is otherwise identical to the distribution form that is being eliminated, including, by way of illustration, the date or event as of which such distribution form was available and without imposing any condition on eligibility for such lump-sum distribution form that did not apply to such distribution form.

The "effective time" that a distribution form is eliminated as described in paragraph (d) above shall be the earliest of the following dates:

- (1) the 90th day following notice to affected Participants in the form of a summary of material modifications or an updated summary plan description; or
 - (2) January 1, 2004.
- 19. The hardship exception for an "Eligible rollover distribution" in Section 8.7(b) is modified for hardship distributions made effective on and after January 1, 2002 by deleting the phrase "under Code section 401(k)(2)(B)(i)(IV)" and substituting "December 31, 2001" for the date "December 31, 1998."
 - 20. The introductory paragraph of Section 8.9 immediately preceding paragraph (a) is amended to read as follows:
 - **8.9** <u>Distribution Rules</u>. Until the "effective time" as described in Section 8.6(d), the provisions of Section 8.9(a) shall apply to the Distributable Benefit of a Participant who does not elect, and has not previously elected, a life annuity under the Plan. The provisions of Section 8.9(b) shall apply to the Distributable Benefit of a Participant who elects (prior to the "effective time" as described in Section 8.6(d)), or has previously elected, a life annuity under the Plan.

- 21. Section 9.1(a) is amended to permit delegation to the Committee and reads as follows in its entirety:
- **9.1 Plan Administration**. The Plan Administrator has authority to control and manage the operation and administration of the Plan, except that the Plan Administrator may delegate certain administrative and operational powers to the Committee.
- 22. The last paragraph of Section 9.2 shall be amended to read as follows:

"Any action taken in good faith by the Plan Administrator or the Committee in the exercise of authority conferred upon it by this Plan shall be conclusive and binding upon the Participants and their Beneficiaries and any Alternate Payees. All discretionary powers conferred upon the Plan Administrator shall be absolute, but shall be exercised in a uniform and nondiscriminatory manner."

23. Section 9.5 is amended to read as follows in its entirety:

9.5 Reliance Upon Documents and Opinions.

- (a) The Plan Administrator, the Board of Directors, the Company, the members of the Committee, and any Employee of the Company delegated under the provisions hereof to carry out any fiduciary responsible under the Plan ("Delegated Fiduciary"), shall be entitled to rely upon any tables, valuations, computations, estimates, certificates and reports furnished by any consultant, or firm or corporation which employs one or more consultants, upon any opinions furnished by legal counsel, and upon any reports furnished by the Trustee. The Plan Administrator, the Board of Directors, the Company, the members of the Committee, and any Delegated Fiduciary shall not be liable in any manner whatsoever for anything done or action taken or suffered in reliance from any such consultant or firm or corporation which employs one or more consultants, trustee, or counsel.
- (b) Any and all such things done or actions taken or suffered by the Plan Administrator, the Board of Directors, the Company, the members of the Committee, and any Delegated Fiduciary shall be conclusive and binding on all Employees, Participants, Beneficiaries, Alternate Payees and any other persons whomsoever, except as otherwise provided by law.
- (c) The Plan Administrator, the members of the Committee and any Delegated Fiduciary may, but are not required to, rely upon all records of the Company with respect to any matter or thing whatsoever, and may likewise treat those records as conclusive with respect to all Employees, Participants, Beneficiaries, and any other persons whomsoever, except as otherwise provided by law.

- 24. Sections 9.8(a), (b) and (c) are amended to read as follows in their entirety:
- (a) Except as provided in Part 4 of Title I of ERISA, neither the Company, the Board of Directors (or any member thereof), the Committee (or any member thereof) nor the Plan Administrator shall be subject to any liability with respect to his duties under the Plan unless he or it acts fraudulently or in bad faith.
- (b) Neither the Company, the Board of Directors (or any member thereof), the Committee (or any member thereof) nor the Plan Administrator shall be liable for any breach of fiduciary responsibility resulting from the act or omission of any other fiduciary or any person to whom fiduciary responsibilities have been allocated or delegated, except as provided in Part 4 of Title I of ERISA.
- (c) Neither the Company, the Board of Directors (or any member thereof), the Committee (or any member thereof), the Plan Administrator, nor the Trustee shall be liable to the extent relief from liability is provided pursuant to ERISA section 404(c).
- 25. Section 9.10(c)(ii) is amended to read as follows:
 - (ii) Adopting Plan amendments (except as otherwise provided in Section 15.1);
- 26. Sections 9.11(a) and (b) are modified to read as follows:
- (a) To the extent prohibited by law, in administering this Plan, neither the Plan Administrator nor the Committee shall discriminate in favor of any class of Employees and particularly neither shall discriminate in favor of Highly Compensated Employees.
- (b) Neither the Plan Administrator nor the Committee shall cause the Plan to engage in any transaction that constitutes a nonexempt prohibited transaction under Code section 4975(c) or ERISA section 406(a).
- 27. New Sections 9.15 through 9.19 are added to the Plan to read as follows in their entirety:

9.15 Committee Procedure.

(a) A majority of the members of the Committee as constituted at any time shall constitute a quorum, and any action by a majority of the members present at any meeting, or authorized by a majority of the members in writing without a meeting, shall constitute the action of the Committee.

(b) The Committee may designate certain of its members as authorized to execute any document or documents on behalf of the Committee, in which event the Committee shall notify the Trustee of this action and the name or names of the designated members. The Trustee, Company, Board of Directors, Plan Administrator, Participants, Beneficiaries, and any other party dealing with the Committee may accept and rely upon any document executed by the designated members as representing action by the Committee until the Committee shall file with the Trustee a written revocation of the authorization of the designated members.

9.16 Compensation of Committee.

- (a) Members of the Committee shall serve without compensation unless the Board of Directors shall otherwise determine. However, in no event shall any member of the Committee who is an Employee receive compensation from the Plan for his services as a member of the Committee.
 - (b) All members shall be reimbursed for any necessary or appropriate expenditures incurred in the discharge of duties as members of the Committee.
- (c) The compensation or fees, as the case may be, of all officers, agents, counsel, the Trustee, or other persons retained or employed by the Committee shall be fixed by the Committee.
- **9.17** <u>Resignation and Removal of Members</u>. Any member of the Committee may resign at any time by giving written notice to the other members and to the Board of Directors effective as therein stated. Any member of the Committee may, at any time, be removed by the Board of Directors.

9.18 Appointment of Successors.

- (a) Upon the death, resignation, or removal of any Committee member, the Board of Directors may appoint a successor.
- (b) Notice of appointment of a successor member shall be given by the Secretary of the Company in writing to the Trustee and to the members of the Committee.
- **9.19** <u>Records</u>. The Committee shall keep a record of all its proceedings and shall keep, or cause to be kept, all such books, accounts, records or other data as may be necessary or advisable in its judgment for the administration of the Plan and to properly reflect the affairs thereof.

28. The second sentence of Section 13.1 is amended to include a reference to Catch-up Contributions and reads as follows in its entirety:

"To the extent necessary to reduce an excess amount in a Participant's Accounts, the Plan Administrator may distribute such Participant's Elective Deferrals and, if applicable, Catch-up Contributions to the Participant."

- 29. A new Section 14.3(k) is added to describe the loan rules that may be promulgated by the Committee and reads as follows in its entirety:
 - (k) In addition to the foregoing, the loan rules promulgated by the Committee shall include the following:
 - (i) The identity of the person or positions authorized to administer the Participant loan program;
 - (ii) The procedures for applying for a loan;
 - (iii) The basis on which loans will be approved or denied;
 - (iv) Limitations on the types and amounts of loans offered;
 - (v) The procedure for determining a reasonable rate of interest;
 - (vi) The types of collateral which may secure a Participant loan; and $% \left(\mathbf{v}_{1}\right) =\left(\mathbf{v}_{1}\right)$
 - (vii) The events constituting default and the steps that will be taken to preserve Plan assets in the event of such default.
 - (viii) Notwithstanding the foregoing provisions of this section 14.3, loan repayments may be suspended in accordance with Code section 414(u)(4).
- 30. The first paragraph of Section 15.1 is amended to authorize the Committee to adopt certain Plan amendments, and reads as follows in its entirety:

"The Board of Directors in its sole discretion may at any time, and from time to time, amend the Plan by an instrument in writing executed in the name of the Company by an officer or officers duly authorized to execute such instrument, and delivered to the applicable Trustee. Notwithstanding the foregoing, effective as of January 1, 2002 the Committee shall have the power to amend the Plan as described in the preceding sentence to (i) comply with changes required by the IRS as a

condition to the issuance of a favorable determination of the qualified status of the Plan under Code section 401(a) or otherwise preserve its status as a qualified plan; or (ii) provide rules of administrative convenience or clarify Plan operations so long as such administrative or clarifying changes would not cause the Company to incur additional material costs to operate the Plan. However, except as provided in Section 15.2 no amendment shall be made at any time, the effect of which would be:"

- 31. Section 16.3(b) is amended to read as follows in its entirety:
- (b) Key Employee: Any Employee or former Employee (and the Beneficiaries of such Employees) who, pursuant to the rules of Code section 416(i) and the Regulations thereunder, is or was:
 - (i) An officer of the Company having an annual Compensation greater than 50% of the dollar limitation under Code section 415(b)(1)(A) (for Determination Dates on and after January 1, 2002, Compensation of greater than \$130,000 (as adjusted under Code section 416(i)(1)));
 - (ii) For Determination Dates on or before December 31, 2001, one of the ten Employees having annual Compensation from the Company of more than the dollar limitation under Code section 415(c)(1)(A), and owning (or considered as owning) under Code section 318 the largest interest in the Company;
 - (iii) A 5% Owner of the Company; or
 - (iv) A 1% Owner of the company having Annual Compensation from the Company of more than \$150,000.

The determination period is the Plan Year containing the Determination Date and the four preceding Plan Years. Effective January 1, 2002, the five-year look-back period is reduced to a one-year period ending on the Determination Date, with the exception of any distribution made for a reason other than separation from service, death, or disability.

- 32. A new Section 17.4(c) is added to the Plan to permit paperless communications and reads as follows in its entirety:
- (c) Notwithstanding any provision of the Plan to the contrary, including the preceding paragraphs (a) and (b) of this section 17.4, the Committee in its absolute discretion may accept or require applications, notices, designations, elections and other communications ("communication") given by facsimile, telex, telegram, telephone or any

form of electronic communication that the Committee reasonably believes in good faith to be genuine. If the Committee chooses to accept one or more alternative methods of communication, the Company, any Affiliated Company, any Participant or Beneficiary or any other person providing communication to the Committee will be required to follow reasonable procedures adopted by the Committee for written confirmation. In addition, the Committee may record oral instructions. If any person providing communication to the Committee fails or refuses to comply with the Committee's confirmation procedures, the Committee will be entitled to refuse to comply with such communications without incurring any liability for such refusal.

33. Effective as of January 1, 2002, Section 17.9 is amended to add the words ", the Committee" immediately following the words "the Plan Administrator."

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

Amendment No. 3

Effective as of January 1, 2002, or such other dates as noted herein, the AMSEC Employees 401(k) Profit Sharing Plan (the "Plan") is hereby amended as follows:

- 1. Subsection 2.19(a) is amended to read as follows in its entirety:
- (a) For purposes of determining the allocation of Profit Sharing Contributions pursuant to Sections 4.3 and 6.10(c) "Compensation" shall mean the amounts described in subsection (b) below paid by the Company during a Plan Year by reason of services performed by an Employee less any compensation paid by reason of services performed during any period in which the Employee is not a Participant under this Plan, but including any elective contributions that are made by the Company on behalf of an Employee pursuant to an arrangement described in Code sections 125, 132(f)(4), 402(a)(8), 402(h) or 403(b). For purposes of determining the allocation of a Participant's permissible Elective Deferrals under Sections 5.2 and 6.10(a) and Matching Contributions under Sections 5.3 and 6.10(b) "Compensation" shall mean the amounts described above with regard to Profit Sharing Contributions but also shall exclude bonuses.
- 2. Subsection 2.60(c) is amended to read as follows in its entirety:
- (c) The Participant's termination of employment with the Company or an Affiliated Company. Notwithstanding the foregoing, a Participant may elect, pursuant to procedures established by the Committee (which procedures may provide that no affirmative election is treated as an election to defer distribution), to defer distribution

until the first to occur of (i) five years from the date of termination of employment or (ii) April 1 following the calendar year in which the Participant attains age 70-1/2.

In the case of a 5% Owner, the Required Payment Commencement Date shall be April 1 following the calendar year in which the Participant attains age 70-1/2, whether or not the Participant has retired. If a Participant becomes a 5% Owner after attaining age 70-1/2, the Required Payment Commencement Date shall not be later than the last day of the calendar year in which the Participant becomes a 5% Owner.

- 3. Section 4.6 is amended by substituting the phrase "the Committee" for each occurrence of the phrase "the Plan Administrator".
- 4. Effective on and after April 1, 2002, Section 8.8 is amended to read as follows in its entirety:

8.8 Timing on Distributions.

- (a) <u>Required Distributions Under the Plan</u>. The Participant's ability to elect the form or timing of payment (or the Beneficiary's ability, in the event of the Participant's death before the date on which distributions commence) shall be subject to the following rules:
 - (i) <u>Distributable Benefit Under \$5,000</u>. Notwithstanding any other provision of the Plan, if the value of the Participant's Distributable Benefit determined as of the Participant's termination date; death or any subsequent Valuation Date does not exceed \$5,000, the Participant (or Beneficiary in the event of the Participant's death) shall receive a single sum distribution of the entire value as soon as practicable following the Participant's termination of employment, receipt of notice of Participant's death or such subsequent Valuation Date, whichever is applicable. If the Participant's Distributable Benefit is zero, the Participant shall be deemed to have received a distribution of his entire Distributable Benefit. Such distribution shall not require consent by the Participant or the Participant's spouse.
 - (ii) <u>Distributable Benefit Upon Termination of Employment</u>. If the value of the Participant's Distributable Benefit determined as of the date the Participant terminates employment and at each subsequent Valuation Date exceeds \$5,000 no distribution shall be made to the Participant without the Participant's consent before the Participant attains Normal Retirement Age, or distributions are made at the termination of the Plan. Unless the Participant elects to defer distribution to the extent authorized, payment shall be made or commence no later than sixty (60) days after the close of the Plan Year in which (A) the Participant attains Normal Retirement Age or (B) the Participant terminates employment with the Company, whichever occurs later. The failure of a Participant to consent to a distribution prior to such time shall be deemed to be an election to defer commencement of payment. In no event shall distributions to a Participant commence later than the Participant's Required Payment Commencement Date

- (iii) <u>Distributable Benefit Upon Death</u>. If the Participant's Distributable Benefit determined as of the date of the Participant's death and at each subsequent Valuation Date exceeds \$5,000, distribution of the Participant's entire benefit shall be completed to the Participant's Designated Beneficiary or estate, as applicable, by December 31 of the calendar year containing the fifth anniversary of the Participant's death except to the extent that the Designated Beneficiary elects, prior to April 1, 2002, to receive distributions in accordance with subparagraph (A) or (B) below.
 - (A) If any portion of the Participant's benefits are payable to a Designated Beneficiary, distributions may be made over the Life Expectancy of the Designated Beneficiary if distributions are made or commence on or before December 31 of the calendar year which immediately follows the calendar year in which the Participant dies. Prior to April 1, 2002, the Designated Beneficiary may elect one of the annuity options described in Section 8.6.
 - (B) If the Designated Beneficiary is the Participant's surviving spouse, the date distributions are required to be made or begin in accordance with subparagraph (A) above shall not be earlier than the later of (1) December 31 of the calendar year immediately following the calendar year in which the Participant dies and (2) December 31 of the calendar year in which the Participant would have attained age seventy and one-half (70-1/2). Prior to April 1, 2002, the Beneficiary may select from the annuity options described in Section 8.6.
 - (C) <u>Distributable Benefit After Death</u>. If the Participant dies after distributions under one of the annuity options have begun, the remaining payments shall continue to be distributed at least as rapidly as under the method of distribution being used prior to the Participant's death.
- (iv) <u>Distributable Benefit Upon Required Payment Commencement Date</u>. If the Participant elects to receive his or her interest as a single sum, such Participant's entire Account(s) must be distributed no later than his Required Payment Commencement Date. If prior to April 1, 2002, the Participant commenced receipt of his or her Distributable Benefit in a form other than a single sum, on or after the Required Payment Commencement Date distributions shall be made in accordance with Treasury regulation section 1.401(a)(9), including the minimum distribution incidental benefit rules of section 1.401(a)(9)-2 as may be amended.
- (v) <u>Distributable Benefit to Surviving Spouse</u>. If the surviving spouse dies after the Participant, but before payments to the spouse begin, the above provisions shall apply as if the surviving spouse were the Participant. Any amount paid to a child of the Participant shall be treated as if it had been paid to the surviving spouse if the amount becomes payable to the surviving spouse when the child reaches the age of majority.

- (b) <u>Commencement of Benefits</u>. Notwithstanding the prior provisions of this Article VIII to the contrary, a Participant's Plan Account and Matching Account Balances shall not be distributable earlier than:
 - (i) termination of the Participant's employment with the Company on or after the date he or she attains Normal Retirement Date or by reason of death;
 - (ii) a determination of Disability in accordance with Section 2.23;
 - (iii) termination of the Participant's employment with the Company and Affiliated Companies for any reason other than those specified in paragraph (i) above, but only if such termination constitutes a separation from service (within the meaning of Code section 401(k)(2)(B)(i)(I);
 - (iv) the Committee's approval of the Participant's application for a hardship or in-service withdrawal, but only to the extent provided in Section 8.7;
 - (v) in accordance with and to the limited extent provided in Sections 5.2(d) and 5.6(a);
 - (vi) if and to the extent permitted by Code section 401(k)(10) in connection with the Company's disposition of corporate assets or a subsidiary; and
 - (vii) the creation or recognition of an Alternate Payee's right to all or a portion of a Participant's Plan and Matching Contribution Accounts under a domestic relations order which the Committee determines is a qualified domestic relations order as provided in Section 14.2, but only as to the portion which is payable to the Alternate Payee.
- 5. Effective on and after April 1, 2002 subsection 14.3(e) is amended to read as follows in its entirety:
- (e) In the event the Participant defaults on the loan and Participant's Accounts are security for the loan, the Account balances will not be used to satisfy the loan obligation prior to the earlier of the Participant's termination of employment with the Company or an event resulting in a permissible distribution of his Accounts under the Plan.
- 6. Effective on and after April 1, 2002, Subsection 14.3(j) is deleted.

EGTRRA AMENDMENT TO

AMSEC EMPLOYEES 401(K) PROFIT SHARING PLAN

A. AMENDMENT OF THE PLAN FOR EGTRRA

- 1. Adoption and Effective Date of Amendment. This Amendment of the Plan is adopted to reflect certain provisions of the Economic Growth and Tax Relief Reconciliation Act of 2001 ("EGTRRA"). This Amendment is intended as good faith compliance with the requirements of EGTRRA and is to be construed in accordance with EGTRRA and guidance issued thereunder. Except as otherwise provided, this Amendment shall be effective as of January 1, 2002 and unless otherwise extended past the "sunset" or otherwise made permanent pursuant to EGTRRA, this amendment will expire as of December 31, 2010.
- 2. **Supersession of Inconsistent Provisions.** This Amendment shall supersede the provisions of the Plan to the extent those provisions are inconsistent with the provisions of this Amendment.

B. SECTIONS 2.62. ROLLOVERS FROM OTHER PLANS

The Plan will accept a direct rollover of an eligible rollover distribution from a qualified plan described in Code sections 401(a) or 403(a), excluding after-tax employee contributions, from a qualified plan described in Code sections 401(a) or 403(a), from a qualified plan or annuity contract described in Code sections 403(a) or (b), excluding after-tax employee contributions, from an eligible plan under Code section 457(b) which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state, and from the portion of a distribution from an individual retirement account or annuity that is described in Code sections 408(a) or (b) funded exclusively with a distribution from another retirement plan qualified under Code section 401(a), 403(a) or (b), or 457(b), and earnings thereon (also known as a conduit IRA) that is eligible to be rolled over and would otherwise be includible in gross income.

C. SECTIONS 8.2. DISTRIBUTION UPON SEVERANCE FROM EMPLOYMENT

- 1. New Distributable Event. This amendment to Section 8.2 of the Plan is intended to reflect the elimination of the "same desk rule". A Participant's elective deferrals, qualified nonelective contributions, qualified matching contributions, and earnings attributable to these contributions shall be distributed on account of the Participant's severance from employment. However, such a distribution shall be subject to the other provisions of the Plan regarding distributions, other than provisions that require a separation from service before such amounts may be distributed.
- 2. Effective Date. This change to section 8.2 shall apply to Plan Years and Limitation Years beginning on and after January 1, 2002.

D. SECTION 8.7. DIRECT ROLLOVERS OF PLAN DISTRIBUTIONS

- **1. Effective Date.** This section shall apply to distributions made after December 31, 2001.
- **2. Modification of definition of Eligible Retirement Plan.** For purposes of the direct rollover provisions of section 8.7, the Plan is amended to add the following to the end thereof as follows:
 - "An eligible retirement plan shall also mean an annuity contract described in Code section 403(b) and an eligible plan under Code section 457(b) which is maintained by the state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and which agrees to separately account for amounts transferred into such plan from this Plan. The definition of eligible retirement plan shall also apply in the case of a distribution to a surviving spouse, or to a spouse or former spouse who is the Alternate Payee under a qualified domestic relations order, as defined in Code section 414(p)."
- 3. Modification of definition of Eligible Rollover Distribution to include After-Tax Employee Contributions. For purposes of the direct rollover provision in section 8.7(b) of the Plan, a portion of a distribution shall not fail to be an eligible rollover distribution merely because the portion consists of after-tax employee contributions which are not includible in gross income. However, such portion may be transferred only to an individual retirement account or annuity described in Code sections 408(a) or (b), or to a qualified defined contribution plan described in Code sections 401(a) or 403(a) that agrees to separately account for amounts so transferred, including separately accounting for the portion of such distribution which is includible in gross income and the portion of such distribution which is not so includible.

E. SECTION 8.8. MINIMUM DISTRIBUTION REQUIREMENTS.

A new subsection 8.8(c) is added to read as follows:

- (c) Required Minimum Distributions under Code section 401(a)(9).
 - (i) General Rules.
 - A. <u>Effective Date</u>. The provisions of this Subsection 8.8(c) will apply for purposes of determining required minimum distributions for calendar years beginning with 2003 calendar year.
 - B. <u>Precedence</u>. Except as earlier or larger distributions may be required under the terms of the Plan, the requirements of this Subsection 8.8(c) will take precedence over any inconsistent provisions of the Plan.

- C. <u>Requirements of Treasury Regulations Incorporated</u>. All distributions required under this Subsection 8.8(c) will be determined and made in accordance with the Treasury regulations under Code section 401(a)(9).
- D. <u>TEFRA Section 242(b)(2)Elections</u>. Notwithstanding the other provisions of this Subsection 8.8(c), distributions may be made under a designation made before January 1, 1984, in accordance with section 242(b)(2) of the Tax Equity and Fiscal Responsibility Act (TEFRA) and the provisions of the Plan that relate to section 242(b)(2) of TEFRA.

(ii) <u>Time And Manner Of Distribution</u>.

- A. Required Payment Commencement Date. The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's Required Payment Commencement Date.
- B. <u>Forms of Distribution</u>. Unless the Participant's interest is distributed in a single sum on or before the Required Payment Commencement Date as of the first Distribution Calendar Year, distributions will be made in accordance with this section 8.8(c).
- C. <u>Lifetime Distributions</u>. During the Participant's lifetime, the minimum amount that will be distributed for each Distribution Calendar Year is the lesser of:
 - 1. the quotient obtained by dividing the Participant's Distributable Benefit by the distribution period in the uniform lifetime table set forth in section 1.401(a)(9)-9 of the Treasury regulations, using the Participant's age as of the Participant's birthday in the Distribution Calendar Year; or
 - 2. if the Participant's sole Designated Beneficiary for the Distribution Calendar Year is the Participant's spouse, the quotient obtained by dividing the Participant's Distributable Benefit by the number in the joint and last survivor table set forth in section 1.401(a)(9)-9 of the Treasury regulations, using the Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the Distribution Calendar Year.
 - 3. Required minimum distributions will be determined under this section 8.8(c)(ii)(C) beginning with the first Distribution Calendar Year and up to and including the Distribution Calendar Year that includes the Participant's date of death.

D. <u>Distributions Upon Death</u>. Upon the Participant's death, distributions of the Participant's Distributable Benefit shall occur in accordance with Subsections 8.8(a)(iii), (v) and (vi).

E. Definitions.

- 1. <u>Designated Beneficiary</u>. The individual who is designated as the Beneficiary under section 8.10 of the Plan and is the Designated Beneficiary under Code section 401(a)(9) and section 1.401(a)(9)-1, Q&A-4, of the Treasury regulations.
- 2. <u>Distribution Calendar Year</u>. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first Distribution Calendar Year is the calendar year immediately preceding the calendar year which contains the Participant's Required Payment Commencement Date. For distributions beginning after the Participant's death, the first Distribution Calendar Year is the calendar year in which distributions are required to begin under Subsection 8.8(a) of the Plan. The required minimum distribution for the Participant's first Distribution Calendar Year will be made on or before the Participant's Required Payment Commencement Date. The required minimum distribution for other Distribution Calendar Years, including the required minimum distribution for the Distribution Calendar Year in which the Participant's Required Payment Commencement Date occurs, will be made on or before December 31 of that Distribution Calendar Year.
- 3. <u>Life Expectancy.</u> Life expectancy as computed by use of the single life table in section 1.401(a)(9)-9 of the Treasury regulations.
- 4. <u>Participant's Distributable Benefit</u>. The Distributable Benefit as of the last Valuation Date in the calendar year immediately preceding the Distribution Calendar Year (valuation calendar year) increased by the amount of any contributions made and allocated or forfeitures allocated to the Participant's Account as of dates in the valuation calendar year after the Valuation Date and decreased by distributions made in the valuation

calendar year after the Valuation Date. The Distributable Benefit for the valuation calendar year includes any amounts rolled over or transferred to the Plan either in the valuation calendar year or in the Distribution Calendar Year if distributed or transferred in the valuation calendar year.

5. Required Payment Commencement Date. The date specified in section 2.60 of the Plan.

F. SECTION 16.6. MODIFICATION OF TOP-HEAVY RULES

1. Effective Date. The provision of this amendment to section 16.6 of the Plan shall apply for Plan Years beginning after December 31, 2001 for purposes of determining whether the Plan satisfies the minimum benefits requirements of Code section 416(c) for such years. This section amends section 16.6 of the Plan.

2. Minimum Benefits.

Matching Contributions. Company Matching Contributions shall be taken into account for purposes of satisfying the minimum contribution requirements of Code section 416(c)(2) and section 16.6 of the Plan. The preceding sentence shall apply with respect to Matching Contributions under the Plan or, if the Plan provides that the minimum contribution requirement shall be met in another plan, such other plan. Company Matching Contributions that are used to satisfy the minimum contribution requirements shall be treated as Matching Contributions for purposes of the actual contribution percentage test and other requirements of Code section 401(m).

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

Amendment No. 6

Effective as of June 1, 2003, or such other dates as noted herein, the AMSEC Employees 401(k) Profit Sharing Plan (the "AMSEC Plan") is hereby amended as follows:

- 1. Subsection 2.1 is amended by adding the following new Subsections (k) through (n) to the end thereof:
- (k) PCI Transfer Deferral Account maintained for each Participant and each former PCI employee who formerly participated in the PCI Plan and had an elective deferral account in that plan which was transferred to the Plan pursuant to Section 5.6.
- (l) PCI Transfer Matching Account maintained for each Participant and each former PCI employee who formerly participated in the PCI Plan and had a matching account in that plan which was transferred to the Plan pursuant to Section 5.6.
- (m) PCI Transfer Profit Sharing Account maintained for each Participant and each former PCI employee who formerly participated in the PCI Plan and had a profit sharing account in that Plan which was transferred to the Plan pursuant to Section 5.6.

- (n) PCI Transfer Rollover Account maintained for each Participant and each former PCI employee who formerly participated in the PCI Plan and had a rollover account in that plan which was transferred to the Plan pursuant to Section 5.6.
- 2. Effective on and after October 1, 2003 Subsection 2.19(a) is amended by adding the following to the end thereof:
- (a) Notwithstanding the provisions of subsections (a) and (b) to the contrary, for purposes of determining the allocation of Profit Sharing Contributions pursuant to Sections 4.3 and 6.10(e) for the Plan Year ending December 31, 2003, Compensation for a PCI Legacy Employee will only include Compensation for the period beginning October 1, 2003 and ending December 31, 2003.
- 3. Section 2.24 is amended to read as follows in its entirety:
- 2.24 <u>Distributable Benefit</u>. "Distributable Benefit" shall mean the interest of a Participant in this Plan that is determined and distributable to him in accordance with the provisions of Article VIII and which shall include his Vested Interest in his Deferral Account, Company Contribution Account, Voluntary Account, Rollover Account, IRP Account, NNS Transfer Deferral Account, NNS Transfer After-Tax Account, NNS Transfer Rollover Account, NNS Transfer Employee Account, PCI Transfer Deferral Account, PCI Transfer Rollover Account, PCI Transfer Matching Account and PCI Transfer Profit Sharing Account, if any. In the case of an Alternate Payee, the Distributable Benefit shall mean the Vested Interest in the Alternate Payee Account.
- 4. Section 2.29 is amended to read as follows in its entirety:
- 2.29 Employment Commencement Date. "Employment Commencement Date" shall mean each of the following:
- (a) The date on which an Employee first performs an Hour of Service in any capacity for the Company or an Affiliated Company with respect to which the Employee is compensated or is entitled to compensation by the Company or the Affiliated Company.
- (b) In the case of an Employee whose employment is terminated and who is subsequently reemployed by the Company or an Affiliated Company, the term "Employment Commencement Date" shall also mean the first day following the termination of employment on which the Employee performs an Hour of Service for the Company or an Affiliated Company with respect to which he is compensated or entitled to compensation by the Company or Affiliated Company.
- (c) Unless the Board of Directors or its delegate shall expressly determine otherwise, an Employee shall not, for the purposes of determining his Employment Commencement Date, be deemed to have commenced employment with an Affiliated Company prior to the effective date on which the entity became an Affiliated Company.

- (d) Notwithstanding the forgoing, with respect to employees of PCI the Employment Commencement Date shall be:
 - (i) the date described in subsection (a) above if such Employee was hired by PCI on or after June 2, 2003; or
 - (ii) October 1, 2003 for all PCI Legacy Employees.
- 5. Effective on and after January 1, 2004, Section 2.45 "Normal Retirement Age" is amended by substituting age 59-1/2 for age 65.
- 6. A new Section 2.54A is added to read as follows in its entirety:
 - 2.54A PCI. "PCI" shall mean Planning Consultants, Inc., which became a wholly-owned subsidiary of the Company effective as of May 23, 2003.
- 7. Effective on and after October 1, 2003 a new Section 2.54B is added to read as follows in its entirety:
- **2.54B** <u>PCI Legacy Employee</u>. "PCI Legacy Employee" shall mean an Employee who was actively employed by PCI effective as of June 1, 2003, continued in employment with PCI on October 1, 2003, and is otherwise eligible to receive benefits under the Plan.
- 8. A new Section 2.54C is added to read as follows in its entirety:
- **2.54C** <u>PCI Plan (the "PCI Plan")</u>. "PCI Plan" shall mean the Planning Consultants, Inc. Savings Plan which was frozen effective on and after September 30, 2003 and merged into this Plan.
- 9. A new Section 2.54D is added to read as follows in its entirety:
- **2.54D** <u>PCI Transfer Deferral Account.</u> "PCI Transfer Deferral Account" shall mean the Account opened up and maintained for each Participant and each former PCI employee who formerly participated in the PCI Plan and had an elective deferral account in that plan which was transferred to the Plan.
- 10. A new Section 2.54E is added to read as follows in its entirety:
- **2.54E** <u>PCI Transfer Matching Account</u>. "PCI Transfer Employer Account" shall mean the Account opened up and maintained for each Participant and each former PCI employee who formerly participated in the PCI Plan and had a matching account in that plan which was transferred to the Plan.

- 11. A new Section 2.54F is added to read as follows in its entirety:
- **2.54F PCI Transfer Profit Sharing Account.** "PCI Transfer Profit Sharing Account" shall mean the Account opened up and maintained for each Participant and each former PCI employee who formerly participated in the PCI Plan and had a profit sharing account in that plan which was transferred to the Plan.
- 12. A new Section 2.54G is added to read as follows in its entirety:
- **2.54G** <u>PCI Transfer Rollover Account</u>. "PCI Transfer Rollover Account" shall mean the Account opened up and maintained for each Participant and each former PCI employee who formerly participated in the PCI Plan and had a rollover account in that plan which was transferred to the Plan.
- 13. Section 2.67 is amended to read as follows in its entirety:
 - 2.67 Vested Interest. "Vested Interest" shall mean the following:
 - (a) the nonforfeitable interest of a Participant in his Profit Sharing, Matching Contribution, NNS Transfer Employee Accounts, PCI Transfer Matching Account and PCI Transfer Profit Sharing Account, if any, which has become vested in accordance with Article VII; and
 - (b) the entire balance of his Deferral Account, Voluntary Account, Rollover Account, NNS Transfer Deferral Account, NNS Transfer After-Tax Account, NNS Transfer Rollover Account, PCI Transfer Deferral Account and PCI Transfer Rollover Account, if any; and
 - (c) the Vested Interest of an Alternate Payee in his Alternate Payee Account shall be determined as set forth in Article VII.
- 14. Section 4.3 is amended to read as follows in its entirety:

4.3 Company Contributions.

(a) The Company shall pay over to the Trustee in cash an amount equal to the aggregate of the amounts that each Participant elects (pursuant to Article V) to have contributed to the Plan on his behalf as Elective Deferrals and, if applicable, Catch-up Contributions. Such contributions shall be made no later than 15 business days after the month that such amounts are withheld from the Participant's Compensation. If necessary or appropriate, the Company, in its sole discretion, may make contributions to the Trust in excess of the amounts required under the preceding sentence as Profit Sharing Contributions or Additional Company Contributions. In addition, the Company shall pay over to the Trustee Matching Contributions in the amount determined under Section 5.3.

- (b) With respect to Participants, who (i) formerly participated in the NNS 401(k) Plan or (ii) are PCI Legacy Employees, the Company, in its sole discretion, may make Profit Sharing Contributions, Additional Company Contributions, or Matching Contributions to those Participants or PCI Legacy Employees alone and in an amount that is different than the amount of similar Contributions available to other Participants.
- (c) Company Contributions shall be made no later than the due date for filing the federal income tax return (including extensions) of the Company for the taxable year with respect to which the contribution is made. The Company may, in its discretion, make contributions at any time to defray Plan expenses as provided in Section 6.13.

15. Subsection 5.2(a) is amended to read as follows in its entirety:

- (a) The amount of an individual's Compensation that may be deferred subject to the election provided in Section 5.1 shall be a percentage of the individual's Compensation, not to exceed the maximum percentage established by the Plan Administrator of 25%, except with respect to Participants who (i) formerly participated in the NNS 401(k) Plan or (ii) are PCI Legacy Employees.
 - (i) With regard to such Participants who formerly participated in the NNS 401(k) Plan, the amount of such Employees' Compensation that may be deferred, subject to the election provided in Section 5.1, shall be an amount equal to not less than one percent (1%) or such percentage of Compensation greater than twelve percent (12%) as may be specified from time to time by the Plan Administrator, provided that the percentage rate for determining such contributions shall be a whole number.
 - (ii) With regard to Participants who are PCI Legacy Employees, the amount of such Employees' Compensation that may be deferred, subject to the election provided in Section 5.1, shall be an amount equal to not less than one percent (1%) or such percentage of Compensation greater than ninety percent (90%) as may be specified from time to time by the Plan Administrator, provided that the percentage rate for determining such contributions shall be a whole number.

16. Subsection 5.3(a) is amended to read as follows in its entirety:

(a) Effective for Plan Years beginning on or after January 1, 1990, the Company may, in its discretion, make a Matching Contribution in an amount equal to a percentage (determined each year in the Company's discretion) for each \$1 of a Participant's Elective Deferrals up to a maximum of \$5,000 of such Elective Deferrals, subject to the limitations specified in Article XIII, except with respect to Participants who (i) formerly participated in the NNS 401(k) Plan or (ii) are PCI Legacy Employees. With regard to such Participants, the Company, in its sole discretion, may make a Matching Contribution to those Participants alone and in an amount which is different than that available to other Participants, subject to the limitations specified in Article XIII.

- 17. Subsection 5.3(b) is amended to read as follows in its entirety:
- (b) The contribution described in Section 5.3(a) shall be paid to the Trustee on the last day of the calendar quarter, or as soon as possible thereafter, but in any event no later than the date which is prescribed by law for filing the Company's federal income tax return, including any extensions for the taxable year with respect to which such contributions are made. A Matching Contribution shall be allocated as provided in Section 6.10(b) and made to each Participant, including former NNS employees and PCI Legacy Employees, who are performing duties as a wage earning Employee for the Company as of the last day of a calendar quarter and any Participant who dies, retires or becomes disabled during such quarter. Notwithstanding the above, for the period of January 1, 1995 through June 30, 1995, Matching Contributions will be paid to the Trustee and allocated to each Participant's account as of June 30, 1995 for those Participants who are employed on June 30, 1995.
- 18. Subsections 5.6(a) and (b) are amended and a new Subsection (c) is added to read as follows in its entirety:
- (a) If it is made to this Plan from a plan which contained Protected Benefits, such Protected Benefits shall continue to be maintained with respect to the amounts transferred whether in the Participant's NNS Transfer Accounts, PCI Transfer Accounts or in the Participant's other Accounts, depending upon the character of the contribution under the transferee plan.
- (b) To the extent required by Code section 411(d)(6), and pursuant to procedures established by the Company, an amount transferred to the Participant's Accounts, including a Participant's (i) NNS Transfer Accounts or (ii) PCI Transfer Accounts, if any, pursuant to this Section 5.6 shall retain the Protected Benefits that were applicable to such Account balance prior to the transfer to the Plan. Notwithstanding the foregoing, the Company reserves the right to eliminate all or any of such Protected Benefits if and to the extent permitted by Code section 411(d)(6) and any regulations or guidance promulgated thereunder.
- (c) Effective on and after October 1, 2003 a Participant or the former employee of an acquired entity whose accounts in a defined contribution plan are transferred to the Plan as the result of an acquisition or merger shall be entitled to make a withdrawal from his Transfer Rollover Account at any time, subject to procedures established by the Plan Administrator.
- 19. A new Section 6.9A is added to read as follows in its entirety:
- **6.9A** <u>PCI Transfer Deferral Account</u>. The Company shall open and maintain a separate PCI Transfer Deferral Account for each Participant and each former PCI employee who formerly participated in the PCI Plan and had an elective deferral account in that plan which was transferred to the Plan.

- 20. A new Section 6.9B is added to read as follows in its entirety:
- **6.9B** <u>PCI Transfer Matching Account</u>. The Company shall open and maintain a separate PCI Transfer Matching Account for each Participant and each former PCI employee who formerly participated in the PCI Plan and had a matching account in that plan which was transferred to the Plan.
- 21. A new Section 6.9C is added to read as follows in its entirety:
- **6.9C PCI Transfer Profit Sharing Account.** The Company shall open and maintain a separate PCI Transfer Profit Sharing Account for each Participant and each former PCI employee who formerly participated in the PCI Plan and had a profit sharing account in that plan which was transferred to the Plan.
- 22. A new Section 6.9D is added to read as follows in its entirety:
- **6.9D** <u>PCI Transfer Rollover Account</u>. The Company shall open and maintain a separate PCI Transfer Rollover Account for each Participant and each former PCI employee who formerly participated in the PCI Plan and had a rollover account in that plan which was transferred to the Plan.
- 23. Subsection 6.10(c) is amended to read as follows in its entirety:
- (c) Any Company Contributions which constitute Profit Sharing Contributions shall be allocated to the Company Contribution Account of each Participant who is performing duties as a wage-earning Employee of the Company on the last day of the Plan Year. An allocation also shall be made for any Participant who dies, retires or becomes disabled during the Plan Year. The Profit Sharing Contribution shall be allocated to each Participant eligible to receive a share of such contribution (except for those Participants who (i) formerly participated in the NNS 401(k) Plan or (ii) are PCI Legacy Employees) in the proportion that the Compensation paid to each such Participants.
 - (i) With respect to Participants who formerly participated in the NNS 401(k) Plan and whose accounts in that plan were transferred to the Plan pursuant to Section 5.6, any Profit Sharing Contribution shall be allocated among each such Participant who is otherwise eligible to receive a share of such contribution in the proportion that the Compensation paid to each such Participant during the Plan Year bears to the Compensation paid to all such Participants who formerly participated in the NNS 401(k) Plan.
 - (ii) With respect to Participants who formerly participated in the PCI Plan and whose accounts in that plan were transferred to the Plan pursuant to Section 5.6, any Profit Sharing Contribution shall be allocated among each such Participant who is otherwise eligible to receive a share of such contribution in the proportion that the Compensation paid to each such Participant during the Plan Year bears to the Compensation paid to all such Participants who formerly participated in the PCI Plan.

- 24. Section 6.12 is amended by adding a new Subsection (f) to the end thereof:
- (f) For the Plan Year ending December 31, 2003, any forfeitures transferred from the PCI Plan to the Plan as a result of the merger shall be used to offset PCI's obligation to make Company Contributions and shall be allocated to the Matching Contribution Accounts of those PCI Legacy Employees who are entitled to received an allocation of Matching Contributions for the 2003 Plan Year in the manner prescribed in Section 6.10(b). Pending allocation, such forfeitures shall be accounted for in the manner prescribed in Section 6.12(d). No such forfeitures shall be allocated to the Voluntary Account, Rollover Account, Deferral Account or Alternate Payee Account of any PCI Legacy Employee.
- 25. Section 7.6 is amended by adding a new Subsection (c) to the end thereof:
- (c) Notwithstanding the above, the Vested Interest in the PCI Transfer Matching Account and/or the PCI Transfer Profit Sharing Account, of each PCI Legacy Employee and any former PCI employee who was never an Employee of the Company, in his Transfer Matching Account and his PCI Transfer Profit Sharing Account, if any, shall be one hundred percent (100%).
- 26. Subsection 8.3(a) is amended to read as follows in its entirety:
- (a) A Participant (except a Participant who formerly participated in either (i) the NNS 401(k) Plan or the (ii) PCI Plan) shall be entitled to make a withdrawal from his Voluntary Account and, effective on and after October 1, 2003, from his Rollover Account at any time, subject to procedures established by the Plan Administrator.
 - (i) A Participant who formerly participated in the NNS 401(k) Plan and had accounts in that plan transferred to the Plan, shall be entitled to make a withdrawal from his NNS Transfer After-Tax Account, NNS Transfer Employer Account, or NNS Transfer Rollover Account at any time, subject to procedures established by the Plan Administrator.
 - (ii) A Participant who formerly participated in the PCI Plan and had accounts in that plan transferred to the Plan, shall be entitled to make a withdrawal from his PCI Transfer Matching Account, PCI Transfer Profit Sharing Account, or PCI Transfer Rollover Account at any time, subject to procedures established by the Plan Administrator.
 - (iii) Notwithstanding the foregoing, a former PCI employee whose accounts in the PCI Plan were transferred to this Plan as a result of the merger but was

never a Participant in this Plan shall be entitled to make a withdrawal from his PCI Transfer Deferral Account, PCI Transfer Matching Account, PCI Transfer Profit Sharing Account or PCI Transfer Rollover Account at any time, subject to procedures established by the Plan Administrator.

- 27. The first paragraph of Subsection 8.3(b) is amended to read as follows:
- (b) Further, a Participant may withdraw amounts from his Deferral Account and, if applicable, his NNS Transfer Deferral Account or his PCI Transfer Deferral Account, that are in each case attributable to Elective Deferrals, excluding all earnings attributable thereto, upon incurring a financial hardship as determined by the Plan Administrator. A withdrawal will be deemed made on account of financial hardship only if the distribution is made on account of an immediate and heavy financial need and is necessary to satisfy such financial need. Any withdrawal made pursuant to this paragraph shall be made in accordance with rules prescribed by the Plan Administrator that are consistent with regulations under Code section 401(k), and the following rules:
- 28. Section 8.3(b)(i) is modified by deleting paragraph (E) effective for hardship withdrawal applications submitted on and after October 1, 2003.
- 29. Effective on and after January 1, 2003, the title of Section 8.4 is amended by adding the phrase "In-Service" in front of the phrase "Withdrawals after age 59-1/2".
 - 30. New Subsections 8.6(e) and (f) are added to read as follows:
 - (e) Subject to Section 8.6(f) and solely to the extent required by Code section 411(d)(6), each PCI Legacy Employee may receive his Distributable Benefit in one of the following forms subject to the provisions of Sections 8.7 and 8.8:
 - (i) a single sum; or
 - (ii) installments.

Any distribution to a PCI Legacy Employee not made in a single sum shall be made in accordance with subsection 8.6(b) above.

(f) The installment form of benefit described in Section 8.6(e)(ii) above shall be eliminated at the "effective time" described below. Thereafter, the only distribution form permitted with respect to amounts in a PCI Legacy Employee's Account shall be a lump-sum distribution; provided, however, that any payment in a lump-sum shall be made in a manner that is otherwise identical to the distribution form that is being eliminated, including, by way of illustration, the date or event as of which such distribution form was available and without imposing any condition on eligibility for such lump-sum distribution form that did not apply to such distribution form.

The "effective time" that a distribution form is eliminated as described in paragraph (e) above shall be January 1, 2004 or, if later, the earliest of the following dates:

- (1) the 90^{th} day following notice to affected PCI Legacy Employees in the form of a summary of material modifications or an updated summary plan description; or
 - (2) October 1, 2004.

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

Amendment No. 7

Effective as of January 1, 2004, or such other dates as noted herein, the AMSEC Employees 401(k) Profit Sharing Plan (the "AMSEC Plan") is hereby amended as follows:

- 1. Section 2.16 is amended to read as follows in its entirety:
- **2.16** Company Contributions. "Company Contributions" shall mean all amounts (whether in cash or other property, including Company Stock) paid by the Company or a Contributing Employer into the Trust Fund established and maintained under the provisions of this Plan for the purpose of providing benefits for Participants and their Beneficiaries, and shall include Elective Deferrals, Profit Sharing Contributions, Matching Contributions, Additional Company Contributions, Catch-up Contributions, Special Contract Contributions and any contributions to pay Plan expenses as provided in Section 6.13.
- 2. A new Section 2.19A is added to the Plan to read as follows in its entirety:
- **2.19A** Contributing Employer. "Contributing Employer" shall mean (a) an Affiliated Company granted permission by the Board of Directors to participate in this Plan, (b) a subsidiary of the Company, or (c) a clearly definable subset of the business of the Company or an Affiliated Company participating in the Plan (which subsets include, but are not limited to, NNS Legacy, PCI and the core AMSEC business) which may make a separate Company Contribution for a Plan Year in accordance with Section 4.3(b).
- 3. A new Subsection 3.2(f) is added to the Plan to read as follows in its entirety:
- (f) Notwithstanding the foregoing to the contrary, an "SCA Eligible Employee" as described in Subsection 5.8(d)(i) who is not an Eligible Employee shall be eligible to commence participation in the Plan solely for the limited purpose of receiving a Special Contract Contribution.
- 4. Subsection 4.3(b) is amended to read as follows in its entirety:
- (b) The Company or a Contributions Employer, with the approval of the Committee, may make Profit Sharing Contributions, Additional Company Contributions, Matching Contributions or Special Contract Contributions to eligible Participants. Contributions made by any Contributing Employer may be made at a separate time from Contributions made by the Company and may be made alone and in an amount that is different than the amount of similar Contributions available to other Participants; provided, however, that such contributions do not discriminate in favor of Highly Compensated Employees pursuant to Code Section 401(a)(4).

5. The title of Article V is amended to read as follows in its entirety:

ELECTIVE DEFERRALS, ROLLOVER CONTRIBUTIONS, PLAN-TO-PLAN TRANSFERS, VOLUNTARY CONTRIBUTIONS AND SPECIAL CONTRACT CONTRIBUTIONS.

- 6. Subsection 5.2(a) is amended to read as follows in its entirety:
- (a) Effective on and after January 1, 2002, the amount of an individual's Compensation that may be deferred subject to the election provided in Section 5.1 shall be a percentage of the individual's Compensation, not to exceed 25%. Notwithstanding the foregoing to the contrary, with regard to Participants who are PCI Legacy Employees, the amount of such Participants' Compensation that may be deferred, subject to the election provided in Section 5.1, shall be an amount equal to not less than one percent (1%) or greater than ninety percent (90%), provided that the percentage rate for determining such contributions shall be a whole number.
- 7. A new Section 5.8 is added to the Plan to read as follows in its entirety:

5.8 Special Contract Contributions.

- (a) <u>Amount of Special Contract Contributions</u>. Effective on and after January 1, 2004, the Company or a Contributing Employer may satisfy the fringe benefit obligations under the McNamara-O'Hara Service Contract Act by electing, with the approval of the Committee, to make a Special Contract Contribution to each SCA Eligible Employee (as defined below) under the Plan who was for the Measuring Period in question performing duties as a wage-earning Employee of the Company or a Contributing Employer. The amount of Special Contract Contribution to be allocated to each SCA Eligible Employee shall be calculated in accordance with the formula specified in 29 CFR §4.175 as follows:
 - (i) the SCA Eligible Employee's fringe benefit determination rate multiplied by the total hours worked under the SCA Contract by the SCA Eligible Employee during the Measuring Period (subject to any caps or limits which may be set forth in the McNamara-O'Hara Service Contract Act or the SCA Contract); less
 - (ii) the total amount contributed during the Measuring Period by the contributor (as applicable, the Company or a Contributing Employer) on behalf of such SCA Eligible Employee for health and welfare and/or pension contribution (including, but not limited to, the employer's portion of contributions for medical, dental, disability, and life insurance, sick leave and retirement savings, including matching contributions under this Plan).
- (b) <u>Multiple Contracts</u>. Notwithstanding the preceding Section 5.8(a) to the contrary, the allocation to an SCA Eligible Employee working on more than one SCA Contract during a Measuring Period may consist of the sum of multiple calculations under (i) and (ii), above; however, each such calculation shall be aggregated

and offset to the full extent permitted under the McNamara-O'Hara Service Contract Act or the SCA Contract.

(c) <u>Timing</u>. It is expected that contributions made by the Company or a Contributing Employer under Section 5.8(a) shall be paid to the Trustee quarterly or at such other intervals as may be required under the McNamara-O'Hara Service Contract Act or a specific SCA Contract, but not less than annually.

(d) Definitions.

- (i) SCA Eligible Employee. "SCA Eligible Employee" means a covered non-exempt Employee, as defined under the Fair Labor Standards Act, working on one or more SCA Contracts and who for the Measuring Period, has a deficiency between (A) the health and welfare and/or pension contributions made by the Company or a Contributing Employer and (B) the Company or a Contributing Employer's fringe benefit obligation under an SCA Contract. Notwithstanding the forgoing to the contrary, if for a Measuring Period the Company or a Contributing Employer elects to fulfill its fringe benefit obligation under an SCA Contract by making hourly cash equivalent or differential payments to its Employees, such Employees shall not qualify as SCA Eligible Employees for such Measuring Period.
- (ii) *SCA Contract*. "SCA Contract" means a contract under the McNamara-O'Hara Service Contract Act or any other contract where a fringe benefit obligation wage determination is established by federal or state law.
- (iii) *Measuring Period*. "Measuring Period" means the time set forth by the Committee for calculating a Special Contract Contribution. The Committee may, in its discretion, establish a separate Measuring Period for the Company and each Contributing Employer which elects to make a contribution described in Section 5.8(a). Notwithstanding the foregoing to the contrary, under no circumstances may the Measuring Period established by the Committee extend beyond the time required under an SCA Contract for the payment of a fringe benefit obligation and/or pension contribution under the McNamara-O'Hara Service Contract Act or a specific SCA Contract.

8. Subsection 6.10(c) is amended to read as follows in its entirety:

- (c) In accordance with Section 4.3, the Company or any Contributing Employer may, with the approval of the Committee, make a separate Profit Sharing Contribution on behalf of its eligible Participants. Any Company Contributions that constitute a Profit Sharing Contribution shall be allocated to the Company Contribution Account of each Participant who is performing duties as a wage-earning Employee of the Company or a Contributing Employer on the last day of the Plan Year and for any Participant who dies, retires or becomes disabled during the Plan Year.
 - (i) Any Profit Sharing Contribution made by the Company shall be allocated to each Participant eligible to receive a share of such contribution in the

proportion that the Compensation paid to each such Participant during the Plan Year by the Company bears to the Compensation paid to all such Participants employed by the Company; and

- (ii) Any Profit Sharing Contribution made by a Contributing Employer shall be allocated to each such Participant who is otherwise eligible to receive a share of such contribution in proportion that the Compensation paid to each such Participant during the Plan Year by the Contributing Employer bears to the Compensation paid to all such Participants employed by the Contributing Employer.
- (iii) Although a Participant may receive allocations from the Company and one or more Contributing Employers, or from two or more Contributing Employers, during a Plan Year the same Compensation shall not be used by more than one Contributing Employer and the Company for allocations made to such Participant's Company Contribution Account.

9. The first paragraph of Section 14.3 is amended to read as follows in its entirety:

The Plan Administrator may authorize one loan from the Trust Fund to Participants (including, for this purpose, Suspended Participants) pursuant to rules prescribed by the Plan Administrator, except with respect to (i) Participants who formerly participated in the NNS 401(k) Plan and (ii) Participants who formerly participated in the NNS 401(k) Plan will be permitted to maintain up to two loans if such loans are outstanding as of September 1, 1999, but after such loans are repaid, such Participants will only be entitled to one loan from the Trust Fund pursuant to the rules prescribed by the Plan Administrator. Participants who formerly participated in the EMA 401(k) Plan will be permitted to maintain up to three loans but only if such loans are outstanding under the EMA 401(k) Plan as of December 1, 2003 and such loans were subsequently rolled over to this Plan in accordance with Section 5.5. After all such EMA 401(k) Plan loans are repaid, such Participants will only be entitled to one loan from the Trust Fund pursuant to the rules prescribed by the Plan Administrator. These rules shall be designed to ensure that all Participant loans satisfy the requirements of Code sections 4975(d)(1), 72(p), and any other provision of law that is, or may become applicable. These rules shall provide that:

AMSEC EMPLOYEES 401 (K) PROFIT SHARING PLAN

Amendment No. 8

The AMSEC Employees 401(k) Profit Sharing Plan is hereby amended as follows:

1. Subsection 5.2(a) is amended to read as follows in its entirety:

Effective on and after July 1, 2004, the amount of an individual's Compensation that may be deferred subject to the election provided in Section 5.1 shall be a percentage of the individual's Compensation not to exceed fifty percent (50%), provided that the percentage rate for determining such contributions shall be a whole number. Notwithstanding the foregoing to the contrary, effective on and after January 1, 2004 the maximum percentage of an individual's Compensation that may be deferred as a Catch-Up Contribution shall be limited to the Participant's "available" Compensation, defined to mean seventy-five percent (75%) of Compensation determined after taking into account the amount of any elective deferrals described in the first sentence of this paragraph (a).

- 2. Section 14.2 is amended by adding the following subparagraph (f) to the end thereof to read as follows effective on and after July 1, 2004:
- (f) The reasonable expense of determining whether a domestic relations order satisfies the requirements of being a qualified domestic relations order shall be charged to the Account of the Participant; provided, however, that if a qualified domestic relations order so provides, such expense may, instead, be charged to the Alternate Payee's share of the Participant's Account or divided equally between the Participant and the Alternate Payee but charged to the Participant's Account.

AMSEC EMPLOYEES 401 (K) PROFIT SHARING PLAN

Amendment No. 9

The AMSEC Employees 401(k) Profit Sharing Plan is hereby amended effective as of January 1, 2006, or such other dates as noted below, as follows:

- 1. Section 2.18 shall be amended to read as follows in its entirety:
- **2.18** Company Stock. "Company Stock" shall mean Class A Common Stock, par value \$.01 per share ("Class A Common Stock"), and to the extent accounted for pursuant to Section 4.2 of the Plan and solely for such purpose, shall also mean Class B Common Stock, par value \$.05 per share ("Class B Common Stock"), of Science Applications International Corporation, or any other security of Science Applications International Corporation or its ultimate parent corporation, if any, designated as Company Stock by the Committee, including Preferred Stock.
- 2. A new Section 2.57A is added to the Plan:
- **2.57A** <u>Preferred Stock</u>. "Preferred Stock" shall mean preferred stock of Science Applications International Corporation or its ultimate parent corporation convertible into Company Stock in accordance with the rules set forth in Section 409(1) of the Code.
- 3. Section 2.59A shall be amended to read as follows in its entirety:
- **2.59A** Quarterly Trade Date. "Quarterly Trade Date" shall mean if there is a "limited market" for the Company Stock, one of the four predetermined times in each calendar year (one each calendar quarter) during which Company Stock may be bought or sold in accordance with the terms and conditions of the "limited market" as set forth in a prospectus prepared by Science Applications International Corporation or its ultimate parent corporation for the Plan. If there is no "limited market" for the Company Stock, the four predetermined date(s) in each calendar year (one each quarter) during which Science Applications International Corporation or its ultimate parent corporation, in its sole discretion, elects to conduct a transaction with respect to the assets of the Plan. If Company Stock is readily tradable on an established securities market (or is Preferred Stock convertible into Company Stock that is readily tradable on an established securities market), then rules established by Science Applications International Corporation or its ultimate parent corporation may designate any or all business days as "Trade Dates;" however, in no event shall such Trade Dates be set any less frequently than four predetermined times per calendar year (one each calendar quarter).

- 4. Subsection 8.3(b)(i) is amended to read as follows effective June 1, 2006 in its entirety:
- (i) The determination of whether an Employee has an immediate and heavy financial need is to be made by the Committee on the basis of all relevant facts and circumstances. A distribution will be deemed to be made on account of an immediate and heavy financial need of the Employee if the distribution is on account of:
 - (A) Medical expenses that would be deductible under Code Section 213(d) (determined without regard to whether the expenses exceed 7.5% of adjusted gross income);
 - (B) Purchase of the principal residence for the Employee (excluding mortgage payments);
 - (C) Payment of tuition and related education fees for the next twelve months of post-secondary education for the Employee's spouse, children or dependents (as defined in Code Section 152, without regard to sections 152(b)(l), (b)(2) and (d)(i)(B));
 - (D) The need to prevent the eviction from the Employee's principal residence or foreclosure on the mortgage of the Employee's principal residence;
 - (E) Payments for burial or funeral expenses for the Employee's deceased parent, spouse, children or dependents (as defined in Code Section 152, without regard to section 152(d)(l)(B));
 - (F) Expenses for the repair of damages to the Employee's principal residence that would qualify for the casualty deduction under Code Section 165 (determined without regard to whether the loss exceeds 10% of adjusted gross income); or
 - (G) Any other event described in Treasury Regulations or rulings as an immediate and heavy financial need and approved by the Committee as a reason for permitting distribution under this section.
- 5. Section 8.6 is amended to read as follows effective June 1, 2006 in its entirety:

8.6 Optional Forms of Payment.

- (a) <u>Form</u>. Effective on and after October 1, 2004 with respect PCI Legacy Employees and effective on and after January 1, 2004 with respect to all other Participants distributions shall be in the form of a cash lump sum or a Direct Rollover, as set forth Section 8.7. For distributions made on and after June 1, 2006, a Participant who has attained age 59 ½ may elect to receive the portion of his or her Distributable Benefit invested in Company Stock in the form of (i) whole shares of Company Stock (plus cash for any fractional shares of Company Stock) or (ii) cash. The value of the amount distributed in the form of cash from the portion of a Participant's Distributable Benefit invested in Company Stock shall be the net proceeds at the sale of the Company Stock liquidated pending distribution, plus the cash attributable to a fractional share.
- (b) <u>Notice of Right to Receive Company Stock</u>. The Committee shall notify the Participant in advance of making the distribution of his right to elect distribution in the form of Company Stock upon reaching age 59 ½. Upon being so notified the Participant shall have a reasonable period of time (at least fourteen days) in which to elect to receive his Distributable Benefit in the form of a Company Stock distribution as calculated under Subsection (c) below.

This election by the Participant shall be made in writing or as otherwise permitted by Section 17.4, shall be irrevocable when made unless the Committee shall approve a revocation thereof, and shall operate to require the Committee to cause the Participant's Distributable Benefit to be made in the form of a Company Stock distribution as calculated under Subsection (c) below.

- (c) <u>Distribution of Company Stock</u>. A Company Stock Distributable Benefit shall be governed by the following rules:
- (i) The amount of such a distribution shall be the number of whole shares of Company Stock held in the Participant's Plan balance (determined as of the Applicable Valuation Date) with the remainder of the value of the Participant's Plan balance distributable in the form of cash.
- (ii) If more than one class of Company Stock is available for distribution to a Participant, the Participant must receive substantially the same proportion of each such class of stock ("Pro Rata Distribution"). The rule in the preceding sentence shall not apply to the extent that the Participant elects, pursuant to rules to be prescribed by Science Applications International Corporation or its ultimate parent corporation, to receive the distribution in a form other than the Pro Rata Distribution. For purposes of the foregoing, Preferred Stock is not considered available for distribution because Preferred Stock is converted prior to distribution.
- (iii) Notwithstanding paragraphs (i) and (ii) above, a Company Stock Distributable Benefit shall not include Preferred Stock. Instead, any such Preferred Stock shall be converted prior to distribution in accordance with the provisions of the applicable Certificate of Incorporation.

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

Amendment No. 10

The AMSEC Employees 401(k) Profit Sharing Plan is hereby amended effective on and after January 1, 2006 as follows:

1. Section 2.18, defining Company Stock, is amended by adding the following to the end of the definition:

"Company Stock shall consist of the 'Exchangeable Company Stock Subaccount,' and the 'Non-Exchangeable Company Stock Subaccount.' If, at the time it authorizes a contribution to the Plan in the form of Company Stock, the Board of Directors specifies that such Company Stock is subject to investment direction by Participants as provided in Section 4.6, then such Company Stock shall be credited to the Exchangeable Company Stock Subaccount. Furthermore, if pursuant to rules established by the Committee under Section 4.6, a Participant directs that a portion of his Accounts (other than Company Stock credited to the Non-Exchangeable Company Stock Subaccount) be invested in Company Stock, then the Company Stock credited as a result of such contributions (along with that portion of any loan repayments by such Participant credited as Company Stock by reason of the Participant's direction to invest Elective Deferrals in Company Stock) shall be credited to the Exchangeable Company Stock Subaccount. All other all Company Stock shall be credited to the Non-Exchangeable Company Subaccount."

2. Section 4.6(a) is amended to read as follows:

"The Committee shall establish rules setting forth the extent to which Participants may determine the manner in which their Accounts (other than Company Stock credited to the Non-Exchangeable Company Stock Subaccount) may be invested. Consistent with such rules, the Committee may establish a choice of investment alternatives from which each Participant may select in determining the manner in which the applicable portions of his Plan Accounts will be invested. The Committee shall prescribe procedures for investment of amounts (other than Company Stock credited to the Non-Exchangeable Company Stock Subaccount) allocated to an Alternate Payee Account. To the extent permitted under rules established by the Committee, a Participant may direct that Company Contributions (other than Company Stock credited to the Non-Exchangeable Company Stock Subaccount) be invested all or in part in a Company Stock fund within the Trust. Company Stock credited to the Non-Exchangeable Company Stock Subaccount shall not be subject to the direction of investments by Participants except as provided in Section 4.6(g) with respect to Participants who are no longer Employees. Fifty percent (50%) of a Participant's Company Matching Contributions shall be invested in the Non-Exchangeable Company Stock Subaccount. Neither the Committee nor the

Trustee shall cause such Company Stock to be invested other than as provided in Section 4.6(g)."

3. The following new Section 4.6(g) is hereby added to the Plan:

"(g) Non-Exchangeable Company Stock Subaccount.

Except for Company Stock credited to the Exchangeable Company Stock Subaccount, the Company's contributions made in the form of Company Stock are made with the expectation and understanding that such contributions shall remain invested in Company Stock until the Participant whose Account is credited with such Company Stock ceases to be an Employee, except to the extent that diversification is otherwise expressly permitted under this Plan. Accordingly, the Non-Exchangeable Company Stock Subaccount shall be invested exclusively in Company Stock (subject only to the power to hold cash pending immediate distribution to Participants, Beneficiaries, and Alternate Payees), and until the Participant whose Account is credited with such Company Stock ceases to be an Employee shall not be transferred to any other investment alternative, except as otherwise expressly provided in this Plan. Notwithstanding any other provisions of the Plan, no fiduciary (including but not limited to the Committee and the Trustees) shall have the authority or duty to cause any Company Stock credited to the Non-Exchangeable Company Stock Subaccount to be invested other than in Company Stock (except for cash held pending immediate distribution to Participants, Beneficiaries, or Alternate Payees)."

AMSEC EMPLOYEES 401(k) PROFIT SHARING PLAN

Amendment No. 11

Effective on and after May 15, 2006 the AMSEC Employees 401(k) Profit Sharing Plan is hereby amended by adding the following new Section 4.6 (h) to read as follows in its entirety:

"(h) Notwithstanding the provisions of this Plan generally prohibiting transfers from the Non-Exchangeable Company Stock Subaccount, subject to rules established by the Committee, and SAIC's right to not purchase all or any portion of the shares of Company Stock offered for sale to it by the Plan, each Participant shall be entitled to direct for the Quarterly Trade Date occurring on or about June 30, 2006 or any postponement thereof (and for no other Quarterly Trade Date), that amounts credited to the Non-Exchangeable Company Stock Account on the Participant's behalf as of such Quarterly Trade Date be invested all or in any part in an investment alternative offered under the Plan other than the Company Stock fund."

AMSEC EMPLOYEES 401(K) PROFIT SHARING PLAN

Amendment No. 12

Effective for distributions occurring on and after March 28, 2005 the AMSEC Employees 401(k) Profit Sharing Plan (the "Plan") is hereby amended to read as follows:

1. Section 7.3(b)(iv) is amended to read as follows in its entirety:

(iv) If a Participant terminates employment with the Company when the value of the Participant's Distributable Benefit is \$5,000 or less, the Participant shall receive a mandatory distribution of the entire value of the Participant's Distributable Benefit as a single sum distribution as soon as practicable following termination of employment. However, effective for distributions occurring on and after March 28, 2005, if the amount of the mandatory distribution, as described above, is greater than \$1,000, then such amount shall be paid as a direct rollover to an individual retirement plan designated by the Company unless the Participant elects, in the manner proscribed by the Company, to either (i) have the mandatory distribution paid directly to an eligible retirement plan specified by the Participant as a direct rollover in accordance with Section 8.7 or (ii) to receive the mandatory distribution directly.

2. Section 8.8(a)(i) is amended to read as follows in its entirety:

(a)(i) <u>Distributable Benefit Under \$5,000</u>. Notwithstanding any other provision of the Plan, if the value of the Participant's Distributable Benefit determined as of the Participant's termination date, death or any subsequent Valuation Date does not exceed \$5,000, the Participant (or Beneficiary in the event of the Participant's death) shall receive a mandatory distribution of the entire value of the Participant's Distributable Benefit as a single sum distribution as soon as practicable following the Participant's termination of employment, receipt of notice of Participant's death or such subsequent Valuation Date, whichever is applicable. However, effective for distributions occurring on and after March 28, 2005, if the amount of the mandatory distribution, as described above, is greater than \$1,000, then such amount shall be paid as a direct rollover to an individual retirement plan designated by the Company <u>unless</u> the Participant elects, in the manner prescribed by the Company, to either (i) have the mandatory distribution paid directly to an eligible retirement plan specified by the Participant as a direct rollover in accordance with Section 8.7 or (ii) to receive the mandatory distribution directly. If the Participant's Distributable Benefit is zero, the Participant shall be deemed to have received a distribution of his entire Distributable Benefit. Distributions in accordance with this Section 8.8(a)(i) shall not require consent by the Participant or the Participant's spouse.

AMSEC EMPLOYEES 401 (k) PROFIT SHARING PLAN

Amendment No. 13

Effective on and after January 1, 2007, except as otherwise noted herein, the AMSEC Employees 401 (k) Profit Sharing Plan (the "Plan") is hereby amended as follows:

- 1. Section 2.19(a) of the Plan is amended to read as follows, effective with respect to payroll periods commencing on or after the date of adoption of this Amendment No. 13:
 - "(a) For purposes of determining the allocation of Profit Sharing Contributions pursuant to Sections 4.3 and 6.10(c) 'Compensation' shall mean the regular (i) salary, (ii) wages, (iii) overtime, (iv) shift differential, hazard pay and similar payments, (v) pay while on leave or vacation, (vi) special pay approved by the Company pursuant to its policies for extended work by salaried employees, (vii) retroactive pay adjustment, and (viii) bonus, in each case to the extent paid by the Company during a Plan Year by reason of services performed by an Employee, and further shall include any elective contributions that are made by the Company on behalf of an Employee pursuant to an arrangement described in Code sections 125,132(f)(4), 402(a)(8), 402(h) or403(b), but shall not include any other amounts (including but not limited to expense reimbursements, sign-on bonuses, special pay other than for salaried non-exempt employees, imputed income and any dividends attributable to Company Stock) and further shall not include any compensation paid by reason of services performed during any period in which the Employee is not a Participant under this Plan. For purposes of determining the allocation of a Participant's permissible Elective Deferrals under Sections 5.2 and 6.10(a) and Matching Contributions under Sections 5.3 and 6.10(b) 'Compensation' shall mean the amounts described above with regard to Profit Sharing Contributions but also shall exclude regular bonuses."
 - 2. Section 5.3(a) of the Plan is amended to read as follows:
 - "(a) Effective for Plan Years beginning on or after January 1, 2007, the Company may, in its discretion, make a Matching Contribution in an amount equal to 45% of a Participant's first \$5,000 of Elective Deferrals for such Plan Year up to a maximum of \$2,000, subject to the limitations specified in Article XIII."

- 3. Section 5.3(b) of the Plan is amended to read as follows:
- "(b) The contribution described in Section 5.3(a) shall be paid to the Trustee as soon as possible after the close of the payroll period to which such contribution relates, but in any event no later than the date which is prescribed by law for filing the Company's federal income tax return, including any extensions for the taxable year with respect to which such contributions are made. A Matching Contribution shall be allocated as provided in Section 6.10(b) and made to each Participant (i) for Plan Years beginning prior to January 1, 2007, who is performing duties as a wage earning Employee for the Company as of the last day of a calendar quarter and any Participant who dies, retires or becomes disabled during such quarter, and (ii) for Plan Years beginning on or after January 1, 2007, who receive Compensation during the relevant payroll period.
- 4. Section 6.10(b) of the Plan is amended to read as follows:

"(b) All Company Contributions which constitute Matching Contributions shall be allocated to Participants' Company Contribution Accounts within ninety (90) days following their contribution to the Trust. An allocation shall be made on behalf of a Participant who is otherwise eligible to receive a Matching Contribution only if (i) for Plan Years beginning prior to January 1, 2007, such Participant is a wage earning Employee of the Company on the last day of the calendar quarter, or if such Participant dies, retires or becomes disabled during such quarter, and (ii) for Plan Years beginning on or after January 1, 2007, such Participant receives Compensation during the relevant payroll period."

October 18, 2006

SAIC, Inc. 10260 Campus Point Drive San Diego, California 92121

Gentlemen:

I am the Senior Vice President, General Counsel and Secretary of SAIC, Inc. (the "Company"). As such, I have acted as your counsel in connection with the Registration Statement on Form S-8 (the "Registration Statement") filed with the Securities and Exchange Commission (the "Commission) on October 19, 2006 under the Securities Act of 1933, as amended (the "Securities Act"). The Registration Statement covers the offer and sale of an aggregate of up to 219,000,000 shares (the "Company Shares") of the Company's Class A Preferred Stock, par value \$0.0001 per share, and Common Stock, par value \$0.0001 per share, by the Company pursuant to the 2006 Equity Incentive Plan; 2006 Employee Stock Purchase Plan; Science Applications International Corporation Retirement Plan; Science Applications International Corporation Plan and Management Stock Compensation Plan; Science Applications International Corporation Key Executive Stock Deferral Plan; Science Applications International Corporation 1999 Stock Incentive Plan; and AMSEC Employees 401(k) Profit Sharing Plan (collectively, the "Plans").

I am generally familiar with the affairs of the Company. In addition, I have examined and am familiar with originals or copies, certified or otherwise identified to my satisfaction, of (i) the Registration Statement, (ii) the Certificate of Incorporation and Bylaws of the Company as currently in effect, (iii) resolutions adopted by the Board of Directors relating to the filing of the Registration Statement and the issuance of the Company Shares thereunder, (iv) the Plans, and (v) such other documents as I have deemed necessary or appropriate as a basis for the opinions set forth below. In my examination, I have assumed the genuineness of all signatures, the legal capacity of natural persons, the authenticity of all documents submitted to me as originals, the conformity to original documents of all documents submitted to me as certified or photostatic copies, and the authenticity of the originals of such copies.

Based upon and subject to the foregoing, I am of the opinion that:

The Company Shares that are being offered and sold directly by the Company have been duly authorized for issuance, and when issued and sold in accordance with the Plans and the Registration Statement, will be legally issued, fully paid, and nonassessable.

SAIC, Inc. October 18, 2006 Page 2

I hereby consent to the use of my name in the Registration Statement under Item 5 and to the filing of this opinion as an exhibit to the Registration Statement. In giving such consent, I do not thereby admit that I come within the category of persons whose consent is required under Section 7 of the Securities Act or the rules and regulations of the Commission thereunder.

Very truly yours,

/s/ Douglas E. Scott

Douglas E. Scott Senior Vice President, General Counsel and Secretary

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the incorporation by reference in this Registration Statement of SAIC, Inc. on Form S-8 of (1) our report relating to the financial statements of Science Applications International Corporation dated April 24, 2006, and our report relating to the balance sheet of SAIC, Inc. dated April 24, 2006, appearing in Registration Statement No. 333-128021 on Form S-1 of SAIC, Inc., (2) our report relating to the financial statements and supplemental schedule of Science Applications International Corporation 401(k) Profit Sharing Plan dated June 22, 2006 (which report expresses an unqualified opinion and includes an explanatory paragraph relating to valuation of investment securities for which market values are not readily ascertainable), appearing in the Annual Report on Form 11-K of Science Applications International Corporation 401(k) Profit Sharing Plan for the year ended December 31, 2005, and (3) our report relating to the financial statements and supplemental schedule of AMSEC Employees 401(k) Profit Sharing Plan dated June 26, 2006 (which report expresses an unqualified opinion and includes an explanatory paragraph relating to valuation of investment securities for which market values are not readily ascertainable), appearing in the Annual Report on Form 11-K of AMSEC Employees 401(k) Profit Sharing Plan for the year ended December 31, 2005.

/s/ DELOITTE & TOUCHE LLP

San Diego, California October 18, 2006